The relationship between demographic variables and consumer confidence levels in the rationales for regulation:

A quantitative study of Scottish retail banking consumers

Claire Lynne McCafferty

A thesis submitted in partial fulfilment of the requirements of Edinburgh Napier University, for the award of Doctor of Philosophy

April 2020

DECLARATION

I hereby declare that the work presented in this thesis has not been submitted for any other degree or professional qualification, and that it is the result of my own independent work.

Claire Lynne McCafferty

April 2020

ABSTRACT

Confidence in the banking system is critical to the health of households, businesses and the overall economy. Government intervention in the financial services sector and the strict regulation of the banking industry is rationalised by the need to minimise the occurrence of events that could undermine the reputation of banks and threaten confidence in the system.

This aim of this research is to answer the question, 'Are Scottish consumers confident in retail banking regulation?' This research has taken a unique approach to answering this question by measuring the levels of confidence in each of the specific rationales identified in the literature for regulating the retail banking system.

Analysis of the findings was undertaken in three distinct stages: individual demographics in relation to levels of confidence; the direction of confidence in relation to these demographics; and the combined demographic factors that have the greatest impact on the levels of confidence in each of the areas that regulation seeks to address.

In designing the survey, a three-stage approach was also adopted; theorists, practitioners and end-users contributed to the development of a consumer survey. The questions were derived from the theoretical framework and developed into the final consumer survey that would gather the demographic information and test the hypotheses.

The main findings are that the youngest age group (18-29 years) have the highest levels of confidence in their retail banks. Those with a higher income and those with higher educational attainment have the least confidence in the areas that regulation seeks to address. This has implications for both industry and the regulators in terms of how they target these specific groups to increase their levels of trust and satisfaction. A lack of trust threatens the stability of the financial system and the wider economy.

Finally, this research has contributed to the continued debate into the use of parametric methods of data analysis when using Likert scales. The finding is that the use of parametric tests for the analysis of Likert scale data is robust.

ACKNOWLEDGEMENTS

I would like to thank my Director of Studies, Malcolm Pettigrew, and my Supervisor, Dr David White, for their guidance, patience, encouragement and support.

I would also like to thank the theorists, practitioners, and end-users who took part in the three stages of the survey design, and the participants who took part in the final study.

Also my family, in particular my husband and Mum, who have been on this journey with me and have kept me going throughout the challenging times.

This is dedicated to my Dad, Alexander William Blackwood (1957-2010).

CONTENTS

| LIST OF TABLES | page xi |
|---|-----------|
| LIST OF FIGURES | page xii |
| LIST OF ABBREVIATIONS | page xvii |
| CHAPTER 1: INTRODUCTION | |
| 1.1 Background | page 1 |
| 1.2 Recent Scandals in the UK Banking Sector | page 5 |
| 1.2.1 Mis-selling | page 6 |
| 1.2.2 Ring Fencing | page 7 |
| 1.2.3 Bank Runs | page 9 |
| 1.3 Who was to Blame? | page 11 |
| 1.4 The Financial Services Authority | page 11 |
| 1.5 The Financial Conduct Authority | page 12 |
| 1.6 Consumer Confidence | page 12 |
| 1.7 Research Aim and Objectives | page 13 |
| 1.8 Research Methods | page 14 |
| 1.9 The Importance of the Research | page 15 |
| 1.10 Structure of the Thesis | page 16 |
| CHAPTER 2: LITERATURE REVIEW | |
| 2.1 Introduction | page 18 |
| 2.2 Retail Banks | page 18 |
| 2.2.1 What is a Retail Bank? | page 18 |
| 2.2.2 The Evolution of the Retail Bank Market in the UK | page 19 |
| 2.2.3 Retail Bank Structure (Business Models) | page 19 |
| 2.2.4 Retail Bank Culture | page 20 |

| 2.2.5 Organisational Ethicspage 22 |
|--|
| 2.2.5.1 The Corporation as a Moral Agentpage 24 |
| 2.2.5.2 Ethical Frameworkspage 26 |
| 2.3 Retail Bank Consumerspage 33 |
| 2.3.1 Who is a Retail Bank Consumer?page 33 |
| 2.3.2 Consumer Protection in Retail Bankingpage 34 |
| 2.3.3 Consumer Protection Lawspage 35 |
| 2.3.4 Consumer Demand for Regulationpage 36 |
| 2.4 Regulationpage 36 |
| 2.4.1 What is Regulation?page 37 |
| 2.4.2 The Evolution of the Regulation of UK Financial Servicespage 38 |
| 2.4.3 Twin Peakspage 49 |
| 2.4.4 Current Regulatory Structure of the UK Financial Services Sectorpage 51 |
| 2.4.5 Ring Fencingpage 52 |
| 2.4.6 Economics for Effective Regulation (EFER)page 53 |
| 2.5 Arguments Against Regulationpage 55 |
| 2.5.1 Hazardspage 58 |
| 2.5.2 Regulatory Capturepage 60 |
| 2.5.3 The Revolving Doorpage 61 |
| 2.6 Rationales For Regulationpage 63 |
| 2.6.1 Demand from the Publicpage 63 |
| 2.6.2 Market Failurespage 65 |
| 2.6.3 Economic Rationales for Regulationpage 65 |
| 2.6.3.1 Externalitiespage 66 |
| 2.6.3.2 Imperfect and Asymmetric Informationpage 70 |
| 2.6.3.3 Moral Hazardpage 72 |

| 2.6.3.4 Fraudulent Activities | page 73 | |
|--|----------|--|
| 2.6.4 Social Rationales for Regulation | page 75 | |
| 2.6.4.1 Paternalism | page 76 | |
| 2.6.4.2 Distributive Justice | page 81 | |
| 2.6.4.3 Community Values | page 83 | |
| 2.8 Conclusions | page 84 | |
| CHAPTER 3: RESEARCH METHODOLOGY AND METHODS | | |
| 3.1 Introduction | page 88 | |
| 3.2 Methodology | page 89 | |
| 3.2.1 Research Paradigm | page 90 | |
| 3.3 Methods | page 92 | |
| 3.3.1 Data Collection | page 96 | |
| 3.3.2 Data Analysis | page 103 | |
| 3.4 Strategy for Analysis | page 109 | |
| 3.5 Research Ethics and Integrity | page 110 | |
| 3.6 Survey Design | page 111 | |
| 3.7 Stage 1: Theorists | page 113 | |
| 3.8 Stage 2: Practitioners | page 114 | |
| 3.9 Feedback from Stage 1 and Stage 2 | page 114 | |
| 3.10 Stage 3: End-userspage 115 | | |
| 3.11 Reliability and Validity Considerations for Data Collection | page 118 | |
| 3.12 Sample | page 118 | |
| 3.13 Data Collection | page 121 | |
| 3.13.1 Gender and Age Groups | page 122 | |
| 3.14 Vulnerable Groups | page 126 | |
| 3.14.1 Older Consumers | page 127 | |
| 3.14.2 Low Income | page 128 | |
| | | |

| 3.14.3 Low Level of Education | page 129 |
|--|----------|
| 3.15 Persons Working in Financial Services | page 131 |
| 3.16 Data Description Conclusion | page 132 |
| CHAPTER 4: DESCRIPTIVE STATISTICS AND CORRELATION A | NALYSIS |
| 4.1 Introduction | page 134 |
| 4.1.1 Tests for Normality | page 134 |
| 4.1.2 Outliers | page 135 |
| 4.1.3 Scale Reliability | page 135 |
| 4.1.4 Labelling of the Variables | page 136 |
| 4.2 Methods of Data Analysis and Presentation of Data | page 138 |
| 4.3 Descriptive Statistics | page 140 |
| 4.3.1 Externalities | page 141 |
| 4.3.2 Asymmetric Information | page 157 |
| 4.3.3 Moral Hazard arising from the Principal-Agent Problem. | page 166 |
| 4.3.4 Fraudulent Activities | page 176 |
| 4.3.5 Paternalism | page 185 |
| 4.3.6 Distributive Justice | page 194 |
| 4.3.7 Community Values | page 203 |
| 4.3.8 Protection | page 211 |
| 4.3.9 Summary of Descriptive Statistics | page 220 |
| 4.4 Correlations | page 223 |
| 4.4.1 Age | page 224 |
| 4.4.2 Education | page 224 |
| 4.4.3 Income | page 228 |
| 4.4.4 Summary of Correlation Analysis Findings | page 230 |
| CHAPTER 5: REGRESSION ANALYSIS AND RESEARCH FINDIN | GS |
| 5.1 Introduction | page 233 |
| | |

| 5.2 Demographic Groups | | page 233 |
|---|---|--|
| 5.3 Levels of Confidence in the Ra | tionales for Regulation. | page 234 |
| 5.3.1 Macro Externalities: system is stable and secure | 'l have confidence e' | <i>the UK retail banking</i> page 235 |
| 5.3.2 Micro Externalities: 'I bank account(s) is protected | have confidence that ti d should my bank fail' | he money in my retail page 236 |
| 5.3.3 Asymmetric Informati provide me with sui decisions' | on: 'I have confidence fficient information | that my retail bank will to make informed page 238 |
| 5.3.4 Moral Hazard: 'I hav right thing' | ve confidence that my | <i>retail bank will do the</i> page 240 |
| 5.3.5 Fraudulent Activities: retail bank misconduct' | 'I have confidence that | t I am protected against page 241 |
| 5.3.6 Paternalism: <i>'I have my retail bank'</i> | confidence that I car | n trust the expertise of page 242 |
| 5.3.7 Distributive Justice: banking products is fair' | 'I have confidence t | <i>hat the price of retail</i> page 244 |
| 5.3.8 Community Values: ' discriminated against by my | l have confidence that / retail bank' | I will not be unlawfully page 245 |
| 5.3.9 Overall Confidence; bank customer I am adequa | 'In general, I have cor ately protected' | nfidence that as a retail |
| 5.4 Summary of Regression Analy | sis Findings | page 248 |
| 5.5 Conclusions | | page 249 |
| CHAPTER 6: OVERALL FINDING | S AND CONCLUSION | IS |
| 6.1 Introduction | | page 250 |
| 6.2 Background | | page 250 |
| 6.3 Theory | | page 250 |
| 6.4 Method | | page 251 |
| 6.5 Data Collection | | page 253 |
| 6.6 Analysis | | page 254 |
| 6.7 Findings and Limitations | | page 254 |
| Claire Lynne McCafferty | PhD April 2020 | Page viii |

| 6.7.1 The consumer is confident the UK retail banking system is stable and securepage 255 |
|---|
| 6.7.2 The consumer is confident that the money in their retail bank account(s) is protected should their bank failpage 256 |
| 6.7.3 The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions |
| 6.7.4 The consumer is confident that their retail bank will do the right thingpage 258 |
| 6.7.5 The consumer is confident that they are protected from retail bank misconductpage 259 |
| 6.7.6 The consumer is confident that they can trust the expertise of their retail bankpage 260 |
| 6.7.7 The consumer is confident that the price of retail banking products is fairpage 262 |
| 6.7.8 The consumer is confident that they will not be unlawfully discriminated against by their retail bank |
| 6.7.9 In general, the consumer is that as a retail bank customer they are adequately protectedpage 264 |
| 6.7.10 Structurepage 265 |
| 6.7.11 Culturepage 266 |
| 6.7.12 Organisational Ethicspage 267 |
| 6.8 Limitationspage 268 |
| 6.9 Practical Implicationspage 269 |
| 6.10 Contributions to Theorypage 271 |
| 6.11 Recommendations for Future Researchpage 271 |
| REFERENCESpage 273 |
| APPENDICES |
| A: EXAMPLE CONSENT FORM AND EMAIL COMMUNICATIONpage 304 |
| B: EXAMPLE SURVEYpage 306 |
| C: SHAPIRO-WILK TEST OF NORMALITYpage 312 |

| D: DESCRIPTIVE STATISTICS | page 319 |
|---------------------------------|----------|
| E: CORRELATION ANALYSIS | page 344 |
| F: STEPWISE REGRESSION ANALYSIS | page 346 |

LIST OF TABLES

- 1.1 Financial Services Compensation Scheme: Compensation Levels
- 2.1 FCA Principles for Business
- 2.2 Systematic Drivers of Poor Market Outcomes
- 3.1 The Paradigm of Positivist Inquiry
- 3.2 Demographic Questions and Variable Categorisation
- 3.3 Theoretical Questions and Variable Categorisation
- 3.4 Changes to the Consumer Survey following Stages 1 and 2 of Survey Design
- 3.5 Changes to the Consumer Survey following Stage 3 of Survey Design
- 3.6 Sample Required for each Strata
- 3.7 Sample Achieved for each Strata
- 4.1 Reliability Statistics
- 4.2 Variable Classification
- 4.3 Observed Outcomes of Parametric *vs.* Non-Parametric Tests
- 4.4 Summary of Findings from First Stage of Analysis
- 5.1 Demographic Groupings
- 5.2 Hypothesis Test 1
- 5.3 Hypothesis Test 2
- 5.4 Hypothesis Test 3
- 5.5 Hypothesis Test 4
- 5.6 Hypothesis Test 5
- 5.7 Hypothesis Test 6
- 5.8 Hypothesis Test 7
- 5.9 Hypothesis Test 8
- 5.10 Hypothesis Test 9
- 5.11 Summary of Regression Analysis Findings

LIST OF FIGURES

- 1.1 Comparison of Real GDP Growth from 2007-2017 in UK and USA
- 1.2 Comparison of Unemployment Rate from 2007-2017 in UK and USA
- 2.1 RBS 'Yes' Checks
- 2.2 FCA Consumer Protection Objective
- 2.3 Triangulated Approach to Research Design
- 3.1 Stages of Survey Design: from Theoretical Model to Empirical Testing
- 3.2 Age Distribution in Scotland
- 3.3 Education Distribution in Scotland
- 3.4 Income Distribution in Scotland
- 3.5 Gender: Electronic Surveys
- 3.3 Gender: Paper Surveys
- 3.7 Age: Electronic Surveys
- 3.8 Age: Paper Surveys
- 3.9 Sample Required *vs.* Sample Obtained
- 3.10 Income Distribution of Sample
- 3.11 Education Levels: Groups
- 3.12 Education Levels: All
- 4.1 I have confidence the UK retail banking system is stable and secure: employment in sector *vs.* non-employment in sector
- 4.2 I have confidence the UK retail banking system is stable and secure: male *vs.* female participants
- 4.3 I have confidence the UK retail banking system is stable and secure: age groups

- 4.4 I have confidence the UK retail banking system is stable and secure: level of education
- 4.5 I have confidence the UK retail banking system is stable and secure: level of income
- 4.6 I have confidence that the money in my retail bank account(s) is protected should my bank fail: employment in sector *vs.* non-employment in sector
- 4.7 I have confidence that the money in my retail bank account(s) is protected should my bank fail: male *vs.* female
- 4.8 I have confidence that the money in my retail bank account(s) is protected should my bank fail: age groups
- 4.9 I have confidence that the money in my retail bank account(s) is protected should my bank fail: education level
- 4.10 I have confidence that the money in my retail bank account(s) is protected should my bank fail: income level
- 4.11 I have confidence that my retail bank will provide me with sufficient information to make informed decisions: employment in sector *vs.* non-employment in sector
- 4.12 I have confidence that my retail bank will provide me with sufficient information to make informed decisions: males *vs.* females
- 4.13 I have confidence that my retail bank will provide me with sufficient information to make informed decisions: age groups
- 4.14 I have confidence that my retail bank will provide me with sufficient information to make informed decisions: education level
- 4.15 I have confidence that my retail bank will provide me with sufficient information to make informed decisions: income level
- 4.16 I have confidence that my retail bank will do the right thing: employment in sector *vs.* non-employment in sector

- 4.17 I have confidence that my retail bank will do the right thing: males *vs.* females
- 4.18 I have confidence that my retail bank will do the right thing: age groups
- 4.19 I have confidence that my retail bank will do the right thing: education levels
- 4.20 I have confidence that my retail bank will do the right thing: income levels
- 4.21 I have confidence that I am protected from retail bank misconduct: employment in sector *vs.* non-employment in sector
- 4.22 I have confidence that I am protected from retail bank misconduct: males vs. females
- 4.23 I have confidence that I am protected from retail bank misconduct: age groups
- 4.24 I have confidence that I am protected from retail bank misconduct: education levels
- 4.25 I have confidence that I am protected from retail bank misconduct: income levels
- 4.26 I have confidence that I can trust the expertise of my retail bank: employment in sector *vs.* non-employment in sector
- 4.27 I have confidence that I can trust the expertise of my retail bank: males *vs.* females
- 4.28 I have confidence that I can trust the expertise of my retail bank: age groups
- 4.29 I have confidence that I can trust the expertise of my retail bank: education levels
- 4.30 I have confidence that I can trust the expertise of my retail bank: income levels

- 4.31 I have confidence that the price of retail banking products is fair: employment in sector vs. non-employment in sector
- 4.32 I have confidence that the price of retail banking products is fair: males *vs.* females
- 4.33 I have confidence that the price of retail banking products is fair: age groups
- 4.34 I have confidence that the price of retail banking products is fair: education levels
- 4.35 I have confidence that the price of retail banking products is fair: income levels
- 4.36 I have confidence that I will not be unlawfully discriminated against by my retail bank: employment in sector *vs.* non-employment in sector
- 4.37 I have confidence that I will not be unlawfully discriminated against by my retail: males vs. females
- 4.38 I have confidence that I will not be unlawfully discriminated against by my retail: age groups
- 4.39 I have confidence that I will not be unlawfully discriminated against by my retail bank: education levels
- 4.40 I have confidence that I will not be unlawfully discriminated against by my retail bank: income levels
- 4.41 I have confidence that I as a retail bank customer I am adequately protected: employment in sector *vs.* non-employment in sector
- 4.42 I have confidence that I as a retail bank customer I am adequately protected: males *vs.* females
- 4.43 I have confidence that I as a retail bank customer I am adequately protected: age groups

4.44 I have confidence that I as a retail bank customer I am adequately protected: education levels

4.45 I have confidence that I as a retail bank customer I am adequately protected: income levels

LIST OF ABBREVIATIONS

| ABCUL | Association of British Credit Unions Limited |
|-----------|--|
| AFI | Alternative Financial Institution |
| ANOVA | Analysis of Variance |
| Basel III | Basel Capital Accord III |
| BBC | British Broadcasting Corporation |
| BCCI | Bank of Credit and Commerce International |
| BIPRU | Banks, Building Societies and Investment Firms Prudential Sourcebook |
| BIS | Bank for International Settlements |
| BoE | Bank of England |
| BSB | Banking Standards Board |
| BSRC | Banking Standards Review Council |
| CAB | Citizens Advice Bureau |
| CEO | Chief Executive Officer |
| CSFI | Centre for the Study of Financial Innovation |
| CLT | Central Limit Theorem |
| CMA | Competition and Markets Authority |
| COB | Conduct of Business |
| CPP | Card Protection Plan |
| CRA | Consumer Rights Act 2015 |
| CRD | Capital Requirements Directive |
| DFA | Dodd-Frank Wall Street Reform |
| DG FISMA | Directorate-General for Financial Stability |
| DPA | Deferred Prosecution Agreement |
| EEA | European Economic Area |
| EFER | Economics for Effective Regulation |
| FCA | Financial Conduct Authority |
| FOS | Financial Ombudsman Service |
| FS | Financial Services |
| FSA | Financial Services Authority |
| FSCS | Financial Services Compensation Scheme |
| FSMA | Financial Services and Markets Act 2000 |
| GDP | Gross Domestic Product |
| | |

| GENPRU | General Prudential Sourcebook |
|-----------|--|
| GfK | Growth from Knowledge |
| HSBC | Hong Kong and Shanghai Banking Corporation |
| ICB | Independent Commission on Banking |
| IMF | International Monetary Fund |
| LIBOR | London Inter-Bank Offered Rate |
| LOLR | Lender of Last Resort |
| MMR | Mortgage Market Review |
| NRAM | Northern Rock Asset Management |
| OFT | Office of Fair Trading |
| ONS | Office for National Statistics |
| PPI | Payment Protection Insurance |
| PRA | Prudential Regulation Authority |
| RBS | Royal Bank of Scotland |
| RDR | Retail Distribution Review |
| RFTS | Ring Fencing Transfer Scheme |
| SIB | Securities and Investments Board |
| SIFI | Systemically Important Financial Institution |
| SPSS | Statistical Package for the Social Sciences |
| SRR | Special Resolution Regime |
| SVM | Shareholder Value Model |
| TBTF | Too Big To Fail |
| TCF | Treating Customers Fairly |
| Tukey HSD | Tukey Honestly Significant Difference |
| UK | United Kingdom |
| USA | United States of America |

CHAPTER 1 INTRODUCTION

1.1 Background

A bank account is a basic necessity to participate in modern society. It is a requirement for employment and housing. With no bank account it would be almost impossible to borrow money. Paying for goods and services, paying utility bills, or transferring money to friends and family are all facilitated through bank account services. Plastic cards, internet banking, mobile payments are all services accessed through bank account provision. Commercial banks provide retail banking services to household and business customers. They play the vital role of channelling funds between savers and borrowers and operating the payments system.

In the facilities described above, commercial banks are undertaking the crucial role of financial intermediation. In this role they are providing the essential economic mechanism of transferring and allocating funds to their most productive use. A bank provides loans to borrowers (deficit units) and accepts deposits from savers (surplus units). Since households make up the largest proportion of surplus units (consumer spending is by far the single biggest part of aggregate demand), they are providing a high proportion of the finance that businesses and governments require through the deposits held in their bank accounts.

Financial intermediation reconciles three disparate needs of deficit units and surplus units which cannot be met through direct lending. The first is the difference in size of deposits to loans. Depositors may be making relatively small, albeit frequent, deposits into their bank accounts, whilst borrowers may by looking to borrow a substantial sum perhaps to finance a home purchase, or to finance a large-scale commercial project. Secondly, depositors may need the flexibility to be able to withdraw their funds at relatively short notice, a liquidity preference, but home finance loans and commercial loans for investment projects tend to require long-term borrowing. Lastly, higher risk is associated with the probability of higher return, consequently banks will seek out contracts which offer the best returns, and conversely depositors may be risk-averse. Additionally, borrowers who are more

inclined to take risk are less resistant to higher interest rates. These unequal requirements are resolved through the transformation function that banks perform.

To overcome the disparity in size of deposits to loans, banks lend the aggregated deposits. To resolve the mismatch of demand deposits to long-term loans banks use maturity transformation. Banks are able to lend for a longer timeframe than they borrow because they maintain a net balance which in normal operating conditions is not significantly affected by the daily inflows and outflows of deposits and withdrawals. Lastly, since two contracts exist when banks accept deposits (between bank and depositor) and make loans (between bank and borrower) the risk is transformed. The risk-averse depositor does not take on any of the loan risk, and the bank is able to diversify their risk by investing in a range of products, industries, geographical areas, and consumer profiles.

Confidence in the system is essential since depositors would demand their deposits and possibly trigger a bank run if they feared that there was a risk of loss. Without these deposits, banks would soon find their capital depleted and could not make loans. Banks that face liquidity and capital shortages may face insolvency and could cause a shock to the overall system, triggering a financial crisis.

Where banks have a shortage of funds to lend, or decide to cut back on lending (such as after Global Financial Crisis) businesses will struggle to raise the finance they need to invest and grow. This in turn will lead to slower economic growth (see Figure 1.1 overleaf) and higher unemployment (see Figure 1.2 overleaf).



Figure 1.1: Comparison of Real GDP Growth from 2007-2017 in UK and USA

Figure 1.2: Comparison of Unemployment Rate from 2007-2017 in UK and USA



(Adapted from Statista, 2018 and 2019)

⁽Adapted from Statista, 2019)

Confidence in the banking system is consequently critical to the health of households, businesses and the overall economy. Hence, government intervention in the financial services sector and the strict regulation of the banking industry is rationalised by the need to minimise the occurrence of events that could undermine the reputation of banks and threaten confidence in the system.

The Global Financial Crisis of 2007-2008 left the United Kingdom's financial services sector with a less than favourable reputation. Bennett and Kottasz (2012, p.128) argued that "a substantial deterioration in the favourability of public attitudes towards the banking industry seems to have occurred following the crisis".

The aftermath of the collapse, the nationalisation of Northern Rock and rescuing of The Royal Bank of Scotland Group by the taxpayer are likely to remain in the forefront of the media and general public's minds for decades to come.

In fact, the situation in the financial sector became so desperate that the United Kingdom's government abandoned the Office of Fair Trading and the Competition Commission rules and allowed Lloyds TSB to take over Halifax Bank Of Scotland in order to avoid their collapse and try to regain financial stability on the grounds of "national interest" (Hall, 2009, p.445). This controversial acquisition became the subject of a court case nearly a decade later when 6000 retail and institutional investors sued the bank for £600m. Shareholders argued that they had been deliberately misled by the bank and its executives about Halifax Bank of Scotland's financial situation. Lloyd's TSB executives argued that the waiving of the competition restrictions was an opportunity that was too good to be missed (George, 2018).

Akinbami (2011, p.134) described it as "the most significant financial crisis to occur since the Great Crash of 1929". To put this comparison into perspective, in 2009 Gross Domestic Product (GDP) fell by 4.8% compared to a fall of 5.1% in 1931 (The Economist, 2010). However, the UK economy was around 400 times larger in 2009 compared to 1931 (Measuring Worth, 2018). Gamble (2009, p.452) wrote of the "economic, financial, organisational, political and ideological" hurdles faced during recovery in the 1930s and the long depression that followed the crash. This differed

from the predictable 10-year boom and bust cycles which characterised the 19th century UK economy; crisis came to mean not just financial struggles, but also the socio-political obstacles to recovery.

In contrast, Richardson (1969) described the social and political struggles faced by the British public in the 1930s as mild compared to their counterparts in the United States of America or Germany, with unemployment rising by not much above 5%, wages increasing by an average of 3% annually, national income barely declining, and an "upswing" from 1932 onwards. In reality, in comparison to the rest of the world, the UK depression was minor.

What has compounded public mistrust during this most recent crisis, are the scandals which some feel amount to, at least metaphorically, criminal acts. For example: Payment Protection Insurance (PPI) mis-selling; London Interbank Offered Rate (LIBOR) fixing; and interest rate swap mis-selling, to name just a few. Gosling (2011, p.40) wrote of the PPI debacle, "In the history of financial crimes perpetrated against the public in this country...the mis-selling of PPI...is up there with the best of them". PPI was sold alongside retail credit as an insurance product that enabled consumers to ensure that their repayments would be met should they become unable to earn income due to accident, illness, death and certain other circumstances.

1.2 Recent Scandals in the UK Retail Banking Sector

There have been a number of scandals in the sector which have been attributed to a failure of regulation. Many of these can be linked to the market failures of asymmetric information, when one party to a transaction has more material knowledge than another. Additionally, the principal-agent problem, which exists due to bank executives acting on behalf of shareholders, can lead to problems of moral hazard and conflicts of interest. Furthermore, the inequality in bargaining power has led to a rise in accusations of fraud and mis-selling. Some examples of these scandals will now be discussed.

1.2.1 Mis-selling

One of the biggest retail banking mis-selling scandals of recent years, in terms of number of consumers affected, amount of compensation paid out and time taken to resolve, was exposed in September 2005 when the Citizens Advice Bureau (CAB) published a report which demonstrated the problems consumers faced with PPI. In the report, CAB reviewed cost, exclusions, sales process, and claims. CAB revealed unfair treatment of consumers in all of these areas. For example, one-off premiums of up to 56% of the loan value were being added to the loan and interest charged on the premiums (Citizens Advice Bureau, 2005). Additionally, CAB found that consumers were being pressurised into taking PPI without having the exclusions explained properly, or sufficient questioning carried out to establish suitability of the product. Consumers who were experiencing financial difficulty were finding their debts increase through consolidation loans with PPI added to them. Often these consumers would have to refinance as they struggled to meet the monthly loan repayments, and a new PPI policy would be sold with each new loan. CAB (2005) also found that there were several obstacles when a consumer tried to make a claim. Firstly, for many consumers this was the earliest point in which policy exclusions came to light. Next, the evidence that had to be provided in order to make a claim, either medical or unemployment related, was expensive and time consuming to obtain. Then, the claims process was protracted, in some cases resulting in missed repayments which resulted in financial detriment to the consumer.

CAB raised a super-complaint with the Office of Fair Trading and urged them, the Financial Services Authority and the Treasury Select Committee, to investigate further. In October 2006 the Office of Fair Trading published its findings and in February 2007 recommended referral to the Competition Commission under section 131 of the Enterprise Act 2002 (Competition Commission, 2009).

The Competition Commission made several recommendations to resolve the adverse effect on competition and consumer detriment:

i. A ban was to be placed on selling PPI at the point of sale of credit;

- ii. A personal quote was to be provided; and
- iii. Information relating to costs, benefits, optional nature, alternative providers and protection products must be stipulated in the marketing material (Competition Commission, 2009).

In April 2011 the British Banker's Association decided to challenge the new rules imposed by the FSA, following the Competition Commission findings, in the High Court, a case which they lost and decided not to appeal (Wells, 2011). Several years later, claims are still being made and in March 2019 the Financial Conduct Authority (FCA), the FSA successor, announced that the total paid since January 2011 was £34.2bn (FCA, 2019b). However, a deadline for claims of August 2019 was imposed by the FCA in 2017.

1.2.2 Ring Fencing

There have also been a number of scandals outwith retail banking which have strengthened the case to ring fence them from investment banks. The current regulatory structure is discussed in more detail in Section 2.4.4. Ring fencing is the separation of retail banking from the riskier global operations and investment banking side of the business. During the bank bail-outs, the taxpayer, via the Bank of England (BoE), was effectively bailing out not just the essential retail part of the business but the much riskier and profitable investment side. In future the BoE would be able to act in its capacity of Lender of Last Resort (LOLR) and save the retail side (to protect consumers) and allow the investment side to fail (FCA, 2016c). There will be further discussion of ring fencing in Section 2.4.5. This echoes the reforms in the United States following the Global Financial Crisis. The Dodd-Frank Wall Street Reform Act and the Consumer Protection Act (DFA) was enacted in July 2010 to bring an end to implicit guarantees of bail-out for those institutions regarded 'Too Big To Fail' (TBTF). Balasubramnian and Cyree (2014, p.156) argue that "for market discipline to be effective there must be the market belief that uninsured creditors and shareholders will bear the losses in the event of failure". However, their research

finds that although market discipline has improved since the passing of the DFA, there are opportunities for further improvements.

i. LIBOR-fixing Scandal

Barclays was accused of fixing the London Interbank Offered Rate (LIBOR), the rate at which international banks in the UK lend to each other, and was fined £59.5 million in June 2012 under section 206 of the Financial Services and Markets Act 2000 (Walker, 2013). Dillow (2012) stated that the biggest cost of the Barclay's scandal was trust, as this in turn depresses long-run economic growth. The reason for this, he explains, is asymmetric information, which is explored in more detail in Section 2.6.3.2. The LIBOR scandal is a textbook example of what Llewellyn (Financial Services Authority, 1999) termed 'grid lock'. Grid lock is discussed in more detail in Section 2.2.4.

ii. Money Laundering

The United States Department of Justice (2012) released a statement reporting that HSBC UK and HSBC USA had admitted to Anti-Money Laundering and sanctions violations. This included permitting drugs traffickers to "launder hundreds of millions of dollars through HSBC subsidiaries, and to facilitate hundreds of millions more in transactions with sanctioned countries". HSBC forfeited \$1.256bn and agreed to enter into a five-year Deferred Prosecution Agreement (DPA). This meant that if HSBC failed to comply with the agreement in any way they could be prosecuted.

In October 2017 the British Broadcasting Corporation (BBC, 2017) reported that the UK financial regulators and the Serious Fraud Office were to review allegations that HSBC and Standard Chartered were linked to a South African corruption scandal. Despite this, HSBC's Press Office (2017) reported that the DPA had expired and that as HSBC had "lived up to all of its commitments" it would be seeking the dismissal of all charges.

Simultaneously, Withers (2017) reported that Lord Hain, in the House of Lords, had advised that a whistle-blower was assisting the FCA in their investigations. However, there has been no further information released from the regulator, in the House of Lords or in the national press.

iii. Interest Rate Hedging Products

The FCA (2016b) identified major failings in the selling of these complex products and ordered a total of nine banks to pay redress amounting to $\pounds 2.2$ bn and over $\pounds 500$ m in losses.

iv. Foreign Exchange Rates

In November 2014 the FCA announced they were fining five banks a total of £1.1 billion for manipulating the spot foreign exchange currency rates (FCA, 2014b).

In January 2018 HSBC entered a further Deferred Prosecution Agreement with the Ministry of Justice and was fined over \$100 million for currency rigging (The United States Department of Justice, 2018).

1.2.3 Bank Runs

Perhaps the most critical of events occurred in 2007. Prior to this it was believed that depositor insurance would always prevent a bank run (Diamond and Dybvig, 1983). BBC News (2007) reported the panic started when they announced on the 13 September 2007 that Northern Rock had asked for assistance from the Bank of England, in its Lender of Last Resort capacity.

On 14 September 2007 the Bank of England agreed to provide emergency funding at a penalty rate to Northern Rock following assurances from the FSA that the bank was not insolvent. However, the panic was perpetuated by media images of consumers queuing outside their branches and did not cease until an announcement from Alistair Darling, then Chancellor of the Exchequer, on 17 September 2007

confirming all deposits held by Northern Rock would be guaranteed (Edmonds, 2008).

On 1 October 2007 the Chancellor attempted to restore consumer confidence further by increasing the level of depositor protection obtainable under the Financial Services Compensation Scheme (FSCS) from 100% of the first £2,000 plus 90% of the next £33,000, to 100% of £35,000, per institution, per consumer (Hall, 2009, p.429).

There are different levels of compensation dependent on the claim category, as at August 2018, these are detailed in Table 1.1 below.

| Compensation Level | Date Effective From |
|---|---|
| £85,000 per person per firm (100%) The above figure is calculated on the exchange rate equivalent of €100,000 in EU countries to prevent regulatory arbitrage | From 30 January 2017 |
| £1M for temporary high balances (100%) | From 3 July 2015 |
| £50,000 per person per firm (100%) | From 1 January 2010 |
| £50,000 per person per firm (100%) | From 1 January 2010 |
| No upper limit Long-term, compulsory, professional, and certain claims for injury, sickness or infirmity (100%) Other types of claims (90%) | From 3 July 2015 |
| No upper limit (90%) Compulsory insurance (100%) | From 14 January 2005 |
| | Compensation Level£85,000 per person per firm(100%)The above figure is calculatedon the exchange rateequivalent of €100,000 in EUcountries to prevent regulatoryarbitrage£1M for temporary highbalances (100%)£50,000 per person per firm(100%)£50,000 per person per firm(100%)No upper limitLong-term, compulsory,professional, and certainclaims for injury, sickness orinfirmity (100%)Other types of claims (90%)No upper limit (90%)Compulsory insurance (100%) |

| Fable 1.1: Financial Services (| Compensation Scheme: | Compensation Levels |
|---------------------------------|-----------------------------|----------------------------|
|---------------------------------|-----------------------------|----------------------------|

(Source: Financial Services Compensation Scheme, 2018)

Investments, Home Finance, Insurance Business and General insurance advice and arranging compensation limits are not in place to protect against bank runs, however,

they are essential to maintain a level of trust and protection against bad advice or mis-selling.

1.3 Who was to Blame?

Arguments as to where the blame lay for these scandals were plentiful. 'Greedy bankers' was a popular theme in the national press, fuelled by coverage of extortionate bonus payments when the rest of the UK population struggled through the credit crunch. Black (2008, p.32) argued that although subprime lending was the 'trigger' of the Global Financial Crisis, the cause was the abundance of credit available to all. He goes on to say that if the regulators had stepped in it would have been tantamount to stopping the "party" and it is "the politicians and the public who should accept the blame for this. We all did it and benefited from it".

Somewhat ironically, the regulator in authority during the above scandals was set up after a series of bank failures in the 1990s. Logan (2001, p.320) referred to the period as "the small banks' crisis...when failure was last widespread and the system faced a potentially systemic threat".

1.4 The Financial Services Authority

The FSA came into power as the sole regulator of the financial services sector in the UK in 2001 as a result of the Financial Services and Markets Act (2000). Gordon Brown, the then Chancellor, later publicly admitted setting up the FSA to monitor individual institutions was a "big mistake". During a key note speech at the Institute for New Economic Thinking he explained, "We set up the FSA believing that the problem would come from the failure of an individual institution". He then went on to say that the understanding of what "global" meant was "incomplete" and misunderstood by almost everyone involved in regulation (Brown, 2011).

On 24 April 2012 Hector Sants made his last speech as CEO of the FSA. In this speech he outlined what he felt were the reasons the FSA had failed. Firstly, inadequate capital and liquidity standards of banks; secondly, many of those in key board positions, both executive and non-executive, lacked the technical skills to

manage the risks in their banks; thirdly, the culture of high rewards for short-term gain and lack of governance; and fourthly, the perception of high performing senior individuals being above the law. Rather unapologetically, there is no mention of the billions of pounds of on-going mis-selling scandals that have caused harm to millions of customers and small businesses. The banks were clearly, in his opinion, to blame for the financial crisis (Monaghan, 2012).

1.5 The Financial Conduct Authority

In 2013 the FSA was abolished and replaced with the FCA. The FCA work alongside the Prudential Regulatory Authority (PRA) to create a "twin peaks regulatory structure in the UK" (Competition and Markets Authority, 2015, p.5). The role of the PRA is to ensure authorised firms are managed appropriately and sufficiently robust financially, whilst the FCA monitors authorised firms' conduct, with particular focus on consumer protection and product intervention. It is hoped that the reduction in span of control and defined areas of regulatory control will prevent further failure to maintain systemic stability and protect consumers.

The FCAs regulatory objectives are:

- i. market confidence
- ii. financial stability
- iii. the protection of consumers
- iv. the reduction of financial crime

In June 2018, the FCA published 'The financial lives of consumers across the UK'. Their findings were that only 40% of UK adults are confident in the UK financial services industry, and only 31% feel that financial firms are honest and transparent.

1.6 Consumer Confidence

Prior to the aforementioned FCA research, Ernst and Young published its third Global Consumer Banking Survey, in 2014. Over 32,000 participants from 43 countries were surveyed, including 767 from the UK. The Ernst and Young press release title '*Two years of declining confidence in UK banking draws to a close*'

perhaps suggests a more positive impression than the data itself. Although almost double the number of participants in 2013 stated that their confidence had increased since 2011 and 2012, this was only 11% of participants, with 52% saying their level of confidence had not changed and 37% saying their confidence had decreased; hence, still a long way to go in terms of meeting the FCA objective of maintaining consumer confidence in the UK financial system. Interestingly, the UK consumer has double the level of distrust in their banks than their counterparts in the United States, Germany, France and Switzerland. Ernst and Young found that the key driver of mistrust in the UK was the national press (67%), whilst the key driver of trust was how the individual consumer was treated by their bank (68%). However, the survey found that trust does not prevent the price-sensitive UK consumer from switching, with 66% of consumers who had switched in the past 12 months citing rates and fees as the reason for doing so.

It is clear that the events of the last decade, such as Payment Protection Insurance and interest-rate hedging product mis-selling scandals, LIBOR-fixing, money laundering and foreign exchange rate fixing have compounded distrust of the industry, the culmination of a bank run has elevated this distrust to suspicion of corruption, destroying confidence in the institutions and those in place to regulate them. Trust and consumer confidence, and why they are so important in financial services will be discussed in more detail in Section 2.4.4.3.

The purpose of this research is to measure specifically the confidence of consumers in the areas which regulation seeks to address, fundamentally the prevention of events such as those mentioned above.

1.7 Research Aim and Objectives

Considering the above issues, and acknowledging the critical role that financial institutions contribute to the health and development of the economy, the aim of this thesis is to answer the question, 'Are Scottish consumers confident in retail banking regulation?' Based on the answer to this question, the implications of the levels of trust afforded by Scottish consumers on the banking sector, wider financial services

sector, and ultimately the economy, will be assessed. The June 2018 FCA Report, 'The financial lives of consumers across the UK' found the differences in levels of confidence between regions to be marginal. Therefore, it could be argued that results from a Scottish study can be more widely applied to the UK.

The specific research objectives are:

- i. Undertake a critical review of the existing theory of financial regulation literature, from the perspective of consumer confidence in retail banking, and identify the areas that retail banking regulation seeks to address;
- ii. Contextualise the theory and develop a set of key research questions through a three-stage survey design with theorists, practitioners and end-users; from theoretical model to empirical testing, to identify relevant theoretical and practical issues;
- iii. Create and conduct a Consumer Survey which measures levels of consumer confidence in the areas that retail banking regulation seeks to address;
- iv. Analyse and evaluate the data to determine the measure of overall levels of confidence, levels of confidence in each of the areas that regulation seeks to address, and differences in levels of confidence between demographic groups, such as age, income and education;
- v. Present the findings and contributions of this study to theory and practical implications for the regulator, banks, consumer groups and government, and make recommendations for future research.

1.8 Research Methods

The study will involve the following research methods:

- i. Secondary data collection through a review of the literature in terms of the theory of banking regulation and applicable economic theories;
- ii. Primary data collection will take a mixed method approach of a three-stage survey design and a survey of over 400 Scottish banking customers.

The survey will be empirically tested with three groups: the first with theorists, the second with practitioners, and the third with end-users.

1.9 The Importance of the Research

The research will provide an insight into the levels of consumer confidence in each of the rationales that regulation seeks to address. As far as the researcher is aware this is the first study that has been conducted to obtain this data. The research has identified the arguments for regulating the banking industry and, specifically, the reasons why retail consumers in particular must be protected. Motivations for regulating the industry are mainly driven by the need to protect against market failures, such as externalities, asymmetric information, moral hazard, and additionally non-economic, social rationales for regulation. Maintaining the confidence levels of consumers is paramount to the safety and stability of the financial system and is a key objective of the UK regulator. The researcher has therefore elected to measure the levels of confidence in each of the areas that regulation seeks to address to establish if regulation is 'fit for purpose' in this regard. Secondly, the researcher has used established demographics, such as age, gender, income level and education level to determine if there are differences in the levels of confidence between the groups.

The methodology adopted in this research has sought to achieve a 360 degree view of the problem by including theorists and practitioners in the initial research stages to ensure the consumer survey questions were a true and accurate interpretation of each of the rationales for regulation.

The importance and value of the potential findings and recommendations from this research are mainly twofold. Firstly, this adds a new dimension to the academic literature on regulation and consumer protection in financial services. The confidence levels of retail banking consumers has been broken down into each specific area that regulation seeks to address, and additionally compares the demographic characteristics of those consumers and how this affects their levels of confidence. Secondly, this adds to practice by improving knowledge of consumers

within particular demographic profiles. This has implications for both the regulator and commercial banks in the industry, which may treat consumers differently based on their profiles and specific needs.

1.10 Structure of the Thesis

The thesis is organised into six chapters. The first chapter has introduced the background of mis-selling and institutional failures which brought about the reform of the UK financial sector regulator, and described the research aims, objectives and methods.

The second chapter is a review of the literature in terms of the key theories and concepts. Each theme will be explored by tracing the theory back to its roots to understand how it has developed over time to become accepted practice. This chapter is split into two distinct sections. The first section contextualises the research and the second section provides the theoretical framework.

The third chapter provides an outline of the research methodology and methods employed by the researcher. The relationship between the research question, the philosophical assumptions and the techniques used to conduct the research are demonstrated. Additionally, there is an overview of the preliminary research, including changes made to the Consumer Survey following examination of the feedback obtained during the three-stage empirical testing with theorists, practitioners and end-users. Further, the data collected in the final study is described, including the characteristics and demographics of the sample population and the validity of the sampling approach is discussed.

The fourth chapter presents the descriptive statistics and correlation analysis. The data from the Consumer Survey theoretical questions is analysed, focusing on specific characteristics of the participants.

The fifth chapter presents a stepwise multiple regression analysis to identify the combination of demographic variables that would be used to predict the levels of confidence in the rationales for regulation.

Claire Lynne McCafferty

The sixth, and final, chapter concludes the thesis and provides the development of the research topic, suggestions for future work, summary, and contribution.
CHAPTER 2 LITERATURE REVIEW

2.1 Introduction

The first chapter introduced the series of events, such as mis-selling and institutional failures, which brought about the reform of the UK financial sector. In this chapter, an analytical overview of the key literature and development of regulation will be undertaken. From this, the theories identified will inform the research method employed to answer the question, 'Are Scottish consumers confident in retail banking regulation?'

This question can only be answered by examining the trilateral relationship between the banks, the consumers and the regulator(s). This chapter is therefore split into three distinct themes. The first is the retail banks, why they exist and their structure and culture, and how this impacts consumer confidence; the second is the consumer and the role of consumer protection in banking; and the last is regulation and the structure and role of regulatory bodies in maintaining consumer confidence. Each of these themes is split into sub-themes that explore the applicable economic theories.

2.2 Retail Banks

The definition of a retail bank will first be examined and how they have evolved over time in the UK market. The structure and culture of these institutions will be explored and how these factors influence the treatment of consumers and impact consumer confidence in the effectiveness of regulation.

2.2.1 What is a Retail Bank?

The definition of a bank has changed considerably over the past few decades as there has been rapid change within the sector. The most up-to-date definition provided in the FCA Handbook (2018a) is:

a. A firm with a Part 4A permission, which includes accepting deposits

and:

Claire Lynne McCafferty

- i. which is a credit institution; or
- whose Part 4A permission includes a requirement that it comply with the rules in GENPRU and BIPRU relating to bank; but which is not a building society, a friendly society or a credit union;
- b. an EEA bank which is a full credit institution.

Part 4A permission refers to the permission granted by the FCA, or PRA under Part 4A of the Financial Services and Markets Act (2000). The Financial Services Act (2012) amended the 2000 Act and implemented the new regulatory framework by replacing the FSA with the FCA and PRA.

2.2.2 The Evolution of the Retail Bank Market in the UK

UK financial institutions underwent significant structural change during the 1970s and 1980s (Bowen, Hoggarth and Pain, 1999) as a result of deregulation and a changing market environment (Bank of England, 1991). The move from a fragmented system, with a large number of providers with clear differentiation in the products and services provided, to a smaller number of large multinationals was facilitated by "the natural economic drivers of economies of scale and scope, interacting with demand-side drivers are financial deregulation" Davies, Richardson, Katinaite and Manning (2010, p.330). "The bank branch network has been declining steadily for nearly 30 years. There were 20,583 branches in 1988 but only 8,837 in 2012" (Edmonds, 2018). Further consolidation of the sector followed the Global Financial Crisis. The growth in the size of firms has resulted in a move away from relationship banking to transactional banking (Llewellyn, 2014).

2.2.3 Retail Bank Structure (Business Models)

In the UK there is dominance in terms of the ownership structure of financial firms, that is, the Shareholder Value Model (SVM). Llewellyn (2014) identified that this lack of diversity is the root cause of a multitude of problems faced by the industry over

Claire Lynne McCafferty

the past two decades. In particular, this has led to a reduction in consumer choice and effective competition. In 2002 the Competition Commission (later replaced by the Competition and Markets Authority) revealed findings of an inquiry into banking services provisions to small and medium-sized enterprises (which fall within the retail banking remit). They concluded the market was operating as a complex monopoly which had an adverse effect on choice and information, and increased consumer detriment through exploitive practices such as excessive pricing (Competition Commission, 2002). The Centre for Policy Studies retaliated with a report which claimed that it was the government's deluge of regulations which had 'obstructed innovation' and 'impeded competition' (Lodge, 2002).

Llewellyn (2014) posits that there are clear economic, systemic and welfare benefits to a more diversified financial system. Mutual firms (which he calls the Stakeholder Value Model), such as Building Societies, focus on their members and as a result of addressing agency problems more efficiently may be more suited to providing financial products with longer-term contractual relationships such as mortgages and life assurance.

The separation of ownership and control in the SVM also means that there is a lack of personal responsibility for decisions (Llewellyn, 2014). Further, the business objectives of SVM focus on short-term rate of return on capital. This gives rise to the creation of incentive structures and a culture that emphasises bonuses based on sales and maximising shareholder wealth (Llewellyn, 2014).

2.2.4 Retail Bank Culture

Spicer et al., (2014) undertook a significant study of the culture in UK retail banking. The motivation for the study was the lack of progress in culture change, compared to other areas such as capital ratios, remuneration, competition and ring fencing. They noted that the financial crisis had not been enough to incite cultural change and it was only after the LIBOR scandal that culture came more into focus. The first of their main conclusions was that an aggressive sales culture, which was reinforced through reward, was to blame for widespread bad practices and bank failure. During

their research, Spicer et al., (2014) found that in some cases the 'dropped sales targets' had simply be rebranded as 'expectations', 'customer needs' or 'customer outcomes'. In the worst cases a return of the 'sales target whiteboard' had been witnessed.

Second, culture change is the responsibility of the banks themselves and not of the regulator. The Banking Standards Review Council (BSRC) has been set up to support the process of cultural change. However, to avoid what Stigler named 'regulatory capture' and "maintain its public credibility and independence, its senior positions must not be solely filled by secondees from banks, nor can there be allowed to develop a revolving door between banks, audit firms and the BSRC" (p.11). Spicer et al., (2014) add that senior leaders in the banks must take responsibility for embedding the new culture and changing behaviour as a priority. However, only by the BSRC assessing the front-line staff can they be sure that the message is not 'lost in the middle'.

Third, the research indicated that more occurrences were likely to be uncovered. This is supported by the findings that in many cases banks were aware of problems for a number of years before they attempted to address them. This is another example of Llewellyn's 'grid lock' as the banks' justification for not doing anything sooner was that everyone else was doing exactly the same. Grid lock occurs when firms know how they should be behaving but adopt hazardous strategies as there are short-run advantages, which they may also believe their competitors are adopting (Llewellyn, 1998).

Fourth, the banks are trying to change their cultures and become more 'customer centric', and all have programmes in place. However, the research revealed that customer outcomes and metrics can be 'gamed' easily. The examples given in the report are:

- i. closing 'easy-win' complaints quickly so they remain non-reportable and
- ii. a satisfied customer at point of sale, does not necessarily mean a satisfied customer later

The illustration used in the report is a loan customer who is happy to get the loan but later cannot afford the repayments when interest rates rise.

Last, the old culture has become so engrained staff are wary and/or so overloaded by change that Spicer et al., (2014) predict it will take a whole generation to accomplish any significant change.

In 2011, the CEOs and Chairmen of five of the UK's biggest banks signed up to a new ethical code of conduct aimed at improving the way they do business. They committed to treat customers and colleagues with respect and integrity and act fairly, paying close attention to risks (Goff, 2011). The next section discusses ethics in organisations.

2.2.5 Organisational Ethics

Jones and George (2008, p.150) define organisational ethics as "the guiding practices and beliefs through which a particular company and its managers view their responsibility toward their stakeholders".

Although it must be acknowledged that there are hundreds of conflicting definitions of the stakeholder (Miles, 2012), it is generally accepted that a consumer can be defined as a stakeholder (see Donaldson and Preston, 1995; Cullum, 1997; Crane and Matten, 2007; Jones and George, 2008). An early definition of stakeholders from Stanford (1963) as cited by Freeman (2010, p.31), "those groups without whose support the organisation would cease to exist", is particularly fitting since banks would not exist without consumers. For the purposes of this thesis, organisational ethics will be investigated in the context of the consumer and the protection of this stakeholder within financial services.

Donaldson and Preston (1995) identified three distinct aspects of stakeholder management theory:

- i. Descriptive/empirical
- ii. Instrumental
- iii. Normative

Claire Lynne McCafferty

Descriptive stakeholder theory is, as the name suggests, used to describe and explain different aspects of the firm and managerial behaviour, such as, how the firm operates, what the managers think about how they manage (Brenner and Molander, 1977, as cited in Donaldson and Preston, 1995), how the firm is actually managed (Clarkson, 1991; Halal, 1990; Kreiner and Bhambri, 1991, as cited in Donaldson and Preston, 1995), and how the interests of the various stakeholders are considered by the board (Wang and Dewhirst, 1992, as cited in Donaldson and Preston, 1995). In short, what does the firm do to manage the interests of their stakeholders?

Instrumental stakeholder theory is used to determine whether taking stakeholders interests into account actually benefits the firm and positively contributes to the successful achievement of corporate objectives. Briefly, the firm takes stakeholders interests into account because it is good for the bottom line.

Normative stakeholder theory considers the moral and philosophical reasons why a firm should take their stakeholders interests into account. In summary, the firm takes stakeholders interests into account because it is the right thing to do. Donaldson and Preston, (1995), find that essentially there is value in using all three approaches, which are complementary, with normative theory being central to the theory.

The stakeholder value model, or mutual model, discussed by Llewellyn (2014) implies that a firm will not pursue financial objectives above all others. Christen, Rosenberg and Jayadeva (2004) previously developed what they describe as 'double bottom line' institutions. That is, they pursue social objectives in addition to financial objectives. They called these institutions 'alternative financial institutions' (AFIs) who serve the needs of consumers who have difficulties accessing microfinance financial institutions, particularly in developing or transitioning economies. In the UK, Credit Unions are a type of financial cooperative who offer savings and loans provisions to those who may otherwise be excluded from the high street. Credit Unions are not-for-profit and have been endorsed by Ethical Consumer magazine as providing truly ethical current accounts (ABCUL, 2012).

There are questions about who is accountable and responsible for a firm's ethical, or otherwise, decisions. Particularly since no executives were prosecuted for the behaviour that led to the 2007/08 crisis. Overall, a strong ethical culture is considered essential to make the banks less vulnerable to misconduct (Vigeo-Eiris, 2017).

2.2.5.1 The Corporation as a Moral Agent

French (1979, p.207) provided the foundations of a theory in which he proposed "corporations can be full-fledged moral persons and have whatever privileges, rights and duties as are, in the normal course of affairs, accorded to moral persons". French (1979, 1984) argued that corporations are both legal and moral persons and therefore are moral agents. According to French, what makes the organisation a moral person is its ability to perform acts separate to any individuals associated with the organisation, and for these acts to have intentions which are also separate to the intentions of the individuals associated with the organisation. Additionally, French (1995) adds to 'intentionality' the ability to make rational decisions and arguments, and the ability to respond to and change behaviour that is detrimental to others and their own interests.

Green (1989) argued that companies do have ethical responsibility and are not protected by limited liability from the consequences of their actions. Sheppard (1994) reiterated the concept that corporations are moral persons. He proposed that corporate culture can be viewed as the organisation's personality and that the board of directors, who provide the emotional capacity to be a moral person, are ultimately responsible for the corporate moral person.

However, after some criticism (Donaldson, 1982; Werhane 1985), French (2005, p.577) later revisited his original work and conceded that organisations are not moral persons as they "fail to make it as intentional agents". He explains, if intentions are understood in terms of desires and beliefs, then organisational policy cannot be understood as intentions.

In more recent work, Hess (2013) argued that corporations are moral agents but not moral persons. He defined a moral agent as having moral obligations and a moral person as being entitled to specific rights and protections. His argument was based on whether corporations should be entitled to rights and protections from the perspective of four different paradigms: legal, political, cultural and business ethics. From a legal position Hess (2013) argued that corporations are recognised as persons for practical reasons only; such as entering contracts, rather than to afford them any rights and protections. From a political standpoint he argues that whilst corporations can be termed rational agents (capable of entering into contracts), are entitled to certain political privileges, rights and duties, this does not make them a moral person with moral privileges, rights and duties. Thirdly, what is culturally accepted as a person; the emotions and feelings experienced, and the 'vulnerabilities' people are susceptible to, he argues, is what separates them from corporations. The fourth and final paradigm relates to business ethics and is derived from French's (1979) seminal theory on the moral person, which he himself later retracted, leading Hess (2013, p.330) to describe it as a "red herring".

Hess (2013) concludes by reasoning that corporations are moral *agents* in that they can and should be expected to act in accordance with their moral obligations, however, they are not moral *persons* entitled to rights and protections as they cannot experience physical and emotional pain and suffering as a result of their vulnerabilities, and therefore, there is no valid argument to grant them rights and protections.

Since the financial crisis there have been reforms within the industry to increase the accountability of the board of financial institutions. The Senior Managers and Certification Regime (SM&CR) introduced a statutory duty of responsibility in March 2017.

2.2.5.2 Ethical Frameworks

There are certain ethical frameworks that can be applied when making ethical business decisions. Jones and George, (2008) present four ethical frameworks for use in a business context:

- i. utilitarian rule
- ii. moral rights rule
- iii. justice rule
- iv. practical rule

An ethical decision made under the utilitarian rule should produce the "greatest good for the greatest number of people" (Jones and George, 2008, p.139). The philosophy of utilitarianism is linked back to "British philosophers and economists Jeremy Bentham (1748-1832) and John Stuart Mill (1806-1873) and has been influential in modern economics in general" (Crane and Matten, 2007, p.94). Utilitarianism is based on the principle of utility; the value of actions. This is a variable which is commonly used in economics and therefore complements its numerical approach. As an ethical decision under the utilitarian approach weighs 'good' against 'bad', Crane and Matten, (2007) relate it to "cost-benefit analysis". In a corporate situation, when management are faced with an ethical dilemma their decision should result in the greatest amount of good for their stakeholders. A shortcoming of the utilitarian approach can be demonstrated when the common good and individual rights clash. Jones and George, (2008) acknowledge that in capitalist societies, such as the UK, the rights of the shareholders will be favoured above the rights of other stakeholders in the case of cost-cutting exercises. This is evidenced by the announcements off-shoring of 4,000 UK jobs by HSBC to Asia in 2003 (Cave, 2003) and 9,000 job losses at Lloyds in 2014 as the business moves towards digital rather than face-to-face banking (Lawson, 2014). Their ethical justification in these situations was had they not undertaken these cost-cutting exercises they would become uncompetitive and may face closure, resulting in greater harm to all stakeholders.

According to Jones and George (2008), moral rights rule is concerned with protecting the rights of the people who will be affected by the decision. "The British philosopher John Locke (1632-1714)...conceptualised the notion of natural rights" (Crane and Matten, 2007, p.100). When faced with a dilemma, managers need to consider the rights of each group of stakeholders and how these would be affected by different decision outcomes. They should choose an outcome that protects the rights of all stakeholders. This can present difficulties when a decision has positive effects for one group and has negative effects for another. As with the utilitarian approach, one of its shortcomings is the clash between individual rights and the common good. Natural rights are often referred to as human rights today and a breach of these in the UK would be breaking the law. HSBC came under fire in 2014 for financing companies such as Wilmar International and Sime Darby which have questionable human rights abuse records (Green Century Funds, 2014). So, although not directly breaching human rights laws, it could be perceived that they are advocating these companies by financing and holding shares in them. It would not be surprising to find that consumers are confused as to how this could be, since HSBC promotes itself as a leading supporter in the campaign against climate change and states that "sustainability underpins our strategic priorities" (HSBC, 2017).

The justice rule is concerned with ensuring that ethical decisions are made fairly and impartially; and therefore address the clash between individual rights and the common good. In the situation of a bank, management would have to ensure that the costs and benefits are fairly distributed to all stakeholders relative to their input into the organisation. Traditionally, their stakeholders can be identified as:

- i. Shareholders
- ii. Customers
- iii. Employees
- iv. Suppliers (Crane and Matten, 2007).

When looking at how a bank distributes its increased profits from the types of costcutting exercises mentioned above, it is clear that shareholders benefit through dividends received and employees through bonuses received. What is not so clear

Claire Lynne McCafferty

Page 27

is how these cost savings are transferred to consumers as some would argue that customer service standards are falling (Which?, 2019) and costs of running a bank account are rising via service charge and penalties. In reality, lower bank borrowing costs have not been passed on to consumers in the form of lower overdraft costs (Rovnick, 2016).

Jones and George, (2008) suggest that a fourth rule has been introduced, the practical rule, in response to the conflict in stakeholder interests the three rules above present. Three questions must be positively satisfied in order for a business decision to be deemed ethical. Firstly, have today's accepted business "standards" and "values" been met? Secondly, do you feel comfortable that your decision, and how it has "affected" individuals and groups, will be "communicated" to the public and press? Lastly, will friends, family and associates with which you have a "significant" connection "approve"? For example, when bank managers decided to pressurise their sales teams into selling PPI to every customer who opened a personal loan or credit card (whether they needed the product or not) could they justify this decision to their families and would they approve? This approach can be used to determine the practical effects and outcomes of any decisions, however, reliance on others perceptions may lead to poor judgement.

The FCA Principles for Businesses (see Table 2.1 overleaf) are founded on ethics. Language used such as, "integrity" (Principle 1), "due care and due diligence" (Principle 2), "treat [customers] fairly" (Principle 6), "clear, fair and not misleading" (Principle 7), and "manage conflicts of interest fairly" (Principle 8) set the foundations of how financial organisations should be applying ethics in their day-to-day processes and procedures and dealings with consumers.

| 1. Integrity | A firm must conduct its business with integrity |
|--|---|
| 2. Skill, care and due diligence | A firm must conduct its business with due skill, care and diligence |
| Management and control | A <i>firm</i> must take reasonable care to organise and control its affairs responsibly and effectively, with adequate risk management systems |
| 4. Financial prudence | A firm must maintain adequate financial resources |
| 5. Market Conduct | A firm must observe proper standards of market conduct |
| 6. Customers' interests | A <i>firm</i> must pay due regard to the interests of its <i>customers</i> and treat them fairly |
| 7. Communications with clients | A <i>firm</i> must pay due regard to the information needs of its <i>clients</i> , and communicate with information to them in a way that is clear, fair and not misleading |
| 8. Conflicts of interest | A <i>firm</i> must manage conflicts of interest fairly, both between itself and its customers and between a <i>customer</i> and another <i>client</i> |
| 9. Customers: relationships of trust | A <i>firm</i> must take reasonable care to ensure the suitability of its advice and discretionary decisions for any customer who is entitled to rely upon its judgement |
| 10. Clients' assets | A <i>firm</i> must arrange adequate protection for <i>clients</i> ' assets when it is responsible for them |
| 11. Relations with regulators | A <i>firm</i> must deal with its regulators in an open and cooperative way, and must disclose to the <i>appropriate regulator</i> appropriately anything relating to the <i>firm</i> of which that regulator would reasonably expect notice |

Table 2.1 FCA Principles for Business

(Source: Financial Conduct Authority, 2018)

In 2013, Black and Anderson produced a paper which examined how an ethical framework could be created for the financial services industry. Firstly, the suggestion of more legislation was dismissed on the grounds that more rules replace and reduce ownership of responsible decision making and "devalues ethics as it then becomes a question about what is required, rather than what is just and the right thing to do" (Black and Anderson, 2013, p.3). Jackman (2012) describes the reduction in the need for firms to think for themselves as "a kind of laundering of conscience; if it's legal, it must be all right…and often focuses attention on minimum standards". He goes on to say there is a "need to create space for ethics – space in the sense of deliberate 'gaps' in regulation". The Black and Anderson (2013) paper acknowledges ethical duties have long since been part of the regulatory rules, dating back to 1988 (with the passing of The Bank of England Act, 1998), and suggested that the problem is more interpretation and application by financial firms and appropriate supervision by the regulator.

Secondly, it was suggested remuneration structures and performance incentives which drive behaviour could include ethical targets. Each firm's performance against these objectives could be used to benchmark strong and weak ethical behaviour which could be used to gain competitive advantage. Presently, firms and individuals who are found to be conducting bad ethical practice are subject to public rebuke, however, there is no such recognition for good behaviour.

In 2013, O'Loughlin reported that HSBC was joining Barclays and the Cooperative Bank in removing sales targets from frontline staff and instead focusing on customer needs. This followed an investigation by the FSA which demonstrated that of the incentive schemes in place at the 22 firms examined, most were likely to lead staff to mis-sell to meet targets and gain bonuses (O'Loughlin, 2013).

Thirdly, it was suggested that an ethical group could be created which would selfregulate the industry as a whole. BBC News (2014) reported on the establishment of The Banking Standards Review Council (BSRC), an independent body set up to report on the behaviour and competence of member banks. It is fully funded by the industry but membership is not mandatory, which led to criticism and headlines such

as "Ethics panel for banks on the brink" due to the lack of commitment from several investment banks and none from foreign banks at the time (Wallace, 2014). However, to date (2019), 34 UK, foreign, and investment banks, and building societies have joined the renamed Banking Standards Board (BSB) which was launched with a new board in May 2015 (BSB, 2015).

Webley and Werner, (2008) found that although most large corporations had ethical codes of conduct there was a gap between the values and principles conveyed in the code and the attitudes and behaviour of the firm. Their research revealed this gap was due to ineffective ethics programmes and deficient corporate culture. However, McKinney, Emerson and Neubert, (2010) found that firms which had a written code of ethics tended to exhibit more ethical responses to common business dilemmas. The 'Big 4' banks all subscribe to a code under various titles. HSBC's is called 'Our Values'; Barclays' is called 'Purpose and Values'; Lloyds are called 'Ethics and Values'; and RBS' are called 'Our Code'.

Painter-Morland (2010) argued that through the process of formulating corporate ethical codes they (the codes) become institutionalised, losing moral responsiveness and undermining ethical responsiveness. Essentially, imposing control mechanisms and rules was counter-intuitive to informing ethical behaviour. Painter-Morland (2010) suggested 'self-questioning' which would "allow the moral conscience to be spurred into action" (2010, p.273). Within RBS' code is a 'Yes check list' (see Figure 2.1 overleaf). This exercise encourages employees to actively engage in a process which enables them to make ethical decisions; this is a step ahead of their competitors and addresses the dilemma highlighted by Painter-Morland (2010).

Figure 2.1: RBS 'Yes' Checks

Ask yourself...

1 Does what I am doing keep our customers and the bank safe and secure? Consider the impact of what you are doing. Rehearse a briefing with your boss.

2 Would customers and colleagues say I am acting with integrity? Consider: would I do this to someone in my family or a friend? Would I do it to myself?

3 Am I happy with how this would be perceived on the outside? Consider the impact of this in the outside world. Try writing the press release – does it sound good for customers?

4 Is what I am doing meeting the standards of conduct required? Think. If you are unsure then seek a second opinion.

5 In 5 years' time would others see this as a good way to work? Will this have a positive impact? Imagine writing it on your CV.

(Source: RBS, This is Our Code, 2013)

Fourthly, if agreement on ethical conduct could be reached, which Black and Anderson, (2003) concede would be difficult, a form of 'scenario analysis and stress-testing' could be conducted; much in the same way that the banking system's capital adequacy is currently tested. The purpose of these exercises would be to identify weaknesses and develop strategies to deal with them, consequently raising the profile of ethics in banking and opening the channel of communication between the regulator and regulated concerning these matters.

Lastly, Black and Anderson, (2013) suggest, with hesitation, the regulators presence within financial organisations may help to overcome ethical issues as and when they arise. However, it is accepted that this strategy would conflict with the FCA approach of reducing contact with the regulated. Additionally, they highlight the possible risk

of regulatory capture. To date, there is no UK banking industry wide ethical code of practice. In the next section, the consumer and the role of consumer protection in banking will be explored.

2.3 Retail Bank Consumers

In this second theme, the definition of retail bank consumer will be considered from the perspective of the industry regulator, and therefore the legal definition. Next, the regulatory bodies and laws which are in place to protect the interests of the consumer will be introduced. Lastly, the reasons why consumers demand regulation will be explored.

2.3.1 Who is a Retail Bank Consumer?

Although the terms consumer and customer are often used interchangeably, the economic definition is generally accepted as the former being defined as the individual who *consumes* a good or service and the latter as the individual who *purchases* a good or service. Generally, in retail banking both are performed simultaneously making it hard to differentiate. Cartwright (2004) regards consumers of financial services as both private bank customers and private investors and are not restricted to their economic definitions. Therefore, for the purpose of this thesis the terms consumer and customer should be treated as 'one and the same'.

The FCA's consumer mandate has been expanded under section 404E of the Financial Services and Markets Act (2000) and covers the following:

Persons using, contemplating using or those who have rights or interests in services provided by an authorised person/regulated firm, either for themselves or on another's behalf. The services include: acceptance of deposits, consumer credit, investment activity, payment services, electronic money and ancillary services (The National Archives, 2018).

2.3.2 Consumer Protection in Retail Banking

What is consumer protection? In the context of retail banking, it is the legislation that is in place to regulate banks and protect the interests of the consumer. The Financial Services Act (2012) outlines the FCA's statutory strategic and operational objectives. The first of the operational objectives, and the only one considered for the purposes of this thesis, is the consumer protection objective:

Figure 2.2: FCA Consumer Protection Objective

(1) The consumer protection objective is: securing an appropriate degree of protection for consumers.

(2) In considering what degree of protection for consumers may be appropriate, the FCA must have regard to—

(a) the differing degrees of risk involved in different kinds of investment or other transaction;

(b) the differing degrees of experience and expertise that different consumers may have;

(c) the needs that consumers may have for the timely provision of information and advice that is accurate and fit for purpose;

(d) the general principle that consumers should take responsibility for their decisions;

(e) the general principle that those providing regulated financial services should be expected to provide consumers with a level of care that is appropriate having regard to the degree of risk involved in relation to the investment or other transaction and the capabilities of the consumers in question;

(f) the differing expectations that consumers may have in relation to different kinds of investment or other transaction;

(g) any information which the consumer financial education body has provided to the FCA in the exercise of the consumer financial education function;

(h) any information which the scheme operator of the ombudsman scheme has provided to the FCA pursuant to section 232A.

(Source: The National Archives, 2018)

In a 2012 FSA report introducing the FCA, Wheatley said of the FCA's new approach, "We will step in earlier, and act faster, when we identify problems that risk harming consumers or the integrity of the market" (FSA, 2012a, p.8). The new regulator intends to take a more proactive approach, however it is important to note that consumers are not protected against all risks or every eventuality, and are responsible for making their own decisions. This is aligned with the views of Llewellyn (1998, p. 315) who stated that "the objectives of regulation should be limited to correcting identified market imperfections, failures and externalities". Further discussion of the rationales for regulation is in Section 2.6.

2.3.3 Consumer Protection Laws

The Consumer Rights Act (CRA) 2015 consolidated eight different general consumer law legislations into one. The Act covers unfair terms, the supply of goods and services, digital content, enforcement powers, and competition law. The FCA is an enforcer under the CRA and has the right to challenge any consumer credit provider. The FCA took over this responsibility from the Office of Fair Trading (OFT) whose services terminated in 2014. From a consumer perspective, complaints do not have to be pursued through a court of law; instead they can be referred to the Financial Ombudsman Service (FOS) if an unsatisfactory response is received from their bank. This is a free service for consumers, subsidised by the retail banks. There is further discussion of FOS powers in Section 2.4.2.

One of the roles of both the OFT and the Competition Commission was to scrutinise the Financial Services Authority rules and practices in the interests of competition and consumer protection (Wu and Lee, 2001). Consumer rights responsibilities were passed to the FCA, and anti-competitive behaviour became the remit of the Competition and Markets Authority (CMA) in 2014. It could be argued that by passing all consumer protection powers to the FCA it has diluted the rights of the consumer, particularly since it was the OFT who campaigned so vigorously for consumers in the PPI mis-selling scandal, insisting the then regulator, FSA, investigate. Conversely, Green and Perry (2016) argue that the new regime is much

stricter and highlight the significant steps the FCA has taken to ensure the fair treatment of consumers by payday lenders and debt management companies.

2.3.4 Consumer Demand for Regulation

There are several reasons why consumers demand regulation of the banking industry. Llewellyn and the FSA (1999) identified that it stems from consumers' need to feel assured. He argues that if consumers do not feel confident enough to purchase suitable financial products it has the same detrimental effect as being missold unsuitable products. For example, protection and retirement products may be avoided due to their long-term nature resulting in a welfare loss (Llewellyn, 2014). He further explains that assurance is sought due to the very nature of financial products: complexity; cost; duration of consumption; and also due to the recent past experiences which have resulted in a reduction of trust. Ring (2012) found that some consumer knowledge or capability is important in assessing whether an adviser can be trusted, and where there are low levels of knowledge trust becomes most important; however is most lacking. The case is made for setting minimum standards, which all regulated firms must meet, to increase consumer confidence. Of course the consumer may be under the impression that regulation is free and for that reason, Llewellyn and the FSA (1999) suggest making them aware there is a cost to avoid demand distortion. Consumer demand as a rationale for the regulation of financial services is discussed in more detail in the next section.

2.4 Regulation

This theme is split into two distinct sections; context and theory. The first section provides a definition of regulation, followed by an analysis of the evolution of regulation of the industry. The development of a 'twin peaks' approach and current regulatory structure of the UK financial services sector is then examined, including the most significant new development of ring fencing.

The second half of this theme will address the theoretical framework, which evaluates the rationales for regulation such as demand from the public, market failures and economic and social justifications. The research questions have been

Claire Lynne McCafferty

developed from this framework and the arguments against regulation are also appraised to ensure a comprehensive evaluation.

2.4.1 What is Regulation?

According to the Oxford Dictionary of Finance and Banking (2014), it is the imposition by a government of controls over the decisions of individuals or firms. It often refers to the control of industries in which there is monopoly or oligopoly; in order to prevent firms from exploiting their market power at the expense of the public...regulation is especially important in the financial services industry.

Certainly the UK financial services sector could be argued to be operating an oligopoly, with four major players, Barclays plc., Lloyds Banking Group plc., the Royal Bank of Scotland Group plc. and HSBC Holdings plc., collectively accounting for approximately 70% of the personal current account market (Competition and Markets Authority, 2015). However, if the purpose of regulation is to protect the public against exploitation from these powerful institutions, then in what form is this exploitation taking place and what is/can the regulator do to mitigate this?

Den Hertog (2010, p.223) defines regulation as "the employment of legal instruments for the implementation of social-economic policy objectives". He identified two traditional economic theories of regulation; 'public interest theories of regulation' and 'private interest theories of regulation'. Public interest theories assume that the regulators pursue the public interest and have sufficient information and enforcement powers to do so. Private interest theories assume that the regulators have imperfect information, and therefore can only imperfectly (if at all) promote the public interest. Additionally, some of the economic agents may pursue their own interests, at the cost of the public interest. In essence, the regulators have the same motivations as those in the private sector, that is, self-interest. Private interest theory

is the public sector equivalent of the principal-agent problem which is examined further in Section 2.6.3.3.

Harnay and Scialom (2016) argue that the economic theories of regulation have played a significant role in moulding banking regulation policy since the 1960s. They suggest that the public interest view of regulation is strongly connected with welfare economics. For example, by promoting public interest and increasing social welfare through correcting market failures such as externalities and natural monopolies (discussed further in Section 2.6.3.1). They assert that by the 1970s there was much criticism towards this theory of regulation and a move towards the private interest view of regulation where it was believed that regulation was actually stifling business and paved the way towards deregulation.

2.4.2 The Evolution of the Regulation of UK Financial Services

Prior to The Banking Act 1979 there was no regulation of the UK financial services sector. The introduction of regulation was intended to protect depositors, partially in response to the secondary banking crisis of the 1970s and primarily due to the need to meet the requirements of the Directive 77/780/EEC (Cartwright, 2004). However, the near collapse of Johnson Matthey Bank in 1984 exposed deficiencies in the Act, particularly the two-tier system of supervision of recognised banks and deposit takers (Cartwright, 2004). This led to the introduction of The Banking Act 1987 which was intended to strengthen the BoE supervisory powers and replaced the 1979 Act.

The Financial Services Act (1986) established a series of self-regulatory organisations for different types of financial institutions and firms under the overall supervision of the Securities and Investments Board (SIB) (The Association of Futures Brokers and Dealers, the Financial Intermediaries, Managers and Brokers Regulatory Association, the Investment Management Regulatory Organisation, the Life Assurance and Unit Trust Regulatory Organisation, and The Securities Association). The deregulation of the financial services sector led to increased competition and an overhaul of the products and services provided by financial institutions. Traditional banks were now able to offer mortgages, investments,

general insurance, pensions and life assurance and no longer acted solely as deposit takers. However, this fragmented approach of having 11 separate bodies led to both overlap and under-lap issues of regulatory supervision.

Yet it was not long until the shortcomings of the Banking Act 1987 were realised. In 1991, the Bank of Credit and Commerce International (BCCI) was forcibly shut down by the Bank of England, owing billions around the world (Hemraj, 2005). The collapse "raised serious questions about the effectiveness of bank regulation in an era of international finance" (Adams and Frantz, 1992, p.9). BCCI was accused of being a corrupt criminal organisation; banking dictators, drug dealers and arms merchants. This led to its closure in a number of jurisdictions due to fraud, with the Bank of England being heavily criticised for failing in its oversight of BCCI.

In 1997 self-regulation was replaced, after several mis-selling scandals and a series of bank failures during that decade, with an independent regulatory body and SIB became known as the Financial Services Authority (FSA). The FSA was granted statutory power by the Financial Services and Markets Act (2000). When this Act came into force in December 2001 the FSA became the single regulatory body for the UK, responsible for banking, insurance and investment business. This formed part of a tripartite regulatory system in which the BoE, the FSA and HM Treasury had collective responsibility for financial stability.

The FSA was responsible for the systemic, prudential and conduct of business regulation for the retail and wholesale markets. It was believed that having one body to oversee the whole market would lead to greater efficiencies. Goodhart, Hartmann, Llewellyn, Rojas-Suarez, and Weisbrod, (1998) suggested these efficiencies could be realised through a shared infrastructure which would reduce duplication, overlap, gaps and inconsistencies; thereby reducing costs. Additionally, it was believed that a more simplified structure would lead to greater regulator accountability and better understanding and recognition by both financial firms and consumers. Furthermore, a single regulator would be able to monitor a firm's full range of activities and there would be no conflict in the different objectives used to manage this. Goodhart et al., (1998) did concede that there were limitations, such as problems of coordination due *Claire Lynne McCafferty PhD April 2020 Page 39*

to the need to create divisions to deal with the different aspects of regulation and different types of financial institution within the single regulator.

The Financial Services and Markets Act (2000) gave the FSA four statutory objectives:

- i. Market (maintaining confidence in the UK financial system).
- ii. Public Awareness (promoting public understanding of the financial system).
- iii. Consumer Protection (securing the appropriate degree of protection for consumers).
- iv. Financial Crime Reduction (reducing the possibility of regulated businesses to be used for purposes connected with financial crime).

The Public Awareness objective was later replaced by Financial Stability (contributing to the protection and enhancement of stability in the UK financial system) by the Financial Services Act 2010.

In the FSA's first Annual Report, 2001/02, the regulator set out what they hoped to achieve as the new single supervisory power. In terms of consumer protection, the FSA vowed to "help consumers achieve a fair deal" (FSA Annual Report 2001/02, p.13). The Handbook of rules and guidance was created which amalgamated the rulebooks of 10 previous regulators and introduced new policies covering money laundering, complaint handling and market abuse. Section 4 of the Annual Report was dedicated to 'a fair deal for consumers'. This section of the report detailed how the FSA undertook a Treating Customers Fairly (TCF) project which examined the after-sales experience. Their key findings suggested that consumers:

- i. did not always fully understand the information they were given about certain products
- ii. were not kept fully informed post-sale
- iii. expectations were not met by either the product and/or firm
- iv. switching between product and firm was discouraged
- v. complaints were not always dealt with fairly.

The FSA took significant steps to guide firms to improve on these shortcomings by highlighting what they considered to be "good and bad practice" (FSA Annual Report 2001/02, p.34) covering fairness in marketing and advertising, disclosure at point of sale, and complaint handling. Products such as annuities, with-profits policies and endowments were reviewed and new disclosure rules introduced to ensure consumers understood they had wider options than those that their provider may offer. The Financial Ombudsman Service (FOS) was established as an independent body to deal with complaints that were not satisfactorily resolved between the consumer and firm. The Financial Services Compensation Scheme was also set up to provide compensation to consumers where a regulated firm is unable to meet its liabilities. Additionally, the FSA took over some of the Unfair Terms in Consumer Contracts Regulations from the OFT. The pensions review undertaken by the FSA required firms to compensate consumers for historic mis-selling. In its first year the FSA seemed to be taking significant steps to safeguard and protect consumers from detriment. Baldwin, Cave and Lodge (2012, p.19) assert that "making information" more extensively available, accurate, and affordable" encourages a strong competitive market and reduces the market failures of imperfect and asymmetric information.

A year later the FSA reported that "38% of consumers now know of the FSA" (FSA Annual Report 2002/03, p.14). They hailed this as "significant" progress and attributed this to its considerable work with consumer bodies such at the Citizens Advice Bureau (CAB). The FSA was now turning its attention to educating young people, increasing financial literacy in adults and alerting consumers to high-risk products and fraudulent firms and individuals. The consumer had a part to play in taking responsibility for their finances and making informed choices. The FSA introduced product comparison tables to encourage consumers to shop around and competition between providers.

From 2003/04 until 2008/09 the FSA Annual Report contained a section dedicated to 'helping retail consumers achieve a fair deal'. Each report contained key themes which were publicised each year.

Claire Lynne McCafferty

PhD April 2020

In the 2003/04 publication the FSA shared their belief that better education is the key to consumers becoming more capable of making better decisions. They suggest that asymmetric information hinders efficient markets. This may be the case; nevertheless, should consumers be expected to possess the same level of knowledge about complex financial products as financial services professionals? Just how much knowledge is *enough* knowledge is not really clear. The FSA set out plans to improve consumer financial capability with advice from the Financial Capability Steering Group. Additionally, the FSA made an announcement about reforming the regulation of the retail market, moving from a polarised system to a de-polarised system. This meant that the restrictions on the range of products and services that an adviser could sell were removed resulting in increased choice for consumers.

It appears slightly contradictory to provide consumers with more choice when at the same time it is suspected they are not sufficiently proficient to make suitable choices. The significance of this is addressed in Section 2.6.4.1.

In 2004/05 the FSA publicised the work they had commenced with partners, the Personal Finance Research Centre at the University of Bristol, to undertake a survey to measure financial capability in the UK. Additionally, they announced the introduction of 'Key Facts' documents which were to be provided to consumers of mortgage and/or investment products as a result of the eradication of polarisation. These documents would provide consumers with simple, clear and user-friendly information about the product and any associated risks and costs. Depolarisation was intended to improve consumer choice by introducing a 'third way'. Previously financial advisers could either be independent or not. Now, advisers could be multitied, meaning that they could offer a range of products from a limited range of providers. However, the introduction of more choice for consumers did not coincide with the development of a financial education strategy or wider advice networks, which was viewed as a missed opportunity which may have led to interim problems (Ring, 2004). The FSA launched a newly designed consumer website which they reported experienced an increase of 20% in average monthly visits and an update

on their TCF initiative. The FSA identified that firms were not taking TCF into account when developing products or when designing sales staff reward and remuneration schemes

In the 2005/06 publication the FSA focused on the outcome of their survey, 'Financial Capability in the UK; Establishing a Baseline, March 2006'. Their findings revealed four key themes:

- i. there is a significant lack of planning ahead for retirement in general by UK consumers
- ii. the moderately small number of people affected by debt are acutely affected and those that are just managing would encounter financial strain if there was a change in their personal circumstances or a downturn in the economy as a whole; as people are not putting money away for a 'rainy day'
- iii. people are not choosing products which are suitable for their needs
- iv. the age range most affected by these finding are those in the 18-40 year old bracket.

This younger generation did not appear to be as financially competent as the over 40s. The FSA believed that their seven-point programme, 'Financial Capability in the UK, Delivering Change' could address these issues by reaching over 10 million people in five years.

The FSA also provided an update on their TCF initiative and 'encouragement' they had given the industry to adopt TCF as part of their organisations culture and values.

In the 2006/07 publication the FSA announced its intention to review the retail investment market, called the Retail Distribution Review (RDR). The intentions of this review were to understand the root causes of the problems in the retail markets for investment products. The FSA estimated that their financial capability programme had reached approximately one million people; although the outcome of this is not clear from the publication. With regards to their TCF initiative the FSA published a number of examples of good and bad practice and announced fines against nine firms for breaches in the selling of PPI policies.

Claire Lynne McCafferty

In 2007/08 the FSA publicised updates on their TCF initiative, the RDR and their financial capability programme. The FSA reported that there was evidence that senior management in most companies were taking TCF seriously and working towards embedding it within their organisation's culture. This was demonstrated through improvements in a few product specific areas such as general insurance and mortgage disclosures. However, there was insufficient evidence of any improvement to customer outcomes for retail customers in general. The RDR had now been underway for a couple of years and the FSA had produced an interim report after working with both small and large firms for the industry and consumer groups. The FSA reported three major findings:

- i. the need for a clear separation between advice and sales
- ii. the need to raise professional standards
- iii. the need to remove product provider influence on remuneration.

With regards to the FSA's progress on their financial capability agenda the five-year 'Delivering Change' strategy was now at the half-way point. The FSA reported the success of the agenda stating that the programme had reached over three million people through schools, universities, youth work organisations, workplaces, health authorities and charities. The target was to reach ten million by 2011. It was between the publication of this Annual Report and the next that the financial crisis took hold in the UK.

In the 2008/09 Annual Report the FSA acknowledge the on-going financial crisis and the difficulties experienced by consumers over the previous year. The FSA response to this was an increased focus on conduct regulation whilst continuing their work on TCF, consumer education and the RDR. The FSA fined many firms for failing to meet the March 2008 deadline to have suitable measures in place to test whether they were treating customers fairly and providing sufficient evidence of embedding TCF within their organisation's culture (the financial penalties incurred due to TCF implementation failure have not been quantified in the report). In response to the findings of the RDR (mentioned in the previous annual report) the FSA proposed the following:

Claire Lynne McCafferty

- i. product providers will no longer be able to offer commission to adviser's for selling their products
- ii. a clear definition of independent and non-independent advice
- iii. an increase in professional standards through increased skills and knowledge, raised ethical and behavioural standards and the governance of standards of practice.

These changes would be expected to be implemented by firms by the end of 2012. The FSA also undertook reviews of the mortgage and pensions market resulting in fines, public censures, prohibitions of individuals and cancellation of firm's permissions for inadequate or unfair practices. Previously the Conduct of Business (COB) of Retail Banking had taken the form of self-regulation via the Banking Code, however, the FSA decided to bring it within its remit following its involvement in the test case between the OFT and several banks concerning unfair overdraft charges. Continuing with their work on consumer education and increasing consumers' ability to manage their own financial affairs, the FSA had now reached over 6.5 million people through schools, universities, youth work organisations, workplaces, health authorities and charities. The FSA also published statistics following some of these initiatives which demonstrated a marked improvement in the confidence of these people to deal with money matters of up to 81% in some participant groups (FSA, 2009a).

In the 2009/10 publication of the Annual Report the 'Helping retail consumers achieve a fair deal' section is replaced with a section entitled 'Delivering consumer protection and education'. The FSA had launched their new consumer protection strategy which aimed to achieve the following:

- i. make the retail market work better for consumers
- ii. avoid the crystallisation of conduct risks that exceed our risk tolerance
- iii. deliver credible deterrence and prompt and effective redress for consumers.

The FSA acknowledged that to achieve these objectives they would have to become more 'proactive' and 'intensive' in their supervisory approach. In addition to increased engagement with firms the FSA also introduced metrics to quantify the effect of their work, these included:

- i. redress paid by firms
- ii. a consumer awareness survey
- iii. complaints made to the Financial Ombudsman.

During 2009 the FSA took action to ensure 'financial promotions and accompanying literature is fair, clear and not misleading' by forcing firms to amend or withdraw 450 financial advertisements after receiving complaints. However, much of the work in this area was undertaken in retrospect; Lehman Brothers-backed products were found to have 'serious deficiencies' in their literature and sales advice.

As the FSA had taken over the COB for retail banking on 1 November 2009 it was now responsible for deposit taking and payment services but announced that their consumer education objective would now be taken over by a new consumer education body following the passing of the Financial Services Act 2010.

In 2010/11 the FSA dropped the 'education' element and focused on 'delivering consumer protection'. The Mortgage Market Review (MMR) tackled irresponsible lending at the same time as setting out how a firm should deal with a consumer in financial difficulties and/or arrears. To some degree these could be partly viewed as a cause and effect of the financial crisis. The Financial Services Compensation Scheme (FSCS) compensation limit was increased to £85,000 per person, per authorised firm and extended to all European Economic Area (EEA) member states at the end of 2010 (agreed by all members at €100,000).

In the 2011/12 publication of the FSA annual report, in addition to the MMR and RDR, the FSA had undertaken a review of packaged bank accounts. They found

cases where the consumer was ineligible or unable to use some of the packaged products, for example insurance products, causing detriment. As a result, the FSA published new rules on the sale of general insurance contracts as part of a packaged bank account. Another significant development was the removal of the two stage complaint handling process. This meant that consumers could go straight to the FOS if they were unhappy with the first response they received from a financial firm when making a complaint. The FSA also announced an increase in the FOS compensation limit to £150,000.

The final annual report published by the FSA was in 2012/13. On the 31 December 2012 the RDR came into force, the main obligations being a higher professional standard, which included professional qualifications and the removal of commission from providers for advice. The creation of the Money Advice Service, launched in 2011, acknowledged the importance of 'guidance' in the new post-RDR landscape (Ring, 2016). In November 2012 Card Protection Plan (CPP) was fined £10.5M for widespread mis-selling of their products through financial firms which was the latest in a long line of product mis-selling scandals.

In January 2013 the FSA published its final guidance on 'Risks to customers from financial incentives' (FSA, 2013). In this report they begin by emphasising the need for consumer trust and confidence in financial services. The main findings of the investigation were that current staff incentives were likely to encourage mis-selling and banks were doing little to mitigate that risk. Indeed, the mis-selling of PPI and packaged bank accounts was primarily due to the 'points' accumulated by frontline staff when they sold these types of products. Conversely, selling a Basic Bank Account (which is not a money-maker for a bank) would accumulate few or no points towards a monthly or annual bonus. The FSA also reported that it welcomed the changes already implemented by some of the banks. Specifically, the changing incentive schemes so they are not based on sales volume, capping the maximum bonus, closer monitoring of high performing sales staff, identifying poor behaviour during face-to-face sales, stronger governance around signing-off incentive

Claire Lynne McCafferty

PhD April 2020

schemes and reviewing past sales for possible systemic mis-selling incidents (FSA, 2013).

It is clear that the FSA carried out substantial work in an attempt to improve the marketplace for consumers and secure better protection for them. However, it was heavily criticised for failing to prevent the events that led to the Global Financial Crisis. The expectations of a single regulator leading to greater efficiencies, lower costs and improved monitoring of firms appeared to have failed. In reality, the FSA had grown in cost and size due to increased activity for FOS and the FSCS in the wake of mis-selling scandals such as PPI and unfair bank charges complaints. In 2010, the Guardian reported that the then Chancellor of the Exchequer, George Osborne, said of the FSA, "at the heart of the crisis was a rapid and unsustainable increase in debt that our macroeconomic and regulatory system utterly failed to identify let alone prevent" (Elliott, 2010). Indeed, the FSA's Turner Review (2009b) concluded that the FSA had concentrated its efforts on the supervision of individual institutions and was lacking in attention to 'wider sectoral and system-wide risks'.

The FSA admitted its failings over Northern Rock in a report published in March 2008. Crucially, the regulator did not probe for a more comprehensive analysis of the risks inherent in the business model and failed to engage at an appropriate level for a high impact firm (FSA, 2008a). The regulator conceded that it must increase its focus on prudential supervision, including liquidity and stress testing.

This reignited the debate for a' twin peaks' approach to financial services regulation with Mervyn King, the then Bank of England governor, championing the approach, stating that prudential regulation, and consumer protection and market conduct must be separated as they require both different skills and approach (Taylor, 2009).

In The Tripartite Review, Sassoon (2009) details five possible options for discussion for a new regulatory model. The first suggestion was that the FSA should retain its current responsibilities but undergo a structural change away from focusing on Retail

and Wholesale divisions to Prudential and Conduct of Business divisions, with particular attention to financial stability issues. The second proposal was to implement the suggestions above, plus grant new supervisory powers to the Bank of England in addition to its new macro-prudential role. This would create a tripartite power, with the Treasury stepping in to resolve any differences between the FSA and the Bank of England. The third proposition was to abolish the FSA and replace it with two regulators, each with a clearly defined purpose and focus. One would be responsible for Prudential Regulation and the other for Conduct of Business regulation. The fourth recommendation was to combine proposals two and three with the Bank of England as the head of micro-prudential regulation in exceptional circumstances. The fifth, and final, suggestion was for the micro-prudential regulator to be merged with and become the joint responsibility of the Bank of England.

The Tripartite Review recommended the ways in which the three regulatory powers, the Bank of England, the FSA and HM Treasury could strengthen their capabilities in delivering their proposed regulatory roles (Sassoon, 2009). The Bank of England's main remit would be enhancing financial stability; the FSA would focus on financial crime and consumer education, with HM Treasury playing a pivotal role in setting the direction of the UK's economic policy.

The third option proposed by Sassoon was preferred, creating two new regulatory bodies (a twin peaks approach); one would be responsible for Prudential Regulation and the other for Conduct of Business regulation.

Cartwright (2004, p.5-6) defines prudential regulation as being "concerned with the solvency, safety and soundness of banks...in relation to consumer protection...even if there is no question of systemic risk" and conduct of business regulation is concerned with "how banks conduct business with their customers".

2.4.3 Twin Peaks

This was not the first time such a regulatory structure had been proposed. Thephrase 'twin peak' was first coined by Michael Taylor in his 1995 paper for the CentreClaire Lynne McCaffertyPhD April 2020Page 49

for the Study of Financial Innovation (CSFI). This publication was calling for reform of the regulatory system pre-FSA. Taylor (1995) argued that the twin peaks approach to regulation would:

- i. eradicate duplication and over-lap
- ii. create clear and precise objectives and focus for each supervisor
- iii. encourage conflict resolution between the objectives of regulation
- iv. encourage an 'open', 'transparent' and 'accountable' regulatory process.

His model proposed a Financial Stability Commission who would take responsibility for the prudential supervision of the system, ensuring it was sound; banks held sufficient capital and managed their risks appropriately (systemic protection). The second peak, the Consumer Protection Commission, would ensure responsibility for conduct of business regulation; ensuring consumers receive a fair deal (consumer protection).

The Deputy Governor of the Bank of England, Howard Davies, considered the arguments for a single or two separate regulatory structures in a speech in 1997. Davies was opposed to both a single regulator and the twin peaks approach due to the costs and 'upheaval' associated with regulatory reform. Davies argued that banks were unique and one regulator overseeing many different types of financial services firms would not be effective as it would need to 'tailor the rules' to meet each type of business. Davies felt that financial firms would not accept two regulators and that there would be a gap created as it was unclear where the regulation of markets fitted into Taylor's proposal (Davies, 1997). Despite these reservations, the FSA was granted statutory power by the Financial Services and Markets Act (2000). When this Act came into force in December 2001 the FSA became the single regulatory body for the UK.

Following the Global Financial Crisis, Taylor (2009) published a further paper advocating the twin-peaks structure; emphasising that prudential and consumer protection goals should indeed be pursued separately. In 2012 this reform finally began to take shape.

2.4.4 Current Regulatory Structure of the UK Financial Services Sector

The Financial Services Act (2012) implemented a new regulatory framework for financial services in the UK. The FCA took over as the regulator alongside the Prudential Regulatory Authority (PRA) to create a twin peaks approach.

A Memorandum of Understanding between the FCA and PRA set out their separate and independent mandates and their relationship to one another. The PRA is part of the Bank of England and is responsible for the prudential regulation of banks, building societies, insurance firms, investment firms and credit unions, it aims to promote 'safety and soundness' and 'protect policyholders'. The PRA's main focus is the harm that firms can cause to the stability of the UK financial system. The FCA is accountable to both HM Treasury and Parliament and is responsible for the conduct regulation of retail, wholesale and financial markets. The FCA is also responsible for the prudential regulation of firms which are outwith the PRA's scope. The aims of the FCA are threefold:

- i. to secure the appropriate degree of protection for consumers
- ii. to protect and enhance the integrity of the UK financial system
- iii. to promote effective competition

Both the PRA and FCA aim to have a forward-looking, proactive and judgementbased approach to their areas of regulation.

In a speech to the British Bankers Association in early 2012, two months before the launch of twin peaks, Hector Sants spoke of 'independent but coordinated regulation' when he set out his vision for the model (Russell, 2012). The two regulators will have different objectives against which they will make their own separate judgements. They will also engage with firms separately but will share and exchange information and data internally ensuring that regulatory data is only collected once. However, firms could find that they are asked similar questions from the two supervisors as they will be gathering information for different purposes. Sants suggested that meeting objectives will be less challenging due to the greater clarity

provided by the Financial Services Act 2010. In addition to a new regulatory framework, there have also been structural changes for the banks.

2.4.5 Ring Fencing

Schwarz (2013, p.82) defines ring fencing that is required by government regulation as, "legally deconstructing a firm to reallocate and reduce risk more optimally, such as by protecting the firm's assets and operations and minimising its internal and affiliate risks".

Perhaps the most significant transformation of the sector since the FCA came into power is the legal deconstruction of core retail deposits from investment banking. The Independent Commission on Banking (ICB) produced its *Final Report* (widely known as the 'Vickers Report') in September 2011 making several recommendations to enhance financial stability and competition; including loss-absorbency and ring fencing.

The ICB (2011) set out three objectives the retail ring fence should achieve 'at the lowest possible cost to the economy'. The first is to make it easier to resolve troubles faced by both ring fenced and non-ring fenced banks, without the need for the taxpayer-funded government support. The second is to protect vital banking services from problems elsewhere in the financial system. The third is to limit the level of government guarantees, which should in turn reduce the risks taken by the banks. To achieve all three of these objectives the ICB (2011) set out five principles which can be summarised as follows. Firstly, only ring fenced banks will be given the regulatory power to provide mandated services (i.e. the taking of deposits from, and the provision of overdrafts to, individuals and small and medium-sized organisations). Secondly, ring fenced banks are prohibited from providing certain services which would expose them to global financial markets and the risks beyond the provision of payments services and financial intermediation. Thirdly, a ring fenced bank's ancillary services are limited to those which are essential to carry out their non-prohibited activities, and there are limits placed on wholesale funding and exposures to non-ring fenced banks and non-bank financial companies. The fourth

and fifth principles relate to operation of a corporate body, where fourthly, the ring fenced and non-ring fenced banks should operate as separate legal entities, and lastly, the ring fenced bank must conduct its business with other entities in the group on a third party basis.

The Financial Services (Banking Reform) Act 2013 was enacted on 18 December 2013 and legislated for a process for transfer of business known as a ring fencing transfer scheme (RFTS). This enabled firms to restructure their businesses to comply with the ring fencing requirements that came into effect on 1 January 2019 (Bank of England, 2019). The largest firms (specifically those with deposits of £25bn or more) were required to fulfil this, and completed at various stages throughout 2018: Barclays, April 2018; Lloyds, May 2018; HSBC, July 2018 Santander, July 2018; and RBS, August 2018. Schwarz (2013) suggests that ring fencing can be used as a regulatory tool to correct market failures and protect against systemic risk. The rationales for regulation are discussed in more detail in Section 2.6.

2.4.6 Economics for Effective Regulation (EFER)

In March 2016 the FCA published Occasional Paper 13, '*Economics for Effective Regulation*' (EFER), which set out a new approach to economic analysis of financial services (FCA, 2016a). In this paper they identified five key market imperfections which could lead to poor outcomes for consumers:

- i. Information asymmetries
- ii. Externalities
- iii. Market power
- iv. Behavioural distortions
- v. Regulatory failure

The first three categories have traditionally been the focus of economic frameworksfor public policy.Information asymmetries are explored in Section 2.6.3.2,externalities in Section 2.6.3.1 and market power was discussed in Section 2.4.1 ofthis thesis.Consideration of behavioural distortions, however, is a relatively newapproach adopted very early on from the inception of the FCA.The use ofClaire Lynne McCaffertyPhD April 2020Page 53
behavioural economics in financial regulation is discussed in greater detail in Section 2.6.4.1.

The EFER paper (FCA, 2016a) identified 11 systematic drivers of poor market outcomes which are detailed in the Table 2.2 below.

| Driver Group | Specific patterns observed |
|--|--|
| Consumer behaviour | 1. Appropriate information about products is not available or not used by customers |
| | 2. Difficulty in comparing products and services, and their value (e.g. due to complexity or bundling) |
| | 3. Behavioural or rational inertia in taking appropriate action (e.g. switching) |
| | 4. Errors by consumers in assessing own long-term needs |
| Supply-side behaviour and market structure | 5. Unrecognised conflicts of interest between provider and their clients, leading to misaligned incentives |
| | 6. Large providers in the market face little or no competitive discipline from rivals (market power) |
| | 7. Barriers to entry and expansion arising from market structure or strategic conduct |
| | 8. Providers coordinating activities in an anti-competitive way |
| | 9. Restrictive agreements or integration between firms at different levels of the supply chain |
| Other market distortions | 10. Market participants act without considering side effects on markets or society |
| | 11. Unintended consequences of past interventions (e.g. perverse incentives or outdated regulation) |

Table 2.2: Systematic Drivers of Poor Market Outcomes

(Source: FCA, Economics for Effective Regulation (EFER), 2016)

The drivers are used by the FCA as a practical tool that helps to structure the analysis. The process of analysis in EFER for designing effective interventions can be broadly separated into three stages:

- i. Problem diagnosis
- ii. Intervention design
- iii. Impact assessment

In diagnosing the root cause of problems the FCA identify the relevant markets, describe the facts about the market, identify the ways in which the market is not working well, assess the nature of harm caused, and evaluate the rationale for intervening. In designing effective interventions the FCA state the goals of intervention based on problem diagnosis, identify alternatives for addressing individual goals, eliminate unfeasible or highly inefficient alternative options, and combine individual actions into packages. In assessing the impact of interventions the FCA define the baseline, state the direct effects on an intervention, consider how market participants will respond, summarise the effects on market outcomes, and assess and, where appropriate, quantify the costs and benefits.

This methodology involves a much more complex analysis and for that reason the FCA would not adopt this approach in every situation. The economic analysis adopted will depend on the specific regulatory problem and are likely to be proportionate only for major interventions (FCA, 2016a). The specific rationales for adopting such a detailed approach will be outlined in Section 2.6. This will bring together the context and practice covered above with the theoretical framework below. First, the arguments against regulation of the sector are considered.

2.5 Arguments Against Regulation

There are several arguments against government intervention into the financial sector, such as inefficiency, costs of compliance, ineffectiveness, circumvention, impeding innovation and capture, to name a few.

First, however, a paper in which Benston (2000) appears to critically respond to the rationales set out by Llewellyn in his 1995 paper, 'Regulation of Retail Investment Services' will be considered.

On consumer protection as a justification for regulation, Benston (2000) sets out very clear arguments against the rationales he has found in the literature. The six consumer protection justifications he identified were:

- i. to maintain consumer confidence in the financial system and financial services
- ii. assure that a financial-services supplier on which consumers rely does not fail
- iii. assure that consumers receive sufficient information to make good decisions and are dealt with fairly
- iv. assure fair prices for financial services
- v. protect consumers from fraud and misrepresentation
- vi. protect invidious discrimination against individuals.

Each of these justifications and the reasons why he rejects them will now be discussed.

Benston (2000) argues that bank and payments systems failures can be eradicated with higher capital requirements. For this reason, he asserts intervention to maintain confidence is not necessary, with the exception of this and deposit insurance. He also argues that holding sufficient capital is the only appropriate regulatory requirement needed to provide assurance that a financial services supplier, on which consumers depend, does not fail. Holding sufficient capital would absorb losses, thereby avoiding insolvency. In a later publication, Benink and Benston (2005) reject the argument that intervention is required to protect against systemic risk since, they argue, retail banks *should*, without direction, hold sufficient capital requirements are necessary to maintain a stable and secure financial system, although he does not agree that there is a necessity to enforce it through regulation. The first

hypothesis that will be tested in the consumer survey is that the consumer is confident the retail banking system is stable and secure.

In the case of Lender of Last Resort, Benston (2003) argues that should one bank fail, it would simply be taken over by another bank and consumers would not notice any difference. Yet, this was not the case with Northern Rock. Despite four expressions of interest (from Virgin Money, J C Flowers, existing Northern Rock management, and a Group led by a former Abbey National chief executive) it was decided that no bid was good value for the taxpayer and the company was nationalised (Milne and Wood, 2009) in February 2008. After nationalisation the bank was split into two parts: Northern Rock plc. and Northern Rock Asset Management (NRAM), which have been described as the 'good' and 'bad' parts of the bank (Osborne, 2011). It wasn't until 2012 that an agreement was reached for Virgin Money to purchase £465 million of mortgage assets from NRAM loans (Virgin Money, 2012). This demonstrates that it is not a straightforward or speedy process.

On consumer confidence in the financial system, Benston (2003) disagrees with the suggestion that one failed bank would cause a run on other banks. Indeed, the September 2007 bank run on Northern Rock did not spill-over and affect other solvent banks which could back his theory that consumers "can distinguish among sound and insolvent banks" (2003, p.33). However, Benston's theory was not entirely proven by the events of 2007 as his belief that consumers do not run when their funds are protected by deposit insurance was incorrect. Depositors of Northern Rock clearly did run, despite compensation levels of 100% of the first £2,000, and 90% between £2000 and £35,000 in place. The average balance held at Northern Rock was £17,500 at 31 December 2006 (Northern Rock Annual Report and Accounts, 2006). Hence, this would suggest, the existence of deposit insurance is not sufficient to maintain consumer confidence where the compensation levels are not deemed to be adequate – or perhaps understood. The second hypothesis to be tested in the consumer survey is how confident consumers are that their money is protected should their bank fail.

PhD April 2020

Clearly there is some level of agreement between Llewellyn and Benston that capital requirements and depositor insurance are necessary for the existence of a stable and secure system and consumer confidence that their deposits are safe. However, financial products, Benston (1998) argues, are no different to any other type of product. To illustrate his point, Benston asserts Llewellyn's 'list of imperfections' could also be applied to electronic goods or cars, where asymmetric information also exists in terms of the manufacturing process, and it may be some time before a fault develops. Nevertheless, some financial products are extremely complex and it could be decades (in the case of mortgages or pensions for example) before potentially harmful issues arise. Benston (2000) also uses the argument that financial products are no different to other products when he argues that consumers do not require regulation to ensure fair prices for financial services. The forces of supply and demand may work well in a perfect market. However, the current oligopoly, or as some argue complex monopoly (Competition Commission, 2002), would suggest that this argument is flawed. The same argument is used for fraud and discrimination, as is the existence of laws to protect all consumers in these areas. However cases of misrepresentation, such as endowment mortgages and payment protection insurance, would suggest that laws are deficient. Additionally, the fact that bank products and services are a basic requirement to function in society necessitates additional protection for the most vulnerable in society to ensure they are not financially excluded.

2.5.1 Hazards

As discussed previously, there are well argued and accepted rationale for the regulation of financial services; both economic and social. Banks are a 'special' case and inefficiencies and externalities such as asymmetric information and contagion are such justifications. However, the provision of financial services regulation gives rise to hazards.

Goodhart et al., (1998) identified the danger of over-regulation as one such hazard. Two causes were identified which explained this occurrence. Firstly, regulation is not supplied through a market mechanism, but rather is enforced through institutional *Claire Lynne McCafferty PhD April 2020 Page 58*

requirements, restrictions and guidelines. A major consequence of this is information loss, specifically how much regulation the consumer wants and needs and how much they are prepared to pay for it. Secondly, most consumers of financial services perceive regulation as a free good. However, this is not the case and there is a cost borne by the sector for regulation and compliance. Where the regulator pursues self-interest there could be 'excessive' regulation, which in turn may result in the cost of regulatory compliance outweighing the benefits. Moosa (2015) claims regulation should never be viewed in terms of cost only, but rather in terms of the costs and benefits. His line of reasoning is that, as good regulation pays off we should not worry about reducing the bonuses of the CEOs of banks.

Dowd (2009) points out that the existence of moral hazard is an underrated problem and it played a central role in the events leading up to the Global Financial Crisis. Dowd claims that the existence of deposit insurance is highly destabilising for the banking system. His argument is based on the premise that inefficient banks will go out of business as consumers will prefer the more prudent banks. Moosa (2015) opposes this view and suggests that inefficient bank theory is actually an example of bad regulation. Undeniably, the failures of Northern Rock and Royal Bank of Scotland were due to poor management decisions and excessive risk-taking encouraged by light-touch regulation. Further, claiming that regulation should be avoided as it impedes innovation is, in his view, counterintuitive. The use of derivatives and other 'synthetic' financial products actually played a major part in the Global Financial Crisis and he points to the use of these products to circumvent regulation.

Some underlying issues have been addressed through legislative or regulatory intervention. Moral hazard has been reduced through the introduction of alternative mechanisms for dealing with failing banks such as bail-ins. The Special Resolution Regime (SRR) is essentially a process for the Central Bank to step in and ensure that a failing bank does so in an orderly way, thus reducing systemic risk (Bank of England, 2019). To avoid a repeat of the bail-out of banks during the Global Financial Crisis, with what was fundamentally taxpayers' money; the Bank of

Claire Lynne McCafferty

PhD April 2020

England has put in place a mechanism that will allow banks to dissolve with the costs of failure falling on the shareholders and creditors, much like other corporations do, but taking into account the special role that they play in the economy. The resolution powers granted to the Bank of England enable them to stabilise the bank so that it's critical functions remain available to its customers and restructure the bank, including replacing the senior executives of the bank. Restructuring may involve selling functions of the banks to competitors or moving them to a bridge bank managed by the government. Another alternative is a bail-in. This is where the balance sheet is repaired by swapping debt for equity (bonds for shares), much like what may be carried out by a corporation facing financial difficulties. The last option available to the Bank of England is to dissolve the bank by transferring customer deposits to another bank, or refunding them through depositor insurance.

2.5.2 Regulatory Capture

The term 'regulatory capture' was coined by George Stigler in 1971 in his paper, 'The Theory of Economic Regulation'. Stigler begins with a powerful statement, "The state – the machinery and power of the state – is potential resource or threat to every industry in the society". Stigler highlighted the fact that the state can choose who to assist and who to harm through penalties, financial or otherwise, or incentives. Stigler (1971, p. 3-6) presented the argument that, "as a rule, regulation is acquired by the industry and is designed and operated primarily for its benefit". Stigler identified three policies where this can be evidenced, "control over entry by new rivals...substitutes and complements...and price-fixing". There has, in recent years, been an emergence of new players in the UK financial services sector, hailed 'Challenger Banks' due to the dominance of the big four within the industry. Lloyds Banking Group, Royal Bank of Scotland, Barclays and HSBC provided 77% of current accounts and 85% of business lending (BBC News, 2014) when the Competition and Markets Authority (CMA) was formed in 2014. The CMA was a merger between the Office of Fair Trading and the Monopolies and Mergers Commission; they launched a competition inquiry into the sector in November 2014, the results of which were published in February 2016. The main findings were that

the sector was not as innovative or competitive as it needs to be and older banks have advantages over new entrants and smaller banks (CMA, 2016). A major consequence of this is that consumers are not switching, even in cases where they could make significant savings.

Baker (2010, p.647) found that a "growing number of respected commentators now argue that regulatory capture of public agencies and public policy by leading banks was one of the main casual factors behind the financial crisis of 2007-2009". Further, he explains that this contributed to pro-cyclical behaviour (during financial booms there is an absence of public or political lobbying) and therefore it is the banks themselves, through industry lobbies, which have the greatest incentives to influence regulatory developments. In particular, he asserts the Basel II regulations that encouraged the use of banks' own internal risk models, and were not robust enough to prevent the Global Financial Crisis, almost completely lacked public scrutiny. Whilst Baker observed this pro-cyclical, or positive, correlation between financial booms and bank incentives to influence regulation, Lucca, Seru and Trebbi (2014) identified countercyclical, negative, net worker inflows from industry to regulator in good times. Their study was interested in the phenomenon of the 'revolving door' which is discussed further in Section 2.5.3 below.

2.5.3 The Revolving Door

Recent studies (Lucca, Seru and Trebbi, 2014; Shive and Forster, 2016), media commentary (Arauzo, 2013; Ginsberg, 2014; Gilbert, 2015; Vassalos, 2018) and government reports (Committee on Standards in Public Life, 2016) have uncovered a growing trend for movement back and forth between industry and regulator. In 2015 Wheatley, the then Chief Executive of the Financial Conduct Authority defended the FCA's revolving door into the private sector, stating that it is beneficial for the regulator to recruit from industry and for staff to take their knowledge into the sector (Norman, 2015). Two years later, Wheatley himself was recruited by a hedge fund based in Hong Kong (Denham, 2017). Wheatley had previously run the Securities and Futures Commission in Hong Kong where he introduced measures

to improve short selling transparency (Cash, 2017), before taking up the post with the UK regulator.

Lucca, Seru and Trebbi (2014) highlight some of the benefits of the revolving door, in particular the potential for regulators to employ higher ability workers. It is clear that the regulator would benefit from the practice knowledge of former industry workers, and, equally, banks would benefit from employing those who have a relevant understanding of compliance. An interesting finding from their research was the theory that regulators preferred to impose stricter and more intense regulations as this increased the likelihood that their skills would be required in industry and they would be highly rewarded in return. They call this 'regulatory schooling' a widely accepted theory in the studies of the revolving door between Washington and Wall Street. They conclude that limiting the mobility between the regulated and the regulators may have the negative impact of the reduction of attainment and retainment of talent.

Commentary in Europe tends to be less forgiving. Arauzo (2013) writes of the threat to public interest, where reform is being stifled and undermined, and causing repeated financial and social crises. Vassalos (2018) identified that one third of top positions at the Directorate-General for Financial Stability, Financial Services and Capital Markets Union (DG FISMA), in the period 2008-2017, were held by people who worked for the financial industry before or after leaving it. The DG FISMA is responsible for initiating and implementing EU policy in the area of Banking and Finance in Europe. He demonstrated that the overreliance of regulators on industry expertise leads to the crucial question of whether those appointed to serve the public interest have the neutrality necessary to fulfil this role, or whether they will naturally consider solutions to regulatory problems that are best for the industry. The Committee on Standards in Public Life (2016) made recommendations to the UK government to manage conflicts of interest when staff revolve in and out of the regulator, including the promotion of transparency and a corporate culture of integrity, however, there is no specific regulation to deal with this risk.

PhD April 2020

In summary, there is clearly a case against regulation. The debate about whether the current regulatory regime is effective is another matter. The rationales for regulating the retail banking market will be identified and will form the theoretical questions that will be used to measure consumer confidence in retail banking regulation in the UK. In the next Section, 2.6, the rationale for regulation will be explored.

2.6 Rationales for Regulation

Why is regulation especially important in the financial services industry? Some economists believe that market solutions are preferable to government solutions in the financial services market (Dowd, 1996; Benston and Kaufman, 1996; Stiglitz, 2010; Moosa and Ramiah, 2014), as discussed in the previous Section, 2.5. Now, the key rationales for regulation that have been identified in the literature will be discussed. These are: demand from the public; market failures; social justifications; and fraudulent activities.

2.6.1 Demand from the Public

The demand for regulation of financial services is linked to consumer confidence and trust (Llewellyn and the FSA, 1999). In fact, it could be argued to have a negative correlation, with demand rising as trust and confidence in the financial system falls. Public demand for regulation to correct market inefficiencies or inequities was developed initially by Pigou (1932) in Public Interest Theory. As discussed earlier, public interest theories assume that the regulators pursue the public interest and have sufficient information and enforcement powers to do so (Den Hertog, 2010). These theories had become unpopular by the 1970s as the UK government repealed their control and the UK moved towards the private interest view of regulation (Harnay and Scialom, 2016).

Llewellyn and the FSA (1999) highlight several reasons why consumers demand regulation. The first being lower transaction costs. Indeed, if each consumer had to investigate the safety and soundness of each institution there would be significant costs, both financial and time. Secondly, he highlights the issue of access to relevant

Claire Lynne McCafferty

information, and whether the consumer has the knowledge or skills to use the information they may be able to gather. Thirdly, Llewellyn suggests that consumers need assurance when carrying out transactions with financial firms, which links to his fourth reason, their past experience of bad behaviour by financial firms. Fifth, due to the nature of financial contracts, the value can be determined after it has been signed and the product purchased, by the firm's behaviour and solvency. Sixth, again related to the nature of contract and products, the consumer is often making a substantial commitment. Seventh, the old adage that prevention is better than cure. Llewellyn suggests that regulation will prevent 'hazardous behaviour' by firms, which is preferable to claiming redress afterwards. Eighth, the regulator can enjoy economies of scale in monitoring, and lastly, a part of the costs of regulation will be borne by the firms themselves. Of course the consumer may be under the impression that regulation is free and for that reason, Llewellyn and the FSA (1999) suggest making them aware there is a cost to avoid demand distortion.

Benston (2000) argues that the preservation of consumer confidence in banks and the banking system is not a valid reason to regulate banks. To put this into context, Benston is debating the need for regulation to protect the consumer against bank failure by regulating *products and services*. As discussed earlier, his theory is that deposit protection and capital requirements should be sufficient.

In addition to regulation being imposed to protect the public against monopolistic exploitation, Goodhart, et al., (1998) highlight two further reasons:

- i. smaller, retail clients are less informed and so must be protected
- ii. the need to safeguard systemic stability.

These are examples of 'market failures' which Baldwin et al, (2012, p.15) state justify the need for regulation as "the uncontrolled market place will, for some reason, fail to produce behaviour or results in accordance with the public interest". As discussed previously, this is due, in part, to the unequal bargaining power of banks and consumers in the current oligopoly market structure.

2.6.2 Market Failures

The view of Baldwin et al, (2012) matched that of Llewellyn (1998, p.315) who not only suggested they were grounds for regulation, but who also stated that "the objectives of regulation should be limited to correcting identified market imperfections, failures and externalities". The market failures highlighted in particular by Llewellyn were externalities and asymmetric information. In his 1999 publication for the FSA he classifies these as economic rationales for regulation. Cartwright (2004) identified that the rationales for regulation can be split into two principal categories, economic and social. It could be argued that consumer demand is in fact a social rationale for regulation since it is concerned with social welfare. On the other end of the spectrum, the Chicago economists, including George Stigler and Richard Posner amongst others, argued that regulation can be used to give benefits to particular individuals or groups. Stigler's (1971, p.3) theory was that regulation is "acquired by industry and is designed and operated primarily for its benefit". Posner (1974, p.344) argued that regulation can be viewed as a product whose allocation is determined by the basic principles of supply and demand. He goes on to explain, that when viewed in this way it can be accepted that, all other things being equal, regulation will be "supplied to those who value it the most". Their views are more aligned to the private interest theory of regulation, or regulatory capture, which was discussed in Section 2.5.2.

2.6.3 Economic Rationales for Regulation

The economic rationales for regulation are those which are fundamental to the economic well-being of the country. As discussed previously, the UK financial system plays a pivotal role in the economy. The commercial banks play the vital role of channelling funds between savers and borrowers and operating the payments system. Confidence in the banking system is critical to the health of households, businesses and the overall economy. Hence, government intervention in the financial services sector and the strict regulation of the banking industry is rationalised by the need to minimise the occurrence of events that could undermine the reputation of banks and threaten confidence in the system.

Claire Lynne McCafferty

2.6.3.1 Externalities

The primary external concern is that of systemic risk, "banks are subject to runs, which have contagion effects, and which can throw solvent banks into insolvency" (Goodhart et al., 1998, p.8-9). In today's global economy this is not geographically restricted. The problem cannot be contained at a national level but becomes a global crisis. As witnessed in the Global Financial Crisis, what began as a problem in the USA soon rippled across the world. There are mechanisms in place to deal with the issue of both macro and micro risks.

Macroprudential financial regulation seeks to safeguard the financial system as a whole by recognising the importance of general equilibrium effects (Hanson, Kashyap, and Stein, 2011). Since the Global Financial Crisis, the necessity to achieve global financial stability has seen macroprudential policy become central to the development of public policy (Kenc, 2016). Kenc (2016) has traced the term macroprudential back to unpublished documents by the Bank for International Settlements (BIS) and the BoE in the late 1970s. By 1986 the BIS was using the term when discussing policies which aimed to secure the safety and soundness of the payments mechanism and the financial system as a whole. The rationale for macroprudential regulation is the existence of externalities which can lead to systemic risk and ultimately market failure. The International Monetary Fund (IMF) classified these externalities into three categories:

- i. externalities related to strategic complementarities
- ii. externalities related to fire sales
- iii. externalities related to interconnectedness (De Nicolo, Favara and Ratnovski, 2012).

Strategic complementarities arise when banks expose themselves to the same types of risk through activities, such as reducing screening intensity and increasing lending activity during boom times, hiding losses or maintaining risky lending, and correlated asset choice. Fire sales can lead to the devaluation of similar assets held by other banks, reducing the capital ratio of those banks and leading to financial distress,

eventually triggering a credit crunch. The interconnectedness of banks can lead to issues of contagion, that is, the failure of one institution can have a spillover effect and lead to the failure of otherwise solvent banks. De Nicolo et al, (2012) emphasise that externalities arising from interconnectedness are particularly great in Systemically Important Financial Institutions (SIFIs). Further, they argue that the macroprudential tool that can address the externalities arising from all three categories is capital requirements.

Banks are required to hold capital reserves to enable them to absorb any unexpected losses. Both the UK and USA had minimum capital requirements in place as far back as 1981 (Bank for International Settlements, 1999). However, these requirements did not extend to non-domestic banks, giving foreign competitors an unfair advantage.

The collapse of German based Herstatt Bank in 1974 posed a systemic threat due to its heavy involvement in the foreign exchange markets (Schenk, 2014). The bank had losses (DM470 million reported, actual losses were much higher) of over 10 times what they held in capital and reserves (DM44 million). This led to discussions between the G10 central bankers (Belgium, Canada, France, Germany, Italy, Japan, the Netherlands, Sweden, the United Kingdom and the United States) to make international efforts to reduce such risks. The Basel Committee on Banking Supervision was launched in 1975.

Early in 1987, the UK and USA proposed an international agreement to impose riskbased capital standards, which by July 1988 had the agreement of the G10 central banks and bank regulators of Switzerland and Luxembourg (Wall, 1989). These new rules were to be effective from 1992 and focused on the credit risk of banks. The agreement became known as the Basel Capital Accord.

The bankruptcy of Barings Bank in 1995 highlighted weaknesses in the Accord. The Barings Bank collapse was mainly due to weaknesses in the bank's internal controls and management systems. Rogue trader, Nicholas Leeson, was the chief trader and floor manager for Barings Bank in Singapore, but, he was also the general

manager of the back office (Canac and Dykman, 2011). Being in control of both the front and back office functions allowed Leeson the opportunity to hide financial futures trade losses and simultaneously publicly record profits. By the time the losses of over £800 million (Stonham, 1996) were discovered they were so huge that they caused the bankruptcy of the over 200 year old merchant bank.

By 1996 there were proposals to amend the Accord to take account of market risks arising from exposure to foreign exchange, traded debt securities, commodities and options, and operational risk (BIS, 2018); this became known as Basel II, the revised international capital framework.

However, this was not sufficiently strong enough to prevent the Global Financial Crisis. Even before the collapse of Lehman Brothers it was apparent that advancements in financial technology and the development of new sophisticated and complex financial products had surpassed the framework of the existing guidelines. Following the Global Financial Crisis, Basel III was proposed which extended several areas of the existing framework. This new framework introduced liquidity ratios to deal with short-term shocks as well as enhancing the capital requirements to take account of different levels of risk exposures and stable funding over the medium- to long-term (BIS, 2018). Basel III has been implemented in Europe through the Capital Requirements Directive (CRD) IV package (European Banking Authority, 2018). CRD IV is an EU legislative package that contains prudential rules for banks, building societies and investment firms.

From past experience this market failure, and the need for regulation to be in place to protect consumers and the economy as a whole is unquestionable. Indeed, even Benston, who asserts that much of the rationale for regulation is weak (2003), concedes that bank capital is the exception (2000). If the consumer is confident that the UK retail banking system is stable and secure, and there is sufficient regulation in place to ensure that it is, then it could be argued that regulation has corrected this market failure. However, it may only become apparent that there is insufficient regulation in place, if and when, another crisis arises.

The first market failure addressed above is that of systemic risk at a macro-prudential level. The regulations are enforced at both a domestic and international level in an effort to reduce the likelihood of contagion, and thus increase the stability and security of the financial system. Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of these macro-prudential regulations in maintaining consumer confidence in the financial system. The first hypothesis that will be tested is:

H1: The consumer is confident that the UK retail banking system is stable and secure

Microprudential regulation is aimed at preventing the failure of individual financial institutions (Hanson, Kashyap, and Stein, 2011). In 1983 Diamond and Dybvig published a seminal paper providing a model of bank runs and associated financial crises. The principal mechanism proposed to avoid bank runs was the government provision of deposit insurance. In their model they assert that deposit insurance is critical for the stability of the banking system. This was based on the premise that consumers would not panic and try to withdraw their demand deposits if they knew that their deposits were safe.

The UK was one of the first countries to introduce a deposit insurance scheme in 1982, which was later revised in 1995 following the failure of BCCI (Goodhart, 2008). The current scheme, operated by the Financial Services Compensation Scheme (FSCS), has been in existence since 2001; however, this did not prevent the run on Northern Rock in 2007. Goodhart (2008) identified two main issues that must be addressed so that deposit insurance is effective. First, the large majority of consumers' balances must be covered, and second, depositors must have immediate access to their funds. At the time of the Global Financial Crisis, UK depositors may have had to wait up to three months, which is not feasible for most individuals. Ayadi and Lastra (2010) found that deposit insurance was ill-designed and insufficiently publicised. Hamalainen, Pop, Hall, and Howcroft (2011) found that the existence of co-insurance did not prevent a bank run as consumers were not

prepared to lose *any* of their funds. The compensation levels covered by the FSCS can be found in Table 1.1.

Diamond and Dyvbig (1983) also propose the LOLR function can act as a preventative tool against bank runs. However, they point out that if the LOLR were to unconditionally bail out every bank that found itself with liquidity problems then this in itself would create an incentive for banks to take greater risks. Conversely, the fact that it is not guaranteed does not then provide 100% confidence to consumers, and therefore, cannot prevent a bank run. Indeed, it was the announcement that Northern Rock was seeking assistance from the BoE that triggered the bank run of 2007. However, if the consumer is confident that the money in their retail bank account(s) is protected should their bank fail, and there is sufficient regulation in place to ensure that it is, then it could be argued that regulation has corrected this market failure.

The second market failure addressed above is that of systemic risk at a microprudential level. The regulations are enforced at a domestic level in an effort to reduce the likelihood of a bank run by guaranteeing the protection of a proportion of consumer deposits should a bank fail. Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of these micro-prudential regulations in maintaining consumer confidence in the financial system. The second hypothesis that will be tested is:

H2: The consumer is confident that the money in their retail bank account(s) is protected should their bank fail

2.6.3.2 Imperfect and Asymmetric Information

In financial services it may be the case that the consumer does not have adequate information to make a fully informed decision. Additionally, the problem of asymmetric information occurs as the consumer has less information than the 'expert' supplying the financial product and/or advice. Consumers are unable to inform themselves adequately so as to avoid making decisions which may disadvantage them (Wood, 2001). The individual consumer does not have the time

Claire Lynne McCafferty

or the resources to sufficiently source information on 'good' and 'bad' products and providers (Llewellyn, 1995). Additionally, the "low level of financial literacy and product comprehension" compounds the issue (Gaskill and Ashton, 2008, p.160). The nature of financial contracts is such that it may be some time before the performance of a product can be determined (Llewellyn, 1995).

Mullineux (2000) believes that regulation is required to protect the consumer against fraud, excessive risk taking, incompetence or bad behaviour on the part of the financial organisation. There were many cases of PPI policies being added to personal loans without the consumers' knowledge, or, the consumer being led to believe the policy was compulsory when in fact it was optional. This misbehaviour was largely target-driven and not at all in the interests of the consumer. This is an example of asymmetric information being used to 'exploit' the consumer; Llewellyn and the FSA (1999) describes this as the 'agency cost'. Asymmetric information occurs in financial services when there is inequality in information; one party knows more than the other (Mishkin and Eakins, 2012). In retail banking it is almost always the case that the consumer will have less (product knowledge) information than the representative of the financial institution. This could, and has in the past, led to a situation where the consumer's lack of knowledge is taken advantage of. However, if the consumer is confident that their retail bank will provide them with sufficient information to make informed decisions and there is sufficient regulation in place to ensure that firms do, then it could be argued that regulation has corrected this market failure.

The third market failure addressed above is that of imperfect and asymmetric information. The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers and the regulator monitors this through using tools such as Economics for Effective Regulation (discussed in Section 2.4.6). Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of conduct of business regulations in reducing information problems and maintaining consumer confidence in their retail bank.

Claire Lynne McCafferty

PhD April 2020

The third hypothesis that will be tested is:

H3: The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions

2.6.3.3 Moral Hazard

One of the problems that arise when the externalities market failure is corrected is the emergence of moral hazard. Krugman (2008, p.63) describes moral hazard as "any situation in which one person makes the decision about how much risk to take, while someone else bears the cost if things go badly". This is based on the theory that greater risks are taken if a party is "insulated" and therefore does not have to fully realise the consequences of their actions (Buckley, 2011, p.58).

By creating systems in which the consumer deposits are protected it may induce a bank to take greater risks (Hellman, Murdock and Stiglitz, 2000), perhaps even provide an incentive to intentionally take on the risk of failure (Boyd and De Nicolo, 2005), and so is not viable without regulation (Keeley, 1990). The introduction of ring fencing in the UK has reduced this risk in retail banking somewhat, since the level of risk that can now be taken by ring fenced banks has been reduced significantly. Ring fencing was introduced as a post-crisis regulatory response to the moral hazard problem surrounding 'Too Big To Fail' (TBTF) banks (Hardie and Macartney, 2016). Moral hazard can undermine the effectiveness of schemes such as LOLR, and magnify costs for the government providing it. The positive effects of ring fencing are two-fold: it reduces the firms' ability to take excessive risk; and the firm is no longer able to 'blackmail' the state into a bailout (Lehmann, 2014).

Moral hazard also arises due to the existence of the principal-agent problem. The consumer frequently requires advice when purchasing financial products and the quality of the product cannot be ascertained at the point of purchase (Llewellyn and the FSA, 1999). The existence of asymmetric information between the principal (the consumer) and the agent (bank staff) can lead to exploitation of the consumer where the agent is motivated by self-interest. Certainly, the mis-selling of PPI and packaged bank accounts was primarily due to incentive structures which *Claire Lynne McCafferty PhD April 2020 Page 72*

encouraged high volume sales of these products by frontline staff in exchange for monetary reward. Thus these products were often sold with little consideration of whether this was the most suitable product for the consumer.

The rationale for regulation is to remove the probability that the moral hazard (caused by schemes such as LOLR and deposit insurance) and the principal-agent problem will be exploited (Llewellyn, 1998). If the consumer is confident that their retail bank will do the right thing and there is sufficient regulation in place to ensure that firms do, then it could be argued that regulation has corrected this market failure.

The fourth market failure addressed above is that of moral hazard. The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers and has published guidance on staff incentives and remuneration. Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of conduct of business regulations in combatting moral hazard and maintaining consumer confidence in their retail bank. The fourth hypothesis that will be tested is:

H4: The consumer is confident that their retail bank will do the right thing

2.6.3.4 Fraudulent Activities

Fraudulent activity is linked to moral hazard and examples of the banks taking excessive risks when the costs are borne by others are plentiful. In the case of subprime mortgage selling in the US, the fraudulent reporting of income to ensure the acceptance of a home loan became standard practice at both small and large institutions (Skinner, 2016). However, the risk of default was not borne by the mortgage provider since the mortgages were then bundled together and their cash flows sold as securities, such as bonds. Likewise, in the UK, Northern Rock was offering a 'Together' mortgage which was made up of a mortgage worth 95% of the value of the property and a personal loan worth up to 30% (capped at £30,000) meaning that no deposit was required and buyers could borrow up to 5.9 times their income (Lythe, 2014). These 'Together' mortgages were then securitised to raise more funds for Northern Rock. Then, in the market for securitised mortgage

Claire Lynne McCafferty

products, bankers were failing to perform adequate due diligence tests on the included loans or practices of the loan providers. RBS was heavily criticised for misrepresenting the risks associated with the underlying collateral of their mortgage securities (Skinner, 2016).

The PPI misconduct case alone amounted to four times the cost (£37bn) of the London Olympics (Treanor, 2016). A culture which focused on sales was also behind the mis-selling of several other products, specifically interest rate hedging products, endowment mortgages, packaged bank accounts, investment products and advice, pensions and ID theft and card protection insurance (Baker, 2016). However, fraud was not limited to the practices of a few banks competing with each other. LIBOR and foreign exchange market rate manipulation was essentially collusive, collaborative and collegial (Skinner, 2016) between the largest of the world's banks. It had become industry accepted practice, essentially grid lock (see Section 2.6.4), and gave rise to significant social and economic costs.

Baker (2010) wrote that in the aftermath of the Global Financial Crisis many commentators cited regulatory capture of public bodies and public policy by leading banks as one of the main casual factors. The appointment of a person from the private business sector (for example, a bank) to a position of responsibility in government or a public body (for example, a regulator), or vice versa, leads to a conflict of interests, "such conflicts of interest may give rise to favouritism, abuse of power, discriminatory treatment, lack of impartiality and objectivity, and, ultimately, waste of public resources" (Cerrillo-i-Martinez, 2017, p.357). The regulator was widely criticised after the Global Financial Crisis for failing to regulate the banks that caused the crisis.

Boyer and Ponce (2012) found splitting supervisory powers between different supervisors to be the optimal arrangement, in terms of social welfare, as it could lower the potential that a single supervisor could be captured by banks. Regulatory capture was discussed in more detail in Section 2.5.2.

The steps taken to restructure the regulatory framework and separate the retail side of banks' business from riskier operations does address some of the causes of fraudulent activity that have been experienced in banking in recent times. Nonetheless, there is still nothing to address the revolving door in the UK; which is not exclusive to the banking industry. However, if the consumer is confident that they are protected from retail bank misconduct and there is sufficient regulation in place to prevent it, then it could be argued that regulation has corrected this market failure.

The fifth market failure addressed above is that of fraudulent activity. The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers and the regulator monitors this through using tools such as Economics for Effective Regulation (discussed in Section 2.4.6). Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of conduct of business regulations in combatting fraudulent activity and maintaining consumer confidence in their retail bank. The fifth hypothesis that will be tested is:

H5: The consumer is confident that they are protected from retail bank misconduct

2.6.4 Social Rationales for Regulation

In the previous section the economic rationales for regulating retail banking were discussed. These are the economic reasons for imposing regulations such as capital requirements, depositor insurance and Conduct of Business rules to reduce externalities, asymmetric information, moral hazard and fraudulent activities. These are justified as they are deemed necessary interventions for the efficient functioning of the economy. As discussed previously, confidence in the banking system is critical to the health of the economy.

The Organisation for Economic Cooperation and Development (OECD, 1997) defines social regulations as those that "protect public interests such as health, safety, the environment, and social cohesion". Further, "the economic effects of *Claire Lynne McCafferty PhD April 2020 Page 75*

social regulations may be secondary concerns or even unexpected, but can be substantial".

Llewellyn (1998, p.315) stated that "the objectives of regulation should be limited to correcting identified market imperfections, failures and externalities". However, Cartwright (2004) identified the following social rationales for regulation:

- i. Paternalism
- ii. Distributive justice
- iii. Community values.

2.6.4.1 Paternalism

Paternalism is when there is intervention in the interests of another, against their will. An example of this is the Mortgage Market Review (FSA, 2010a). This put an end to self-certified mortgages, consumers could no longer submit false income and expenditure information in order to pass the affordability checks. Additionally, it saw the introduction of mandatory advice on regulated mortgage products. These interventions were introduced solely to secure better outcomes for the consumer by reducing the likelihood that they will make decisions to their detriment; whether it was desired or not. Behavioural economics has been referred to as 'light' paternalism (Thaler and Sunstein, 2008).

As stated above, in its strictest sense, paternalism "removes choice and replaces it with the choice of the state" (Cartwright, 2004, p.19). The behavioural economists practice a libertarian version of paternalism by 'nudging' people to make more rational decisions (this is discussed in more detail below). Although an economic theory, it is important to distinguish between behavioural economics and neoclassical economics. In Section 2.4.1 the two traditional theories of economic regulation were discussed. The early neoclassical approach of welfare economics, also a neoclassical school of economic thought, from which private interest theory emerged. Neoclassical economics assumes that agents act rationally and in their own self-interests. Conversely, behavioural economics does not assume that

people make rational decisions, and instead studies the behavioural biases that can affect the decision making process. "A principal difficulty with describing a provision as paternalistic is that most apparently paternalistic laws can be justified on other grounds" (Cartwright, 2004, p.20). Although an economic theory, the use of behavioural economics in this context falls under a social rationale as the primary justification for intervention is not an economic one.

The FCA published its first Occasional Paper in April 2013 where it set out how it will apply behavioural economics to meet its overall strategic objective "to make markets for financial services work well for consumers" and use the insights to help "choose the right interventions" and "secure better outcomes for consumers" and "promote effective competition" (Erta, Hunt, Iscenko, Brambley and the FCA, 2013, p.3, p.51-52).

The FCA believed that by gaining better insight into consumer behaviour, specifically on how behavioural biases can affect the decision making process, particularly in financial services decisions, they could help consumers avoid making decisions to their detriment.

In the second occasional paper, also published in April 2013, called 'Encouraging consumers to claim redress: evidence from a field trial'. An anonymous firm reproduced the experiment the government's Behavioural Insights Team (also known as the 'Nudge Unit') undertook with non-payers of vehicle tax (The Economist, 2012). In this experiment they made subtle changes to the language used in their mailings to encourage a higher response rate. In the government experiment they found that by personalising the letter they experienced an increase in response rate of a third. In the FCA experiment changes were made to letters advising consumers of redress owed to them, due to an error identified in the sales process. The results were a sevenfold increase in response rate. The changes made were:

- i. envelope marked 'act quickly'
- ii. FSA logo in the letterhead
- iii. salient bullet points

Claire Lynne McCafferty

PhD April 2020

- iv. simplified text, reduced by 40%
- v. stated in letter that claims process would take only '5 minutes';
- vi. CEO signed the letter instead of generic team
- vii. a reminder letter was sent three to six weeks afterwards.

The above is a demonstration of the FCA's commitment to "test remedies before implementation more frequently than previously and conduct research to gain deeper insights into specific markets" (Erta et al, 2013, p.52). The outcome of such research, using behavioural economics, will then be used by the FCA to decide the most appropriate intervention.

Behavioural economics "has emerged as an alternative view of financial markets" (Shleifer, 2000, p.2) and is the study of psychology and economics (DellaVigna, 2009). This relatively new field of study developed due to rising scepticism, increased by events such as the 1987 stock market crash, about the efficient market hypothesis (Mishkin and Easkins, 2012). Fama (1970, p.388) developed the efficient market hypothesis, "security prices at any point in time fully reflect all available information". In doing so he asserted the financial markets are efficient and cannot be beaten. Theory developed over time to demonstrate that individuals are not fully rational (Shleifer, 2000) and therefore cannot make fully rational decisions. Consequently, the efficient market hypothesis cannot be applied to financial markets in such a way as they can be to alternative economic markets.

Altman (2012) suggests that due to behavioural biases and errors which are 'hardwired in the brain'; financial education and the drive to improve financial literacy of the UK population by the FCA (and its predecessor the FSA) will actually do little to improve the decisions of consumer of financial services. Instead, Altman (2012) suggests nudging consumers into making decisions which 'experts' deem to be better for them under this approach to behavioural economics. Altman suggests that the 'alternative' approach (which he dubs the 'Simon-March' approach after cognitive psychologist Herbert A. Simon and behavioural theorist James G. March who collaborated on several publications on organisational theory) is more suitable for

government intervention to improve financial education and literacy and therefore there is less need for 'nudges', and quality of information is key.

The UK government have been using behavioural economics to 'protect' the general public or 'persuade' them to make better decisions. In a report sponsored by Aviva (previously, Norwich Union) Dixon (2006) suggested that as many as 30.5 million adults in the UK in early 2006 were experiencing some form of difficulty when it came to financial capability. Dixon (2006, p.2) suggested financial capability was more of "social welfare issue" for the government than an issue for the FSA. He then went on to suggest various ways in which the government could apply behavioural economics to improve consumers' financial capability, such as, automated debt repayments from benefits and automated transfer of salary increases to savings accounts.

The UK government have been using behavioural economics to make policy decisions for some time via the 'Nudge Unit'. Harford (2009, p.13) described his reservations in 'nudging' the public to "behave as the nanny state preferred" when he believes that the popularity of the theory has meant that it has become "mutated". Harford's reservations stem from his observation that behavioural economics has quickly become public policy panacea, and his belief that perhaps a more cautious approach by government would be to understand what people would prefer if they could be persuaded to give some proper thought to decisions.

Ormerod (2010) demonstrated why despite government assertion that behavioural economics can be used to 'nudge' the general public into making better decisions they are failing to successfully apply it in practice. For example, in the case of workless families the government hopes to change behaviour by cutting or changing the benefits they receive. Additionally, in the case of binge drinking the government approach is to increase the cost of alcohol. These approaches do not take into account that we are not "autonomous individuals" and that we are influenced by others and what our circle considers to be socially acceptable. This demonstrates that changing behaviour is not simple or straightforward and a deeper insight into

what influences consumers is required if the benefits of behavioural economics are to be realised.

Following the FCA's first occasional paper, Miller (2013) published an article summarising how the FCA intended to use behavioural economics to resolve consumer issues and the four key ways in which they intend to do this. Firstly, the information provided to consumers would be more closely monitored so that it was marketed in a particular way and certain practices were forbidden; such as misleading information. Secondly, the choice that consumers are presented with when making financial decisions would be more closely scrutinised by the regulator. Thirdly, product distribution would be more closely supervised to ensure that products were only sold through the appropriate channels to the appropriate target market. Lastly, products would be more controlled so that their features could not lead to the exploitation of consumers.

Watt (2013) commented that the PPI mis-selling scandal is a perfect example of "people's implied acceptance that the product was appropriate for their circumstances". If the above four principles had been applied to PPI products, then it becomes clear that mis-selling on the scale that was seen would have unlikely occurred.

Chung (2013) suggests that using behavioural economics will only be effective if financial institutions have a good understanding of each consumer group's behaviour and motivations, and that they can only do this by undertaking a comprehensive review of their profile, specifically: 'demographics', 'channel use', 'affluence', 'ability to bear loss', 'financial sophistication and capability', 'investment appetite, investment objectives and willingness to accept loss', and 'risk type and emotional ability to accept loss'.

Spicer et. al., (2014) found that customer inertia exists due to information asymmetries and lack of transparency in the market. They suggest that the use of league tables and rankings would help consumers to compare offers and identify the

best option. This could alleviate some of the concerns expressed by Harford (above) and further enhance good competition in retail banking.

The sixth rationale for regulation is that of paternalism. Since coming into power the FCA has developed a keen interest in the use of behavioural economics to choose the right interventions, secure better outcomes for consumers and promote effective competition (Erta et al, 2013). Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of paternalism in maintaining consumer confidence in their retail bank. The sixth hypothesis that will be tested is:

H6: The consumer is confident that they can trust the expertise of their retail bank

2.6.4.2 Distributive Justice

Distributive justice is concerned with distributing resources on the basis of what is fair, and when applied to the regulatory system it is concerned with the protection of consumers (Cartwright, 2004). Examples of regulatory protection of the most vulnerable financial services consumers are policies of financial inclusion and the FCA investigation into payday lenders which resulted in extra rules for firms providing high-cost, short-term credit which took effect on 1 July 2014 (FCA, 2014a).

The FCA published its eighth Occasional Paper, 'Consumer Vulnerability' in February 2015. The FCA conceded that much consumer protection legislation is "underpinned by the notion of the average or typical consumer" (Coppack, Raza, Sarkar, Scribbins, and the FCA, 2015, p.6). Their study found that many consumers in vulnerable circumstances are not being served consistently or well. The FCA defines a vulnerable consumer as, "someone who, due to their personal circumstances, is especially susceptible to detriment, particularly when a firm is not acting with appropriate levels of care" (Coppack et al, 2015, p.7). Some of the examples of risk factors for vulnerability identified in the FCA's research were:

i. Low literacy, numeracy and financial capability skills

Claire Lynne McCafferty

- ii. Physical disability
- iii. Severe or long-term illness
- iv. Mental health problems
- v. Low income and/or debt
- vi. Caring responsibilities (including operating a power of attorney)
- vii. Being 'older old' for example over 80, although this is not absolute (may be associated with cognitive or dexterity impairment, sensory impairments such as hearing or sight, onset of ill-health, not being comfortable with new technology)
- viii. Being young (associated with less experience)
- ix. Change in circumstances (e.g. job loss, bereavement, divorce)
- x. Lack of English skills
- xi. Non-standard requirements or credit history (e.g. armed forces personal returning from abroad; ex-offenders; care-home leavers; recent immigrants).

It is beyond the scope of this research to examine all of the factors identified by the FCA. The researcher, for ethical reasons, is unable to access particular vulnerable groups and also does not have access to other specific groups, for example from the 'non-standard' list above. The specific demographic groups that will be tested for any significant difference in levels of confidence in this thesis are: age, education and income. A common problem encountered by each of these groups is access to affordable products (Coppack et al, 2015). In July 2018 the FCA set out a framework for considering both economic and distributive fairness considerations in the practice of price discrimination (Starks, Reynolds, Gee, Burnik, Vass, and FCA, 2018). If the consumer is confident that the price of retail banking products is fair, and there is sufficient regulation in place to ensure it is, then it could be argued that regulation has corrected this market failure.

The seventh rationale for regulation is the principal of distributive justice. Access to financial services should be equal for all and inclusive. Price tends to be a common problem which creates inequality of access. The FCA has published several papers to help firms identify and understand the specific needs of vulnerable groups and

challenge financial exclusion. Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the principal of distributive justice in maintaining consumer confidence in their retail bank. The seventh hypothesis that will be tested is:

H7: The consumer is confident that the price of retail banking products is fair

2.6.4.3 Community Values

Community values are the values which society holds and wish regulation to protect and promote; such as trust, honesty and fair dealing (Cartwright, 2004). Due to market failures such as asymmetric information, trust is paramount in retail financial services, as without trust demand for products and services may decrease (Llewellyn, 2005). One of the FCA's statutory objectives is to 'maintain confidence in the UK financial system' under the terms of the Financial Services and Markets Act 2000.

In 2004 the FOS found that not all consumers were receiving fair and equal treatment and their 37th newsletter focused on types of discrimination that can occur, namely disability, race and sex, which were protected under the Disability Discrimination Act (1995), the Race Relations Act (1976), and the Sex Discrimination Act (1975). In their 123rd newsletter published in 2015, the FOS revisited the issues which had been brought together, by this previously separate legislation, under the Equality Act (2010). They identified several case studies of discrimination at the hands of financial services providers. The FOS outlined the cases and the outcomes, clearly explaining how they had made their decision and how the service provider had fallen short of both expectations and the law. The purpose of these newsletter case studies is to provide those working within the industry with essential information on how to prevent and/or settle complaints, before escalation to FOS. Clearly, there is UK legislation in place to protect all citizens, not only consumers of financial services. However, the FOS has taken on the role of out-of-court settler of complaints which means that the consumer does not incur any legal costs (the firms pay a fee for each complaint against them). The FOS does have to consider the law when making a

ruling but can depart from it if it would lead to a conclusion contrary to what is fair and reasonable (Taylor, 2017). If the consumer is confident that they will not be unlawfully discriminated against by their retail bank, and there is sufficient regulation in place to prevent it, then it could be argued that regulation has corrected this market failure.

The eighth, and final, rationale for regulation studied in this thesis is the preservation of community values. In addition to the guidance and regulations in place, as discussed above, there is also an arbitrator, the Financial Ombudsman. The purpose of FOS is to ensure that consumers receive a fair deal and where this is not the case it is put right. This is particularly important for groups who could find they are discriminated against but may lack a voice. Therefore, as part of this investigation, the confidence levels of consumers will be measured to assess the impact of community values in maintaining consumer confidence in their retail bank. The eighth hypothesis that will be tested is:

H8: The consumer is confident that they will not be unlawfully discriminated against by their retail bank

When applying the philosophies of distributive justice and community values, it is clear there are ethical considerations as well as social ones. The organisational culture and ethics of UK financial institutions has come into question for decades, but more recently due to the Global Financial Crisis of 2007/08.

The above has set out the economic and social rationales for regulation of the financial services sector.

2.8 Conclusions

The structure and culture of retail banks has been explored, and how these factors influence the treatment of consumers and impact consumer confidence in the effectiveness of regulation. Firstly, the consolidation of the retail bank market has given rise to a complex monopoly which by its very nature is not conducive to the enhancement of consumer welfare. Additionally, the reduction in diversification of

Claire Lynne McCafferty

business models gave rise to the incentive structures and cultures which were blamed for widespread bad practices and bank failure (Spicer et. al., 2014). Conversely, a strong ethical culture is considered to make banks less vulnerable to misconduct (Vigeo-Eiris, 2017); despite this there is still no mandatory ethical code for the industry (Jenkins, 2016).

It is important to consider the above factors since they heavily influence the behaviour of the firm which has consequences for the consumer. For example, whether the firm 'will do the right thing', 'provide them with sufficient information to make informed decisions', can be trusted 'in their expertise', and will ensure they are protected against widespread 'misconduct' and 'discrimination' is inherently down to the structure and culture of the organisation.

The research aim is to answer the question, 'Are Scottish consumers confident in retail banking regulation?'. After undertaking a critical review of the existing theory of financial regulation literature, from the perspective of consumer confidence in retail banking, it is clear the areas that regulation seeks to address are the market failures which cause economic and/or social detriment.

Therefore, the theories have been contextualised to develop the following hypotheses:

H1: The consumer is confident that the UK retail banking system is stable and secure

H2: The consumer is confident that the money in their retail bank account(s) is protected should their bank fail

H3: The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions

H4: The consumer is confident that their retail bank will do the right thing

H5: The consumer is confident that they are protected from retail bank misconduct

H6: The consumer is confident that they can trust the expertise of their retail bank

H7: The consumer is confident that the price of retail banking products is fair

H8: The consumer is confident that they will not be unlawfully discriminated against by their retail bank

H9: In general, the consumer is confident that as a retail bank customer they are adequately protected

These hypotheses will be used to develop a set of consumer survey questions which will be evaluated using a triangulated approach. The evaluation process is part of the survey design (from theoretical model to empirical testing) and is undertaken in three distinct steps, see Figure 2.3 below.

Figure 2.3 Triangulated Approach to Research Design



The researcher held a three-stage survey design, the first with subject knowledge academics (theorists), the second with industry experts (practitioners) and the third with a group of consumers (end-users) who would not be included in the main study. The rationale for separating the groups was due to the different levels of technical

Claire Lynne McCafferty

Page 86

knowledge, additionally, corrections and amendments could be made before 'testing' on the next group.

There are numerous studies, in particular those undertaken by the current regulator, where there is participation from either industry or consumer. The regulator and academics often conduct collaborative research, as evidenced in this chapter. However, the researcher could not find any specific studies where theorists, practitioners and consumers had direct input into the design of the study. The next chapter will provide an outline of the research methodology and the methods employed by the researcher in this thesis.

CHAPTER 3: RESEARCH METHODOLOGY AND METHODS

3.1 Introduction

In the previous chapter, the researcher explored the structure and culture of both the retail banking regulator(s) and the retail banking institutions and how this has changed over recent decades to become the system known today. Some of the changes were a result of innovation, others the result of failures, but essentially the rationales for regulating the sector have remained the same. The emphasis on the importance of confidence in the financial system has intensified since the Global Financial Crisis of 2007/8. There are of course still advocates of free-banking, and the regulation versus deregulation debate may never cease. Both the social and economic rationales for regulation are widely debated in the literature, and as mentioned above, although the industry has transformed over the decades the principles behind these have not. Nonetheless, the measurement of confidence in the processes to enforce these appears to be lacking in the literature. This research aims to measure consumer levels of confidence in the rationales for regulation and identify whether demographic factors such as age, education and income have any relationship to these levels of confidence. These are factors that the Financial Conduct Authority has identified as risk factors for vulnerability (Coppack at el, 2015).

- i. Being young, or being 'older old'
- ii. Low literacy, numeracy and financial capability skills
- iii. Low income

This chapter will provide an outline of the research methodology and methods employed by the researcher in this thesis. The relationship between the research question, the hypotheses previously identified in the Literature Review Chapter, the philosophical assumptions and the techniques used to conduct the research will be demonstrated. The researcher's overall research philosophy is stated in Section 3.2 Methodology. The research approach follows a positivist paradigm and there is a discussion of the methods deemed appropriate for the primary research and the

techniques adopted for analysis, including consideration of academic ethical issues arising from the research.

Finally, an overview of the preliminary research is provided, including changes made to the Consumer Survey following examination of the feedback obtained during the three-stage survey design.

A description of the data collected in the final study, including the characteristics and demographics of the sample population is defined and the validity of the sampling approach is discussed.

3.2 Methodology

The terms methodology and methods are often, incorrectly, used interchangeably. For example, Trochim (2000) defines epistemology as the philosophy of how we come to know, and methodology as the practice of how we come to know. There is no separation of methodology and methods in his definition. McGregor and Murnane (2010) offer concise definitions as follows: methodology refers to any philosophical assumptions that underpin the study, and methods are the techniques or procedures followed to conduct the study. Moreover, they explain that each research paradigm (discussed in section 3.2.1 below) has an associated group of methodological principles. Wainwright (1997) defines methodology as the philosophical analysis of research strategies, and the methods as the techniques used to gather and analyse the data. Furthermore, the philosophy of a paradigm is made up of the ontology, epistemology and methodology.

Based on the research outcomes of the study, and a comprehensive review of prior literature, the researcher has identified that to describe and explain the methodology and methods used there must be two distinct parts. First, the philosophical approach is discussed as the research paradigm; which constitutes ontology, epistemology and methodology. Second, the methods are discussed in terms of data collection procedures and data analysis strategy.
3.2.1 Research Paradigm

Kuhn (1970) coined the phrase 'research paradigm' in his book, *The Structure of Scientific Revolution.* He defined the paradigm as a source of guidance for conducting and evaluating research which is based on a shared set of theoretical assumptions. A paradigm, therefore, is the identity of a research community based on their shared belief system or set of principles (Cohen, Manion, and Morrison, 2013). The theoretical assumptions that characterise the research paradigms are ontology, epistemology and methodology (Guba, 1990).

Hitchcock and Hughes (1995) proposed that ontological assumptions lead to epistemological assumptions; these, in turn, lead to methodological considerations; and these, in turn, lead to issues of data collection (as cited in Cohen et al, 2013).

Ontology refers to the study of the nature of reality (Guba, 1990), the nature of existence, and understanding 'what is' (Gray, 2007). Velardo (2016) attempts to answer the question, 'What is reality?', by approaching the question from two disciplines, science and philosophy. He posits that while science focuses on empirical evidence and observation, philosophy tends to be more speculative and based upon argument and reasoning from values. Ontology determines what can be rationally understood (Poli and Seibt, 2010). The researcher's position is that the world is external and objective, and through research one single reality, or truth, can be discovered. Objectivism is the epistemology which underpins the positivist stance (Crotty, 1998).

Epistemology refers to the study of knowledge and understanding what it means to know (Gray, 2007), how we know what we know (Crotty, 1998), and how we legitimise and validate this knowledge. Essentially, it is the theory of knowledge (Benton and Craib, 2011). In line with the aim and research objectives of this thesis the researcher has a positivist theoretical perspective as it is posited that reality can be measured in terms of this specific research project; facts are facts. Moreover, the researcher is independent from the study, hence it is purely objective. Guba (1990) proposed that all paradigms can be characterised by the responses to three

basic questions. Table 3.1 overleaf illustrates how the researcher formed a positivist philosophical position.

| Theoretical assumption | Basic questions | Positivist inquiry |
|------------------------|---------------------|---|
| meoretical assumption | Dasic questions | r ositivist inquiry |
| Outstan | | |
| Ontology | what is the nature | Realist – reality exists fout there |
| | of reality? | |
| | | |
| Epistemology | What is the nature | Dualist/objectivist - distant and non- |
| | of the relationship | interactive. Values and other biases are |
| | between the | automatically excluded from influencing the |
| | researcher and | outcome |
| | reality? | |
| | , | |
| Methodology | How should the | Experimental/manipulative – questions or |
| | researcher go | hypotheses are stated in advance in |
| | about finding out | propositional form and subjected to |
| | the reality? | empirical tests under carefully controlled |
| | | conditions |
| | 1 | |

Table 3.1: The Paradigm of Positivist Inquiry

(Source: adapted from Guba, 1990)

'Positive' or confirmatory data is a term used to describe hypothesis-driven research at the heart of science (Anderson, Sprott, and Olsen, 2013). Positivism is the philosophy that all knowledge comes from observable experiences. The positivists are driven by their belief that society could be studied like a natural science.

Criticisms of the positivist approach are that it fails to consider the subjective nature of participants and relies on experience as a valid source of knowledge (Scotland, 2012). Additionally, research findings are only descriptive; they will not produce indepth insight into the issue being examined. Lastly, the reliance of experience as a valid source of knowledge has been criticised. On the contrary, for this study these

points are valid reasons for adopting this approach. In this research the aim is to measure the levels of confidence that retail bank consumers have in eight specific areas of regulation. Although individual opinions are subjective, collectively they form an objective reality that can be measured, the reality is 'out there' and the researcher cannot influence this or bias the results. The researcher will then undertake statistical analysis of the captured data; this forms part of the method and is discussed in more detail in the next section.

3.3 Methods

Owing to the researcher's positivist ontology and epistemology, a quantitative approach to data collection was employed. Cohen et al, (2013), claim that positivist approach researchers will choose from 'traditional' tools, such as surveys or experiments.

However, most studies will be a mixture of both qualitative and quantitative categories with more emphasis on one or the other (Hackley, 2003). The researcher has opted to use a traditionally qualitative method, in the three-stage survey design, to validate the consumer survey questions. One of the main purposes of the process is to ensure the questions were not misleading or likely to be misinterpreted. A disadvantage of a questionnaire is that due to its standardised questions with at least one group of people knowledgeable about the issues being examined. The researcher held a three-stage survey design; the first with subject knowledge academics (theorists), the second with industry experts (practitioners) and the third with a group of consumers (end-users) who would not be included in the main study. The rationale for separating the groups was due to the different levels of technical knowledge, additionally, corrections and amendments could be made before 'testing' on the next group, see Figure 3.1 overleaf.

There are numerous studies; in particular those undertaken by the current regulator, where there is participation from **either** industry **or** consumer, however, the

researcher could not find any specific studies where theorists, practitioners **and** consumers had direct input into the design of the study.

Figure 3.1: Stages of Survey Design: from Theoretical Model to Empirical Testing



For this study the primary data collection method used was a consumer survey in the form of a questionnaire. The researcher designed a consumer survey that would collect both demographic data, and, measure levels of confidence using a Likert scale. The data was collected online, over the telephone and face-to-face.

These three channels were used to ensure no group(s) was excluded from participation due to immobility, geographic location, or lack of internet access. Face-to-face surveys were conducted in shopping centres in the north, east and central

Scotland. Additionally, a pensioners coffee morning in north east Scotland was attended to capture this age demographic. The participants who were contacted via telephone were selected at random from phone books purchased from BT which covered households in the south west of Scotland. Primarily, the researcher sought to ensure the widest range of ages and geographic locations were covered. There were practical limits on time and budget in relation to postcode data and specialised software (or employing an external company to conduct the research, such as the FCA has done in previous studies).

The advantages and disadvantages of the survey method differ slightly depending on the channel utilised to conduct the questionnaire and collect the data. It's widely recognised that surveys tend to be straightforward, quick and cheap to administer (Ackroyd and Hughes, 1992). However, this very much depends on the chosen channel. In terms of simplicity, an online survey will be easier to analyse than paper format conducted via telephone or face-to-face. In this research, the use of Novi Survey removed the requirement for data entry prior to coding. Another advantage the online survey has over paper completion is that the survey can be set up in such a way that questions cannot be missed. For example, one of the clauses on the consent form was that an incomplete questionnaire would be treated as a wish to withdraw from the research. It could be the case that some of the face-to-face surveys had unintentional gaps; since these were completed independently by the participants.

The online and face-to-face surveys were completed independently by the participants; however, the telephone surveys were completed by the researcher on the participants' behalf. The researcher endeavoured to ensure they did not influence the outcome in any way, however, it must be acknowledged that a major potential disadvantage of this channel is interviewer bias (Szolnoki and Hoffmann, 2013).

It has been argued that an online survey is quicker than face-to-face or telephone research (Gunter, Nicholas, Huntington and Williams, 2002). However, the online survey for this research was actually open for 9 months, whereas the face-to-face *Claire Lynne McCafferty PhD April 2020 Page 94*

and telephone surveys were completed in a matter of days. Certainly, the researcher had to dedicate their full attention and time to the latter two methods as these were labour intensive. Nonetheless, this concurs with Duffy, Smith, Terhanian and Bremer (2005), that online surveys actually take some time if a wide-reach is required.

In terms of monetary cost, there was nothing incurred for the online or telephone surveys (the university had purchased the software licence, and the calls were made as part of a contract plan, so the researcher did not incur further charges), however, the face-to-face surveys did incur travel costs and other incidental costs.

In terms of sample representativeness, face-to-face surveys tend to deliver the best results, followed by telephone and then online surveys (Szolnoki and Hoffmann, 2013). In this research, face-to-face surveys were carried out after most of the online surveys had been completed. This enabled the researcher to target particular demographic groups that were lacking via the online method. The telephone surveys also targeted a particular demographic: males. The telephone method had the best response rate, with 100% of successful contact resulting in a completed questionnaire. Some face-to-face requests were declined, although not many, and it is impossible to determine how many potential participants did not click on the survey link, or clicked it and then decided against participation before commencing. However, around one fifth of surveys that were started were never completed.

A clear disadvantage of the online method was sample representativeness; this was made-up through the other two channels. It is clear that some of the disadvantages of one method can be overcome by adopting another. Using three methods to collect the survey data ensured a representative sample was achieved, which was the researcher's highest priority.

The survey was analysed using quantitative methods. The principal reason for choosing to do this was to be able to capture a large, random sample which could allow the overall research findings to be generalisable. Additionally, the aim of the research was to address the key issue 'Are Scottish consumers confident in the retail

banking system?' Based on the answer to this question, the implications of the levels of trust afforded by Scottish consumers on the banking sector, wider financial services sector, and ultimately the economy, will be assessed. Therefore, collecting data from any other stakeholder was not required to meet the aim and objectives of this research. Furthermore, due to the large sample required, a survey was deemed more appropriate than interviews, which are time-consuming and subject to more personal biases (both respondent and interviewer). The advantages of questionnaires are that they are relatively inexpensive and quick to administer and can be used to capture data from a large number of respondents in different geographical locations (Walliman, 2015).

3.3.1 Data Collection

The techniques applied for data gathering will be described in this section. Revisiting the research focus will add clarity to the aim of the method; 'Are Scottish consumers confident in retail banking regulation?' This can only be answered by gathering data on the key themes that have been identified in the banking regulation and consumer protection literature. Based on the answer to this question, the implications of the levels of trust afforded by Scottish consumers on the banking sector, wider financial services sector, and ultimately the economy, will be assessed.

Accordingly, the reasons for and against the regulation of financial services have been examined. An interesting finding was the consensus, from both sides of the debate, that the need for consumer confidence is grounds for regulating the banking industry (Llewellyn and the FSA, 1999); although the level of regulation deemed appropriate is still debated (Benston, 2000).

Secondly, the rationales identified in the pro-regulation literature have been categorised, and a survey question has been derived from each specific rationale. Each of the identified rationales has been split into one of two categories: economic rationales and social rationales. In total nine hypotheses have been derived from the literature.

Lastly, as consumer confidence is the main driver of regulation of the system, the measure of success in achieving the rationales is not that the regulator or industry perceives they are meeting them, but that consumers do. For that reason, the main focus that is being investigated through a number of research hypotheses can only be answered by measuring consumer perceptions that each of the rationales for regulation is being met. The results of the three-stage survey design, based on feedback from the groups, are discussed in detail in Sections 3.7 - 3.10. The method for obtaining this data will now be described.

In order to collect the data in a way which did not affect the participants, the researcher undertook an observational study in the form of a survey. The population of interest was individuals over 18 years old in Scotland; since retail banking is a service which should be available to every adult in Scotland. Per the CAB Scotland (2010) 'Banking on the Basics' report approximately 5% of the Scottish adult population was unbanked in 2007/8. Therefore, a question was added to the demographics section of the survey to establish if a personal current account was held by the participant. In June 2018, the FCA published 'The financial lives of consumers across the UK' (FCA, 2018b). In their findings differences between regions of the UK were marginal. Therefore, although this research is focused on Scotland it could be applied more widely to the UK.

The sample selected and presented with the questionnaire represented a crosssection of the Scottish adult population. Selecting a random cross-section avoids bias and supports legitimate results. The sampling frame was created from data obtained from the National Records of Scotland official government website. This defined target population did not statistically favour or exclude any groups (based on age, gender, and income or education level). The wording of the survey questions was carefully considered to ensure they were not misleading or created bias. Feedback from the Theorist group was received around the neutrality of the questions and this was acted upon and updated, for example, to change 'good' financial decisions to 'informed' financial decisions. A test of the validity and suitability of the questions was carried out by three groups:

Claire Lynne McCafferty

PhD April 2020

- i. Theorists
- ii. Practitioners
- iii. End-users

One purpose was to 'sense-check' the survey, and obtain feedback, which may have led to amendments to the questionnaire before data collection began.

The sampling frame is the Scottish adult (18+ years) population. The data used to create this was obtained from the National Records of Scotland's Mid-2016 population estimates. The sample was stratified using the subset of age group to ensure appropriate numbers were drawn from each set. The total population size is 4,372,939 (National Records of Scotland, 2017). The sample size was then determined, the characteristics of which are discussed in full from Section 3.11 onwards.

The survey was distributed in two main ways. First, the software Novi, was used to create an electronic questionnaire that could be emailed to participants and posted on websites to request participation (see Section 3.13 Data Collection for more detail). Secondly, a paper format was available. The researcher then visited several public locations in Scotland (also see Section 3.13) with the paper version with the aim to specifically target the strata of the target population that may not have been covered proportionally through the distribution of the electronic questionnaire, e.g. elderly respondents, people with limited access to the internet. The researcher decided to utilise a man on the street survey as the results may be biased to only those who had access to the internet and therefore some of the target population may not have the opportunity to respond if this method had not also been undertaken. Further to the man on street data collection the researcher also undertook telephone surveys in areas which it was not possible to visit.

The main limitations of the study are:

 Firstly the location. Data was collected from residents of Scotland only, and therefore the results are specific to Scotland (although previous studies in this field have found differences between UK regions were marginal)

Claire Lynne McCafferty

- ii. Secondly, a survey can only report the relationships between variables and cannot claim cause and effect, that is, the research will be able to answer the 'what' but not the 'why'.
- iii. Lastly, a major limitation of stepwise regression is the increased likelihood of committing a type 1 error (Pandit and Khairullah, 2015), i.e. rejecting a true null hypothesis, due to a sampling error. However, sampling error is less of an issue with a larger sample size (Thompson, 1995).

Potentially, future research should be undertaken to encompass a wider participation in terms of location and examine the 'why' questions that this research may uncover.

There are two question sections in the surveys. The first section is comprised of demographic questions such as gender, age, education level, employment status and income. Several studies measuring trust and confidence in the financial services sector since the Global Financial Crisis of 2007-2009 did not report any significant gender differences in their findings, Moin, Devlin and McKechnie, 2017; the FCA Financial Lives survey, 2018; YouGov on behalf of Positive Money survey, 2018; Edelman Trust Barometer (UK results), 2019; and GfK Consumer Confidence Index, 2019, are a few recent examples.

The FCA's Financial Lives Study (2017) found that confidence in managing financial matters increased with age. However, in addition to finding the youngest age group (18-24 year olds: 30%) had the least confidence in the financial services industry they also found the oldest group (age over 85: 31%) had low confidence.

Bannier and Schwarz (2018) found that a higher educational attainment increases actual and perceived financial knowledge; however, they qualified this by stating "there is hardly any evidence of a link between education and confidence in the finance context" (p.67).

Moin et. al. (2017) carried out a study where they measured the influence of demographics on trust in financial services. They found that levels of trust generally

increase with income. This corroborated the findings of research carried out by Alesina and La Ferrara in 2002.

Age, education and income were considered important variables owing to the research conducted by the regulators into consumer vulnerability (Coppack et al, 2015), access to financial services (Collard et al, 2016) and ageing populations (McLoughlin et al, 2017). Clearly, the regulator and academics consider these demographics to be of such significance to warrant collaborative research on several occasions. To re-iterate, the demographics this research focus on are age, education and income. This follows the risk factors for vulnerability identified by Coppack et al, (2015).

- i. Being young, or being 'older old'
- ii. Low literacy, numeracy and financial capability skills
- iii. Low income

Additionally, the researcher included the question '*Do you currently work in the financial services sector*'? The reason this demographic was included was to ensure the results were not skewed by a higher proportion of responses from those employed in financial services. This was verified in Chapter 4. The researcher was unable to find any studies that measured the difference (if any) in confidence levels between those employed in financial services and those who are not.

The second section of the survey covers the main theoretical questions which have been developed from the Literature Review. Overleaf and on page 101 and 102 are two tables which illustrate what the independent and dependent variables are and how they have been categorised.

| Demographics | Variable | Categorisation |
|-------------------------------|-------------|----------------|
| Gender | Independent | Nominal |
| Age | Independent | Ordinal |
| Education level | Independent | Ordinal |
| Income level | Independent | Ordinal |
| Work in Financial Services | Independent | Nominal |

Table 3.2: Demographic Questions and Variable Categorisation

| Rationales for Regulation | Variable | Categorisation | Hypothesis |
|------------------------------|-----------|----------------|---|
| Capital Requirements | Dependent | Ordinal | H1: The consumer is confident that the UK retail banking system is stable and secure |
| Depositor Insurance | Dependent | Ordinal | H2: The consumer is confident that the money in their retail bank account(s) is protected should their bank fail |
| Asymmetric Information | Dependent | Ordinal | H3: The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions |
| Moral Hazard | Dependent | Ordinal | H4: The consumer is confident that their retail bank will do the right thing |
| Corruption | Dependent | Ordinal | H5: The consumer is confident that they are protected from retail bank misconduct |
| Paternalism | Dependent | Ordinal | H6: The consumer is confident that they can trust the expertise of their retail bank |
| Distributive Justice | Dependent | Ordinal | H7: The consumer is confident that the price of retail banking products is fair |
| Community Values | Dependent | Ordinal | H8: The consumer is confident that they will not be unlawfully discriminated against by their retail bank |
| Overall Protection | Dependent | Ordinal | H9: In general, the consumer has confidence that as a retail bank customer they are adequately protected |

Table 3.3: Theoretical Questions and Variable Categorisation

The dependent variable questions were subject to scrutiny in the three-stage empirical testing and this is discussed in detail in Sections 3.7 - 3.10. As a Likert scale was used in Section 2 of the survey these variables are categorical and ordinal. In the next section the strategy for analysing the collected data is discussed.

3.3.2 Data Analysis

The data was analysed using IBM SPSS Statistics Software. The researcher consulted several SPSS sources to understand the general assumptions that apply to all of the parametric techniques. They can be summarised as follows:

- i. The level or scale of measurement is of equal interval or ratio scale (Bryman and Cramer, 2001; Hinton, McMurray and Brownlow, 2014; Pallant, 2016)
- ii. Random sampling (Hinton, McMurray and Brownlow, 2014; Pallant, 2016)
- iii. Independence of observations (Pallant, 2016)
- iv. The data is drawn from a normally distributed population (Bryman and Cramer, 2001; Hinton, McMurray and Brownlow, 2014; Pallant, 2016)
- v. Homogeneity of variance (Bryman and Cramer, 2001; Hinton, McMurray and Brownlow, 2014; Pallant, 2016)

The first assumption is determined by the classification of the dependent variables in the study. Traditionally, Likert scale data is discrete categorical but this has sparked some debate. Analysing and interpreting this data has caused some controversy in the academic literature (Glass, Peckham and Sanders, 1972; Kuzon, Urbanchek and McCabe, 1996; Blaikie, 2003, Jamieson, 2004; Carifio and Perla, 2008; Norman 2010; Murray, 2013; Bishop and Herron, 2015; and, Knapp, 2016). The Likert Scale was developed by Rensis Likert in 1932 to measure attitudes on an ordinal scale where respondents can rate the degree to which they agree or disagree with a statement (Sullivan and Artino, 2013), or a level satisfaction/dissatisfaction; importance/unimportance.

In this research, participants were asked to rate the degree to which they agreed or disagreed with a statement on a 6-point ordinal scale. The researcher intentionally chose a 6-point rather than a 5- or 7-point scale to avoid participants defaulting to neutral and force them "to use cognitive effort to think about their true feelings on the subject" (Garland, 1991). Previous research revealed that in uneven scales where a middle/neutral response is offered it is much more likely to be chosen

(Moors, 2008); further, it is also claimed that respondents answer differently if forced to choose a side (Kalton, Roberts and Holt, 1980; Bishop, 1987).

The debate around analysing and interpreting Likert scale data has arisen from the principle that ordinal data, where the differences between responses (Agree -Disagree) cannot be assumed to be equidistant (Sullivan and Artino, 2013), and therefore cannot be treated as interval continuous data. Norman (2010) argues that studies dating back to the 1930s consistently show that parametric statistics are robust even when the data is not normally distributed. Jamieson (2004) maintains that due to the unknown distances between the categories, only non-parametric tests will produce valid results. Conversely, concurring with Norman (2010), Murray (2013) found no differences in the conclusions drawn from Likert data when using parametric and non-parametric analysis. Carifo and Perla (2008) rationalise that this debate only exists due to the problematic classification of Likert data, and, whether it is ordinal or interval. The researcher believes that the scale used in this research (Strongly Disagree, Disagree, Slightly Disagree, Slightly Agree, Agree, Strongly Agree) can be classified as interval data, much as if the responses had been numbered 1 - 6 on a scale. Bishop and Herron (2015) make the case that there is little need to use Likert scales at all in research and that a Visual Analog Scale (VAS) is more appropriate and suitable for parametric analysis. There is some consensus for the use of parametric tests; this assumption could be met with supporting justifications.

As the literature was inconclusive (Glass et al, 1972; Kuzon et al, 1996; Blaikie, 2003, Jamieson, 2004; Carifio and Perla, 2008; Norman 2010; Murray, 2013; Bishop and Herron, 2015; and, Knapp, 2016), the researcher decided to take a pragmatic approach and tested the data against the remaining assumptions of parametric tests first.

The second assumption is that the sample is not a biased selection. The researcher made all possible efforts to ensure the sample was randomly selected and therefore this assumption is met. In addition, checks were undertaken to ensure the characteristics of the respondents in the sample matched those of the wider *Claire Lynne McCafferty PhD April 2020 Page 104*

population in terms of age, gender, income level etc. This is discussed in more detail in Section 3.11 onwards.

The third assumption is that the observations or measurements are independent. The survey was designed in such a way that they would be completed independently without influence from the researcher or any other individual. For the face-to-face and telephone surveys this can be confirmed, for the online surveys the researcher was not present to police this, however, given the structure of the questions it is clear that the survey is seeking an independent opinion. Therefore it is reasonable to assume that this assumption is also met.

The fourth assumption is that the data is drawn from a normally distributed population. The populations that the researcher is interested in are those that are distributed in terms of age, educational level and income level. None of those could be described as normally distributed. First of all, the sampling frame is the adult population of Scotland so clearly there will be a positive skew for each of the three main independent variables. For example, there are more young adults than old (see Figure 3.2), there are more adults educated below higher education level than at Undergraduate or Postgraduate level (see Figure 3.3), and there are more adults earning lower incomes than high incomes (see Figure 3.4).



Figure 3.2: Age Distribution in Scotland

(Source: adapted from National Records of Scotland, 2017)

Figure 3.2 above demonstrates the positive skew of the adult population in terms of age.





(Source: adapted from National Records of Scotland, 2018)

Figure 3.3 above illustrates the positive skew of the adult population in terms of level of educational attainment.





(Source: adapted from Oxfam, 2018)

Figure 3.4 above demonstrates the positive skew of the adult population in terms of income.

Pallant (2016) argues that with a large enough sample size (over 30) violation of this assumption is immaterial as most of the techniques are tolerant. The sample size for this research is much larger than 30, therefore it can be argued that this assumption does not need to be met.

The fifth, and final, assumption is that the samples are obtained from populations which have equal variances. In order to check that this assumption is met the researcher will carry out a Levene's test for equality of variances once the data is collected (Pallant, 2016). This is also an assumption that must be met for certain non-parametric tests.

Having considered the assumptions that must be met to perform parametric tests, that researcher is confident that their use would be statistically sound. However, in

addition to parametric tests, non-parametric tests will be used for robustness checks. The overall data sample selection and population are detailed from Section 3.11 onwards.

3.4 Strategy for Analysis

The first part of the analysis looks at each question based on the individual demographic factors in turn:

- i. Employed in financial services sector
- ii. Gender
- iii. Age
- iv. Education
- v. Income

Correlation analysis will determine relationships between the above variables. Excluding the first demographic characteristic (the reason for including this was discussed on page 100), these demographic variables have been selected for two key reasons. The first is that they have been used in previous academic research on trust and confidence in the financial services sector (Alesina and La Ferrara in 2002; Moin, Devlin and McKechnie, 2017; and Bannier and Schwarz, 2018). The second is that they have been used in industry research of the same nature (the FCA Financial Lives survey, 2018; YouGov on behalf of Positive Money survey, 2018; Edelman Trust Barometer (UK results), 2019; and GfK Consumer Confidence Index, 2019). The results of these studies were discussed in more detail on pages 99-100. It is important to select the same demographics as previous studies so that the results can be more accurately compared and the resultant research findings contextualised with existing knowledge in the area. Additionally, it is important in primary research, such as this, that the sample is representative of the population (this was verified in Chapter 4).

Finally, step-wise multiple regression analysis will identify only those significant variables for each of the eight theoretical questions, and a final question which asks about general levels of confidence, in turn:

H1: The consumer is confident that the UK retail banking system is stable and secure

H2: The consumer is confident that the money in their retail bank account(s) is protected should their bank fail

H3: The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions

H4: The consumer is confident that their retail bank will do the right thing

H5: The consumer is confident that they are protected from retail bank misconduct

H6: The consumer is confident that they can trust the expertise of their retail bank

H7: The consumer is confident that the price of retail banking products is fair

H8: The consumer is confident that they will not be unlawfully discriminated against by their retail bank

H9: In general, the consumer is confident that as a retail bank customer they are adequately protected

This may remove some of the significant relationships found in the correlation analysis; because certain independent variables are naturally correlated with others. So, while age, education and income may each individually be significantly related to one particular hypothesis, when they are included in a regression model only one of these is significant.

3.5 Research Ethics and Integrity

Ethical approval for this study was granted through formal application to the Faculty Research Ethics and Governance Committee of the Business School at Edinburgh Napier University.

Signed consent was taken from all participants in the Survey Design Groups and from all participants in the main study. Participants were free to withdraw from the

study at any time before submission (afterwards it is impossible to retrieve their specific data since it was anonymous). Additionally, any incomplete questionnaire was treated as a wish to withdraw from the research. This resulted in a significant reduction of valid questionnaires (approximately 20% reduction) that were taken forward for the formal analysis, however, this is deemed to be preferable than including data that cannot be confirmed as being valid.

The University's Codes of Practice, Research Integrity and Data Protection, were adhered to at all times.

The ethical committee did have further questions regarding the involvement of participants aged over 65 years. To exclude, marginalise or treat these participants as if they were less than fully competent would be unethical. No new, or unconsidered, ethical issues arose during the three-stage survey design process or data collection.

3.6 Survey Design

Although conducting a pilot study does not guarantee success in the main study it does increase the probability (van Teijlingen and Hundley, 2001). The main purpose of the three-stage survey design conducted for this research was to empirically test the validity of the questions in the consumer survey.

Ruel (2016) distinguishes between the pretesting and piloting of surveys. Pretesting, he explains, are the methods of validating the survey instrument and its measurements. Pilot testing, however, is the trial of survey administration and procedures. Stage 1 and 2 of the survey design were expert-driven pretests and stage 3 was a respondent-driven pretest. The purpose of the pretest was to ensure the questions measured what they intended to measure, were unbiased, and free of unnecessary technical language and jargon (Ruel, 2016). The researcher did not carry out a small-scale pilot and collect data during the pretests; instead this was an intentional three-stage validation process for the design of the survey questions. The key driver of this method was to close any gaps in theoretical and practical knowledge and understanding by ensuring each question had the same meaning

Claire Lynne McCafferty

regardless of the background of the participants and most crucially, that the meaning was not lost when translated from a theoretical hypothesis into a simple question for the layman. The researcher could find no other studies where this practice had been adopted. This is particularly important when conducting a study with consumers who may not be familiar, or even aware of, regulations or industry-specific terms. It is vital, however, that the denotation is not lost in translation. There is great value in future researchers adopting this approach for closing any gaps between principle, process and experience.

Since the survey questions were derived from the specific rationales for regulation found in the literature there was no need for a focus group to ascertain the areas to survey, but rather, a process to test the actual questions designed for the survey. The survey consisted of two parts; the first section is comprised of demographic questions such as gender, age, education level, employment status and income. These were considered important variables owing to the research conducted by the regulator into consumer vulnerability (Coppack et al, 2015), access to financial services (Collard et al, 2016) and ageing populations (McLoughlin et al, 2017). Clearly, the regulator and academics consider these demographics to be of such significance to warrant collaborative research on several occasions. To re-iterate, the demographics this research focus on are age, education and income.

This follows the risk factors for vulnerability identified by Coppack et al (2015):

- i. Being young, or being 'older old'
- ii. Low literacy, numeracy and financial capability skills
- iii. Low income

The second section of the survey covers the main theoretical questions which were developed from the Literature Review.

A three-stage survey design was undertaken. The first group, 'Theorists' was undertaken with subject academics. The second group, 'Practitioners' was undertaken with industry professionals. The participants in the Theorist and Practitioner groups were provided with a background to the literature in order that

Claire Lynne McCafferty

they could understand the context of the study and which area of the literature each specific question related to. The purpose of this was to 'test' that the rationale for regulation identified translated into the question being asked and, as with the third group, 'End-users', to check that the wording of the question was such that it could be easily understood.

Since no data was collected during these initial studies, and additionally the participants were ineligible to take part in the final study, there was no risk of contamination (van Teijlingen and Hundley, 2001). Ethical approval was obtained for each stage of the survey design and the final data collection.

3.7 Stage 1: Theorists

Three academics from Edinburgh Napier University were invited via email to take part in stage one of the survey design, two of whom agreed and provided feedback. Both participants were male, aged over 55 and previously worked in the financial services industry before moving to academia. The participants were asked to complete consent forms and sign and return these. The participants were then provided with a copy of the intended consumer survey, including a background to the study. All exchanges took place remotely via email and telephone where further clarification was required.

The following questions were asked:

- 1. Do you believe other people may have difficulty in answering this questionnaire? If so, which question may pose a problem?
- 2. Given the purpose of the study outlined in the introduction section, would you suggest a question may perhaps be added or removed to enhance the study?
- 3. Did you read the instructions? Were the instructions clear? Did you object to answering any of the questions?
- 4. Was the layout of the questionnaire clear and logical?

- 5. Is there any question that the wording could be enhanced to clarify?
- 6. How long did it take you to answer the questionnaire?

3.8 Stage 2: Practitioners

In total, 12 industry professionals were invited to take part in stage two of the survey design, four of whom agreed to take part and provided feedback. All invitees were contacts from the retail banking industry that the researcher had through their own professional network. Three participants were male (40-44, 45-49, 60-64 age groups), and one was female (35-39 age group). Three participants are currently employed in the retail banking sector and one had recently retired. Each participant signed and returned a consent form. All exchanges took place remotely via email exchanges and telephone where appropriate. The industry professionals were asked the same questions as the academics (see above).

3.9 Feedback from Stage 1 and Stage 2

The feedback provided from the participants was mostly around clarity of the survey's second section of questions. The first section of the survey covered demographic questions such as gender, age, education level, employment status and income.

The second section covered the main questions which had been developed from the Literature Review. This was helpful to the researcher and appropriate amendments were made prior to Stage 3 being carried out. Changes to the relevant questions are detailed in Table 3.4 overleaf.

| Table | 3.4: | Changes | to t | he | Consumer | Survey | following | Stages | 1 | and | 2 | of |
|-------|-------|---------|------|----|----------|--------|-----------|--------|---|-----|---|----|
| Surve | y Des | sign | | | | | | | | | | |

| RATIONALE | ORIGINAL QUESTION/HYPOTHESIS | REVISED QUESTION/HYPOTHESIS |
|--------------------------|--|--|
| Externalities (macro) | I have confidence in the UK banking system | I have confidence in the UK retail banking system |
| Externalities (micro) | I have confidence that my money is protected should my bank fail | I have confidence that the money in my bank account(s) is protected should my bank fail |
| Asymmetric information | I have confidence that I have enough information from my bank to make good financial decisions | I have confidence that my bank will provide me with sufficient information to make informed decisions |
| Moral hazard | I have confidence that I am protected against bank fraud and misrepresentation | I have confidence that my bank will do the right thing |
| Corruption | I have confidence that, if required, my bank will face the consequences of its actions | I have confidence that I am protected from bank misconduct |
| Paternalism | I have confidence that my bank will help me to make good financial decisions | I have confidence that I can trust the expertise of my bank |
| Distributive Justice | I have confidence that the price of banking products is fair | No change |
| Community Values | I have confidence that I will not be discriminated against by my bank | No change |
| N/A - General | I have confidence that as a bank customer I am protected | I have confidence that as a bank customer I am adequately protected |

3.10 Stage 3: End-users

Five consumers, from across different age groups and both genders, were invited to take part in stage three of the survey design, using the revised core questions from the feedback captured in stages one and two. These were acquaintances of the researcher who resided in England and who therefore would not be taking part in the final study. Two males (Group 1 and Group 2, see below) and three females (two in Group 2 and one in Group 4) provided feedback. The participants were

required to complete and sign a consent form (see Appendix A), four studies took place remotely and one face-to-face study was undertaken. After careful consideration of the feedback, further changes were made and are detailed in Table 3.5 overleaf.

- i. Group 1: 18-29 year olds are defined as 'Young Professionals'
- ii. Group 2: 30-44 year olds are defined as 'Married Professionals'
- iii. Group 3: 45-59 year olds are defined as 'Empty Nesters'
- iv. Group 4: 60+ year olds are defined as 'Retired'

The above groups have been segmented by age. This is a simplified life cycle model historically used by financial institutions due to the close association between age and product usage (Stanley, Ford and Richards, 1985). Of course there are deviations from this in reality and product marketing strategies have moved on to become much more sophisticated, taking into account larger quantities of data and using increasingly advanced techniques to extract and interpret it (Ionut, 2017). However, age profile remains a key demographic variable used in a variety of academic research, industry surveys and government statistics. The purpose of segmenting by age in this research is to ensure that a proportionate number of participants from each group have been included.

| Table 3.5: Changes to the Consumer | Survey Core Questions following Stage |
|------------------------------------|---------------------------------------|
| 3 of Survey Design | |

| RATIONALE | ORIGINAL QUESTION | REVISED QUESTION |
|---------------|-------------------------------------|--|
| Externalities | I have confidence in the UK | I have confidence the UK retail banking |
| (macro) | retail banking system | system is stable and secure |
| Externalities | I have confidence that the | I have confidence that the money in my |
| (micro) | money in my bank account(s) | retail bank account(s) is |
| | is protected should my bank | protected should my |
| | fail | bank fail |
| Asymmetric | I have confidence that my | I have confidence that my retail bank |
| information | bank will provide me with | will provide me with sufficient |
| | sufficient information to make | information to make informed decisions |
| | informed decisions | |
| Moral hazard | I have confidence that my | I have confidence that my retail bank will |
| | bank will do the right thing | do the right thing |
| Corruption | I have confidence that I am | I have confidence that I am protected from |
| | protected from bank | retail bank misconduct |
| | misconduct | |
| Paternalism | I have confidence that I can | I have confidence that I can trust the |
| | trust the expertise of my bank | expertise of my retail bank |
| Distributive | I have confidence that the | I have confidence that the price of retail |
| Justice | price of banking products is | banking products is fair |
| | fair | |
| Community | I have confidence that I will not | I have confidence that I will not be |
| Values | be discriminated against by my bank | unlawfully discriminated against by my |
| | | retail bank |
| N/A - | I have confidence that as a | In general, I have confidence that as a |
| General | bank customer I am | retail bank customer I am adequately |
| | adequately protected | protected |

The term 'retail' was added to all the questions and clarification of what this term means was included in the introduction to the survey (see Appendix B). 'Don't know' was added to the last question as one of the consumers expressed this during the End-User test. The consumers who took part in the design group were not eligible to participate in the final study.

3.11 Reliability and Validity Considerations for Data Collection

Reliability of data potentially enables study findings to be generalised to the wider population (Braham and Finch, 2004). The level of confidence in the validity of the data gives it credence as empirical evidence for a claim (Roberts and Johnson, 2015). The next sections will establish if the data can justly be claimed to be a true representation of reality and if there were any intervening variables which might reduce confidence in the accuracy of the data collected.

3.12 Sample

The first consideration for the researcher was the population to be studied. The aim of the research was to measure retail bank consumer confidence levels. The research was limited to Scotland and therefore the adult population (18+ years) residing in Scotland was the target population (since almost all adults in Scotland are 'banked') CAB Scotland (2010). According to the 2016 Scottish government data (National Records of Scotland, 2017) the adult population was 4,372,939. To calculate a representative sample the researcher used Slovin's Formula (Tejada and Punzalan, 2012). Slovin's Formula is used to calculate an appropriate sample size from a population.

Slovin's formula: $n = N/1 + Ne^2$

Where *n* is the sample size, *N* is the total population (4,372,939) and *e* is the margin of error. Following the study by Tejada and Punzalan (2012), and well established practices in the wider research area, the researcher determined a standard confidence level of 95% to be appropriate for this research. Thus, the error tolerance, or margin of error, is 0.05.

1 + 4,372,939 x 0.05²

n = <u>399.93</u>

Claire Lynne McCafferty

PhD April 2020

Using this formula, a minimum sample size of 400 useable surveys was required for a representative sample. In total, 538 surveys were completed, and of these, 432 were usable. Incomplete questionnaires were discounted as per the instructions on the consent form. Any incomplete questionnaire was treated as a wish to withdraw from the research. This resulted in a significant reduction of valid questionnaires (approximately 20% reduction) that were taken forward for the formal analysis, however, this is deemed to be preferable than including data that cannot be confirmed as being valid.

- i. 138 paper questionnaires, with 126 being useable
- ii. 400 electronic questionnaires, with 306 being useable

The target sample size (see Table 3.6 below) was exceeded with a total of 432 useable surveys being collected. A comparison of the group population and sample representation can be seen in Table 3.7 overleaf.

| Group | Population | %age of Population | Sample Required |
|-------|------------|--------------------|-----------------|
| 18-29 | 865,468 | 20% | 80 |
| 30-44 | 1,017,304 | 23% | 92 |
| 45-59 | 1,169,763 | 27% | 108 |
| 60+ | 1,320,404 | 30% | 120 |

Table 3.6: Sample Required for each Strata

The National Records of Scotland Mid-2016 population statistics were used as the sampling frame. The researcher then created a basic segmentation model as discussed in Section 3.10. Using the figure of 4,372,939 obtained for the total adult population residing in Scotland (National Records of Scotland, 2017) the sample requirements were calculated (see Table 3.6 above).

| Group | Sample Size | %age of target sample achieved | Target %age of population | Actual %age achieved |
|-------|-------------|--------------------------------------|---------------------------|-------------------------|
| 18-29 | 71 | 89% | 20% | 16% |
| 30-44 | 125 | 136% | 23% | 29% |
| 45-59 | 150 | 138% | 27% | 35% |
| 60+ | 86 | 72% | 30% | 20% |

| Table 3.7: Sample | Achieved for | each Strata |
|-------------------|--------------|-------------|
|-------------------|--------------|-------------|

A chi-square goodness-of-fit test indicates there was a significant difference in the proportions of the age groups identified in the current sample (18-29, 16%; 30-44, 29%; 45-59, 35%; 60+, 20%) as compared with the values (18-29, 20%; 30-44, 23%; 45-59, 27%; 60+, 30%) that were obtained from official published population statistics, x^2 (3, n = 432) = 33.53, p <0.001. However, due to the large population, and as the sample required is only 0.01% of the population, there will always be a significant difference.

Table 3.7 above illustrates the differences in representation of the four groups in the population and sample. Although the sample does not match the population age categorisation exactly, it reflects the sample in the FCA Financial Lives Survey (2017).

- i. My study: age 18-29 16%; FCA study: age 18-24 11%
- ii. My study: age 30-44 29%; FCA study: age 25-44 33%
- iii. My study: age 45-59 35%; FCA study: age 45-65 33%
- iv. My study: age 60+ 20%; FCA study: age 65+ 23%

Additionally, the FCA Financial Lives Survey (2017) used postcode, rather than targeting age or gender, indicating that these demographics are not considered critical to a relevant and valid survey in this particular subject area. For this research, postcode targeting was not possible due to the time and cost involved in obtaining the data and software. The FCA hired an external firm to gather this information for them (McLoughlin et al, 2017). However, in this study every effort to capture data from various regions of Scotland was made, both face to face and over the telephone.

3.13 Data Collection

The researcher employed a mixed data collection, whereby 71% of responses were achieved via an online survey, with 29% paper surveys attained to ensure the views of non-internet users were represented. These figures represent the total number of useable surveys.

Any incomplete questionnaires were deemed unusable. Clause 5 of the participants consent form stated that 'a partial completion will indicate a wish to withdraw. Partial completions will not be included in the research and will be void'.

The data collection ran for a total of nine months. The online survey tool Novi Survey was utilised and the link distributed via email and shared on social media platforms such as Twitter, Facebook, Mumsnet, PistonHeads, Digital Spy, LinkedIn and Money Saving Expert. The link was removed from PistonHeads forum as it contravened their rules of 'promotion' as it was a link to a survey. The post was also removed from the Digital Spy forum as it contained a link to a third-party site. The LinkedIn share was not publicly posted but was instead sent via private message to the researcher's connections that do not work in the financial services industry. Since the researcher has previously worked in the industry for over a decade this was done to avoid skewed results.

The paper questionnaires were distributed in shopping centres in Dundee, Edinburgh and Inverness. Additionally, a pensioner's coffee morning was attended in a village in Angus to capture this age group's opinions. Lastly, telephone surveys

were conducted with households in the South West of Scotland. The researcher made efforts to capture participants from different geographic areas and non-internet users.

The next sections of the data description concentrates on gender and age groups, and in particular on the 'at risk' groups identified by the FCA, older consumers, low income levels and low level of education. Additionally, those currently employed in the financial services sector are considered.

3.13.1 Gender and Age Groups

Figure 3.5 below illustrates the gender split of participants in the electronic survey, a ratio of females to males of 2.36:1.



Figure 3.5: Gender: Electronic Surveys

The paper survey (Figure 3.6, overleaf) reveals a more balanced split of 54% females, 44% males (1.2:1) and 2% did not wish to reveal their gender. Combined, approximately one third of participants were male, and almost two thirds were female

(1.92:1). Clearly, this is not representative of the population; however, as this study was not specifically looking at differences between these variables it was not considered a significant limitation. Additionally, several studies measuring trust and confidence in the financial services sector since the Global Financial Crisis of 2007-2009 did not report any significant gender differences in their findings. Including a study by Moin, Devlin and McKechnie (2017); the FCA Financial Lives survey, 2018; YouGov on behalf of Positive Money survey, 2018; Edelman Trust Barometer (UK results), 2019; and GfK Consumer Confidence Index, 2019, are a few recent examples.



Figure 3.6: Gender: Paper Surveys

The bar charts overleaf represent the proportion of respondents by age group for both electronic (Figure 3.7) and paper format (Figure 3.8).

The FCA's Financial Lives Study (2017) found that confidence in managing financial matters increased with age. However, in addition to finding the youngest age group

(18-24 year olds: 30%) had the least confidence in the financial services industry they also found the oldest group (age over 85: 31%) had low confidence.



Figure 3.7: Age: Electronic Surveys





These have been combined and are compared to the total population in Figure 3.9 overleaf. This demonstrates the representativeness of the sample by age group. Overall, the sample is representative, with only Group 4 being significantly under-represented. The researcher targeted males and older consumers with the paper version of the Consumer Survey to address the sapling problems of the online version. This will be discussed in more detail in the next section.




3.14 Vulnerable Groups

Although the term 'vulnerable' is hard to define the FCA consider a vulnerable person to be someone who, "due to their personal circumstances, is especially susceptible to detriment" (Coppack et al, 2015, p.5). The FCA have carried out several studies into these groups, specifically consumer vulnerability (Coppack et al, 2015), access to financial services (Collard et al, 2016) and ageing populations (McLoughlin et al, 2017).

To reiterate, some of the examples of risk factors for vulnerability identified in the FCA's research were:

- i. Low literacy, numeracy and financial capability skills
- ii. Physical disability
- iii. Severe or long-term illness
- iv. Mental health problems

- v. Low income and/or debt
- vi. Caring responsibilities (including operating a power of attorney)
- vii. Being 'older old' for example over 80, although this is not absolute (may be associated with cognitive or dexterity impairment, sensory impairments such as hearing or sight, onset of ill-health, not being comfortable with new technology)
- viii. Being young (associated with less experience)
- ix. Change in circumstances (e.g. job loss, bereavement, divorce)
- x. Lack of English skills
- xi. Non-standard requirements or credit history (e.g. armed forces personal returning from abroad; ex-offenders; care-home leavers; recent immigrants)

It is beyond the scope of this research to examine all of the factors identified by the FCA. The researcher, for ethical reasons, is unable to access particular vulnerable groups and also does not have access to other specific groups, for example armed forces personnel, ex-offenders, and immigrants. The specific demographic groups that will be tested for any significant difference in levels of confidence in this thesis are: age, low income and low level of education.

3.14.1 Older Consumers

Although older consumers are not necessarily vulnerable it has been recognised by the FCA (Collard et al, 2016, p.7) that, "There are risks that older consumers' financial services needs are not being fully met, resulting in exclusion, poor customer outcomes and potential harm". While the FCA Occasional Paper looked at consumers aged 55 years and over, as this is the age at which people can draw a defined benefit pension, the researcher will include only those aged over 60 in this group (Group 4) since this research is not product focused. There are a broad range of products and services covered by regulation which could be considered to be a more specialist or particularly focused area for potential mis-selling due to their very nature: complexity; cost; duration of consumption.

In total, 20% of the participants who took part in this study were aged over 60. This is lower than the 28% of over 60s in the Scottish population (National Records of Scotland, 2017). However, as the researcher only visited public locations and the sample was randomly selected access to this age group was restricted.

3.14.2 Low Income

According to the Office for National Statistics (ONS) Annual Survey of Hours and Earnings (2016), 0.8% of the Scottish adult population earned below the minimum wage. As data on hours worked has not been collected in this research the figure is based on respondents who indicated that they work full-time but earn less than \pounds 13,500 per annum. The figure for participants in this study is 2.8% which is close to the population figure of 2.4%.

According to the ONS the median annual salary was £27,600 in 2016. Participants in this study, who work full-time but earn below the national median made up 36% of the sample. This indicates that, on average, the participants in this study were better-off than the general population.

Other income figures are:

- i. 26.4% of participants in this study earn above the national median but below the higher tax rate threshold (£43,000 in tax year 2017/18).
- ii. The HM Revenue and Customs statistics (2015) estimate 85.1% of the population will be non-higher rate tax payers in 2017/18. A similar figure of 84.3% of participants in the study fell into this category.
- 14.4% of participants in this study earn above the higher rate tax rate threshold but below the additional rate threshold. This is in line with HMRC statistics of 13.7% estimated for 2017/18.
- iv. 1.4% of participants in this study earn above the additional rate tax threshold (over £150,000 in tax year 2017/18). This is in line with the HMRC statistics of 1.2% estimated for 2017/18.

The income distribution for all participants is presented overleaf in Figure 3.10.





3.14.3 Low Level of Education

A study undertaken by the FSA (2006a) in collaboration with Bristol University's Personal Finance Research Centre found that there was a direct correlation between levels of education and qualifications, and levels of financial capability. This is important to this study as a low level of capability could lead to financial exclusion. Llewellyn and the FSA (1999) argue that if consumers do not feel confident enough to purchase suitable financial products it has the same detrimental effect as being mis-sold unsuitable products.

In total, 108 of all respondents held a university degree, this is 25% compared to the Scottish national average of 26% recorded in the Scottish 2011 Census. This indicates that the participants who took part in this research were representative of the national average.

Participants who had obtained only the mandatory level of education (Secondary School level, not including current students) represented 23% of the sample.

Claire Lynne McCafferty

Page 129







Figure 3.12: Education Levels: All

3.15 Persons Working in Financial Services

Persons working in the financial services sector have been a focus of this research as it could be expected that they would have more knowledge about the industry and the regulatory requirements and consumer protection in place than the average layman.

The financial services sector is important to the Scottish economy, employing almost 200,000 people and generating over £8.8bn for the UK and Scottish economies (Office of the Secretary of State for Scotland, 2014). Based on the adult population figures, from the Mid-2016 population estimates by the Scottish government (National Records of Scotland, 2017), of 4,372,939, this is equal to 4.6% of the population working in the industry.

In total, 18 of the 432 participants, or 4.2%, currently work in financial services, which is in line with the national average.

3.16 Data Description Conclusion

The above has described the data obtained in terms of its reliability to represent the wider Scottish population and validity with respect to the appropriateness of the data being gathered in this study. It can be concluded that the sample of data collected in this study is representative is terms of:

- i. Size
- ii. Income distribution of participants
- iii. Education levels of participants
- iv. Participants employed in the financial services sector

In terms of the groups used to stratify the sample, Groups 1, 2 and 3 are representative, however Group 4 (over 60+ years), was under-represented due to the random sampling technique employed. It could be expected that fewer elderly participants are likely to participate online, and secondly, they may not be so readily available in busy public places. The researcher did attend a pensioner's coffee morning in an attempt to capture the opinions of this age group. Gender was not considered a dependent variable and therefore, the study did not seek to stratify based on this demographic.

In the next chapter an analysis of the collected data has been undertaken focusing on the groups described above. The participants were asked to rank their confidence levels; the hypotheses were derived from the literature, and are as follows:

H1: The consumer is confident that the UK retail banking system is stable and secure

H2: The consumer is confident that the money in their retail bank account(s) is protected should their bank fail

H3: The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions

H4: The consumer is confident that their retail bank will do the right thing

H5: The consumer is confident that they are protected from retail bank misconduct

H6: The consumer is confident that they can trust the expertise of their retail bank

H7: The consumer is confident that the price of retail banking products is fair

H8: The consumer is confident that they will not be unlawfully discriminated against by their retail bank

H9: In general, the consumer is confident that as a retail bank customer they are adequately protected

In the next chapter an analysis of the collected data has been undertaken focusing on the groups outlined in this chapter:

- i. Employed in the financial services sector
- ii. Gender
- iii. Age
- iv. Education
- v. Income

The analysis will establish differences in levels of confidence between each demographic group, the direction of confidence, and the combination of demographic factors that have the most impact on levels of confidence in the specific areas of regulation.

CHAPTER 4: DESCRIPTIVE STATISTICS AND CORRELATION ANALYSIS

4.1 Introduction

In this chapter the data from the core theoretical survey questions will be analysed, focusing on specific characteristics of the participants, as discussed in the Chapter 3:

- i. Employed in the financial services sector
- ii. Gender
- iii. Age
- iv. Level of education
- v. Level of income

In the Methodology chapter the debate around the use of parametric and nonparametric analysis for Likert scale data was discussed. This is based on the arguments concerning whether Likert scale data can be treated as interval variables rather than ordinal variables. One of the assumptions that must be met for parametric tests to be valid is that the data is interval or ratio data (Field, 2013). As the literature was inconclusive (Glass et al, 1972; Kuzon et al, 1996; Blaikie, 2003, Jamieson, 2004; Carifio and Perla, 2008; Norman 2010; Murray, 2013; Bishop and Herron, 2015; and, Knapp, 2016), the researcher tested the data against the remaining assumptions of parametric tests first. After screening and cleaning the data and describing the characteristics of the sample to ensure it was representative of the population (discussed in the previous chapter) the data was tested to check that it met the assumptions of:

- i. a normally distributed sampling distribution
- ii. homogeneity of variance, and
- iii. independence (Field, 2013)

4.1.1 Tests for Normality

In order to check this assumption of parametric statistical procedures is met there are various tests that can be conducted. However, due to the large sample size,

tests such as Skewness, Kurtosis and Kolmogorov-Smirnov were not required (Tabachnick and Fidell, 2014). Accordingly, the researcher undertook a visual inspection of the histograms for the total sample responses to each of the survey's 'levels of confidence' questions. The data presented as moderately negatively skewed. The researcher undertook one further test of normality, the Shapiro-Wilk test, which confirmed that the data was not normally distributed (see Appendix C).

As discussed in the previous chapter, the central limit theorem (CLT) indicates that if a sufficiently large random sample size is taken from a population with a determinate level of variance, the mean of the samples (aggregated) will be approximately equal to the mean of the population (Kwak and Kim, 2017). Further, the CLT indicates that whether the underlying distribution is normal or skewed, in samples of over 30, the sample will be normal (Pallant, 2016; Seaman, Allen and Zetumer, 2018).

4.1.2 Outliers

The boxplots produced in SPSS were visually inspected for outliers. In four of these questions (44.4%) outliers were found. However, the 5% trimmed mean, where the lowest 5% and highest 5% of the data is excluded, for each question was very close to the mean so did not require further investigation (Pallant, 2016).

4.1.3 Scale Reliability

Lastly, a test to check the reliability of the scale was undertaken. The participants were asked to rate their levels of confidence on a scale of 1 (Strongly Disagree) to 6 (Strongly Agree). It is important that the scale selected in a study is reliable in terms of its 'internal consistency' (Pallant, 2016). That is, how well the survey is measuring what the study required it to measure. In addition to the three-stage empirical testing of the validity of the questions, the researcher has also applied one of the most commonly used indicators of internal consistency, Cronbach's alpha coefficient to test reliability (Pallant, 2016). A low score may indicate a low number of questions, or poor correlation between items, conversely, a high score may indicate that some items are redundant (Tavakol and Dennick, 2011). Tavakol and

Claire Lynne McCafferty

Dennick (2011) cited studies which suggested acceptable values of alpha, ranging from 0.70 to 0.95 (see, Nunnally and Bernstein, 1994 and Bland and Altman, 1997). Cronbach's alpha score for the 'level of confidence' questions in this research is 0.922 (see Table 4.1 below).

Table 4.1: Reliability statistics

| Cronbach's Alpha | Cronbach's Alpha Based on Standardised Items | No of Items |
|------------------|--|-------------|
| 0.922 | 0.922 | 9 |

In summary, the data met the assumptions, where required, of distribution, homogeneity of variance and independence

4.1.4 Labelling of the Variables

Table 4.2 overleaf, clearly classifies the variables into Nominal and Ordinal. Levels of confidence were measured using a Likert scale. It has become common practice, although controversial, to assume interval status for Likert derived data (Blaikie, 2003).

| Survey Question: Demographics | Variable | Classification |
|---|-------------|----------------|
| Gender | Independent | Nominal |
| Age | Independent | Ordinal |
| Education level | Independent | Ordinal |
| Employment status | Independent | Nominal |
| Income level | Independent | Ordinal |
| Work in Financial Services | Independent | Nominal |
| Survey Question: Theoretical | Variable | Classification |
| H1: The UK retail banking system is stable and secure | Dependent | Ordinal |
| H2: The money in my retail bank account(s) is protected should my bank fail | Dependent | Ordinal |
| H3: My retail bank will provide me with sufficient information to make informed decisions | Dependent | Ordinal |
| H4: My retail bank will do the right thing | Dependent | Ordinal |
| H5: I am protected from retail bank misconduct | Dependent | Ordinal |
| H6: I can trust the expertise of my retail bank | Dependent | Ordinal |
| H7: The price of retail banking products is fair | Dependent | Ordinal |
| H8: I will not be unlawfully discriminated against by my retail bank | Dependent | Ordinal |
| H9: In general, as a retail bank customer I am adequately protected | Dependent | Ordinal |

Table 4.2 Variable Classification

As can be seen from the above table, all of the hypotheses are classified as ordinal scale data which, as discussed below, has some contradictory advice concerning analysis.

Claire Lynne McCafferty

4.2 Methods of Data Analysis and Presentation of Data

Following the statistical tests of the assumptions the researcher adopted parametric statistical techniques to compare groups for this research; the independent-samples t-test and one-way between-group analysis of variance. Due to the controversy surrounding the treatment of ordinal scales and interval scales (Jamieson, 2014) non-parametric tests were used as robustness checks, and as an opportunity to contribute to the literature in this area.

Table 4.3 overleaf, outlines the tests adopted in the preliminary analysis, and any observed variance.

| Demographic | Statistical Test | Robustness Check | Outcome |
|--------------------|------------------|----------------------|-----------------------------|
| Employed in the | Parametric test: | Non-parametric test: | Both the parametric and |
| financial services | Independent- | Mann-Whitney U | non-parametric tests |
| sector | samples t-Test | Test | revealed the same |
| | | | findings |
| Gender | Parametric test: | Non-parametric test: | Both the parametric and |
| | Independent- | Mann-Whitney U | non-parametric tests |
| | samples t-Test | Test | revealed the same |
| | | | findings |
| Age | Parametric test: | Non-parametric test: | Both the parametric and |
| | One-way | Kruskal-Wallis Test | non-parametric tests |
| | between-group | | revealed the same |
| | analysis of | | findings |
| | variance | | |
| | Doromotrio tooti | Non noromotria tooti | Both the nerometric and |
| Level of education | | Kruekel Wellie Test | Boin the parametric tests |
| | One-way | Kluskal-wallis Test | non-parametric tests |
| | between-group | | |
| | analysis of | | indings |
| | variance | | |
| Level of income | Parametric test: | Non-parametric test: | Non-parametric test |
| | One-way | Kruskal-Wallis Test | recorded a difference |
| | between-group | | across the groups under |
| | analysis of | | 'fraudulent activities' but |
| | variance | | no difference recorded |
| | | | between the groups. |
| | | | All other tests revealed |
| | | | the same findings |

Table 4.3 Observed Variance of Parametric vs. Non-Parametric Tests

To explore the differences between groups an independent sample t-test was used where there were two different groups of variables (employed in financial services sector/not employed in financial services sector and male/female) and a one-way analysis of variance (ANOVA) was used where there were more than two independent groups (age, education and income). As ANOVA can only determine whether groups differ and not where the significant difference is, post-hoc comparisons were performed to establish which groups were significantly different from one another. The non-parametric alternatives adopted were the Mann-Whitney U Test (for the Independent-samples t-Test) and Kruskal-Wallis Test (for the ANOVA).

To detect and describe the relationship among variables and explore the association between pairs of variables, correlation analysis was used. Pearson productmoment correlation coefficient (r) was used to describe the strength and direction of the linear relationship between an independent variable and dependent variable. Further, Spearman Rank Order Correlation (rho), the non-parametric alternative, was used as a robustness test.

4.3 Descriptive Statistics

In this section each hypothesis based on each of the individual demographic factors is examined in turn. There are certain assumptions based on previous studies conducted by the regulator:

- i. Employed in the financial services sector. Although the researcher could find no specific studies which compare the levels of confidence in these two groups, it would not be unreasonable to surmise that those employed may be more likely to have a higher level of confidence, based on their industry knowledge and expertise.
- ii. Gender. The researcher could find no recent studies which suggested that gender played a significant role in the levels of consumer confidence in the retail banking sector. Additionally, recent studies did not base their methodology on gender differences or present their results by analysing the

effect of gender. For these reasons, it would be rational to conclude that there would be no difference found in the results of this study, which looks specifically at confidence in regulation of the sector.

- iii. Age. The FCA associate being young with less experience and being 'older old' (over 80) with increased vulnerability (Coppack et al, 2015). Although this does not lead to a conclusion about whether these groups will have lower or higher levels of confidence in retail banking regulation, it does indicate that the results of these groups would be of particular relevance.
- iv. Education. The FCA associate low literacy, numeracy and financial capability skill with financial vulnerability. The results of this group would be relevant to both the regulator (who looks to protect the most vulnerable groups) and the banks (who profile consumers using factors such as education since it often leads to higher net worth). However, higher levels of confidence in dealing with financial matters does not necessarily equate to higher levels of confidence in retail banking regulation. Given the focus of regulator initiatives on vulnerable groups it may actually suggest lower education groups would have higher levels of confidence in regulation.
- v. Income. Low income has been identified as a higher risk factor for financial vulnerability. Given the profiling used by banks to attract higher net worth consumers the results would be of particular interest to them, and to the regulator who look to ensure vulnerable groups are protected. It would not be entirely unreasonable to surmise that higher income consumers, who have typically more exposure to the system, may feel more confident in dealing with financial matters. However, as mentioned above, it is not possible to predict whether those on a higher income have higher levels of confidence in regulation.

4.3.1 Externalities

The first two theoretical questions of the survey tested how confident consumers were that they were protected against failure at a macro and micro level. Question

one concerned the risk of contagion, or systemic risk. That is, the failure of one institution could cause the collapse of other, solvent institutions.

As discussed in Chapter 2, the interconnectedness of banks can lead to issues of contagion, that is, the failure of one institution can have a spillover effect and lead to the failure of otherwise solvent banks. De Nicolo et al (2012) emphasise that externalities arising from interconnectedness are particularly great in Systemically Important Financial Institutions (SIFIs). Further, they argue that the macroprudential tool that can address the externalities arising from all three categories is capital requirements. Banks are required to hold capital reserves to enable them to absorb any unexpected losses. The first consumer survey question will test the hypothesis that consumers are confident in this area that regulation seeks to address. The statement participants were asked to consider was, *"I have confidence the UK retail banking system is stable and secure"*.

Currently employed in the Financial Services sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is a significant difference in the scores for non-Financial Services employees (M = 3.86, SD = 1.190) and Financial Services employees (M = 4.83, SD = 0.857; t(430) = -3.45, p = 0.001, two-tailed).

A Mann-Whitney U Test was used as a robustness check and also revealed there is a significant difference in the confidence levels of FS employees (Md = 5.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 5410.5, z = 3.374, p = 0.001, r = 0.16.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a significant difference in levels of confidence of those who are employed in financial services and those who are not
- ii. Participants who are employed in the financial services sector *agree*, whilst participants who are not employed in the financial services sector

slightly agree that they have confidence the UK retail banking system is stable and secure

iii. Therefore, those employed in the sector have more confidence

This is graphically represented in Figure 4.1 below.

Figure 4.1 I have confidence the UK retail banking system is stable and secure: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

This finding is not unexpected since those employed in the financial services sector are more likely to be aware of the Capital Requirements regulations.

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.01, SD = 1.22) and females (M = 3.85, SD = 1.17; t (427) = 1.34, p = 0.180, two-tailed).

A Mann-Whitney U Test was used as a robustness check and also revealed no significant difference in the confidence levels of males (Md = 4.00, n = 147) and females (Md = 4.00, n = 282), U = 18,981.5, z = -1.488, p = 0.137, r = -0.072.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no significant difference in the levels of confidence of males or females
- ii. Both males and females *slightly agree* that they have confidence the UK retail banking system is stable and secure

This is graphically represented in Figure 4.2 below.

Figure 4.2: I have confidence the UK retail banking system is stable and secure: male *vs.* female participants



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was no statistically significant difference at the *p* < 0.05 level in confidence scores for the four age groups: *F* (3, 428) = 1.483, p = 0.218.

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across the four different age groups (Group 1, n = 71: 18-29yrs, Group 2, n = 126: 30-44yrs, Group 3, n = 149: 45-59yrs, Group 4, n = 86: 60+yrs, x_2 (3, n = 432) = 4.19, p = 0.241, all groups recorded median values of 4.00.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence between age groups.
- ii. All age groups *slightly agree* that they have confidence the UK retail banking system is stable and secure.

This is graphically represented in Figure 4.3 overleaf.

Figure 4.3: I have confidence the UK retail banking system is stable and secure: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< 0.05 level in confidence scores for the four education groups: F(3, 428) = 2.689, p = 0.046. However, post-hoc comparisons using the Tukey HSD test indicated that the mean score for each group did not differ significantly from any groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 8.30, p = 0.040, however, all groups recorded a median value of 4.00. Dunn's pairwise comparisons

with adjusted *p*-values (Bonferroni correction) revealed that despite the significant overall difference, none of the specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Both the parametric and the non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence between education levels.
- ii. Participants with different levels of education *slightly agree* they are confident the UK retail banking system is stable and secure.

This is graphically represented in the chart Figure 4.4 below.

Figure 4.4: I have confidence the UK retail banking system is stable and secure: level of education



Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2:

£13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was no statistically significant difference at the p< 0.05 level in confidence scores for the five income groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 7.21, p = 0.125, all groups recorded median values of 4.00.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence between income levels.
- ii. Participants at all levels of income *slightly agree* they have confidence the retail banking system is stable and secure.

This is graphically represented in Figure 4.5 overleaf.

Figure 4.5: I have confidence the UK retail banking system is stable and secure: level of income



Independent-Samples Median Test

4.3.1.1 Microprudential regulation is aimed at preventing the failure of individual financial institutions (Hanson, Kashyap, and Stein, 2011). In 1983 Diamond and Dybvig published a seminal paper providing a model of bank runs and associated financial crises. The principal mechanism proposed to avoid bank runs was the government provision of deposit insurance. In their model they assert that deposit insurance is critical for the stability of the banking system. This was based on the premise that consumers would not panic and try to withdraw their demand deposits if they knew that their deposits were safe. In the UK the deposit insurance scheme is run by the Financial Services Compensation Scheme (FSCS). The second question focused on the rationale that consumer deposits should be protected if a bank becomes insolvent, *"I have confidence that the money in my retail bank account(s) is protected should my bank fail"*.

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is a significant difference in the scores for non-FS employees (M = 4.45, SD = 1.112) and FS employees (M = 5.17, SD = 0.707; t (20.8) = -4.10, p = 0.001, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed there is a significant difference in the confidence levels of FS employees (Md = 5.00, n = 18) and non-FS employees (Md = 5.00, n = 414), U = 5186, z = 2.991, p = 0.003, r = 0.144.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in the levels of confidence of those who are employed in FS and those who are not
- ii. The parametric test reported a higher mean score (5.17) for those employed in FS compared to those not employed in FS (4.45), however, the nonparametric test recorded a median of 5, *agree* for both groups
- iii. This is due to the small sample size of the group employed in the sector (n = 18)
- iv. Figure 4.6 overleaf reveals that although the medians were the same, a higher proportion of respondents in the FS Group recorded a level of confidence above the median than in the non-FS Group.

Figure 4.6: I have confidence that the money in my retail bank account(s) is protected should my bank fail: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.55, SD = 1.16) and females (M = 4.45, SD = 1.07; t (427) = 0.928, p = 0.354, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 5.00, n = 147) and females (Md = 5.00, n = 282), U = 19,215.5, z = -1.318, p = 0.188, r = -0.064.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence of males and females.
- ii. Both groups *agree* they have confidence that the money in their retail bank account(s) is protected should their bank fail.

This is graphically represented in Figure 4.7 below.

Figure 4.7: I have confidence that the money in my retail bank account(s) is protected should my bank fail: male *vs.* female



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was no statistically significant difference at the *p* < 0.05 level in confidence scores for the four age groups: *F*(3, 428) = 1.90, *p* = 0.129.

Claire Lynne McCafferty

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29yrs, Group 2, n = 126: 30-44yrs, Group 3, n = 149: 45-59yrs, Group 4, n = 86: 60+yrs, x_2 (3, n = 432) = 4.57, p = 0.206, all Groups recorded median values of 5.00.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the confidence levels across the age groups.
- ii. All age groups *agree* they have confidence that the money in their retail bank account(s) is protected should their bank fail.

This is graphically represented in Figure 4.8 below.

Figure 4.8: I have confidence that the money in my retail bank account(s) is protected should my bank fail: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided

Claire Lynne McCafferty

into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was no statistically significant difference at the p< 0.05 level in confidence scores for the four education Groups: F(3, 428) = 1.7, p = 0.157.

A Kruskal-Wallis Test revealed no significant difference in levels of confidence across the four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 5.57, p = 0.135, Groups 1, 3 and 4 recorded the same median score (Md = 5.00), and Group 2 recorded a median value of 4.50.

The parametric and non-parametric tests revealed the same findings:

- i. There is no statistically significant difference in confidence levels between education levels.
- ii. Only Group two (College) recorded a lower level of confidence: between *slightly agree* and *agree*
- iii. Overall, the participants *agree* they have confidence that the money in their retail bank account(s) is protected should their bank fail.

This is graphically represented in Figure 4.9 overleaf.

Figure 4.9: I have confidence that the money in my retail bank account(s) is protected should my bank fail: education level



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the p < 0.5 level in confidence scores for the five income groups: F(4, 427) = 4.567, p = 0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean scores for the groups did not differ significantly from each other; Group 1 (M = 4.41, SD = 1.075), Group 2 (M = 4.20, SD = 1.206), Group 3 (M = 4.64, SD = 1.032), Group 4 (M = 4.79, SD = 1.010), and Group 5 (M = 5.17, SD = 0.408).

A Kruskal-Wallis Test revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62:

Claire Lynne McCafferty

£43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 17.76, p = 0.001. Group 2 (£13,501-£27,600) recorded a lower median score (Md = 4.00) than the other four groups, which all recorded median values of 5.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 2 (£13,501-£27,600) was statistically different to Group 3 (£27,601-£43,000), p =0.029, r = -0.194. Group 2 was also statistically different to Group 4 (£43,001-£150,000), p = 0.009, r = -0.246. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

The parametric and non-parametric tests revealed slightly different findings:

- i. There is a statistically significant difference in the levels of confidence across the income level groups
- ii. The parametric test found no significant difference between the groups
- iii. The non-parametric test found a statistically significant difference between those earning £13,501-£27,600 (*slightly agree*) and two other income Groups: £27,601-£43,000 and £43,001-£150,000 (*agree*)
- iv. Overall, the respondents *agree* they are confident that the money in my retail bank account(s) is protected should my bank fail: education level

This is graphically represented in Figure 4.10 overleaf.

Figure 4.10: I have confidence that the money in my retail bank account(s) is protected should my bank fail: income level



Independent-Samples Median Test

4.3.2 Asymmetric Information

"I have confidence that my retail bank will provide me with sufficient information to make informed decisions"

In financial services it may be the case that the consumer does not have adequate information to make a fully informed decision. Additionally, the problem of asymmetric information occurs as the consumer has less information than the 'expert' supplying the financial product and/or advice. Consumers are unable to inform themselves adequately so as to avoid making decisions which may disadvantage them (Wood, 2001). This survey question was designed to test how confident consumers are that regulation can overcome this market failure.

Currently employed in the FS sector: An independent-samples t-test wasconducted to compare the confidence scores of those employed in the financialClaire Lynne McCaffertyPhD April 2020Page 157

services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 4.06, SD = 1.16) and non-FS employees (M = 4.18, SD = 1.15; t (430) = 0.455, p = 0.649, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 4.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 3362.5, z = -0.738, p = 0.461, r = -0.36.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence of the two groups
- ii. Both those who are employed in financial services and those who are not, *slightly agree* they have confidence that their retail bank will provide them with sufficient information to make informed decision.

This is graphically represented in Figure 4.11 overleaf.

Figure 4.11: I have confidence that my retail bank will provide me with sufficient information to make informed decisions: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.21, SD = 1.18) and females (M = 4.17, SD = 1.12; t (427) = 0.380, p = .704, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 5.00, n = 147) and females (Md = 4.00, n = 282), U = 20,002.5, z = -0.626, p = 0.532, r = -0.03.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in levels of confidence of males and females
- ii. However, males *agreed* and females *slightly agreed* they have confidence that their retail bank will provide them with sufficient information to make informed decision
- iii. Therefore although males recorded a higher level of confidence in this question it is not of statistical significance
- iv. Overall, the results indicate they *slightly agree* with this statement.

This is graphically represented in Figure 4.12 below.

Figure 4.12: I have confidence that my retail bank will provide me with sufficient information to make informed decisions: males *vs.* females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups

according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < .05 level in the confidence scores for the four age groups: F(3, 428) = 6.7, p = .000. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.676, SD = 0.92) was statistically different from Group 2 (M = 3.952, SD = 1.26) and Group 3 (M = 4.094, SD = 1.08). Group 4 (M = 4.233, SD = 1.12) did not differ statistically from Groups 1, 2, or 3.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29yrs, Group 2, n = 126: 30-44yrs, Group 3, n = 149: 45-59yrs, Group 4, n = 86: 60+yrs, x_2 (3, n = 432) = 20.46, p = < 0.001. Groups 2 and 3 both recorded a lower median score (Md = 4.00) than Groups 1 and 4, which both recorded median values of 5.00. Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (18-29yrs) was statistically different to Group 2 (30-44yrs), p = < 0.001, r = 0.299. Group 1 was also statistically different to Group 3 (45-59yrs), p = 0.001, r = 0.264. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in confidence levels between the age groups
- ii. The lowest age group (18-29yrs) *agreed* they have confidence that their retail bank will provide them with sufficient information to make informed decision
- iii. Group 2 (30-44 years) and Group 3 (45-59 years) only *slightly agreed* they have confidence that their retail bank will provide them with sufficient information to make informed decision.
- iv. Group 4, (60+ years) who *slightly agreed*, is not statistically different from any other group.

This is graphically represented in Figure 4.13 overleaf.
Figure 4.13: I have confidence that my retail bank will provide me with sufficient information to make informed decisions: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< .05 level in confidence scores for the four education groups: F(3, 428) = 3.5, p = 0.017. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.42, SD = 1.025) was statistically different from Group 4 (M = 3.93, SD = 1.152). Group 2 (M = 4.25, SD = 1.077) and Group 3 (M = 4.14, SD = 1.264) did not differ significantly from either Group 1 or 4.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 12.74, p = 0.005. *Claire Lynne McCafferty PhD April 2020 Page 162*

Groups 3 (Undergraduate) and 4 (Postgraduate) recorded the lowest median score (Md = 4.00). Group 2 recorded a median value of 4.50 and Group 1 (Secondary School education) recorded the highest median value of 5.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 4 (Postgraduate), *p* = 0.003, *r* = 0.243. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence levels across the groups
- ii. Between the groups there is a statistically significant difference between the lowest and highest education level groups
- iii. Group 1 (Secondary School) recorded the highest level of confidence; they agreed with the statement that their retail bank will provide them with sufficient information to make informed decisions
- iv. Group 2 (College) recorded a level of confidence between *slightly agree* and *agree*
- v. Group 3 (Undergraduate) and Group 4 (Postgraduate) only *slightly agreed* that their retail bank will provide them with sufficient information to make informed decisions
- vi. The group with the highest level of education has the lowest level of confidence that their retail bank will provide them with sufficient information to make informed decisions. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.14 overleaf.

Figure 4.14: I have confidence that my retail bank will provide me with sufficient information to make informed decisions: education level



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the p < 0.5 level in confidence scores between the five groups: F(4, 427) = 4.787, p = 0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.48, SD = 1.050) was significantly different to Group 3 (M = 3.92, SD = 1.184). Group 2 (M = 4.17, SD = 1.155), Group 4 (M = 4.10, SD = 1.097) and Group 5 (M = 3.33, SD = 1.211) did not differ significantly from any other group.

A Kruskal-Wallis Test revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 21.91, p = <0.001.

Claire Lynne McCafferty

Page 164

The lowest earning group (<£13,500) recorded the highest median score (Md = 5.00), Groups 2, 3 and 4 recorded median values of 4.00, and the highest earning Group (£150,001+) recorded the lowest median score (Md = 3.50). Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (<£13,500) was statistically different to Group 3 (£27,601-£43,000), *p* = <0.001, *r* = 0.266. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence across the five groups
- ii. The lowest earning group (<£13,500) recorded the highest level of confidence; they agree with the statement, '*I have confidence that my retail bank will provide me with sufficient information to make informed decisions*'
- iii. The highest earning group (£150,001+) recorded the lowest level of confidence; their confidence score falls between *slightly disagree* and *slightly agree*
- iv. Groups 2, 3 and 4 slightly agree with the statement
- v. There is a statistically significant difference in confidence between groups one and three (£27,601-£43,000)
- vi. The level of confidence is lowest for the highest income group and confidence is highest for the lowest income group. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.15 overleaf.

Figure 4.15: I have confidence that my retail bank will provide me with sufficient information to make informed decisions: income level



Independent-Samples Median Test

4.3.3 Moral Hazard arising from the Principal-Agent problem.

'I have confidence that my retail bank will do the right thing'.

One of the problems that arise when the externalities market failure is corrected is the emergence of moral hazard. Krugman (2008, p.63) describes moral hazard as "any situation in which one person makes the decision about how much risk to take, while someone else bears the cost if things go badly". This is based on the theory that greater risks are taken if a party is "insulated" and therefore does not have to fully realise the consequences of their actions (Buckley, 2011, p.58).

By creating systems in which the consumer deposits are protected it may induce a bank to take greater risks (Hellman et al, 2000), perhaps even provide an incentive to intentionally take on the risk of failure (Boyd and De Nicolo, 2005), and so is not viable without regulation (Keeley, 1990). The introduction of ring fencing in the UK has reduced this risk in retail banking somewhat, since the level of risk that can now

be taken by ring fenced banks has been reduced significantly. Ring fencing was introduced as a post-crisis regulatory response to the moral hazard problem surrounding 'Too Big To Fail' banks (Hardie and Macartney, 2016). Moral hazard can undermine the effectiveness of schemes such as LOLR, and magnify costs for the government providing it (Allen, Carletti, Goldstein and Leonello, 2015). The positive effects of ring fencing are two-fold: it reduces the firms' ability to take excessive risk; and the firm is no longer able to 'blackmail' the state into a bailout (Lehmann, 2014).

Moral hazard also arises due to the existence of the principal-agent problem. The consumer frequently requires advice when purchasing financial products and the quality of the product cannot be ascertained at the point of purchase (Llewellyn and the FSA, 1999). The existence of asymmetric information between the principal (the consumer) and the agent can lead to exploitation of the consumer where the agent is motivated by self-interest. Certainly, the mis-selling of PPI and packaged bank accounts was primarily due to incentive structures which encouraged high volume sales of these products by frontline staff in exchange for monetary reward. Thus these products were often sold with little consideration of whether this was the most suitable product for the consumer.

The rationale for regulation is to remove the probability that the moral hazard (caused by schemes such as LOLR and deposit insurance) and the principal-agent problem will be exploited (Llewellyn, 1998). If the consumer is confident that their retail bank will do the right thing and there is sufficient regulation in place to ensure that firms do, then it could be argued that regulation has corrected this market failure.

The fourth market failure addressed above is that of moral hazard. The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers and has published guidance on staff incentives and remuneration. Therefore as part of this investigation, the confidence levels of consumers will be measured to assess the success of conduct of business regulations in combatting moral hazard and maintaining consumer confidence in their retail bank.

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 3.83, SD = 1.34) and non-FS employees (M = 3.88, SD = 1.21; t (430) = 0.157, p = 0.876, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 4.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 3581, z = -0.290, p = 0.772, r = -0.14.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in the levels of confidence of the two groups
- ii. Both groups *slightly agree* with the statement '*I have confidence that my retail bank will do the right thing*'.

This is graphically represented in Figure 4.16 overleaf.

Figure 4.16: I have confidence that my retail bank will do the right thing: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 3.97, SD = 1.27) and females (M = 3.84, SD = 1.18; t (427) = 1.103, p = 0.271, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 4.00, n = 147) and females (Md = 4.00, n = 282), U = 19,104.5, z = -1.382, p = 0.167, r = -0.067.

Both the parametric and non-parametric tests reveal the same findings:

i. There is no statistically significant difference in the levels of confidence of the two groups

Claire Lynne McCafferty

ii. Males and females both *slightly agree* with the statement '*I have confidence that my retail bank will do the right thing*'.

This is graphically represented in Figure 4.17 below.

Figure 4.17: I have confidence that my retail bank will do the right thing: males *vs.* females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < 0.05 level in the confidence scores for the four age groups: F(3, 428) = 10.5, p = <0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.465, SD = 1.03) was statistically different from Group 2 (M = 3.524, SD = 1.28) and Group 3 (M = 3.80, SD = 1.16). Group 2 (M = 1.28)

3.524, SD = 1.28) was also statistically different from Group 4 (M = 3.88, SD = 1.22). There was no statistical difference between Groups 1 and 4, Groups 2 and 3, or Groups 3 and 4.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistical difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 32.23, p = <0.001. Groups 2, 3 and 4 all recorded a lower median score (Md = 4.00) than the youngest age Group (18-29yrs), which recorded a median value of 5.00. Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (18-29yrs) was statistically different to Group 2 (30-44yrs), p = < 0.001, r = 0.386. Group 1 was also statistically different to Group 3 (45-59yrs), p = < 0.001, r = 0.274. Group 2 was statistically different to Group 4 (60+yrs), p = 0.008, r = -0.220. None of the other specific comparisons between Groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in the confidence levels across the four groups
- ii. The youngest group (18-29 years) recorded the highest level of confidence; they agreed with the statement '*I have confidence that my retail bank will do the right thing*'
- iii. The other age groups *slightly agreed* with the statement
- iv. There was a statistical difference in levels of confidence across the four groups
- v. There was a statistical difference in levels of confidence between Group 1 (18-29 years) and Group 2 (30-44 years), Group 1 and Group 3 (45-59 years), and Group 2 and Group 4 (60+ years). This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.18 overleaf.

Figure 4.18: I have confidence that my retail bank will do the right thing: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< .05 level in confidence scores for the four education Groups: F(3, 428) = 7.2, p = <0.001. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.1, SD = 1.07) and Group 3 (M = 3.83, SD = 1.31) and Group 4 (M = 3.54, SD = 1.19). Group 2 (M = 3.89, SD = 1.18) did not differ significantly from any of the groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108:

Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 24.55, p = <0.001. Group 1 (Secondary School education) recorded the highest median score (Md = 5.00), the other three groups all recorded median values of 4.00. Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 2 (College), p = 0.033, r = 0.191. Group 1 was also statistically different to Group 3 (Undergraduate), p = 0.027, r = 0.198. Group 1 was also statistically different to Group 4 (Postgraduate), p = <0.001, r = 0.341. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant different level of confidence across the four groups
- ii. Group 1 (Secondary School) recorded the highest level of confidence; they agree with the statement 'I have confidence that my retail bank will do the right thing'
- iii. Group 2 (College education), Group 3 (Undergraduate) and Group 4 (Postgraduate) all *slightly agree* with the statement
- iv. There is a statistically different level of confidence between Group 1 and all the other groups
- v. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.19 overleaf.

Figure 4.19: I have confidence that my retail bank will do the right thing: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the p < 0.5 level in confidence scores between the five groups: F(4, 427) = 5.262, p = <0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.21, SD = 1.050) was significantly different to Group 3 (M = 3.75, SD = 1.149), Group 4 (M = 3.53, SD = 1.264), and Group 5 (M = 2.83, SD = 1.169). However, there was no statistical difference recorded between Group 2 (M = 3.87, SD = 1.206) and any other groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114:

£27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 23.86, p = <0.001. The lowest earning group (<£13,500) recorded the highest median score (Md = 5.00), Groups 2 and 3 recorded a median value of 4.00, Group 4 (£43,001-£150,000) recorded a median value of 3.50, and the highest earning group (£150,001+) recorded the highest median score (Md = 3.00). Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (<£13,500) was statistically different to Group 3 (£27,601-£43,000), p =0.008, r = 0.216. Group 1 was also statistically different to Group 4 (£43,001-£150,000), p = 0.001, r = 285. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in the confidence levels across the income groups
- ii. The parametric test recorded a significant difference between the lowest and highest earners
- iii. The lowest income group recorded the highest level of confidence, *agreeing* with the statement that their retail bank will do the right thing.
- iv. The highest earners recorded the lowest level of confidence, *slightly disagreeing* with the statement.
- v. The non-parametric test did not record a significant difference between the lowest and highest earners, however, this is likely due to the small sample size of Group 5 (n = 6)
- vi. Although the level of confidence recorded lowered as earnings increased, overall, the participants *slightly agreed* that their retail bank will do the right thing.

This is graphically represented in Figure 4.20 overleaf.

Figure 4.20: I have confidence that my retail bank will do the right thing: income levels



Independent-Samples Median Test

4.3.4 Fraudulent Activities

'I have confidence that I am protected from retail bank misconduct'.

Fraudulent activity is linked to moral hazard and as discussed in the Literature Review, the examples of the banks taking excessive risks when the costs are borne by others are plentiful. The steps taken to restructure the regulatory framework and separate the retail side of banks' business from riskier operations does address some of the causes of fraudulent activity that have been experienced in banking in recent times. The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers and the regulator monitors this through using tools such as Economics for Effective Regulation (discussed in Section 2.4.6).

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 4.06, SD = 1.16) and non-FS employees (M = 4.03, SD = 1.17; t (430) = -0.077, p = 0.939, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 4.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 3741.5, z = 0.031, p = 0.975, r = 0.002.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between those who work in the financial sector and those who don't
- ii. Both groups *slightly* agree with the statement '*I* have confidence that *I* am protected from retail bank misconduct.

This is graphically represented in Figure 4.21 overleaf.

Figure 4.21: I have confidence that I am protected from retail bank misconduct: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.102, SD = 1.25) and females (M = 4.01, SD = 1.18; t(268) = 0.743, p = 0.458, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 4.00, n = 147) and females (Md = 4.00, n = 282), U = 19,362.5, z = -1.169, p = 0.242, r = -0.056.

Both the parametric and non-parametric tests reveal the same findings:

i. There was no statistically significant difference in the levels of confidence recorded between males and females

ii. Both groups *slightly* agree with the statement '*I* have confidence that *I* am protected from retail bank misconduct'.

This is graphically represented in Figure 4.22 below.

Figure 4.22: I have confidence that I am protected from retail bank misconduct: males vs. females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < 0.05 level in the confidence scores for the four age groups: F(3, 428) = 3.721, p = 0.012. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.408, SD = .99) was statistically different from

Group 2 (M = 3.905, SD = 1.26) and Group 3 (M= 3.913, SD = 1.10). Group 4 (M = 4.128, SD = 1.23) did not differ statistically from Groups 1, 2 or 3.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 11.60, p = 0.009. Groups 2, 3 and 4 all recorded a lower median score (Md = 4.00) than the youngest age group (18-29 years), which recorded a median value of 5.00. Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (18-29 years) was statistically different to Group 2 (30-44 years), p = 0.038, r = 0.195. Group 1 was also statistically different to Group 3 (45-59 years), p = 0.012, r = 0.208. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was a statistically significant difference in the levels of confidence recorded across the four age groups
- ii. The youngest group (18-29 years) recorded the highest confidence score; they agree with the statement '*I have confidence that I am protected from retail bank misconduct*'
- iii. The three other age groups all *slightly agree* with the statement
- iv. There was a statistical difference in levels of confidence between Group 1 and Group 2 (30-44 years), and Group 1 and Group 3 (45-59 years). This is discussed in more detail in Section 5.3.9.

This is graphically represented in Figure 4.23 overleaf.

Figure 4.23: I have confidence that I am protected from retail bank misconduct: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< 0.05 level in confidence scores across the four education groups: F(3, 428) = 3.8, p = 0.01. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.33, SD = 1.12) was statistically different from Group 4 (M = 3.79, SD = 1.18). Group 2 (M = 3.97, SD = 1.16) and Group 3 (M = 4.08, SD = 1.17) did not differ significantly from any of the groups.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically significant difference in levels of confidence across the four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 14.85, p

Claire Lynne McCafferty

= .002. Group 1 (Secondary School education) recorded the highest median score (Md = 5.00), the other three groups all recorded median values of 4.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 2 (College), *p* = 0.049, *r* = 0.182. Group 1 was also statistically different to Group 4 (Postgraduate), *p* = 0.001, *r* = 0.275. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in the levels of confidence across the levels of education
- ii. Between groups, both tests revealed a significant difference between the lowest (*agree*) and highest education groups (*slightly agree*).
- iii. The non-parametric test also recorded a significant difference between Groups 1 and 2 which was not recorded in the parametric test
- iv. All groups *slightly agreed* they are protected from retail bank misconduct.

This is graphically represented in Figure 4.24 overleaf.

Figure 4.24: I have confidence that I am protected from retail bank misconduct: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less (M = 4.20, SD = 1.23); Group 2: £13,501-£27,600 (M = 4.01, SD = 1.09); Group 3: £27,601-£43,000 (M = 4.01, SD = 1.15); Group 4: £43,001- £150,000 (M = 3.89, SD = 1.23); Group 5: £150,001 and above (M = 3.17, SD = 0.98). There was no statistically significant difference at the p < 0.5 level in confidence scores between the five groups: F(4, 427) = 1.715, p = 0.146. Post-hoc comparisons using the Tukey HSD test indicated no statistical differences in the means between groups.

A Kruskal-Wallis Test was used as a robustness test and revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 9.78, p = 0.044. The highest earning group (£150,001+) recorded *Claire Lynne McCafferty PhD April 2020 Page 183*

the lowest median score (Md = 3.50), Groups 2, 3 and 4 recorded median values of 4.00, and the lowest earning group (<£13,500) recorded the highest median score (Md = 5.00). Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that despite the significant overall difference, none of the specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

There was a difference in the findings reported by the parametric and non-parametric tests:

- i. The parametric tests found no statistically significant differences across or between the income groups
- ii. The non-parametric tests revealed a statistically significant difference across the groups but not between the groups
- iii. Overall, the levels of confidence recorded fell as income increased
- iv. Across the income groups the median recorded they *slightly agree* they have confidence they are protected from retail bank misconduct. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.25 overleaf.

Figure 4.25: I have confidence that I am protected from retail bank misconduct: income levels



Independent-Samples Median Test

The above has described the findings for the economic rationales for regulation. The non-economic rationales, specifically social rationales, are arguments for regulatory intervention to protect the social values of a group of individuals. Where economic rationales seek to address market failures which could ultimately damage the economy, social rationales arise due to the very nature of the UK's highly developed and market-oriented economy.

SOCIAL RATIONALES

4.3.5 Paternalism (opposite of freedom of choice)

"I have confidence that I can trust the expertise of my retail bank"

Paternalism can exist in the form of interventions in the interests of another, against their will, such as the FCA's Mortgage Market Review directions. Behavioural economics has been referred to as 'light' paternalism (Thaler and Sunstein, 2008).

Claire Lynne McCafferty

Page 185

Since coming into power the FCA has developed a keen interest in the use of behavioural economics to choose the right interventions, secure better outcomes for consumers and promote effective competition (Erta et al, 2013). In particular, retail bank consumers should be able to trust that their retail bank will, through their expertise, help them avoid making decisions which are detrimental to their welfare.

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 3.83, SD = 1.51) and non-FS employees (M = 3.94, SD = 1.17; t(17.91) = 0.282, p = 0.781, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 4.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 3649.5, z = -0.153, p = 0.878, r = -0.007.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between those who work in the financial sector and those who don't
- ii. Both groups *slightly* agree with the statement '*I* have confidence that I can trust the expertise of my retail bank'.

This is graphically represented in Figure 4.26 overleaf.

Figure 4.26: I have confidence that I can trust the expertise of my retail bank: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 3.93, SD = 1.29) and females (M = 3.94, SD = 1.13; t(427) = -0.035, p = 0.972, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 4.00, n = 147) and females (Md = 4.00, n = 282), $U = 20,337.5 \ z = -0.332$, p = 0.740, r = -0.016.

Both the parametric and non-parametric tests reveal the same findings:

i. There was no statistically significant difference in the levels of confidence recorded between males and females

Claire Lynne McCafferty

ii. Both groups *slightly* agree with the statement '*I* have confidence that *I* can trust the expertise of my retail bank'.

This is graphically represented in Figure 4.27 below.

Figure 4.27: I have confidence that I can trust the expertise of my retail bank: males *vs.* females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < 0.05 level in the confidence scores for the four age groups: F(3, 428) = 10.955, p = <0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.606, SD = .978) was statistically different from Group 2 (M = 3.690, SD = 1.26), Group 3 (M = 3.779, SD = 1.11) and Group 4 (M = 1.26).

Claire Lynne McCafferty

3.988, SD = 1.18). Group 2 did not differ statistically from Groups 3 or 4, and Group 3 did not differ statistically from Group 4.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 35.17, p = <0.001. Groups 2, 3 and 4 all recorded a lower median score (Md = 4.00) than the youngest age group (18-29 years), which recorded a median value of 5.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (18-29 years) was statistically different to Group 2 (30-44 years), p = <0.001, r = 0.389. Group 1 was also statistically different to Group 3 (45-59 years), p = <0.001, r = 0.327. Group 1 was statistically different to Group 4 (60+ years), p = 0.002, r = 0.284. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in the levels of confidence recorded across the groups
- ii. The youngest Group (18-29 years) recorded the highest level of confidence; they agree with the statement 'I have confidence that I can trust the expertise of my retail bank'
- iii. Group 2 (30-44 years), Group 3 (45-59 years) and Group 4 (60+ years) all slightly agreed with the statement
- iv. There is a statistically significant difference in the levels of confidence recorded between Group 1 and Groups 2, 3 and 4. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.28 overleaf.

Figure 4.28: I have confidence that I can trust the expertise of my retail bank: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< 0.05 level in confidence scores for the four education groups: F(3, 428) = 4.91, p = 0.002. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.18, SD = 1.19) was statistically different from Group 4 (M = 3.63, SD = 1.16). Group 2 (M = 4.09, SD = 1.07) was also statistically different from Group 4. Group 3 (M = 3.85, SD = 1.27) did not differ significantly from any of the groups.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 19.95, p = .000.

Group 1 (Secondary School education) recorded the highest median score (Md = 5.00), the other three groups all recorded median values of 4.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 4 (Postgraduate), p = <0.001, r = .275. Group 2 was also statistically different to Group 4 (Postgraduate), p = 0.012, r = 0.206. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence scores across the four groups.
- ii. The group with the lowest level of education (Secondary School) recorded the highest level of confidence; they *agreed* with the statement '*I have confidence that I can trust the expertise of my retail bank*'.
- iii. Group 2 (College), Group 3 (Undergraduate) and Group 4 (Postgraduate) all *slightly agreed* with the statement.
- There is a statistically significant difference in confidence scores between Group 1 and Group 4, and Group 2 and Group 4. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.29 overleaf.

Figure 4.29: I have confidence that I can trust the expertise of my retail bank: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the p < 0.5 level in confidence scores for the five income groups: F(4, 427) = 5.052, p = 0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.21, SD = 1.141) was significantly different from Group 3 (M = 3.76, SD = 1.192), Group 4 (M = 3.61, SD = 1.150) and Group 5 (M = 2.83, SD = 1.329). Group 2 (M = 4.01, SD = 1.168) did not differ significantly from any other groups.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, *Claire Lynne McCafferty PhD April 2020 Page 192*

 x_2 (4, n = 432) = 22.70, p = <.001. The lowest earning group (<£13,500) recorded the highest median score (Md = 5.00), Groups 2 and 3 recorded median values of 4.00, and Groups 4 and 5 recorded median values of 3.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (<£13,500) was statistically different to Group 3 (£27,601-£43,000), p = 0.015, r =0.205. Group 1 was also statistically different to Group 4 (£43,001-£150,000), p =0.002, r = 0.274. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

Both the parametric and non-parametric tests revealed the same findings:

- i. Both tests revealed a significant difference in levels of confidence across the income groups
- ii. Between groups, the parametric test revealed a significant difference between Groups 1, 3, 4 and 5
- iii. The non-parametric test did not record a significant difference between Groups 1 and 5, however, this can be explained by the small sample size of Group 5 (n = 6)
- iv. The level of confidence recorded reduced as earnings increased
- v. Overall, the participants *slightly agreed* they could trust the expertise of their retail bank
- vi. The findings are discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.30 overleaf.

Figure 4.30: I have confidence that I can trust the expertise of my retail bank: income levels



Independent-Samples Median Test

4.3.6 Distributive Justice (moral guidance)

'I have confidence that the price of retail banking products is fair'.

Distributive justice is concerned with distributing resources on the basis of what is fair, and when applied to the regulatory system it is concerned with the protection of consumers (Cartwright, 2004). A common problem encountered by the vulnerable groups identified by the FCA, and discussed in the Literature Review, is access to affordable products (Coppack et al, 2015).

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M =

3.61, SD = 1.46) and non-FS employees (M = 3.57, SD = 1.24; t (430) = -0.128, p = 0.898, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 4.00, n = 18) and non-FS employees (Md = 4.00, n = 414), U = 3751, z = 0.050, p = 0.960, r = 0.002.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between those who work in the financial sector and those who don't
- ii. Both groups *slightly* agree with the statement '*I* have confidence that the price of retail banking products is fair'.

This is graphically represented in Figure 4.31 overleaf.

Figure 4.31: I have confidence that the price of retail banking products is fair: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 3.66, SD = 1.36) and females (M = 3.53, SD = 1.20; t (427) = 1.002, p = 0.317, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 4.00, n = 147) and females (Md = 4.00, n = 282), U = 19,406, z = -1.114, p = .265, r = -0.054.

Both the parametric and non-parametric tests reveal the same findings:

i. There was no statistically significant difference in the levels of confidence recorded between males and females

Claire Lynne McCafferty

ii. Both groups *slightly* agree with the statement '*I* have confidence that *I* can trust the expertise of my retail bank'.

This is graphically represented in Figure 4.32 below.

Figure 4.32: I have confidence that the price of retail banking products is fair: males *vs.* females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < .05 level in the confidence scores for the four age groups: F(3, 428) = 10.936, p = .000. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4,296, SD = 1.11) was statistically different from Group 2 (M = 3.365, SD = 1.29), Group 3 (M = 3.383, SD = 1.13) and Group 4 (M = 1

Claire Lynne McCafferty

Page 197
3.616, SD = 1.29). Group 2 did not differ statistically from Groups 3 or 4, and Group 3 did not differ statistically from Group 4.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 32.28, p = <0.001. The youngest age group (18-29 years recorded the highest median score (Md = 5.00), the oldest age group (60+ years) recorded a median value of 4.00, and Groups 2 and 3 both recorded the lowest median score (Md = 3.00). Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (18-29 years) was statistically different to Group 3 (45-59 years), p = <0.001, r = 0.362. Group 1 was also statistically different to Group 4 (60+yrs), p = 0.005, r = 0.267. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is a statistically significant difference in confidence scores across the groups.
- ii. The youngest group (18-29 years) recorded the highest level of confidence; they agree with the statement '*I have confidence that the price of retail* banking products is fair'
- iii. The oldest group (60+ years) *slightly agree* with the statement.
- iv. Group 2 (30-44 years) and Group 3 (45-59 years) recorded the lowest levels of confidence; they *slightly disagree* with the statement.
- v. There is a statistically significant difference between Group 1 and Groups 2, 3 and 4. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.33 overleaf.

Figure 4.33: I have confidence that the price of retail banking products is fair: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p < 0.05 level in confidence scores for the four education groups: F(3, 428) = 3.95, p = 0.008. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 3.90, SD = 1.17) was statistically different from Group 4 (M = 3.32, SD =1.22). Group 2 (M = 3.61, SD = 1.27) and Group 3 (M = 3.50, SD = 1.28) did not differ significantly from any of the groups.

A Kruskal-Wallis Test was used as a robustness test and revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 12.61, p = 0.006.

Group 4 (Postgraduate education) recorded the lowest median score (Md = 3.00), the other three groups all recorded median values of 4.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 4 (Postgraduate), *p* = 0.003, *r* = 0.239. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence scores across the groups
- ii. The group with the highest level of education attainment (Postgraduate) recorded the lowest confidence score; they *slightly disagree* with the statement '*I have confidence that the price of retail banking products is fair*'
- iii. The other three groups *slightly agree* with the statement
- iv. There is a statistically significant difference between Group 1 and Group 4.This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.34 overleaf.

Figure 4.34: I have confidence that the price of retail banking products is fair: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the p < 0.5 level in confidence scores for the five income groups: F(4, 427) = 5.425, p = <0.001. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 3.98, SD = 1.204) was significantly different from Group 2 (M = 3.44, SD = 1.220), Group 3 (M = 3.45, SD = 1.198) and Group 4 (M = 3.24, SD = 1.302). Group 5 (M = 3.33, SD = 1.633 did not differ significantly from any other groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+,

 x_2 (4, n = 432) = 22.04, p = <0.001. The lowest earning group (<£13,500) recorded the highest median score (Md = 4.00), Groups 2 and 4 recorded median values of 3.00, and Groups 3 and 5 recorded median values of 3.50. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (<£13,500) was statistically different to Group 2 (£13,501-£27,600), p = 0.004, r =0.225. Group 1 was also statistically different from Group 3 (£27,601-£43,000), p =0.004, r = 0.228, and Group 1 was also statistically different from Group 4 (£43,001-£150,000), p = 0.001, r = 0.280. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence scores across the groups
- ii. The group with the lowest level of income (<£13,500) recorded the highest confidence score; they *slightly agree* with the statement '*I have confidence that the price of retail banking products is fair*'
- iii. Group 2 (£13,501- £27,600) and Group 4 (£43,001-£150,000) recorded the lowest confidence scores; they *slightly disagree* with the statement
- iv. There is a statistically significant difference between Group 1 and Groups 2, 3, and 4
- v. The highest income group is not statistically significantly different from the other groups.

This is graphically represented in Figure 4.35 overleaf.

Figure 4.35: I have confidence that the price of retail banking products is fair: income levels



Independent-Samples Median Test

4.3.7 Community Values (core values of society)

'I have confidence that I will not be unlawfully discriminated against by my retail bank'.

Community values are the values which society holds and wish regulation to protect and promote; such as trust, honesty and fair dealing (Cartwright, 2004). Due to market failures such as asymmetric information, trust is paramount in retail financial services, as without trust demand for products and services may decrease (Llewellyn, 2005). One of the FCA's statutory objectives is to 'maintain confidence in the UK financial system' under the terms of the Financial Services and Markets Act 2000. This is particularly important for vulnerable groups who could find they are discriminated against but may lack a voice.

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 4.44, SD = 1.46) and non-FS employees (M = 4.43, SD = 1.05; t (430) = -0.047, p = 0.963, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 5.00, n = 18) and non-FS employees (Md = 5.00, n = 414), U = 4114, z = 0.802, p = 0.422, r = 0.039.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between those who work in the financial sector and those who don't
- ii. Both groups agree with the statement 'I have confidence that will not be unlawfully discriminated against by my retail bank'

This is graphically represented in Figure 4.36 overleaf.

Figure 4.36: I have confidence that I will not be unlawfully discriminated against by my retail bank: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.47, SD = 1.14) and females (M = 4.42, SD = 1.03; t (427) = 0.436, p = 0.663, two-tailed).

A Mann-Whitney U Test was used a as robustness test and also revealed no significant difference in the confidence levels of males (Md = 5.00, n = 147) and females (Md = 5.00, n = 282), U = 19,554, z = -1.032, p = 0.302 r = -0.05.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between males and females
- ii. Both groups agree with the statement 'I have confidence that I will not be unlawfully discriminated against by my retail bank'.

The mean recorded was slightly lower, however, this can be explained by the outliers. 1.4% (N=6) of the participants strongly disagreed with the statement and 4.7% (N=20) of the participants disagreed with the statement. As discussed previously, the 5% trimmed mean, where the lowest 5% and highest 5% of the data is excluded, for each question was very close to the mean so did not require further investigation (Pallant, 2016).

This is graphically represented Figure 4.37 below.

Figure 4.37: I have confidence that I will not be unlawfully discriminated against by my retail: males vs. females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was no statistically significant difference at the *p* < 0.05 level in confidence scores for the four age groups: *F* (3, 428) = 2.35, p = 0.071.

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 6.35, p = 0.096, all groups recorded median values of 5.00.

Both the parametric and non-parametric tests reveal the same findings:

- i. There is no statistically significant difference in confidence scores across the groups
- ii. All age groups agree with the statement 'I have confidence that I will not be unlawfully discriminated against by my retail bank'.

This is graphically represented in Figure 4.38 overleaf.

Figure 4.38: I have confidence that I will not be unlawfully discriminated against by my retail: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was no statistically significant difference at the p< .05 level in confidence scores for the four education Groups: F(3, 428) = 1.4, p = 0.248. The mean score for the groups was: Group 1 (M = 4.56, SD = .942), Group 2 (M = 4.51, SD = 1.07), Group 3 (M = 4.38, SD = 1.08) and Group 4 (M = 4.29, SD = 1.15).

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 4.05, p = 0.257, all groups recorded a median value of 5.00.

Parametric and non-parametric tests revealed the same findings:

- i. There is no statistically significant difference in confidence scores across the groups
- ii. All education levels agree with the statement 'I have confidence that I will not be unlawfully discriminated against by my retail bank'.

This is graphically represented in Figure 4.39 below.

Figure 4.39: I have confidence that I will not be unlawfully discriminated against by my retail bank: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was a statistically significant difference at the *p*< 0.05 level in confidence scores for the five income groups: F(4,427) = 3.507, p = 0.008.

Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.69, SD = 0.945) was significantly different from Group 2 (M = 4.32, SD = 1.054) and Group 4 (M = 4.13, SD = 1.235). Group 3 (M = 4.44, SD = 1.081) and Group 5 (M = 4.33, SD = 0.816) did not differ significantly from any other groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 14.22, p = .007. Groups 1, 2 and 3 recorded the higher median score (Md = 5.00). Group 4 (£43,001-£150,000) recorded the lowest median value of 4.00; the highest earners (£150,001+) recorded a median value of 4.50. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (<£13,500) was statistically different to Group 2 (£13,501-£27,600), p = .037, r = .183. Group 1 was also statistically different to Group 4 (£43,001-£150,000), p = .009, r = .241. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of income.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence scores across the groups
- ii. Group 4 (£43,001-£150,000) recorded the lowest confidence score; they slightly agree with the statement 'I have confidence I will not be unlawfully discriminated against by my retail bank'
- iii. The other income groups agree with the statement
- iv. There is a statistically significant difference between Group 1 and Groups 2 and 4. This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.40 overleaf.

Figure 4.40: I have confidence that I will not be unlawfully discriminated against by my retail bank: income levels



Independent-Samples Median Test

4.3.8 Protection

'In general, I have confidence that as a retail bank customer I am adequately protected'.

The final question in the consumer survey tested overall confidence that consumers felt they were protected by retail banking regulation

Currently employed in the FS sector: An independent-samples t-test was conducted to compare the confidence scores of those employed in the financial services sector and those who are not employed in the financial services sector. There is no statistically significant difference in the scores for FS employees (M = 4.28, SD = 1.045) and non-FS employees (M = 4.44, SD = 1.097; t (430) = -0.661, p = 0.509, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of FS employees (Md = 5.00, n = 18) and non-FS employees (Md = 4.50, n = 414), U = 4152.5, z = 0.876, p = 0.381, r = -0.042.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between those who work in the financial sector and those who don't
- ii. Those who work in financial services agree with the statement 'I have confidence that as a retail bank customer I am adequately protected
- iii. Those who do not work in the sector recorded a confidence level between *slightly agree* and *agree*

This is graphically represented in Figure 4.41 below.

Figure 4.41: I have confidence that I as a retail bank customer I am adequately protected: employment in sector *vs.* non-employment in sector



Independent-Samples Median Test

Gender: An independent-samples t-test was conducted to compare the confidence scores of males and females. There was no significant difference in scores for males (M = 4.41, SD = 1.046) and females (M = 4.22, SD = 1.038; t (427) = 1.809, p = 0.071, two-tailed).

A Mann-Whitney U Test was used as a robustness test and also revealed no significant difference in the confidence levels of males (Md = 5.00, n = 147) and females (Md = 4.00, n = 282), U = 18,467, z = -1.976, p = 0.048, r = -0.095.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was no statistically significant difference in the levels of confidence recorded between males and females
- ii. Males agree with the statement '*I have confidence that that as a retail bank customer I am adequately protected*'
- iii. Females *slightly agree* with the statement.

This is graphically represented in Figure 4.42 overleaf.

Figure 4.42: I have confidence that I as a retail bank customer I am adequately protected: males *vs.* females



Independent-Samples Median Test

Age: A one-way between-group analysis of variance was conducted to explore the impact of age on levels of confidence. Participants were divided into four groups according to their age (Group 1: 18-29 years; Group 2: 30-44 years; Group 3: 45-59 years; Group 4: 60 years and above). There was a statistically significant difference at the p < 0.05 level in the confidence scores for the four age groups: F(3, 428) = 4.671, p = 0.003. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.606, SD = .95) was statistically different from Group 2 (M = 4.063, SD = 1.15). Groups 3 and 4 did not differ significantly from any other groups.

A Kruskal-Wallis Test was used to test robustness and also revealed a statistically significant difference in levels of confidence across four different age groups (Group 1, n = 71: 18-29 years, Group 2, n = 126: 30-44 years, Group 3, n = 149: 45-59 *Claire Lynne McCafferty PhD April 2020 Page 214*

years, Group 4, n = 86: 60+ years, x_2 (3, n = 432) = 11.73, p = 0.008. Groups 2 and 3 both recorded a lower median score (Md = 4.00) than Groups 1 and 4, which both recorded median values of 5.00. Dunn's pairwise comparisons with adjusted p-values (Bonferroni correction) revealed that Group 1 (18-29 years) was statistically different to Group 2 (30-44 years), p = 0.011, r = 0.221. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to difference in age.

Both the parametric and non-parametric tests reveal the same findings:

- i. There was a statistically significant difference in confidence scores across the groups
- ii. The lowest age group (18-29 years) and the highest age group (60 years and above) both agree with the statement '*I have confidence that that as a retail bank customer I am adequately protected*'
- iii. The other two groups *slightly agree* with the statement
- iv. There is a statistically significant difference between Group 1 and Group 2 (30-44 years). This is discussed in more detail in Section 4.3.9.

This is graphically represented in Figure 4.43 overleaf.

Figure 4.43: I have confidence that I as a retail bank customer I am adequately protected: age groups



Independent-Samples Median Test

Education level: A one-way between-group analysis of variance was conducted to explore the impact of education on levels of confidence. Participants were divided into four groups according to their level of educational attainment (Group 1: Secondary School; Group 2: College; Group 3: Undergraduate; Group 4: Postgraduate). There was a statistically significant difference at the p< 0.05 level in confidence scores for the four education groups: F(3, 428) = 4.7, p = 0.003. Posthoc comparisons using the Tukey HSD test indicated that the mean score for Group 1 (M = 4.57, SD = .908) was statistically different from Group 4 (M = 4.04, SD = 1.15). Group 2 (M = 4.26, SD = 1.06) and Group 3 (M = 4.31, SD = 0.990) did not differ significantly from any of the groups.

A Kruskal-Wallis Test was used as a robustness test and also revealed a statistically significant difference in levels of confidence across four different levels of education (Group 1, n = 98: Secondary School, Group 2, n = 114: College, Group 3, n = 108: Undergraduate, Group 4, n = 112: Postgraduate, x_2 (3, n = 432) = 15.27, p = 0.002.

Claire Lynne McCafferty

Group 1 (Secondary School education) recorded the highest median score (Md = 5.00), the other three groups all recorded median values of 4.00. Dunn's pairwise comparisons with adjusted *p*-values (Bonferroni correction) revealed that Group 1 (Secondary School) was statistically different to Group 4 (Postgraduate), *p* = 0.001, *r* = 0.266. None of the other specific comparisons between groups indicate a significant difference in levels of confidence due to differing levels of education.

Parametric and non-parametric tests revealed the same findings:

- i. There is a statistically significant difference in confidence scores across the groups
- ii. The group with the lowest level of education (Secondary School) recorded the highest confidence score; they agree with the statement 'I have confidence that I as a retail bank customer I am adequately protected'
- iii. The other three groups *slightly agree* with the statement
- iv. There is a statistically significant difference between Group 1 (Secondary School) and Group 4 (Postgraduate).

This is graphically represented in Figure 4.44 overleaf.

Figure 4.44 I have confidence that I as a retail bank customer I am adequately protected: education levels



Independent-Samples Median Test

Income level: A one-way between-group analysis of variance was conducted to explore the impact of income on levels of confidence. Participants were divided into five groups according to their annual income: Group 1: £13,500 or less; Group 2: £13,501-£27,600; Group 3: £27,601-£43,000; Group 4: £43,001- £150,000; Group 5: £150,001 and above. There was no statistically significant difference at the *p*< 0.05 level in confidence scores for the five income groups: *F* (4,427) = 1.261, *p* = 0.285.

A Kruskal-Wallis Test was used as a robustness test and also revealed no significant difference in levels of confidence across five different income groups (Group 1, n = 128: <£13,500, Group 2, n = 122: £13,501-£27,600, Group 3, n = 114: £27,601-£43,000, Group 4, n = 62: £43,001-£150,000, Group 5, n = 6: £150,001+, x_2 (4, n = 432) = 7.52, p = 0.111.

Parametric and non-parametric tests revealed the same findings:

- i. There is no statistically significant difference in confidence scores across the groups
- ii. The lowest income group (£13,500 or less) recorded the highest confidence score; they agree with the statement '*I have confidence that I as a retail bank customer I am adequately protected*'
- iii. Groups 2 (£13,501-£27,600), Group 3 (£27,601-£43,000) and Group 4 (£43,001- £150,000) *slightly agree* with the statement
- iv. The highest earning group (£150,001+) recorded a confidence level between *slightly agree* and *agree*.

This is graphically represented in Figure 4.45 below.

Figure 4.45 I have confidence that I as a retail bank customer I am adequately protected: income levels



Independent-Samples Median Test

4.3.9 Summary of Descriptive Statistics

There is a pattern of findings within the demographic groups. First of all, the youngest group tended to record the highest levels of confidence. For example, in:

- i. 'I have confidence that my retail bank will do the right thing'
- ii. 'I have confidence that I am protected from retail bank misconduct'
- iii. 'I have confidence that the price of retail banking products is fair'
- iv. 'I have confidence that that as a retail bank customer I am adequately protected'.

The oldest age group also recorded higher levels of confidence in the last two statements above. In 2011, Blanchflower and Oswald conducted an international study that found that the most happy, confident and satisfied people are disproportionately the young and old (not middle-aged). Their findings also produced the U-curve found in the results of this study for the areas mentioned above.

Secondly, in terms of levels of educational attainment the group with the lowest level (Secondary School) tended to record the highest levels of confidence. For example, in:

- i. 'I have confidence that my retail bank will provide them with sufficient information to make informed decisions'
- ii. 'I have confidence that my retail bank will do the right thing'
- iii. 'I have confidence that I can trust the expertise of my retail bank'.

The group with the highest level of education attainment (Postgraduate) recorded the lowest confidence score:

i. 'I have confidence that the price of retail banking products is fair'.

It is important to question why those who are educated to a higher level have less confidence in retail banking regulation. Although it is beyond the scope of this research to answer the 'why' questions, some possible explanations for the findings are presented in Chapter 6.

The last demographic group was that of income. Again, some interesting and unexpected findings were recorded. The levels of confidence recorded tended to decrease as levels of income increased. For example:

- i. 'I have confidence that my retail bank will provide me with sufficient information to make informed decisions'
- ii. 'I have confidence that I can trust the expertise of my retail bank'
- iii. 'I have confidence I will not be unlawfully discriminated against by my retail bank'.

This finding is important as higher income consumers tend to purchase products and services which generate a higher income for banks. Although these products may not fall into the category of 'retail' products as they may be investments or other wealth products, loyal current account customers provide an opportunity for cross-selling. Therefore, it is vital that these consumers are confident that they can trust their retail bank and have confidence in the regulation that is in place to protect them. As mentioned above, it is beyond the scope of this research to answer the 'why' questions, some possible explanations for the findings are presented in Chapter 6. Table 4.4 overleaf outlines the findings presented above:

| | WORK IN FS | GENDER | AGE | EDUCATION | INCOME LEVEL | RELATED |
|-----|----------------|----------|----------------|----------------|---------------------|---------|
| | | | | LEVEL | | SECTION |
| H1 | Yes: Agree | Both: | All: Slightly | All: Slightly | All: Slightly agree | 5.3.1 |
| | | Slightly | agree | agree | | |
| | No: Slightly | agree | | | | |
| | agree | | | | | |
| H2 | Both: Agree | Both: | All: Agree | All: Agree | All: Agree | 5.3.1 |
| | | Agree | | | | 5.0.0 |
| H3 | Both: Slightly | Both: | Youngest: | Lowest: Agree | Lowest: Agree | 5.3.2 |
| | agree | Slightly | Agree | Llichest | Mid. compare. | |
| | | agree | Post: Slightly | Rightly agree | Slightly agree | |
| | | | agree | Slightly agree | Silgrilly agree | |
| H4 | Both: Slightly | Both: | Youngest: | Lowest: Aaree | Lowest: Aaree | 533 |
| 117 | agree | Slightly | Agree | Rest: Slightly | Lowest. Agree | 0.0.0 |
| | agroo | agree | , igi ee | agree | Highest: Slightly | |
| | | | Rest: Slightly | | disagree | |
| | | | agree | | | |
| H5 | Both: Slightly | Both: | Youngest: | Lowest: Agree | All: Slightly agree | 5.3.4 |
| | agree | Slightly | Agree | _ | | |
| | | agree | | Highest: | | |
| | | | Rest: Slightly | Slightly agree | | |
| | | | agree | | | |
| H6 | Both: Slightly | Both: | Youngest: | Lowest: Agree | Lowest: Agree | 5.3.5 |
| | agree | Slightly | Agree | | | |
| | | agree | | Rest: Slightly | Rest: Slightly | |
| | | | Rest: Slightly | agree | disagree | |
| 117 | | Duth | agree | La sat | | 500 |
| H/ | Both: Slightly | Both: | Youngest: | Lowest: | Lowest: Slightly | 5.3.6 |
| | agree | Slightly | Agree | Slightly agree | agree | |
| | | agree | Oldost: | Highest: | Mid-high earners: | |
| | | | Slightly agree | Slightly | Slightly disagree | |
| | | | enginity agree | disagree | | |
| H8 | Both: Aaree | Both: | All: Agree | All: Agree | Lowest: Aaree | 5.3.7 |
| | 5 | Agree | 5 | 3 | <u> </u> | |
| | | | | | High earners: | |
| | | | | | Slightly agree | |
| H9 | Both: Agree | Both: | Youngest: | Lowest: Agree | All: Agree | 5.3.8 |
| | | Agree | Agree | | | |
| | | | | Highest: | | |
| | | | Mid-age: | Slightly agree | | |
| | | | Slightly agree | | | |

 Table 4.4: Summary of Findings from First Stage of Analysis

The table on the previous page reveals the following:

- i. There is an overwhelming response of slightly agree to most statements
- ii. There doesn't appear to be any difference in the levels of confidence of those employed and not employed in the sector
- iii. There doesn't appear to be any difference in the levels of confidence between the genders
- iv. The younger group tends to be most confident of the age groups
- v. The lowest education group tends to be most confident of the education groups
- vi. The lowest income group tends to be the most confident of the income groups.

From an academic/theorist perspective a number of significant differences have been found from the hypothesis tests when examining the influence of certain demographic factors on certain hypothesis statements. The data is now analysed using correlations to identify if any such patterns exist.

4.4 Correlations

In this section the relationship between variables will be explored using correlation analysis. To detect and describe the relationship among variables and explore the association between pairs of variables, correlation analysis was used. Correlation analysis was calculated on the independent variables described as 'ordinal' in Table 3.2, page 101, Section 3.3.1.

Pearson product-moment correlation coefficient (r) was used to describe the strength and direction of the linear relationship between an independent variable and dependent variable. Further, Spearman Rank Order Correlation (rho), the nonparametric alternative, was used as a robustness test. There were no differences in the findings of the parametric and non-parametric tests and therefore only the parametric tests are recorded below. The tabulated data can be found in Appendix E.

4.4.1 Age

Question six tested the social rationale for regulation, paternalism. Specifically, that retail bank consumer's should be able to trust that their retail bank will, through their expertise, help them avoid making decisions which are detrimental to their welfare. The question asked consumers how confident they were that they 'can trust the expertise of their (my) retail bank'. A Pearson product-moment correlation coefficient was computed to assess the relationship between age and confidence in their bank's expertise. There was a weak negative correlation between these two variables, r = -0.120, n = 432, p = 0.012. That is, older consumers tended to be less confident that they could trust the expertise of their retail bank.

Question seven tested the social rationale for regulation, distributive justice. This relates to the perceived fairness of outcomes. Explicitly, how confident consumers are '*that the price of retail banking products is fair*'. There was a moderate negative correlation between these two variables, r = -0.136, n = 432, p = 0.005. That is, older consumers tended to be less confident that the price of retail banking products is fair. This differs to the ANOVA test, where a U-shaped curve was found. The other variables did not establish a correlation; they did not tend to either increase or decrease.

4.4.2 Education

Question one concerned the risk of contagion, or systemic risk. That is, the failure of one institution could cause the collapse of other, solvent institutions. The statement participants were asked to consider was, *"I have confidence the UK retail banking system is stable and secure"*. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that consumers are protected against the market failure, contagion. There was a weak negative correlation between these two variables, r = -0.098, n = 432, p = 0.042. That is, a higher level of education tended to decrease confidence that the banking system is stable and secure.

Question three tested the economic rationale for regulation, asymmetric information. Particularly that the seller of a financial product or service has greater knowledge than the consumer. The consumer frequently requires advice when purchasing financial products and the quality of the product cannot be ascertained at the point of purchase (Llewellyn, 1999). The question asked consumers how confident they were that their retail bank will provide them '*sufficient information to make informed decision*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that consumers are protected against the market failure, asymmetric information. There was a weak negative correlation between these two variables, r = -0.211, n = 432, p = < 0.001. That is, a higher level of education tended to decrease confidence that their retail bank will provide them with sufficient information to make informed decisions.

Question four tested the economic rationale for regulation, moral hazard. One of the problems that arise when the externalities market failure is corrected is the emergence of moral hazard. Krugman (2008, p.63) describes moral hazard as "any situation in which one person makes the decision about how much risk to take, while someone else bears the cost if things go badly". This is based on the theory that greater risks are taken if a party is "insulated" and therefore does not have to fully realise the consequences of their actions (Buckley, 2011, p.58).

By creating systems in which the consumer deposits are protected it may induce a bank to take greater risks (Hellman et al, 2000), perhaps even provide an incentive to intentionally take on the risk of failure (Boyd and De Nicolo, 2005), and so is not viable without regulation (Keeley, 1990). The survey asked consumers how confident they were that *'retail bank will do the right thing'*. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that consumers are protected against the market failure, moral hazard. There was a weak negative correlation between these two variables, r = -0.211, n = 432, p = < 0.001. That is, a higher level of education tended to decrease confidence that their retail bank will do the right thing.

Claire Lynne McCafferty

PhD April 2020

Question five tested the economic rationale for regulation, fraudulent activity. Dishonest or fraudulent behaviour by those in power has become something of a norm in modern banking. LIBOR and foreign exchange market rate manipulation was essentially collusive, collaborative and collegial (Skinner, 2016) between the largest of the world's banks. It had become industry accepted practice and gave rise to significant social and economic costs. Add to this the ubiquitous mis-selling of Payment Protection Insurance and subprime mortgages for no reason other than profit. Baker (2010) wrote that in the aftermath of the Global Financial Crisis many commentators cited regulatory capture of public bodies and public policy by leading banks as one of the main casual factors. Fraudulent activity is linked to moral hazard and examples of the banks taking excessive risks when the costs are borne by others are plentiful.

The survey asked consumers how confident they are *'that they are protected from retail bank misconduct'*. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that consumers are protected from retail bank misconduct. There was a weak negative correlation between these two variables, r = -0.138, n = 432, p = 0.004. That is, a higher level of education tended to decrease confidence that fraudulent activity was reduced.

Question six tested the social rationale for regulation, paternalism. Specifically, that retail bank consumer's should be able to trust that their retail bank will, through their expertise, help them avoid making decisions which are detrimental to their welfare. The question asked consumers how confident they were that they '*can trust the expertise of their (my) retail bank*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that the expertise of their retail bank could be trusted. There was a weak negative correlation between these two variables, r = -0.180, n = 432, p = < 0.001. That is, a higher level of education tended to decrease confidence that positive paternalism (that is decisions that are detrimental to consumer's welfare are avoided) is increased by regulation.

Claire Lynne McCafferty

PhD April 2020

Question seven tested the social rationale for regulation, distributive justice. This relates to the perceived fairness of outcomes. Explicitly, how confident consumers are that 'the price of retail banking products is fair'. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that the price of retail banking products is fair. There was a weak negative correlation between these two variables, r = -0.161, n = 432, p = 0.001. That is, a higher level of education tended to decrease confidence that distributive justice was maintained by regulation.

Question eight tested the social rationale for regulation, community values. Community values are the values which society holds and wish regulation to protect and promote; such as trust, honesty and fair dealing (Cartwright, 2004). Due to market failures such as asymmetric information, trust is paramount in retail financial services, as without trust demand for products and services may decrease (Llewellyn, 2005). This is particularly important for vulnerable groups who could find they are discriminated against but may lack a voice. The survey asked consumers how confident they are '*that they will not be unlawfully discriminated against by their retail bank*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that they will not be unlawfully discriminated against by their retail bank. There was a weak negative correlation between these two variables, r = -0.097, n = 432, p = 0.044. That is, a higher level of education tended to decrease confidence that community values were protected by regulation.

The final question tested overall confidence that consumers felt they were protected by retail banking regulation, '*In general, I have confidence that as a retail bank customer I am adequately protected*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between education levels and levels of confidence that they were adequately protected. There was a weak negative correlation between these two variables, r = -0.164, n = 432, p = 0.001. That is, consumers with a higher level of education tended to be less confident that, in general, they were adequately protected.

Claire Lynne McCafferty

PhD April 2020

The only variable which did not establish a correlation, that did not tend to either increase or decrease, was the second survey question that focused on the rationale that consumer deposits should be protected if a bank becomes insolvent, '*I have confidence that the money in my retail bank account(s) is protected should my bank fail*'.

4.4.3 Income

Question two focused on the rationale that consumer deposits should be protected if a bank becomes insolvent, "*I have confidence that the money in my retail bank account(s) is protected should my bank fail*". A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that the system was stable and secure. There was a weak positive correlation between these two variables, r = 0.153, n = 432, p = 0.001. That is, the higher the level of income the higher the level of confidence that deposits were protected should the bank fail.

Question three tested the economic rationale for regulation, asymmetric information. Particularly that the seller of a financial product or service has greater knowledge than the consumer. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that consumers are protected against the market failure, asymmetric information. There was a weak negative correlation between these two variables, r = -0.175, n = 432, p = < 0.001. That is, a higher level of income tended to decrease confidence that their retail bank will provide them with sufficient information to make informed decisions.

Question four tested the economic rationale for regulation, moral hazard. The survey asked consumers how confident they were that '*retail bank will do the right thing*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that consumers are protected against the market failure, moral hazard. There was a weak negative correlation between these two variables, r = -0.208, n = 432, p = < 0.001. That is,

a higher level of income tended to decrease confidence that their retail bank will do the right thing.

Question five tested the economic rationale for regulation, fraudulent activity. The survey asked consumers how confident they are *'that they are protected from retail bank misconduct'*. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that consumers are protected from retail bank misconduct. There was a weak negative correlation between these two variables, r = -0.105, n = 432, p = 0.029. That is, a higher level of income tended to decrease confidence that fraudulent activity was reduced.

Question six tested the social rationale for regulation, paternalism. The question asked consumers how confident they were that they 'can trust the expertise of their (my) retail bank'. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that the expertise of their retail bank could be trusted. There was a weak negative correlation between these two variables, r = -0.206, n = 432, p = < 0.001. That is, a higher level of income tended to decrease confidence that positive paternalism (that is decisions that are detrimental to consumer's welfare are avoided) is increased by regulation.

Question seven tested the social rationale for regulation, distributive justice. This relates to the perceived fairness of outcomes. Explicitly, how confident consumers are '*that the price of retail banking products is fair*'. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that the price of retail banking products is fair. There was a weak negative correlation between these two variables, r = -0.192, n = 432, p = < 0.001. That is, a higher level of income tended to decrease confidence that distributive justice was maintained by regulation.

Question eight tested the social rationale for regulation, community values. The survey asked consumers how confident they are '*that they will not be unlawfully*

Claire Lynne McCafferty

discriminated against by their retail bank'. A Pearson product-moment correlation coefficient was computed to assess the relationship between income levels and levels of confidence that they will not be unlawfully discriminated against by their retail bank. There was a weak negative correlation between these two variables, r = -0.143, n = 432, p = 0.003. That is, a higher level of income tended to decrease confidence that community values were protected by regulation.

There were two hypotheses which did not establish a correlation; they did not tend to either increase or decrease:

- i. 'I have confidence the UK retail banking system is stable and secure'
- ii. 'In general, I have confidence that as a retail bank customer I am adequately protected'.

From the industry/practitioner perspective not only have the relationships between demographic factors and the research hypotheses been confirmed, but now the direction of these relationships have been identified. This can be used to inform the sector of these consumer behaviours, e.g. young; low education; low income people have higher confidence, so old; well-educated; high income people need to be the focus of industry advertising and education.

4.4.4 Summary of Correlation Analysis Findings

Below is a summary of the main findings grouped by demographic:

- i. Age: Older customers tended to be less confident that they could trust the expertise of their retail bank and that the price of retail products was fair.
- ii. Education: The higher the level of educational attainment the lower the confidence level that the banking system is stable and secure; that their retail bank will provide them with sufficient information to make informed decisions; that their retail bank will do the right thing; that they were protected from retail bank misconduct; that they can trust the expertise of their retail bank; that the price of retail banking products is fair; that they will not be unlawfully

discriminated against by their retail bank; and that they were adequately protected overall.

iii. Income: The higher the level of income, the higher the level of confidence that deposits are protected should their retail bank fail. However, the lower the level of confidence that their retail bank will provide them with sufficient information to make informed decisions; that their retail bank will do the right thing; that they are protected from retail bank misconduct; that they can trust the expertise of their retail bank; that the price of retail banking products is fair; and that they will not be unlawfully discriminated against by their retail bank.

In this chapter, the survey data was analysed using two different approaches. In the first instance, and with reference to an academic/theorist standpoint, parametric and non-parametric tests were used to identify if there are any statistically significant differences in the hypotheses results based on an underlying demographic factor(s). It was established that for many of the research hypotheses there were indeed different responses based on groupings established within the demographic factors.

There were no recorded differences in the levels of confidence in the economic rationales for regulation or the social rationales for regulation. The economic rationales for regulation measured as slightly agree, as did the social rationales. Only one economic rationale recorded higher, consumers agree, that the money in their retail bank account(s) is protected should their bank fail. The regulation that is in place to correct the market failure of systemic risk at a micro-prudential level is the Financial Services Compensation Scheme (FSCS). During the Global Financial Crisis this particular area of regulation gained considerable exposure due to the bank run on Northern Rock. This may explain the higher level of confidence recorded.

Only one social rationale recorded higher, consumers agreed, that they will not be unlawfully discriminated against by their retail bank. The Equality Act (2010) is the UK anti-discrimination law. Given the current climate for diversity and inclusion, both at organisational and societal levels, it would be surprising to find that consumers

did feel discriminated against. However, as discussed earlier, the Financial Ombudsman Service (FOS) did find cases in a study in 2015.

The second analysis approach then attempted to identify the direction of these relationships. This was achieved using correlation analysis and the ordinal demographic factors, namely, age; education level and income level. The results from this analysis therefore provide industry/practitioners with a key insight into the nature of the differences to responses that can then be used to inform education, customer support, and even marketing initiatives to address any issues pertinent to the focus of this study area. In particular, older respondents; those with relatively high income levels; and those with higher educational attainment had broadly lower trust and satisfaction levels for many of the key hypotheses.

In the next chapter, the final analysis using multiple linear regressions focuses on the end-user/consumer. This analysis will determine which of the behavioural factors, in terms of the key demographics, are ultimately relevant in the overall appraisal of trust and satisfaction in the industry and regulator as they relate to the consumers own bank experiences.

CHAPTER 5 REGRESSION ANALYSIS AND RESEARCH FINDINGS

5.1 Introduction

The correlation analysis in Chapter 4 looked at each independent variable (demographic data) in turn and its relationship with each dependent variable (levels of confidence measured by the hypotheses). The limitation of correlation analysis is that it only looks at the relationship between two variables at a time. Regression analysis helps to overcome this limitation by looking at the relationship between more than one variable. In this chapter, stepwise multiple regression analysis is used to identify the combination of independent (demographic) variables that would be used to predict the levels of confidence in the dependent variable (rationales for regulation).

5.2 Demographic Groups

Below is a table which presents the groupings used for each demographic variable.

| Demographic Variable | Group 1 | Group 2 | Group 3 | Group 4 | Group 5 |
|----------------------------|---------------------|---------------------|---------------------|----------------------|-----------|
| Work in financial services | No | Yes | - | - | - |
| Gender | Male | Female | - | - | - |
| Age | 18-29 | 30-44 | 45-59 | 60+ | - |
| Education | Secondary School | College | Undergraduate | Postgraduate | - |
| Income | <£13,500 | £13,501- £27,600 | £27,601- £43,000 | £43,001- £150,000 | £150,001+ |

Table 5.1: Demographic Groupings
5.3 Levels of Confidence in the Rationales for Regulation

In this section the significant demographic factors that predict the level of confidence in each rationale for regulation will be presented. Although regression will tell us which independent variables influence the dependent variable, it will not tell us the reasons why. Nonetheless, as the factors that are not important are removed, future studies can be carried out to understand the underlying qualitative reasons.

In order to assess the impact of demographics on confidence levels a multiple regression model must be adopted. Logistic regression is not suitable in this study as the dependent variable of interest is not categorical (Pallant, 2016). Multiple stepwise regressions were used in an attempt to find the best regression model. This analysis tool adds or deletes variables one at a time using the p value as a predictor for each step (Frey, 2018). There is criticism of this method; however, it is useful for exploratory research such as this (Field, 2013). Stepwise regression simply allows the data to reveal potential predictors; in this research there were no theoretical predictions based on the demographics of the participants. Stepwise regression was selected in order to explore avenues for future research and not to test hypotheses (Field, 2013). Had the aim of the research been to predict, or determine the order of prediction, a standard or hierarchical regression model may have been more appropriate since a major limitation of stepwise regression is the increased likelihood of committing a type 1 error (Pandit and Khairullah, 2015), i.e. rejecting a true null hypothesis, due to a sampling error. However, sampling error is less of an issue with a larger sample size (Thompson, 1995). Stepwise regression has the advantage over hierarchical regression of the software algorithm selecting the order of the variables, rather than the researcher having to determine this based on previous theory and research (Lewis, 2007). In addition, stepwise regression has been recommended for exploratory and predictive research (Menard, 1995); although Lewis (2007) advises that it should only be used for preliminary research, such as this. The purpose of this research was to determine if there was a relationship between demographic variables and consumer confidence in retail banking regulation and the findings are that there is a negative correlation between

confidence and income and education levels, and a positive correlation between confidence and age.

5.3.1 Macro externalities: 'I have confidence the UK retail banking system is stable and secure'

When confidence the UK retail banking system is stable and secure was predicted it was found that working in financial services₁ (Beta = -1.0456, p <0.001), level of education₁ (Beta = 0.397, p = 0.003), and age₁ (Beta = 0.376, p = 0.013) were significant predictors. The overall model fit was R² = 0.059. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ work in FS₁ + b_2 education₁ + b_3 age₁

$$= 4.745 + (-1.045 \text{ work in } FS_1) + (0.397 \text{ education}_1) + (0.376 \text{ age}_1)$$

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed

The demographic variables that decrease confidence in this area are:

• Not being employed in financial services

This confirms the independent sample t-test in Section 4.3.1, where mean of workers in Financial Services sector is 4.83 and non-workers is 3.86.

| Dependent Variable | The consumer is confident that the UK retail banking system is stable and secure | | | | | |
|-------------------------|---|---|---|--|--|--|
| Independent Variable | Does not work in the Financial Services IndustryAged between 18-29Educated to Secondary School | | | | | |
| Confidence | \downarrow | 1 | 1 | | | |

Table 5.2: Hypothesis Test 1

The first dependent variable concerns the capital requirement regulations that ensure banks hold sufficient capital. These regulations are in place to avoid systemic risk, in particular contagion. Those who are not employed in the financial services industry have lower levels of confidence in this intervention. This could be expected since those working in the banking sector would perhaps be more aware of these regulations. The second dependent variable relates to the age group. The youngest age group (18-29) recorded the higher levels of confidence. The FCA (2015) identified being young as a risk factor for vulnerability. They justified this as this group are associated with having less experience. The increased confidence could perhaps be explained by naivety rather than proficiency. The third dependent variable concerns the education level of the consumer. The lowest level of educational attainment group (Secondary School) recorded higher levels of confidence in this intervention. The FCA (2015) identified low literacy, numeracy and financial capability skills as a risk factor for vulnerability. Possibly, a lack of understanding of the issues and risks in the financial system may result in a higher level of confidence being recorded.

5.3.2 Micro externalities: 'I have confidence that the money in my retail bank account(s) is protected should my bank fail'

When confidence that the money in my retail banks account(s) is protected should my bank fail was predicted it was found that working in financial services₂ (Beta =

Claire Lynne McCafferty

Page 236

0.598, p = 0.023), level of income₂ (Beta = -0.369, p = 0.002), and age₁ (Beta = -0.324, p = 0.022) were significant predictors. The overall model fit was $R^2 = 0.049$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ work in FS₂₊ b_2 income₂ - b_3 age₁

 $= 4.609 + (0.598 \text{ work in } FS_2) + (-0.369 \text{ income}_2) + (-0.324 \text{ age}_1)$

The demographic variables that increase confidence in this area are:

Being employed in financial services

The demographic variables that decrease confidence in this area are:

- Being aged between 18-29
- Having an annual income between £13,501-£27,600

| Dependent Variable | The consumer is confident that the money in their retail bank account(s) is | | | |
|--------------------|---|--------------------|-----------------|--|
| | protected should their bank fail | | | |
| Independent | Works in the Financial | Aged between 18-29 | Income between | |
| Variable | Services Industry | | £13,501-£27,600 | |

1

Table 5.3: Hypothesis Test 2

The second dependent variable concerns depositor insurance regulations that ensure consumer deposits are guaranteed should a bank fail. These regulations are in place to avoid a bank panic and ultimately contagion. Those who are employed in the financial services industry have higher levels of confidence in this intervention.

Confidence

This could be expected since those working in the banking sector would perhaps be more aware of these regulations. Certainly in retail banking the customer-facing staff would be well-versed in this type of protection. The second dependent variable relates to the age group, this has been discussed above. The third dependent variable concerns the income level of the consumer. Although not the lowest income group, this income bracket is below the national average. The FCA (2015) identified low income as a risk factor for vulnerability. Possibly, a lack of understanding of the issues and risks in the financial system may result in a higher level of confidence being recorded, or conversely, the lack of high value savings imply they are not exposed to this risk.

5.3.3 Asymmetric information: 'I have confidence that my retail bank will provide me with sufficient information to make informed decisions'

When confidence that my retail bank will provide me with sufficient information to make informed decisions was predicted it was found that level of $education_1$ (Beta = 0.313, p = 0.015), level of income₃ (Beta = -0.243, p = 0.050), and age₁ (Beta = 0.557, p <0.001) were significant predictors. The overall model fit was R² = 0.061. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ education₁ - b_2 income₃ + b_3 age₁

 $= 4.077 + (0.313 \text{ education}_1) - (0.243 \text{ income}_3) + (0.557 \text{ age}_1)$

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed

The demographic variables that decrease confidence in this area are:

• Having income between £27,601- £43,000

Claire Lynne McCafferty

| Dependent Variable | The consumer is confident that their retail bank will provide them with sufficient | | | | | |
|--------------------|--|---|--|--|--|--|
| | information to make informed decisions | | | | | |
| Independent | Aged between 18-29 | Aged between 18-29 Educated to Secondary Income between | | | | |
| Variable | School level £27,601-£43,000 | | | | | |
| Confidence | 1 | $\uparrow \qquad \downarrow$ | | | | |

Table 5.4: Hypothesis Test 3

The third dependent variable concerns the market failure, asymmetric information. The problem of asymmetric information occurs as the consumer has less information than the 'expert' supplying the financial product and/or advice. Consumers are unable to inform themselves adequately so as to avoid making decisions which may disadvantage them (Wood, 2001). The individual consumer does not have the time or the resources to sufficiently source information on 'good' and 'bad' products and providers (Llewellyn, 1995). The FCA conduct of business regulations set out how a firm must behave in its day-to-day dealing with consumers including how it markets products and services and the literature provided to consumers. Those in the youngest age group and with the lowest levels of education have higher levels of confidence in this intervention. This could be due to a combination of naivety and better quality information being available for basic products, or as a result of the extensive initiatives the regulator has run for this classification of consumers. The third dependent variable concerns the income level of the consumer. The middle income group (between national average and higher-rate tax payers) have lower levels of confidence that their retail bank will provide them with sufficient information. This income level group could find themselves in need of advice for more complex products, but priced out of the market for independent advice.

5.3.4 Moral hazard: 'I have confidence that my retail bank will do the right thing'

When confidence that my retail bank will do the right thing was predicted it was found that level of education₁ (Beta = 0.562, p = < 0.001) and age₁ (Beta = 0.723, p < 0.001) were significant predictors. The overall model fit was $R^2 = 0.084$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ education₁ + b_2 age₁

 $= 3.631 + (0.562 \text{ education}_1) + (0.723 \text{ age}_1)$

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed

Table 5.5: Hypothesis Test 4

| Dependent | The consumer is confident that their retail bank will do the right thing | | | |
|-------------|--|------------------------------------|--|--|
| Variable | | | | |
| Independent | Aged between 18-29 | Educated to Secondary School level | | |
| Variable | | | | |
| | | | | |
| Confidence | \uparrow | 1 | | |
| | | • | | |

The fourth dependent variable concerns the market failure moral hazard. One of the problems that arise when the externalities market failure is corrected is the emergence of moral hazard. By creating systems in which the consumer deposits are protected it may induce a bank to take greater risks (Hellman et al, 2000),

Claire Lynne McCafferty

Page 240

perhaps even provide an incentive to intentionally take on the risk of failure (Boyd and De Nicolo, 2005), and so is not viable without regulation (Keeley, 1990). The rationale for regulation is to remove the probability that the moral hazard (caused by schemes such as LOLR and deposit insurance) and the principal-agent problem will be exploited (Llewellyn, 1998). The independent variables of lowest age and lowest education level presented higher levels of confidence that they are protected against these risks. The possible reasons behind this have been discussed above.

5.3.5 Fraudulent activities: 'I have confidence that I am protected from retail bank misconduct'

When confidence that I am protected from retail bank misconduct was predicted it was found that level of education₁ (Beta = 0.390, p = 0.003) and age₁ (Beta = 0.461, p = 0.002) were significant predictors. The overall model fit was $R^2 = 0.040$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ education₁ + b_2 age₁ = 3.870 + (0.390 education₁) + (0.461 age₁)

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed

| Dependent Variable | The consumer is confident that they are protected from retail bank misconduct | | | |
|-------------------------|---|---|--|--|
| Independent Variable | Aged between 18-29 Educated to Secondary School level | | | |
| Confidence | 1 | 1 | | |

Table 5.6: Hypothesis Test 5

The fifth dependent variable concerns protecting consumers against the fraudulent activities of retail banks. Fraudulent activity is linked to moral hazard and examples of the banks taking excessive risks when the costs are borne by others are plentiful, and are discussed in the Literature Review chapter. Fraudulent activity was not limited to the practices of a few banks competing with each other. LIBOR and foreign exchange market rate manipulation was essentially collusive, collaborative and collegial (Skinner, 2016) between the largest of the world's banks. It had become industry accepted practice, and gave rise to significant social and economic costs. The steps taken to restructure the regulatory framework and separate the retail side of banks' business from riskier operations does address some of the causes of fraudulent activity that have been experienced in banking in recent times. The independent variables of lowest age and lowest education level presented higher levels of confidence that they are protected against these risks. The possible reasons behind this have been discussed above.

5.3.6 Paternalism: 'I have confidence that I can trust the expertise of my retail bank'

When confidence that I can trust the expertise of my retail bank was predicted it was found that level of education₁ (Beta = 0.464, p = 0.001), education₂ (Beta = 0.349, p = 0.008), age₁ (Beta = 0.801, p = <0.001) and income₅ (Beta = -0.975, p = 0.037)

were significant predictors. The overall model fit was $R^2 = 0.103$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ education + b_2 education + b_1 age - b_5 income

 $= 3.615 + (0.464 \text{ education}_1) + (0.349 \text{ education}_2) + (0.801 \text{ age}_1) - (0.975 \text{ income}_5)$

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed
- College education being the highest level completed

The demographic variables that decrease confidence in this area are:

• Having income of £150K+

Table 5.7: Hypothesis Test 6

| Dependent Variable | The consumer is confident that they can trust the expertise of their retail bank | | | | | |
|-------------------------|--|---|---|--------------|--|--|
| Independent Variable | Aged between 18- 29Educated to Secondary School levelEducated to | | | | | |
| Confidence | 1 | 1 | 1 | \downarrow | | |

The sixth dependent variable concerns the social justification for intervention by the regulator of protectiveness. An example of paternalistic regulation is the FCA's Mortgage Market Review directions. Since coming into power the FCA has

developed a keen interest in the use of behavioural economics to choose the right interventions, secure better outcomes for consumers and promote effective competition (Erta et al, 2013). The lowest age and education groups presented higher levels of confidence in this area. The speculative reasons have been discussed above. Interestingly, the highest income group reported lower levels of confidence in this area of regulation. This could perhaps be explained by the fact that they potentially have a much greater need for specialist advice, the type they cannot perhaps access via their retail bank, and are likely to have access to an Independent Financial Adviser.

5.3.7 Distributive justice: 'I have confidence that the price of retail banking products is fair'

When confidence that the price of retail banking products is fair was predicted it was found that level of education₁ (Beta = 0.378, p = 0.008), income₁ (Beta = .0294, p = 0.035) and age₁ (Beta = 0.744, p <0.001) were significant predictors. The overall model fit was $R^2 = 0.097$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ education₁ + b_2 income₁ + b_3 age₁

 $= 3.279 + (0.378 \text{ education}_1) + (0.294 \text{ income}_1) + (0.744 \text{ age}_1)$

The demographic variables that increase confidence in this area are:

- Being aged between 18-29
- Secondary education being the highest level completed
- Having income of <£13,500

| Dependent Variable | The consumer is confident that the price of retail banking products is fair | | | | | |
|-------------------------|--|---|---|--|--|--|
| Independent Variable | Aged between 18-29Educated to Secondary School levelIncome less than £13,500 | | | | | |
| Confidence | 1 | 1 | 1 | | | |

| Table | 5.8: | Hypothesis To | est 7 |
|-----------|------|---------------|-------|
| 1 4 9 1 9 | 0.0. | | |

The seventh dependent variable is concerned with distributing resources on the basis of what is fair, and when applied to the regulatory system it is concerned with the protection of consumers (Cartwright, 2004). Examples of regulatory protection of the most vulnerable financial services consumers are policies of financial inclusion and the FCA investigation into payday lenders which resulted in extra rules for firms providing high-cost, short-term credit which took effect on 1 July 2014 (FCA, 2014a). A common problem encountered by each of these groups is access to affordable products (Coppack et al, 2015). The three independent variables groups (young, low income, low educational attainment) have been identified by the FCA (2015) as at risk for vulnerability. These groups have recorded higher levels of confidence in this particular area. This could be due to the significant number of regulator initiatives aimed at protecting and informing these groups, or, it could be that the basic products these groups have access to do not attract high fees (possibly due to regulator interventions to create basic accounts and reduce fees, charges and interest).

5.3.8 Community values: 'I have confidence that I will not be unlawfully discriminated against by my retail bank'

When confidence that I will not be unlawfully discriminated against was predicted it was found that level of income₁ (Beta = 0.362, p = 0.001) was the only significant predictor. The overall model fit was $R^2 = 0.024$. See Appendix F for the tabulated data.

Confidence = $b_0 + b_1$ income₁

 $= 4.207 + (0.143 \text{ income}_1)$

The demographic variables that increase confidence in this area are:

• Having income of <£13,500

Table 5.9: Hypothesis Test 8

| Dependent Variable | The consumer is confident that they will not be unlawfully discriminated against by their retail bank | | |
|----------------------|---|--|--|
| Independent Variable | Income less than £13,500 | | |
| Confidence | 1 | | |

The eighth dependent variable relates to the values which society holds and wish regulation to protect and promote; such as trust, honesty and fair dealing (Cartwright, 2004). There is UK legislation in place to protect all citizens, not only consumers of financial services. However, the Financial Ombudsman Service is also available to banking consumers as an independent arbitrator for no fee where they feel they have been mistreated by their retail bank (banks are subject to fees for cases brought before them). The lowest income group reported higher levels of confidence that they are protected. The potential reasons have been discussed above.

5.3.9 Overall confidence: 'In general, I have confidence that as a retail bank customer I am adequately protected'

When overall confidence that as a retail bank customer I am adequately protected was predicted it was found that level of $education_1$ (Beta = 0.382, p = 0.001) and age_1 (Beta = 0.398, p = 0.003) was the only significant predictor. The overall model fit was R² = 0.042. See Appendix F for the tabulated data.

 $Confidence = b_0 + b_1 education_1 + b_2 age_1$

 $= 4.133 + (0.382 \text{ education}_1) + (0.398 \text{ age}_1)$

The demographic variables that increase confidence in this area are:

- Secondary education being the highest level completed
- Being aged between 18-29

Table 5.10: Hypothesis Test 9

| Dependent Variable | The consumer is confident that as a retail bank customer they are adequately protected | | | |
|-------------------------|--|---|--|--|
| Independent Variable | Aged between 18-29 Educated to Secondary School level | | | |
| Confidence | 1 | 1 | | |

The final dependent variable measured levels of confidence that the consumer feels adequately protected overall. The lowest age and education level groups recorded

higher levels of confidence, possible reasons why these particular groups may have higher levels of confidence have been discussed above.

5.4 Summary of Regression Analysis Findings

As can be seen in Table 5.11 below, age, education and income are the most important demographic variables when predicting levels of confidence in the rationales for regulation.

| | Work in FS | Gender | Age | Education | Income |
|---|------------|--------|--------------|--------------|--------|
| The UK retail banking system is stable and secure | 1 | × | \ | \checkmark | X |
| The money in my retail bank account(s) is protected should my bank fail | √ | × | > | × | 1 |
| My retail bank will provide me with sufficient information to make informed decisions | × | × | > | √ | 1 |
| My retail bank will do the right thing | × | × | | 1 | × |
| I am protected from retail bank misconduct | X | X | \checkmark | ✓ | × |
| I can trust the expertise of my retail bank | × | × | \checkmark | ✓ | 1 |
| The price of retail banking products is fair | × | × | \checkmark | ✓ | 1 |
| I will not be unlawfully discriminated against by my retail bank | × | × | X | X | 1 |
| As a retail bank customer I am adequately protected | × | × | \checkmark | \checkmark | × |

Table 5.11: Summary of Regression Analysis Findings

5.5 Conclusions

This chapter of the thesis has set out what the best combination of independent (demographic) variables would be to predict the levels of confidence in the dependent variable (rationales for regulation). The significant demographic factors that predict the level of confidence in each rationale for regulation has been presented. Although regression has identified which independent variables influence the dependent variable, it has not revealed the underlying reasons; although hypothetical explanations have been presented.

Being young, earning a low income and having a low level of educational attainment are clearly important factors in the level of confidence the consumer has in retail banking regulation. Future qualitative studies should be undertaken to determine the causes behind this. This would be of particular interest to Consumer Groups, the regulator and the retail banks.

The next, and final chapter, will revisit the research question, methodology, hypotheses and analysis. The overall findings will be presented along with the contributions, limitations and recommendations.

CHAPTER 6: OVERALL FINDINGS AND CONCLUSIONS

6.1 Introduction

In this final chapter, the aim and objectives of the research will be revisited and the findings and limitations will be discussed. The contributions to theory, practical implications and recommendations for future research will also be presented.

6.2 Background

Confidence in the banking system is critical to the health of households, businesses and the overall economy. Hence, government intervention in the financial services sector and the strict regulation of the banking industry is rationalised by the need to minimise the occurrence of events that could undermine the reputation of banks and threaten the stability of the system. The purpose of this research was to answer the question, '*Are Scottish consumers confident in retail banking regulation*'? This question is important for the reasons mentioned above, confidence in the banking system is critical to the health of households, businesses and the overall economy. This research has taken a unique approach to answering this question by measuring the levels of confidence in each of the specific rationales for regulating the retail banking system. This is the first of three significant contributions this study has made and is discussed further in Section 6.3 below.

6.3 Theory

The first objective of this research was to undertake a critical review of the existing theory of financial regulation literature to determine the specific rationales for monitoring and supervision. The justifications for regulating the banking system largely fall into two categories: correcting economic market failures and preventing social detriment. The economic rationales identified were:

- i. Externalities: systemic risk caused by problems in the wider system and contagion caused by events such as a bank run
- ii. Asymmetric Information: the inequality of information between bank and consumer

Claire Lynne McCafferty

- iii. Moral Hazard: incentives for banks to take greater risks, created by interventions into the market (such as lender of last resort and deposit insurance)
- iv. Fraudulent Activities: misconduct and mis-selling scandals which have plagued the industry over recent decades.

The social rationales identified were:

- i. Paternalism: to avoid less well-informed consumers making decisions which are detrimental to their welfare
- ii. Distributive Justice: ensuring fair outcomes for, the less powerful, consumers
- iii. Community Values: protecting and promoting trust, honesty and fair dealing, particularly those who may lack a voice.

6.4 Method

The second objective of this research was to contextualise the above and develop a set of key research questions. Each question would be designed to measure the level of confidence in the outcome of the corrective or protective intervention. The researcher approach to survey design was original, at least in the financial regulation research context, as all three stakeholders played a crucial role in modelling the questions. Three stages of survey design were undertaken to develop a consumer survey from the theoretical model established in Chapter 2. At each stage, empirical testing was undertaken with three distinct participant groups, and in a logical order that considered theory, practice and end-users, see Figure 6.1 overleaf.



Figure 6.1: Stages of survey design: from theoretical model to empirical testing

The rationale for separating the groups was due to the different levels of knowledge; theory traditionally informs practice (academics and practitioners often collaborate to improve standards). Additionally, corrections and amendments could be made before 'testing' on the next group. The final group of end-users were consumers who would not be included in the final study and could feedback on the suitability and validity of the questions whilst understanding the context behind them. There are numerous studies, in particular those undertaken by the current regulator, where there is participation from either industry or consumer. However, the researcher couldn't find any specific studies were theorists, practitioners and end-users had direct input into the design of the study. At each stage of the empirical testing of the

survey, recommendations for change to the questions were made and acted upon (see Tables 3.4 and 3.5 in Chapter 3).

6.5 Data Collection

The researcher employed a mixed data collection, whereby 71% of responses were achieved via an online survey, with 29% paper surveys attained to ensure the views of non-internet users were represented. The research was limited to Scotland and therefore the adult population (18+ years) residing in Scotland was the target population (since almost all adults in Scotland are 'banked') CAB Scotland (2010). According to the 2016 Scottish government data (National Records of Scotland, 2017) the adult population was 4,372,939. Previous FCA research (Coppack et al, 2015) had found the differences in results between UK regions to be marginal so it could be argued that a survey undertaken in Scotland would produce results that can be applied to the UK as a whole. The sample was inspected in terms of its reliability to represent the wider Scottish population and validity with respect to the appropriateness of the data being gathered in this study. It was concluded that the sample of data collected in this study was representative is terms of:

- i. Size
- ii. Income distribution of participants
- iii. Education levels of participants
- iv. Participants employed in the financial services sector

The survey consisted of two parts; the first section was comprised of demographic questions such as gender, age, education level, and income. These were considered important variables owing to the research conducted by the regulators into consumer vulnerability (Coppack et al, 2015), access to financial services (Collard et al, 2016) and ageing populations (McLoughlin et al, 2017).

The second section of the survey covered the main theoretical questions which have been developed from the rationales highlighted in Section 6.3 above.

6.6 Analysis

The third objective of the research was to analyse and evaluate the data to determine:

- i. differences in the hypotheses results based on underlying demographic factors
- ii. the direction of the relationship between each demographic and each rationale for regulation
- iii. which of the demographic factors have the most impact on confidence in each of the hypotheses.

The second contribution this research has made is in respect of the continued debate into the use of parametric methods of data analysis when using Likert scales. As discussed in Chapter 3, having considered the assumptions that must be met to perform parametric tests the researcher was confident that their use would be statistically sound. To verify this, non-parametric tests were used as robustness checks. As Table 4.3 in Chapter 4 illustrates, no meaningful differences were found. Therefore, this study also contributes to this field of the literature.

In summary, the first method of analysis looked at each theoretical question based on the individual demographic factors in turn. The second method of analysis looked at correlations between certain demographic variables. The third method of analysis used multiple stepwise regressions to assess which demographics had the biggest influence on each theoretical question. Lastly, which factors had the biggest influence across the whole range of questions was evaluated.

6.7 Findings

In this section, each hypothesis and a summary of the findings from each of the statistical data analysis methods will be presented in turn.

6.7.1 The consumer is confident that the UK retail banking system is stable and secure

- i. Results of individual demographic factors:
 - a. Those employed in the sector *agreed*, those not employed in the sector *slightly agreed*
 - b. Both males and females *slightly agree*
 - c. All age groups slightly agree
 - d. Participants with different levels of education *slightly agree*
 - e. Participants at all levels of income slightly agree
- ii. Results of correlation analysis:
 - a. Working in the financial services sector tended to increase confidence that the banking system is stable and secure
 - b. The higher the level of education, the lower the confidence that the banking system is stable and secure
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Being aged between 18-29
 - Secondary education being the highest level completed
 - b. The demographic variables that decrease confidence in this area are:
 - Not being employed in financial services

Overall, consumers *slightly agree* that the UK retail banking sector is stable and secure. The correlation and regression analysis results concur, however, the regression analysis also identified the youngest age group as an impact factor in confidence levels.

The main implication of consumers from all demographic backgrounds only *slightly agreeing* that they are confident that the retail banking system is stable and secure is the threat to the stability of the financial system itself. As discussed in the Literature Review, when levels of trust and confidence are high banks can rely on stable funding. However, when levels of trust and confidence are low, systemic risk is higher. Younger consumers with low levels of education tended to present higher *Claire Lynne McCafferty PhD April 2020 Page 255*

levels of confidence, yet traditionally these demographics are not likely to hold the majority of the wealth in the system. Therefore, if already low levels of confidence were to be impacted by yet another scandal, this could cause significant stress to the system.

6.7.2 The consumer is confident that the money in their retail bank account(s) is protected should their bank fail

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector agreed
 - b. Both males and females agree
 - c. All age groups agree
 - d. Overall, all education levels agree
 - e. Those earning above the minimum wage but below the national average, *slightly agree*
 - f. Those earning above the national average, and those earning above the higher rate *agree*
- ii. Results of correlation analysis:
 - a. Working in financial services tended to increase confidence that deposits were protected should the bank fail
 - b. The higher the level of income, the higher the level of confidence that deposits were protected should the bank fail
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Being employed in financial services
 - b. The demographic variables that decrease confidence in this area are:
 - Being aged between 18-29
 - Having an annual income between £13,501-£27,600

The correlation and regression analysis results concur, however, the regression analysis also identified the youngest age group as an impact factor in confidence

levels and detected the specific income range which had the lowest level of confidence; the group earning between minimum wage and the UK national average. Overall, consumers *agreed* they are confident that the money in their retail bank account(s) is protected should their bank fail. Even so, those with higher levels of income tended to record lower levels of confidence. The main implication of this is the risk of bank runs. This could cause significant stress to individual banks if higher-net-worth consumers suddenly withdrew their funds. Conversely but predictably, those working in financial services, which are most likely to be well-versed in the Financial Services Compensation Scheme, tended to record higher levels of confidence.

6.7.3 The consumer is confident that their retail bank will provide them with sufficient information to make informed decisions

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector *slightly agreed*
 - b. Males *agreed* and females *slightly agreed* although there was no statistical difference between the groups
 - c. The youngest and oldest age groups *agreed*, however, the two age groups between *slightly agreed*
 - d. The group with the highest level of education has the lowest level of confidence, *slightly agree*
 - e. The level of confidence is lowest, between *slightly disagree* and *slightly agree*, for the highest income group, and highest, *agree*, for the lowest income group
- ii. Results of correlation analysis:
 - a. A higher level of education tended to decrease confidence that their retail bank will do the right thing
 - A higher level of income tended to decrease confidence that their retail banks will provide them with sufficient information to make informed decisions

iii. Regression analysis:

- a. The demographic variables that increase confidence in this area are:
 - Being aged between 18-29
 - Secondary education being the highest level completed
- b. The demographic variables that decrease confidence in this area are:
 - Having income between £27,601- £43,000

The correlation and regression analysis results concur, however, the regression analysis also detected the youngest age group as an impact factor in confidence levels. Additionally, the income group earning over the national average but under the higher-rate tax band was also identified as a significant factor.

Overall, consumers only *slightly agreed* that their retail bank will provide them with sufficient information to make informed decisions. The main implication of those earning above the UK national having the lowest levels of confidence is the potential for loss of business for the retail banks. There is a risk that these consumers could turn to alternative providers to assist them with financial matters. This could indirectly impact other consumers since a loss of profit for the retail banks may see a rise in fees and charges in other areas to 'balance the books'.

6.7.4 The consumer is confident that their retail bank will do the right thing

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector *slightly agree*
 - b. Both males and females slightly agree
 - c. The youngest age group recorded the highest level of confidence, *agree*, however, the other age groups *slightly agree*
 - d. The group with the lowest level of education recorded the highest level of confidence, *agree*
 - e. The highest earners recorded the lowest level of confidence, *slightly disagree*, however, the lowest earners recorded the highest level of confidence, *agree*

- ii. Results of correlation analysis:
 - a. A higher level of education tended to decrease confidence that their retail bank will do the right thing
 - b. A higher level of income tended to decrease confidence that their retail bank will do the right thing
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Being aged between 18-29
 - Secondary education being the highest level completed

The correlation analysis identified a relationship with income that is not considered to be a significant impact factor. Regression analysis also identified the youngest age group as an impact factor in confidence levels.

Overall, consumers only *slightly agreed* they were confident that their retail bank will do the right thing. Education and income were significant factors, with increased levels of these factors having a negative correlation with levels of confidence. The main implication of this is the potential for consumer detriment if they do not buy suitable financial products due to the belief that their retail bank will not behave in their best interests.

6.7.5 The consumer is confident that they are protected from retail bank misconduct

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector *slightly agree*
 - b. Both males and females slightly agree
 - c. The youngest group recorded the highest level of confidence, agree
 - d. The group with the lowest level of education recorded the highest level of confidence, *agree*
 - e. The lowest earners recorded the highest level of confidence, agree
- ii. Results of correlation analysis:

- a. A higher level of education tended to decrease confidence that fraudulent activity was reduced
- b. A higher level of income tended to decrease confidence that fraudulent activity was reduced
- iii. Regression analysis:
 - b. The demographic variables that increase confidence in this area are:
 - Being aged between 18-29
 - Secondary education being the highest level completed

A relationship with income was detected in the correlation analysis which is not considered to be a significant impact factor; again age was a significant impact factor in confidence levels, however, there doesn't appear to be a directional relationship. Overall, consumers only *slightly agreed* they are confident that they are protected against retail bank misconduct. The financial crises and numerous mis-selling scandals caused a substantial deterioration in the reputation of the banking industry, which even after more than a decade, is still recovering. The main implications of this include those discussed above, namely, unstable funding, increased risk of bank runs, loss of business for retail banks and reduction in consumer welfare. There is again a negative correlation with levels of confidence and levels of income and educational attainment. The youngest age group recorded the highest levels of confidence which may be a result of being too young to remember the crisis; at least for some.

6.7.6 The consumer is confident that they can trust the expertise of their retail bank

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector *slightly agree*
 - b. Both males and females slightly agree
 - c. The youngest age group recorded the highest level of confidence, *agree*, however, the other age groups *slightly agree*

- d. The group with the lowest level of education recorded the highest level of confidence, *agree*
- e. The highest earners recorded the lowest level of confidence, *slightly disagree*, however, the lowest earners recorded the highest level of confidence, *agree*
- ii. Results of correlation analysis:
 - a. Older consumers tended to be less confident that they could trust the expertise of their retail bank
 - b. A higher level of education tended to decrease confidence that they could trust the expertise of their retail bank
 - c. A higher level of income tended to decrease confidence that they could trust the expertise of their retail bank
- i. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - i. Being aged between 18-29
 - ii. Secondary education being the highest level completed
 - iii. College education being the highest level completed
 - b. The demographic variables that decrease confidence in this area are:
 - Having income of £150K+

The correlation and regression analysis results concur; however, the regression analysis also identified the highest earning group, the additional tax rate band, is a significant impact factor in confidence levels.

Overall, consumers only *slightly agreed* they are confident that they can trust the expertise of their retail bank. There is, again, a negative correlation with levels of confidence and levels of income and educational attainment. Also, age was found to be negatively correlated. The highest level of income, the additional rate band, was a significant impact factor for decreased confidence. The highest earners responded that they *slightly disagree* that they are confident that they can trust the expertise of their retail bank. As discussed above the implications of this include

unstable funding, increased risk of bank runs, loss of business for retail banks and reduction in consumer welfare.

6.7.7 The consumer is confident that the price of retail banking products is fair

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector *slightly agree*
 - b. Both males and females *slightly agree*
 - c. The youngest group recorded the highest level of confidence, agree
 - d. The group with the highest level of education recorded the lowest level of confidence, slightly disagree
 - e. The lowest earners recorded the highest level of confidence, slightly agree
- ii. Results of correlation analysis:
 - a. Older consumers tended to be less confident that the price of retail banking products was fair
 - b. A higher level of education tended to decrease confidence that distributive justice was maintained by regulation
 - c. A higher level of income tended to decrease confidence that distributive justice was maintained by regulation
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Being aged between 18-29
 - Secondary education being the highest level completed
 - Having income of <£13,500

The correlation and regression analysis results concur, however, the correlation analysis also demonstrates the negative relationship between the independent and dependent variables.

Overall, consumers only slightly agreed they are confident that the price of retailbanking products is fair.There is, again, a negative correlation with levels ofClaire Lynne McCaffertyPhD April 2020Page 262

confidence and levels of income, educational attainment and age. Ernst and Young (2014) found that 66% of consumers who had switched in the past 12 months had done so due to fees and charges. The main implication of this is that the big four could lose some of their market share to the 'Challenger Banks'.

6.7.8 The consumer is confident that they will not be unlawfully discriminated against by their retail bank

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector agree
 - b. Both males and females agree
 - c. All age groups agree
 - d. All education groups agree
 - e. The higher rate additional rate income group recorded the lowest level of confidence, *slightly agree*
- ii. Results of correlation analysis:
 - a. A higher level of education tended to decrease confidence that they will not be unlawfully discriminated against by their retail bank
 - b. A higher level of income tended to decrease confidence that community values were protected by regulation
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Having income of <£13,500

Education was detected in the correlation analysis as having a negative relationship; however, this has not been identified as a significant impact factor for confidence in the regression analysis.

Overall, consumers *agreed* they are confident that they will not be unlawfully discriminated against by their retail bank. This is particularly important for vulnerable groups who could find they are discriminated against but may lack a voice. Income was a significant factor, with those on the lowest income having the highest levels of *Claire Lynne McCafferty PhD April 2020 Page 263*

confidence, the additional rate taxpayer *slightly agreed* that they are confident that they will not be unlawfully discriminate against by their retail bank. All other demographic groups, and categories within the groups, *agreed* with the statement. The UK has legislation in place to protect all citizens, not only consumers of financial services. The Financial Ombudsman Service (2004; 2015) has found examples of discrimination of consumers at the hands of their retail banks. The main implication of this is that if any particular bank appears to repeatedly fall short of expectations and the law they could suffer both reputational and financial loss, leading to unstable funding, increased risk of bank runs, and loss of business and increased consumer detriment.

6.7.9 In general, I have confidence that as a retail bank customer I am adequately protected

- i. Results of individual demographic factors:
 - a. Both those employed in the sector, and those not employed in the sector agree
 - b. Males *agree* and females slightly *agree*, although there was no statistical difference in the results
 - c. The lowest age group and the highest age group *agree,* the other two groups slightly agree
 - d. The lowest level of education group, *agree*, has a higher level of confidence than the other groups
 - e. The lowest income group recorded the highest level of confidence, agree
- ii. Results of correlation analysis:
 - a. A higher level of education tended to decrease confidence that, in general, they were adequately protected
- iii. Regression analysis:
 - a. The demographic variables that increase confidence in this area are:
 - Secondary education being the highest level completed
 - Being aged between 18-29

Age was identified as a significant impact factor for overall confidence levels, however, no directional relationship was found.

This is the third contribution of this study. The three methods of analysis inform the three key stakeholders of certain phenomena discovered in this research:

- i. Hypothesis tests. These examined the differences in levels of confidence between each demographic group. This would be of interest to a range of consumer groups who may support consumers of different ages, and those who have require specific support due to lower literacy and numeracy skills, or those facing financial difficulties. An interesting finding, however, is that higher education and income consumers reported lower levels of confidence. This is discussed further in Section 6.11.
- ii. Correlation analysis. The regulator would potentially be very interested to understand the demographic factors that determine whether confidence is increased or decreased in a specific area of regulation. Particularly as they tend to focus initiatives based on targeting demographic factors such as age, education and income. This is discussed in more detail in Section 6.9.
- iii. Regression analysis. This analysis determined which demographic factors have the most impact on levels of confidence in specific areas of regulation. This would be of particular interest to banks who are taking a more holistic approach to consumer segmentation. This is discussed in more detail in Section 6.10

6.7.10 Structure

As discussed in the Literature Review, the structure of an organisation can influence consumer confidence levels. With levels of confidence across almost all areas low, consideration should be given to Llewellyn's (2014) argument that a more diversified financial system would bring economic, systemic and welfare benefits. Indeed, it could lead to a reduction in some of the market failures that regulation seeks to address. For example, as mutual firms focus on their members the agency problem is better addressed and consumers can be more confident that they will be provided with sufficient information to make informed decisions. This was the third hypothesis

tested in the study and consumers only slightly agreed they felt confident in this area. Additionally, the Competition Commission (2002) found that the complex monopoly that is currently operating in the UK has led to consumer detriment due to reduced competition and through exploitative practices such as excessive pricing. Diversification of the market could lead to fairer pricing practices. Confidence that the price of retail banking products is fair was tested under hypothesis seven in the study; again consumers were only slightly confident that this was the case.

6.7.11 Culture

As discussed in the Literature Review, Spicer (2014) found that retail bank culture was to blame for widespread bad practices and failures. Specifically, aggressive sales techniques were encouraged, if not a direct result of, incentive structures which were designed to reward maximum sales with uncapped bonuses; the ultimate goal being shareholder wealth maximisation (due to the shareholder value model structure). All the while that the sales culture in retail banking was growing, the FSA was trying, with limited success, to embed 'Treating Customer's Fairly' (TCF) into product and remuneration design. The FSA undertook substantial work to improve the culture of regulated firms, including their TCF initiative which is discussed in detail in Chapter 2. Following their demise in 2013 the FCA published an update of their thematic review, 'Risks to customers from financial incentives' (2014c). This report promoted the need for firms to develop a culture that focuses on customers, rather than maximising profits and bonuses for sales staff. The correlation between poorly designed incentive structures and numerous mis-selling scandals could not be denied. The FSA's previous investigations had revealed that incentive schemes were likely to lead staff to mis-sell to meet targets and gain bonuses. As discussed in Chapter 2, this prompted Barclays, the Cooperative Bank and HSBC to remove sales targets from frontline staff. The FCA have continued to promote the need for cultural change within the industry, a recent publication, 'FCA Mission: Approach to Supervision' (2019), details how they expect both the individual and the firm to behave and the roles they play with respect organisational culture. In particular senior managers, who 'lead and shape' the culture of the organisation, are subject

Claire Lynne McCafferty

to minimum standards checks and must be approved by the regulator as 'fit and proper' to perform their role. The Senior Managers and Certification Regime (SM&CR) introduced a statutory duty of responsibility in March 2017 and is discussed in more detail in the next section. Even the financial crisis did not appear to have provided sufficient lessons. Spicer (2014) found that sales targets had simply been rebranded 'customer needs' or 'customer outcomes' in some cases. He predicted it would be a whole generation before a marked change would be realised. Hypothesis four tested how confident the consumer was that their retail bank will do the right thing. Consumers slightly agreed that they were confident in this area. Doing the right thing may come down to an individual's moral compass. However, when taken in an organisational context, it broadly comes down to the culture of the company, their ethical code. Yet, as discussed in the Literature Review there is no mandatory ethical code for the banking industry. Hypothesis five tested how confident the consumer was that they were protected against retail bank misconduct. Consumers only slightly agreed that they were confident in this area. Interestingly, Vigeo-Eiris (2017) found that a strong ethical culture decreased a bank's vulnerability to misconduct. Hypothesis six tested how confident the consumer was that they can trust the expertise of their retail bank. The FCA has published several papers (2014; 2015; 2018) which identify the groups that are most likely to depend on this expertise. Yet again, consumers recorded that they only slightly agreed they were confident in this question. It would appear that the issue of culture cannot be successfully addressed without considering the ethics of the organisation. This is discussed in more detail in the next section.

6.7.12 Organisational Ethics

As discussed in the Literature Review, there are questions about who is accountable and responsible for a firm's ethical, or otherwise, decisions. Since the financial crisis there have been reforms within the industry to increase the accountability of the board, and of senior managers, of financial institutions. In March 2017, the SM&CR introduced a statutory duty of responsibility. Each senior manager must have an agreed Statement of Responsibility which sets out in detail the areas of the business

Claire Lynne McCafferty

they are accountable for. The aims of the regime are twofold: to foster a culture where staff at all levels take personal responsibility for their actions; and to ensure firms and their staff can clearly demonstrate where the responsibility lies. Senior managers are also responsible for ensuring that employees - who do not have to be approved by the FCA, but whose roles could cause significant detriment to the firm or consumers –undergo an annual fit and proper check for their role. The FCA intend to pursue individual responsibility rigorously (FCA, 2019a).

However, there is a general consensus that regulation should not be expected to ensure an ethical industry (Jackman, 2012; Black and Anderson, 2013); rather the institutions themselves should be striving to create a visibly ethical culture from the 'top-down'. Every regulated firm must adhere to the FCA Principles for Business, which set out 'fundamental obligations'. Additionally, an industry-wide ethical group, the Banking Standards Board, shares best practice across the industry and conducts an annual survey, the last three years results of which have been somewhat mixed (BSB, 2018). Nonetheless, the use of ethical codes within an organisation has been found to lead to more ethical responses to common business dilemmas (McKinney, Emerson and Neubert, 2010); each of the 'Big 4' subscribe to such codes. Such codes should seek to address some of the market failures; however, the results of this research reveal that levels of confidence in these areas are low. For example, taking advantage of the consumer's lack of knowledge (H3), actions that are not in the best interests of the consumer (H4), protection against misconduct (H5); protection of vulnerable consumers (H6); and protection against unfair pricing (H7). Hypothesis three tested how confident consumers were that their retail bank will provide them with sufficient information to make informed decisions. It is clear that there is some overlap between regulation, culture and ethics and that the industry still has some way to go to find the right balance and repair the reputation and trustworthiness of the sector.

6.8 Limitations

The main limitations of the study are:

- Firstly the location. Data was collected from residents of Scotland only, and therefore the results are specific to Scotland (although previous studies in this field have found differences between UK regions were marginal).
- ii. Secondly, a survey can only report the relationships between variables and cannot claim cause and effect, that is, the research can answer the 'what' but not the 'why'.
- iii. Lastly, a major limitation of stepwise regression is the increased likelihood of committing a type 1 error (Pandit and Khairullah, 2015), i.e. rejecting a true null hypothesis, due to a sampling error. However, sampling error is less of an issue with a larger sample size (Thompson, 1995).

6.9 Practical Implications

The hypothesis tests revealed that those with higher incomes and those with higher educational attainment recorded the lowest confidence scores. Given that Consumer Groups are mainly, although not always, targeted at vulnerable consumers such as those on low incomes and facing financial difficulties, or those with lower literacy and numeracy skills this may indicate that some of these more vulnerable consumers are not accessing these specialist services due to a misplaced confidence. Consumer Groups may be interested to find out the reasons why there is this disparity so that they can target their interest groups accordingly.

The correlation analysis revealed a negative correlation between level of confidence in retail banking regulation and educational attainment and levels of confidence in retail banking and income levels. The FCA's Consumer Protection Objective should address issues which cause consumer's detriment. The FCA has stated that this is their most important objective. Perhaps then, their targeted initiatives for improving financial education and ensuring that those who face financial problems are dealt with fairly by their retail bank are having a positive impact. However, this does indicate that higher income and higher educational attainment groups may have needs that are not currently being met. The regulator may be interested to find out the reasons why they have low confidence in these areas of regulation.
The regression analysis revealed that age, income and education were significant factors when measuring confidence in retail banking regulation. Older consumers, higher income and more highly educated consumers were less confident than their younger, lower income, lower educational attainment counterparts. Banks now use big data to understand their consumers and are interested in where their segments, such as age, gender, income, education, channel usage, loyalty, life stage etc., overlap. This research demonstrates that some of the banks' potentially most profitable consumers are the least confident in them. In today's challenging and highly competitive market, addressing this could be a significant differentiator.

The stability of the economy depends on confidence in the banking system. Indeed, the root of government intervention stems from the need to ensure households and businesses are healthy and the threat of destabilising events is reduced. The low levels of confidence that the banking system is stable and secure, and marginally higher levels of confidence that consumer money is protected should their bank fail should indicate to government that the schemes they have in place are perhaps not common knowledge for consumers and could be better promoted to raise awareness; particularly to avoid a future bank run.

One of the most interesting findings in this research is that consumers who have a higher level of income and a higher level of educational attainment have lower levels of confidence in retail banking regulation. Clearly, if these groups do not feel their interests are protected there is a potential loss of custom from the segment of consumers that are likely to be highly profitable. Targeted marketing for these groups should be more focused on building trust and confidence in the institution and its products and services. Focusing on the double bottom line, pursuing social objectives in addition to financial objectives, is one method that would improve the image of some of the larger institutions.

Those with lower education levels and young consumers consistently recordedhigher levels of confidence in each of the areas that regulation seeks to address.This research cannot answer why this is.However, this could be due to naivety orthe implicit trust that these groups tend to have towards others they may feel haveClaire Lynne McCaffertyPhD April 2020Page 270

a greater understanding or experience. The regulator should approach this with caution, as many of the past mis-selling and misconduct scandals of recent decades have occurred due to the belief that banks were working in the consumers best interests when clearly this wasn't the case.

Cultural and regulatory reforms have gone some way to address this; however, as proposed by Llewellyn (2014) a restructure of the financial system to increase the number of mutuals could reduce the imbalance of powerful shareholder value model institutions which dominate the UK retail banking market. This would reduce the complex monopoly structure and the lack of personal responsibility for decisions which has been partially blamed for the Global Financial Crisis.

The Government should also take appropriate steps to reduce the risk of regulatory capture by addressing the revolving door practice in the financial sector. This would reduce the conflicts of interest and reduce another factor in public distrust; the belief that there is corruption or collusion at the top of these institutions. These beliefs were reinforced by events such as price-fixing and market-rigging. Complex products, such as protection and retirement products, may be avoided due to mistrust which increases both individual consumer detriment and overall society as the government has to support those who cannot support themselves during illness and/or old-age. Therefore, increased transparency should be enforced by the government. The introduction of a compulsory ethical code would go some way to improve this.

Overall, consumers *slightly agree* that they have confidence in retail banking regulation. Undoubtedly there is still a long way to go to improve the image of the industry and ensure that consumers are accessing appropriate products and receiving a service that is beneficial to their welfare.

6.10 Contributions to Theory

There are four main contributions to the theory and literature that this study makes:

- i. The finding that those with a higher income and higher educational attainment have less confidence in retail banking regulation
- ii. The use of theorists, practitioners and end-users in the stages of survey design,from theoretical model to empirical testing of the consumer survey
- iii. The finding that the use of parametric tests for the analysis of Likert scale data is robust
- iv. The use of three stages of analysis to inform which considered, individual demographics in relation to levels of confidence; the direction of confidence in relation to these demographics; and the combined demographic factors that have the greatest impact on the levels of confidence. Each of these stages of analysis would be of interest to different stakeholders as discussed above.

6.11 Recommendations for Future Research

Potentially, future research should be undertaken to encompass a wider participation in terms of location and examine the 'why' questions that this research uncovers.

- i. Why are those employed in the sector no more confident than the general public?
- ii. Why do the youngest consumers have the most confidence?
- iii. Why do those who have a higher level of educational attainment have less confidence in the system?
- iv. Why do those who earn a higher income have less confidence in the system?

Additionally, there is likely to be a strong relationship between educational attainment and income level. This has not been tested for in this study, but may be crucial to understanding the 'whys' for these demographic groups.

Answering the above questions could improve the effectiveness of regulation, confidence in the financial system and the profitability of financial institutions. Most importantly, it could reduce consumer detriment and increase confidence in managing day-to-day their retail banking affairs.

REFERENCES

Ackroyd, S. and Hughes, J. (1992). Data collection in context. London: Longman.

Adams, J.R., and Frantz, D. (1992). *A Full Service Bank: How BCCI Stole Billions Around the World*. California: Pocket Books.

Akinbami, F. (2011). Financial services and consumer protection after the crisis. *The International Journal of Bank Marketing*, 29(2), 134-147.

Alesina, A., and La Ferrara, E. (2002). "Who Trusts Others?" *Journal of Public Economics*, 85(2), 207-34.

Allen, F., Carletti, E., Goldstein, G., and Leonello, A. (2015). Moral Hazard and Government Guarantees in the Banking Industry. Journal of Financial Regulation, 1(1), 30-50.

Altman, M. (2012). Implications of behavioural economics for financial literacy and public policy. *Journal of Socio-Economics*, 41(5), 677-690.

Anderson, G., Sprott, H., and Olsen, B.R. (2013). Opinion: Publish Negative Results. *The Scientist Online,* Retrieved from: <u>http://www.the-</u> <u>scientist.com/?articles.view/articleNo/33968/title/Opinion--Publish-Negative-</u> <u>Results/</u>

Anonymous. (2010). Britain: The recovery and the election; End of recession. *The Economist*, 394(8667), p. 65-66. Retrieved from: <u>https://www.proquest.com/</u>

Arauzo, E. (2013). *Banking on the revolving door: Rules full of loopholes for former finance officials*. Corporate Europe Observatory. Retrieved from:

https://corporateeurope.org/en/news/banking-revolving-door-rules-full-loopholesformer-finance-officials

Association of British Credit Unions Limited (ABCUL). (2012). *Ethical Consumer endorses Credit Unions.* Retrieved from: <u>http://www.abcul.coop/media-and-</u> <u>research/news/view/221</u>

Ayadi, R. and Lastra, R.M. (2010). Proposals for reforming deposit guarantee schemes in Europe. *Journal of Banking Regulation*, 11(3), 210-222.

Baker, A. (2010). Restraining regulatory capture? Anglo-America, crisis politics and trajectories of change in global financial governance. *Royal Institute of International Affairs*, 86(3), 647-663.

Balasubramnian, B. and Cyree, K. B. (2014). Has market discipline on banks improved after the Dodd-Frank Act? *Journal of Banking and Finance*, 41, 155-166.

Baldwin, R., Cave, M., and Lodge, M. (2012). *Understanding Regulation: Theory, Strategy, and Practice.* (2nd ed.). Oxford: Oxford University Press.

Bank for International Settlements. (1999). *Capital Requirements and Bank Behaviour: the impact of the Basle Accord.* Working paper no. 1. Retrieved from: <u>https://www.bis.org/publ/bcbs_wp1.pdf</u>

Bank of England. (1991). The Performance of Major British Banks, 1970-90. *Quarterly Bulletin,* 31(4), 508-515. Retrieved from: <u>https://www.proquest.com/</u>

Bank of England. (2019). *Resolution*. Retrieved from: https://www.bankofengland.co.uk/financial-stability/resolution

Bank of England. (2019). *Structural reform*. Retrieved from: <u>https://www.bankofengland.co.uk/prudential-regulation/key-initiatives/structural-reform</u>

Banking Standards Board. (2015). *BSB starts its work and announces line- up of new board.* Retrieved from: <u>http://www.bankingstandardsboard.org.uk/bsb-starts-its-work-and-announces-line-up-of-new-board/</u>

Bannier, C.E., and Schwarz, M. (2018). Gender- and Education-Related Effects of Financial Literacy and Confidence on Financial Wealth. *Journal of Economic Psychology*, 67, 66-86.

Barclays plc. (n.d.). *Purpose and values.* Retrieved from:

https://home.barclays/who-we-are/our-strategy/purpose-and-values/

BBC News. (2007). *Run on the Bank: Northern Crock*. Retrieved from: http://news.bbc.co.uk/1/hi/business/7086473.stm

BBC News. (2014). *Banking standards oversight body to be set up*. Retrieved from: <u>https://www.bbc.co.uk/news/business-27465994</u>

BBC News. (2014). '*Big Four' UK banks set to face CMA competition probe*. Retrieved from: <u>https://www.bbc.co.uk/news/business-28357675</u>

BBC News. (2017). *UK banks 'exposed to money laundering in South Africa'*. Retrieved from: <u>https://www.bbc.com/news/business-41672793</u>

Benink, H., and Benston, G. (2005). The Future of Banking Regulation in Developed Countries: Lessons from and for Europe. *Financial Markets, Institutions and Instruments*, 14(5), 289-328.

Bennett, R. and Kottasz, R. (2012). Public attitudes towards the UK banking industry following the global financial crisis. *The International Journal of Bank Marketing*, 30(2), 128-147.

Benston, G. (1998). Regulating Financial Markets: A Critique and some Proposals, *Hobart Paper 135, London: The Institute of Economic Affairs.* Retrieved from: <u>https://iea.org.uk/publications/research/regulating-financial-markets-a-critique-and-some-proposals</u>

Benston, G. (2000). Consumer Protection as Justification for Regulating Financial-Services Firms and Products. *Journal of Financial Services Research*, 17(3), 277-301.

Benston, G. (2003). How Much Regulation of Financial Services Do We Really Need?' *Business Economics*, 38(1), 31-33.

Benston, G.J., and Kaufman, G.G. (1996). The Appropriate Role of Bank Regulation. *Economic Journal*, 106(436), 688-697.

Benton, T., and Craib, I. (2011) *Philosophy of Science: the philosophical Foundations of social thought.* (2nd ed.). Basingstoke: Palgrave Macmillan.

Bishop, G. F. (1987). Experiments with the middle response alternative in survey questions. *Public Opinion Quarterly*, 51(2), 220-232.

Bishop, P. A., and Herron, R. L. (2015). Use and Misuse of Likert Item Responses and Other Ordinal Measures. *International Journal of Exercise Science*, 8(3), 297-302.

Black, D. (2008). Banking & Finance: The blame game. *The Lawyer,* p. 32. Retrieved from <u>https://www.proquest.com/</u>

Black, J. and Anderson, K. (2016). Creating an ethical framework for the financial services industry. *The Herbert Smith Freehills and London School of Economics Regulatory Reform Forum*. Retrieved from:

http://www.lse.ac.uk/law/people/academic-staff/julia-black/Documents/black10.pdf

Blaikie, N. (2003). *Analysing Quantitative Data. From Description to Explanation.* London: SAGE Publications Limited.

Blanchflower, D.G., and Oswald, A.J. (2011). International Happiness. *NBER Working Paper Series*. w16668.

Bland, J. M., and Altman, D. (1997). Statistics Notes: Cronbach's Alpha. *British Medical Journal Clinical Research*, 314(7080), 572.

Bowen, A., Hoggarth, G., and Pain, D. (1999). 'The recent evolution of the UK banking industry and some implications for financial stability'. *The Monetary and Regulatory Implications of Changes in the Banking Industry*, BIS Conference Papers, BIS (1999), vol. 7, March. Retrieved from: https://www.bis.org/publ/confp07.pdf

Boyd, J. H., and De Nicolo, G. (2005). The Theory of Bank Risk Taking and Competition Revisited. *The Journal of Finance*, 60(3), 1329-1343.

Boyer, P. C., and Ponce, J. (2012). Regulatory capture and banking supervision reform. *Journal of Financial Stability*, 8(3), 206-217.

Braham, R. A., and Finch, C. F. (2004). The reliability of team-based primary data collectors for the collection of exposure and protective equipment use data in community sport. *British Journal of Sports Medicine*, 38(4), 1-2.

Brown, G. (2011). Keynote at Crisis and Renewal. In *INET's Bretton Woods Conference* (Bretton Woods, New Hampshire, 2011, April 8-11). Institute for New Economic Thinking. Retrieved from:

http://ineteconomics.org/net/video/playlist/conference/bretton-woods/V

Bryman, A., and Cramer, D. (2001). *Quantitative Data Analysis with SPSS Release for Windows 10: A Guide for Social Scientists*. East Sussex: Routledge.

Buckley, A. (2011). *Financial Crisis: Causes, Context and Consequences.* Harlow: Pearson Education Limited.

Canac, P., and Dykman, C. (2011). The tale of two banks: Societe Generale and Barings. *Journal of the International Academy for Case Studies*, 17(7), 11-32.

Carifio , J., and Perla, R. (2008). Resolving the 50-year debate around using and misusing Likert Scales. *Medical Education*, 42(12), 1150-1152.

Cartwright, P. (2004). Banks, Consumers and Regulation. Oxford: Hart Publishing.

Cash, J. (2017). *Ex-FCA boss Wheatley returns to finance at hedge fund*. Retrieved from: <u>https://www.moneymarketing.co.uk/ex-fca-boss-wheatley-returns-finance-hedge-fund/</u>

Cave, A. (2003, October, 17). HSBC exporting 4,000 jobs to Asian centres. *The Telegraph*. Retrieved from: <u>https://www.telegraph.co.uk/finance/2866150/HSBC-exporting-4000-jobs-to-Asian-centres.html</u>

Cerrillo-i-Martinez, A. (2017). Beyond Revolving Doors: The Prevention of Conflicts of Interests Through Regulation. *Public Integrity*, 19(4), 357-373.

Christen, R.P., Rosenberg, R., and Jayadeva, V. (2004). *Financial institutions with a "double bottom line": implications for the future of microfinance.* Consultative Group to Assist the Poor (CGAP). Retrieved from:

https://www.cgap.org/sites/default/files/CGAP-Occasional-Paper-Financial-Institutions-with-a-Double-Bottom-Line-Implications-for-the-Future-of-Microfinance-Jul-2004.pdf

Chung, W. (2013). Consumer behaviour, *Professional Adviser*, August/September 2013,10.

Citizens Advice Bureau. (2005). *Protection racket. CAB evidence on the cost and effectiveness of payment protection insurance.* Retrieved from: <u>https://www.citizensadvice.org.uk/Global/Migrated_Documents/corporate/protection_n-racket-final.pdf</u>

Citizens Advice Bureau, Scotland. (2010). Banking on the basics. Retrieved from: <u>https://www.cas.org.uk/system/files/publications/banking-on-the-basics.pdf</u>

Cohen, L., Manion, L., and Morrison, K. (2010). *Research methods in education.* (7th ed.). Abingdon: Routledge.

Collard, S., Coppack, M., Lowe, J., Sarkar, S., and FCA. (2016). *Access to Financial Services in the UK.* (FCA Occasional Paper No.17). Retrieved from: https://www.fca.org.uk/publication/occasional-papers/occasional-paper-17.pdf

Committee on Standards in Public Life. (2016). *Striking the balance: Upholding the Seven Principles of Public Life in Regulation.* Sixteenth Report Cm9327. Retrieved from:

https://www.ocsc.go.th/sites/default/files/attachment/article/16thstriking_the_balanc e_web_-_v3_220916_0.pdf

Competition and Markets Authority. (2015). *Retail banking market investigation: Regulatory framework applicable to the retail banking industry in the UK*. Retrieved from:

https://assets.publishing.service.gov.uk/media/5548cde8e5274a1575000047/Regul atory_framework_working_paper.pdf

Competition and Markets Authority. (2016). *Retail banking market investigation.* Retrieved from: <u>https://www.gov.uk/cma-cases/review-of-banking-for-small-and-</u> medium-sized-businesses-smes-in-the-uk

Competition Commission. (2002). A report on the supply of banking services by clearing banks to small and medium sized enterprises within the UK. Retrieved from: <u>https://webarchive.nationalarchives.gov.uk/+/http://www.competition-</u>commission.org.uk//rep_pub/reports/2002/462banks.htm

Competition Commission. (2009). *Market investigation into payment protection insurance.* Retrieved from:

https://webarchive.nationalarchives.gov.uk/20140402202346/http://www.competitio n-commission.org.uk/our-work/directory-of-all-inquiries/ppi-market-investigationand-remittal/final-report-and-appendices-glossary

Coppack, M., Raza, Y., Sarkar, S., Scribbins, K., and FCA (2015). *Consumer Vulnerability.* (FCA Occasional Paper No.8). Retrieved from: <u>https://www.fca.org.uk/publication/occasional-papers/occasional-paper-8.pdf</u>

Crane, A. and Matten, D. (2007). *Business Ethics* (2nd ed.). New York: Oxford University Press Inc.

Crotty, M. (1998). *The Foundations of Social Research: Meaning and Perspective in the Research Process*. London: SAGE Publications Ltd.

Cruikshank, D. (2000). Competition in UK Banking: A Report to the Chancellor of the Exchequer. Norwich: The Stationery Office.

Cullum, P. (1997). Consumers and stakeholder. *Consumer Policy Review*, 7(3), 100-104.

Davies, H. (1997). Market and Regulatory Structures in a Global Environment Speech. In *Federal Reserve Bank of Atlanta's Financial Markets Conference*. (Atlanta, 1997, 22 February). Bank for International Settlements. Retrieved from: <u>https://www.bis.org/review/r970305e.pdf</u>

Davies, R., Richardson, P., Katinaite, V., and Manning, M. (2010). Evolution of the UK Banking System. *Bank of England Quarterly Bulletin,* 50(4), 321-332.

De Nicolo, G., Favara G., and Ratnovski, L. (2012). *Externalities and Macroprudential Policy.* International Monetary Fund Research Department. Retrieved from:

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.432.4905&rep=rep1&typ e=pdf

DellaVigna, S. (2009). Psychology and Economics: Evidence from the Field. *Journal of Economic Literature*, 47(2), 315-372.

Den Hertog, J. (2000). General Theories of Regulation, *in B. Bouckaert and G. De Geest, Encyclopaedia of Law and Economics (Volume I: The History and Methodology of Law and Economics).* Cheltenham: Edward Elgar.

Denham, K. (2017, May 23). Former FCA boss Wheatley takes on hedge fund role. *FT Adviser.* Retrieved from:

https://www.ftadviser.com/regulation/2017/05/23/former-fca-boss-wheatley-takeson-hedge-fund-role/

Diamond, D. W. and Dybvig, P.H. (1983). Bank Runs, Deposit Insurance, and Liquidity. *The Journal of Political Economy*, 91(3), 401-419.

Dillow, C. (2012). The Cost of the Barclay's scandal. *Investors Chronicle*. Retrieved from: <u>https://www.investorschronicle.co.uk/2012/07/02/comment/chris-</u> <u>dillow/the-cost-of-the-barclays-scandal-YCbkgeW3WHrpZP4b1J4lvJ/article.html</u>

Dixon, M. (2006). Rethinking Financial Capability. Lessons from economic psychology and behavioural finance. *Institute for Public Policy Research*. Retrieved from:

https://www.ippr.org/files/images/media/files/publication/2011/05/financial_capabiliti es_1515.pdf

Donaldson, T. (1982). Corporations and Morality. New Jersey: Prentice Hall.

Donaldson, T. and Preston, L.E. (1995). The Stakeholder Theory of Corporation: Concepts, Evidence and Implications. *The Academy of Management Review*, 20(1), 65-9.

Dowd, K. (1996). The Case for Financial Laissez-Faire. *The Economic Journal*, 106(436), 679-687.

Dowd, K. (2009). Moral Hazard and the Financial Crisis. *Cato Journal*, 29(1), 141-166.

Duffy, B., Smith, K., Terhanian, G., & Bremer, J. (2005). Comparing Data from Online and Face-to-face Surveys. *International Journal of Market Research*, *47*(6), 615–639.

Edelman Trust Barometer. (2019). *UK results.* Retrieved from <u>https://www.edelman.com/sites/g/files/aatuss191/files/2019-</u> 02/2019 Edelman_Trust_Barometer_Global_Report.pdf

Edmonds, T. (2008). Northern Rock and financial supervision. *House of Commons Library.* Retrieved from:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN04478#fullre port

Edmonds, T. (2018). Bank branch closures. *House of Commons Library Briefing Paper No. 385*. Retrieved from:

https://researchbriefings.parliament.uk/ResearchBriefing/Summary/SN00385#fullre port

Elliott, L. (2010, June 16). George Osborne unveils sweeping City reforms. *The Guardian*. Retrieved from:

https://www.theguardian.com/business/2010/jun/16/george-osborne-city-fsabanking

Ernst and Young. (2014). *Two years of declining confidence in UK banking draws to a close.* Retrieved from: <u>http://www.ey.com/uk/en/newsroom/news-releases/14-</u> 03-10---two-years-of-declining-confidence-in-uk-banking-draws-to-a-close

Erta, K., Hunt, S., Iscenko, Z., Brambley, W, and Financial Conduct Authority. (2013). *Applying behavioural economics at the Financial Conduct Authority.* (Occasional Paper No.1). Retrieved from:

https://www.fca.org.uk/publication/occasional-papers/occasional-paper-1.pdf

Fama, E. (1970). Efficient Capital Markets: A Review of Theory and Empirical Work. *The Journal of Finance*, 25(2), 383-417.

Field, A. (2013). Discovering Statistics using SPSS: And sex and drugs and rock 'n' roll. (4th ed.). London: SAGE *Publications Limited.*

Financial Conduct Authority. (2014a). *FCA confirms price cap rule for payday lenders.* Retrieved from: <u>https://www.fca.org.uk/print/news/press-releases/fca-</u> <u>confirms-price-cap-rules-payday-lenders</u>

Financial Conduct Authority. (2014b). *FCA fines five banks £1.1 billion for FX failings and announces industry-wide remediation programme*. Retrieved from: <u>https://www.fca.org.uk/news/press-releases/fca-fines-five-banks-%C2%A311-billion-fx-failings-and-announces-industry-widef</u>

Financial Conduct Authority. (2014c). *Risks to customers from financial incentives an update.* Retrieved from: <u>https://www.fca.org.uk/publication/thematic-</u> <u>reviews/tr14-04.pdf</u>

Financial Conduct Authority. (2016a). *Economics for Effective Regulation. Occasional Paper 13.* Retrieved from:

https://www.fca.org.uk/publication/occasional-papers/occassional-paper-13.pdf

Financial Conduct Authority. (2016b). *Interest rate hedging products (IRHP).* Retrieved from: <u>https://www.fca.org.uk/consumers/interest-rate-hedging-products</u>

Financial Conduct Authority. (2016c). *Ring fencing – what certain high street bank customers need to know. Retrieved from:* <u>https://www.fca.org.uk/consumers/ring fencing</u>

Financial Conduct Authority. (2018a). *FCA Handbook*. Retrieved from: <u>https://www.handbook.fca.org.uk/handbook</u>

Financial Conduct Authority. (2018b). *The financial lives of consumers across the UK Key findings from the FCA's Financial Lives Survey 2017*. Retrieved from: https://www.fca.org.uk/publication/research/financial-lives-consumers-across-uk.pdf

Financial Conduct Authority. (2019a). *FCA Mission: Approach to Supervision.* Retrieved from: <u>https://www.fca.org.uk/publication/corporate/our-approach-</u> <u>supervision-final-report-feedback-statement.pdf</u>

Financial Conduct Authority. (2019b). *Monthly PPI refunds and compensation*. Retrieved from: <u>https://www.fca.org.uk/data/monthly-ppi-refunds-and-</u> <u>compensation</u>

Claire Lynne McCafferty

Financial Services Authority. (2002). *FSA Annual Report 2001-02*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar01_02.pdf

Financial Services Authority. (2003). *FSA Annual Report 2002-03*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar02_03/ar02_03.pdf

Financial Services Authority. (2004). *FSA Annual Report 2003-04*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar03_04/ar03_04.pdf

Financial Services Authority. (2005). *FSA Annual Report 2004-05*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar04_05/ar04_05.pdf

Financial Services Authority. (2006a). *Financial Capability in the UK: Establishing a Baseline*. Retrieved from: <u>http://www.fsa.gov.uk/pubs/other/fincap_baseline.pdf</u>

Financial Services Authority. (2006). *FSA Annual Report 2005-06.* Retrieved from: <u>http://www.fsa.gov.uk/pubs/annual/ar05_06/ar05_06.pdf</u>

Financial Services Authority. (2007). *FSA Annual Report 2006-07*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar06_07/ar06_07.pdf

Financial Services Authority. (2008). *FSA Annual Report 2007-08.* Retrieved from: <u>http://www.fsa.gov.uk/pubs/annual/ar07_08/ar07_08.pdf</u>

Financial Services Authority (2008a). *The supervision of Northern Rock: a lessons learned review*. Retrieved from: <u>https://www.fca.org.uk/publication/corporate/fsa-nr-report.pdf</u>

Financial Services Authority. (2009a). *FSA Annual Report 2008-09*. Retrieved from: <u>http://www.fsa.gov.uk/pubs/annual/ar08_09/ar08_09.pdf</u>

Financial Services Authority. (2009b). *The Turner Review: A regulatory response to the global banking crisis.* Retrieved from: http://www.fsa.gov.uk/pubs/other/turner_review.pdf

Financial Services Authority. (2010). *FSA Annual Report 2009-10*. Retrieved from: http://www.fsa.gov.uk/pubs/annual/ar09_10/ar09_10.pdf

Claire Lynne McCafferty

Financial Services Authority. (2010a). *FSA Mortgage Market Review: Responsible Lending.* Retrieved from: <u>https://www.fca.org.uk/publication/consultation/fsa-cp10-</u>16.pdf

Financial Services Authority. (2011). *FSA Annual Report 2010-11*. Retrieved from: <u>http://www.fsa.gov.uk/pubs/annual/ar10_11/ar10_11.pdf</u>

Financial Services Authority. (2012). *FSA Annual Report 2011-12.* Retrieved from: <u>http://www.fsa.gov.uk/pubs/annual/ar11-12/ar11-12.pdf</u>

Financial Services Authority. (2012a) *Journey to the FCA*. Retrieved from: <u>https://www.fca.org.uk/publication/corporate/fsa-journey-to-the-fca.pdf</u>

Financial Services Authority. (2013). *FSA Annual Report 2012-13.* Retrieved from: <u>https://www.fca.org.uk/publication/annual-reports/fsa-annual-report-12-13.pdf</u>

Financial Services Compensation Scheme. (2018). *Compensation Limits*. Retrieved from: <u>https://www.fscs.org.uk/what-we-cover/</u>

Freeman, R. E. (2010). *Strategic Management. A Stakeholder Approach* (2nd ed.). Cambridge: Cambridge University Press.

French, P. A. (1979). The Corporation as a Moral Person. American Philosophical *Quarterly*, 16(3), 207-215.

French, P. A. (1984). *Collective and Corporate Responsibility*. New York: Columbia University Press.

French, P.A. (1995). Corporate Ethics. Fort Worth, Tex: Harcourt Brace College.

French, P.A. (2005). Inference gaps in moral assessment: capitalism, corporations and individuals. *International Social Science Journal*, 57(185), 573-584.

Frey, B.B. (2018). Stepwise Regression. *The SAGE Encyclopedia of Educational Research, Measurement, and Evaluation.* London: SAGE Publications Ltd.

Gamble, A. (2009). British politics and the financial crisis. *British Politics*, 4(4), 450-462.

Garland, R. (1991). The Mid-Point on a Rating Scale: Is it desirable? *Marketing Bul n*, 2(1), 66-70.

Gaskill, J., and Ashton, J. (2008). Developing a financial services planning profession in the UK. *Journal of Financial Regulation and Compliance*, 16(2), 159-172.

George, H. (2018). *Lloyds Reaches Last Leg of Crisis-Era Clash Over HBOS Takeover*. Retrieved from <u>https://www.bloomberg.com/news/articles/2018-03-02/lloyds-reaches-last-leg-of-crisis-era-clash-over-hbos-takeover</u>

Gilbert, L. (2015). *Reforming the financial services revolving door*. The Hill. Retrieved from: <u>https://thehill.com/blogs/pundits-blog/finance/247962-reforming-the-financial-services-revolving-door</u>

Ginsberg, J. (2014). *How "revolving doors" between banks and regulators could improve oversight.* Chicago Booth Review. Retrieved from:

http://review.chicagobooth.edu/magazine/fall-2014/how-revolving-doors-betweenbanks-and-regulators-could-improve-oversight

Glass, G. V., Peckham, P. D., Sanders, J.R. (1972). Consequences of Failure to Meet Assumptions Underlying the Fixed Effects Analyses of Variance and Covariance. Review of Educational Research, 42(3), 237-288.
Goff, S. (2011, October 25). Banks sign up to 'ethical' code. *The Financial Times*.
Retrieved from: <u>https://www.ft.com/content/6b67e516-ff0e-11e0-9b2f-</u>00144feabdc0

Goodhart, C. (2008). The regulatory response to the Financial crisis. *Journal of Financial Stability*, 4(4), 351-358.

Goodhart, C. Hartmann, P. Llewellyn, D. Rojas-Suarez, L. Weisbrod, S. (1998). *Financial Regulation. Why, how and where now?* London: Routledge.

Gosling, L. (2011). I'm not H.A. PPI. *Investment Week,* p.40. Retrieved from: <u>https://www.investmentweek.co.uk/investment-week/opinion/2070775/i-m-ha-ppi</u>

Gray, D. E. (2007). *Doing Research in the Real World* (4th ed.). London: SAGE Publications Ltd.

Green, C.F. (1989). Business Ethics in Banking. *Journal of Business Ethics*, 8(8), 631-634.

Green, J., and Perry, J. (2016). *OFT to FCA: two years of FCA regulation of the consumer credit industry.* Retrieved from: <u>https://www.ashurst.com/en/news-and-insights/legal-updates/oft-to-fca-two-years-of-fca-regulation-of-the-consumer-credit-industry/</u>

Green Century Funds. (2013). *Institutional Investors Press Palm Oil Industry to Stop Burning Rainforests and Driving Climate Change*. Retrieved from: <u>https://www.greencentury.com/institutional-investors-press-palm-oil-industry-to-</u> <u>stop-burning-rainforests-and-driving-climate-change/</u>

Growth from Knowledge. (2019). *Consumer Confidence Barometer*. Retrieved from:

https://www.gfk.com/fileadmin/user_upload/dyna_content/GB/documents/Press_R eleases/2018/CCB/GfK_Consumer_Confidence_Barometer_Charts_January_2019 .pdf

Guba, E.G. (1990). The Paradigm Dialog. London: SAGE Publications Ltd.

Gunter, B., Nicholas, D., Huntington, P., and Williams, P. (2002). Online versus offline research: implications for evaluating digital media, *Aslib Proceedings*, 54(4), 229 – 239.

Hackley, C. (2003). *Doing Research Projects in Marketing, Management and Consumer Research.* Abingdon: Routledge.

Hall, M.J.B. (2009). The sub-prime crisis, the credit crunch and bank "failure". An assessment of the UK authorities' response. *Journal of Financial Regulation and Compliance*, 17(4), 427-452.

Hamalainen, P., Pop, A., Hall, M., and Howcroft, B. (2011). Did the Market Signal Impending Problems at Northern Rock? An Analysis of Four Financial Instruments. *European Financial Management*, 18(1), 68-87.

Hanson, S., Kashyap, A. K., and Stein, J. C. (2011). A Macroprudential Approach to Financial Regulation. Journal of Economic Perspectives, 25(1), 3-28.

Hardie, I., and Macartney, H. (2016). EU ring fencing and the defence of too-big-tofail banks. *West European Politics*, 39(3), 503-525.

Harford, T. (2009, September 19). To Nudge Is One Thing, To Nanny Quite Another: The Undercover Economist. *The Financial Times.* Retrieved from: <u>https://www.ft.com/content/ac77105a-a0f5-11de-a88d-00144feabdc0</u>

Harnay, S., and Scialom, L. (2016). The influence of the economic approaches to regulation on banking regulations: a short history of banking regulations. *Cambridge Journal of Economics*, 40(2), 401-426.

Hellman, T. F., Murdock, K. C., and Stiglitz, J.E. (2000). Liberalization, moral hazard in banking, and prudential regulation: Are capital requirements enough? *The American Economic Review*, 90(1), 147-165.

Hemraj, M. B. (2005). The Regulatory Failure: The Saga of BCCI. *Journal of Money Laundering Control*, 8(4), 346-353.

Hess, K.M. (2013). "If you tickle us....": How corporations can be moral agents without being persons. *Journal of Value Inquiry*, 47(3), 319-335.

Hinton, P. R., McMurray, I., and Brownlow, C. (2014). *SPSS Explained*. (2nd ed.) East Sussex: Routledge.

Hitchcock, G., and Hughes, D. (1995). *Research and the Teacher: A Qualitative Introduction to School-Based Research.* (2nd ed.). London: Routledge.

HM Revenue and Customs. (2015). *UK Income Tax Liabilities Statistics: 2015-16 Survey of Personal Incomes, with projections to 2018-19.* Retrieved from: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attac</u> <u>hment_data/file/710866/Income_Tax_Liabilities_Statistics_May_2018.pdf</u>

HSBC Bank plc. (2017). Sustainability: Building our business for the long term. Retrieved from: <u>http://www.hsbc.com/citizenship/sustainability</u>

HSBC Bank plc. (2019). *Growing our business in the right way.* Retrieved from: <u>http://www.hsbc.com/our-approach/our-values</u>

HSBC Media Releases. (2017). *HSBC Holdings plc Expiration of 2012 Deferred Prosecution Agreement*. Retrieved from: <u>https://www.hsbc.com/media/media-</u> <u>releases/2017/hsbc-holdings-plc-expiration-of-2012-deferred-prosecution-</u> <u>agreement</u>

Independent Commission on Banking. (2011). *Final Report: Recommendations.* Retrieved from:

http://webarchive.nationalarchives.gov.uk/20131003105424/https:/hmtsanctions.s3.amazonaws.com/icb%20final%20report/icb%2520final%2520report% 5B1%5D.pdf

Ionut, P. A. (2017). Evolution of Customers' Segmentation Techniques in Retail Banking. *Annals of the Constantin Brâncuşi University of Târgu Jiu, Economy Series*, Special Issue, volume I/2017. Retrieved from: http://www.utgiiu.ro/revista/ec/pdf/2017-01.Volumul 1 Special/27 Pascu.pdf

Jackman, D. (2013, July 22). Business won't be ethical until it shares society's values again. *The Independent.* Retrieved from:

https://www.independent.co.uk/voices/commentators/david-jackman-businesswont-be-ethical-until-it-shares-societys-values-again-7965044.html

Jamieson, S. (2004). Likert scales: how to (ab)use them. *Medical Education*, 38(12), 1217-1218.

Jenkins, P. (2016, March 7). Bank ethics under scrutiny as UK sets up a standards body. *The Financial Times*. Retrieved from: <u>https://www.ft.com/content/1d2c3272-e44b-11e5-a09b-1f8b0d268c39</u>

Jones, G.R., and George, J.M. (2008). *Contemporary Management* (5th ed.). New York: McGraw-Hill Irwin.

Kalton, G., Roberts, J., and Holt, D. (1980). The Effects of Offering Middle Response Option with Opinion Questions. *The Statistician*, 29(1), 65-79.

Keeley, M. C. (1990). Deposit Insurance, Risk, and market Power in Banking. *The American Economic Review*, 80(5), 1183-1200.

Kenc, Turalay. (2016). Macroprudential Regulation: History, Theory and Policy. Bank for International Settlements Paper no. 86c.

Knapp, T.R. (2016). Treating Ordinal Scales as Interval Scales: an attempt to resolve the controversy. *Nursing Research*, 39(2), 121-123.

Krugman, P. (2008). *The Return of Depression Economics and the Crisis of 2008.* London: Penguin Books Ltd.

Kuzon, W. M., Urbanchek, M. G., and McCabe, S. (1996). The Seven Deadly Sins of Statistical Analysis. *Annals of Plastic Surgery*, 38(1), 265-272.

Kwak, S. G., and Kim, J. H. (2017). Central limit theorem: the cornerstone of modern statistics. *Korean Journal of Anaesthesiology*, 70(2), 144-156.

Lawson, A. (2014, October 28). Lloyds to axe 9,000 jobs and close 200 branches as it eyes digital growth. *Evening Standard*. Retrieved from:

https://www.standard.co.uk/business/business-news/lloyds-to-cut-9000-jobs-andclose-150-branches-9822362.html

Lehmann, M. (2014). Volcker Rule, Ring fencing or Separation of Bank Activities: Comparison of Structural Reform Acts around the World. LSE Law, Society and Economy Working Papers 25/2014. Retrieved from:

http://eprints.lse.ac.uk/60570/1/WPS2014-25_Lehmann.pdf

Lewis, M. (2007). Stepwise versus Hierarchical Regression: Pros and Cons. *Annual Meeting of the Southwest Educational Research Association.* Retrieved from:

http://www.academia.edu/1860655/Stepwise_versus_hierarchical_regression_Pros______and_cons_____

Llewellyn, D. (1995). Consumer Protection in Retail Investment Services: Protection against what? *Journal of Financial Regulation and Compliance*, 3(1), 43-53.

Llewellyn, D. (2005). Trust and confidence in financial services: a strategic challenge. *Journal of Financial Regulation and Compliance*, 13(4), 333-346.

Llewellyn, D. (2014). Reforming the Culture of Banking: Restoring Trust and Confidence in Banking. *Journal of Financial Management, Markets and Institutions,* 2(2), 221-235.

Llewellyn, D., and Financial Services Authority. (1999). *The economic rationale for financial regulation* (Occasional Paper no. 1). Retrieved from: <u>http://www.fsa.gov.uk/pubs/occpapers/op01.pdf</u>

Llewellyn, D. T. (1998). Principles of effective regulation and supervision of banks. *Journal of Financial Regulation and Compliance*, 6(4), 312-319.

Lloyds Banking Group. (n.d.). Our Values. Retrieved from:

http://www.lloydsbankingGroup.com/careers/life-with-us/ethics--values/

Lodge, O. (2012). A Review of the UK Banking Industry. Centre for Policy Studies. Retrieved from: <u>https://www.cps.org.uk/files/reports/original/111220143147-</u> <u>TheUKBankingIndustryJuly2003OliverLodgeandSeanWilliams.pdf</u>

Logan, A. (2001). The United Kingdom's small banks' crisis of the early 1990s: What were the leading indicators of failure? *Bank of England Working Paper no. 139.* Retrieved from: <u>https://www.bankofengland.co.uk/-/media/boe/files/working-paper/2001/the-uks-small-banks-crisis-of-the-early-</u> <u>1990s.pdf?la=en&hash=0EBB6BF99F442D91E41662494702F8F4281BF3C5</u>

Lucca, D., Seru, A., and Trebbi, F. (2014). The Revolving Door and Worker Flows in Banking Regulation. *National Bureau of Economic Research Working Paper 20241.* Retrieved from: https://www.nber.org/papers/w20241.pdf

Lythe, R. (2014). Buyers who took out 125% mortgages with bailed-out Northern Rock now face crippling rates of 13%. Retrieved from: https://www.thisismoney.co.uk/money/mortgageshome/article-2775515/Northern-Rock-home-loans-crippling-rates-13.html

McGregor, S.L.T., and Murnane, J.A. (2010). Paradigm, methodology and method: Intellectual integrity in consumer scholarship. *International Journal of Consumer Studies*, 34(4), 419-427.

McKinney, J.A., Emerson, T.L., and Neubert, M.J. (2010). The Effects of Ethical Codes on Ethical Perceptions of Actions Toward Stakeholders. *Journal of Business Ethics*, 97(4), 505-516.

McLoughlin, V., Stern, S., and FCA (2017). *Ageing population and financial services.* FCA Occasional Paper No.31. Retrieved from:

https://www.fca.org.uk/publication/occasional-papers/occasional-paper-31.pdf

Measuring Worth. (2018). *What Was the U.K. GDP Then? Historic Series*. Retrieved from: <u>https://www.measuringworth.com/datasets/ukgdpir/result.php</u>

Menard, S. (2002). *Applied Logistic Regression Analysis* (2nd ed.). London: SAGE Publications.

Miles, S. (2012). Stakeholder: Essentially Contested or Just Contested? *Journal of Business Ethics*, 108(3), 285-298.

Miller, L. (2013). The human face of regulation. *Professional Adviser*, 8-9.

Milne, A. and Wood, G., 2009. Shattered on the Rock? British financial stability from 1866 to 2007. *Journal of Banking Regulation*, 10(2), 89-127.

Mishkin, F.S. and Eakins, S. G. (2012). *Financial Markets and Institutions* (7th ed.) Harlow: Pearson Education Limited.

Moin, S.M., Devlin, J.F., and Mckechnie, S. (2017). Trust in financial services: the influence of demographics and dispositional characteristics. *Journal of Financial Services Marketing*, *22*, 64-76.

Monaghan, A. (2012, April 24). FSA chief Hector Sants blames financial sector for global crisis. *The Telegraph*. Retrieved from:

https://www.telegraph.co.uk/finance/financialcrisis/9224451/FSA-chief-Hector-Sants-blames-financial-sector-for-global-crisis.html

Moors, G. (2008). Exploring the effect of a middle response category on response style in attitude measurement. *Quality and Quantity*, 42(6), 779-794.

Moosa, I. A. (2015). *Good Regulation, Bad Regulation: The Anatomy of Financial Regulation.* London: Palgrave Macmillan.

Claire Lynne McCafferty

Moosa, I. A., and Ramiah, V. (2014). *The Costs and Benefits of Environmental Regulation*. Cheltenham: Edward Elgar Publishing Limited.

Mullineux, A. (2000). Re-Regulating banks: The unfinished agenda. *Journal of Financial Regulation and Compliance*, 8(1), 9-15.

Murray, J. (2013). Likert Data: What to use, Parametric or Non-Parametric? *International Journal of Business and Social Science*, 4(11), 258-264.

National Records of Scotland. (2017). *Mid-2016 population estimates Scotland*. Retrieved from: <u>https://www.nrscotland.gov.uk/statistics-and-</u> <u>data/statistics/statistics-by-theme/population/population-estimates/mid-year-</u> <u>population-estimates/mid-2016</u>

Norman, G. (2010). Likert Scales, Levels of Measurement, and the "Laws" of Statistics. *Advances in Health Sciences Education: Theory and Practice*, 15(5), 625-632.

Norman, T. (2015). *Wheatley defends FCA's revolving door into private sector*. Retrieved from: <u>https://www.moneymarketing.co.uk/wheatley-defends-fcas-</u><u>revolving-door-into-private-sector/</u>

Northern Rock plc. (2006). *Northern Rock plc. Annual Report and Accounts.* Retrieved from: <u>http://www.n-ram.co.uk/~/media/Files/N/NRAM-</u> <u>PLC/documents/corporate-reports/res2006pr-annualreportandaccounts.pdf</u>

Nunnally, J.C. and Bernstein, I.H. (1994). The Assessment of Reliability. *Psychometric Theory*, 3(1), 248-292.

Office for National Statistics. (2016). *Annual Survey of Hours and Earnings: 2016 provisional results*. Statistical Bulletin. Retrieved from: <u>https://www.ons.gov.uk/employmentandlabourmarket/peopleinwork/earningsandwo</u> <u>rkinghours/bulletins/annualsurveyofhoursandearnings/2016provisionalresults</u>

Office of Fair Trading. (2006). *Payment protection insurance.* Report on the market study and proposed decision to make a market investigation reference. *OFT869.* Retrieved from: <u>https://londoneconomics.co.uk/wp-</u>

content/uploads/2011/09/70-Research-into-Payment-Protection-Insurance-in-the-UK.pdf

Office of the Secretary of State for Scotland. (2014). *Scotland Analysis.* <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attac</u> <u>hment_data/file/301160/Financial_services_and_banking_summary.pdf</u>

O'Loughlin, D. (2013, February 21). HSBC latest bank to scrap commission sales incentives. *The Financial Times.* Retrieved from:

https://www.ftadviser.com/2013/02/21/ifa-industry/companies-and-people/hsbclatest-bank-to-scrap-commission-sales-incentives-WL6VCH2gn3NUlgBo7gwSbP/article.html

Ormerod, P. (2010, September 16). A network is as good as a nudge for a Big Society. *The Financial Times.* Retrieved from: https://www.ft.com/content/3143e5f0-c0fe-11df-99c4-00144feab49a

Osborne, H. (2011, November 17). Northern Rock sale: The deal: Not the ideal outcome for the Treasury, but was anyone ever likely to pay more? *The Guardian. Retrieved from:* <u>https://www.theguardian.com/politics/reality-check-with-polly-curtis/2011/nov/17/northern-rock-sale-good-deal-for-taxpayers</u>

Oxfam Scotland Research Report. (2017). *Building a More Equal Scotland Designing Scotland's Poverty and Inequality Commission*. Retrieved from: <u>https://oxfamilibrary.openrepository.com/bitstream/handle/10546/620264/rr-building-more-equal-scotland-270417-</u>

en.pdf;jsessionid=951C4D0585A510123F921F41C7A52D47?sequence=1

Oxford Dictionary of Finance and Banking. (2014). (5th ed.). Oxford: Oxford University Press.

Painter-Morland, M. (2010). Questioning corporate codes of ethics. *Business Ethics: A European Review*, 19(3), 265-279.

Pallant, J. (2016). *SPSS Survival Manual*. (6th ed.). Berkshire: Mcraw Hill Education.

Pandit, V. and Khairullah, Z.Y. (2015). Stepwise Regression Choosing the Proper Level of Significance. Proceedings of the 1985 Academy of Marketing Science Annual Conference. Developments in Marketing Science: Proceedings of the Academy of Marketing Science. London: Springer

Pigou, A. C. (1932). The Economics of Welfare. London: Macmillan and Co.

Poli, R., and Seibt, J. (2010). *Theory and Applications of Ontology: Philosophical Perspectives*. London: Springer.

Posner, R. (1974). Theories of Economic Regulation. *Bell Journal of Economics*, 5(2), 335-358.

RBS. (2019). *Our Values.* Retrieved from: <u>https://www.rbs.com/rbs/about/our-values.html</u>

Richardson, H.W. (1969). The Economic Significance of the Depression in Britain. *Journal of Contemporary History*, 4(4), 3-19.

Ring, P. (2004). A Critical Analysis of Depolorisation. *Journal of Financial Regulation and Compliance*, 12(3), 248-262.

Ring, P. (2012). Trust: a challenge for private pension policy. *Journal of Comparative Social Welfare*, 28(2), 119-128.

Ring, P. (2016). The Retail Distribution Review: retail financial services; regulation; financial advice market review. Journal of Financial Regulation and Compliance, 24(2), 140-153.

Roberts, R. and Johnson, P.M. (2015). Understanding the quality of data: a concept map for `the thinking behind the doing' in scientific practice. *The Curriculum Journal*, 26 (3). 345-369.

Rovnick, N. (2016, September 22). UK banks raise overdraft fees whilst cutting savings rates. *The Financial Times.* Retrieved from: https://www.ft.com/content/d6ee7536-80b2-11e6-8e50-8ec15fb462f4

Ruel, E., Wagner, W., & Gillespie, B. (2016). *The practice of survey research*. Thousand Oaks, London: SAGE Publications Ltd.

Russell, J. (2012, February 6). Financial services companies must be proactive on compliance, says FSA chief. *The Telegraph*. Retrieved from: https://www.telegraph.co.uk/finance/newsbysector/banksandfinance/9065077/Financel-says-FSA-chief.html

Sassoon, J. (2009). *The tripartite review. Preliminary Report.* Retrieved from: <u>https://www.conservatives.com/~/media/.../tripartite-review-preliminary-report.ashx</u>

Schenk, C. R. (2014). Summer in the City: Banking Scandals of 1974 and the Development of International Banking Supervision. *English Historical Review*, 129(540), 1129-1156.

Schwarz, S. (2013). Ring fencing. *Southern California Law Review*, 87(69), 69-110. Retrieved from:

https://scholarship.law.duke.edu/cgi/viewcontent.cgi?referer=&httpsredir=1&article =5531&context=faculty_scholarship

Scotland, J. (2012). Exploring the Philosophical Underpinnings of Research: Relating to Ontology, Epistemology to the Methodology and Methods of the Scientific, Interpretive, and Critical Research Paradigms. *English Language Teaching*, 5(9), 9-16.

Scotland's Census 2011. (2018). *Education*. Retrieved from: <u>https://www.scotlandscensus.gov.uk/education-0</u>

Claire Lynne McCafferty

Seaman, J., Allen, I. E., and Zetumer, S. (2018). Staying Relevant The central limit theorem remains a key concept in probability theory. *Quality Progress, Statistics Highlight.* Retrieved from: <u>http://asq.org/quality-</u> <u>progress/2018/02/statistics-spotlight/staying-relevant.html</u>

Sheppard, J. (1994). The Corporate Moral Person: The Organization's Personality and Its Board. *Journal of Business & Society*, 7 (2), 151-164.

Shive, S. A., and Forster, M. M. (2016). The Revolving Door of the Financial Regulators. Review of Finance, 21(4), 1445-1484.

Shleifer, A. (2000). *Inefficient Markets. An Introduction to Behavioural Finance* Oxford: Oxford University Press.

Skinner, C. P. (2016). Misconduct Risk. Fordham Law Review, 84(4), 1559-1610.

Spicer, A., Gond, J. P., Patel, K., Lindley, D., Fleming, P., Mosonvi, S., Benoit, C., and Parker, S. (2014). *A report on the culture of British retail banking.* Retrieved from: <u>http://newcityagenda.co.uk/wp-content/uploads/2014/11/Online-version.pdf</u>

Stanley, T. O., Ford, J. K., and Richards, S. K. (1985). Segmentation of Bank Customers by Age. *International Journal of Bank Marketing*. 3(3), 56-63.

Starks, M., Reynolds, G., Gee, C., Burnik, G., and Vass, L., and Financial Conduct Authority. (2018). Price discrimination in financial services. How should we deal with questions of fairness? *(FCA Research Note)*. Retrieved from:

https://www.fca.org.uk/publication/research/price_discrimination_in_financial_servi ces.pdf

Statista: The Statistics Portal. (2019). *Gross domestic product (GDP) year-on-year growth in the United Kingdom (UK) from 2000 to 2018. Retrieved from:* <u>https://www.statista.com/statistics/281734/gdp-growth-in-the-united-kingdom-uk/</u>

Statista: The Statistics Portal. (2019). *Real GDP growth of the United States from 1990 to 2018*. Retrieved from: <u>https://www.statista.com/statistics/188165/annual-</u> <u>gdp-growth-of-the-united-states-since-1990/</u>

Statista: The Statistics Portal. (2019). *Unemployment rate in the United Kingdom (UK) from 2000 to 2018.* Retrieved from:

https://www.statista.com/statistics/279898/unemployment-rate-in-the-unitedkingdom-uk/

Statista: The Statistics Portal. (2018). *United States: Unemployment rate from 2007 to 2017.* Retrieved from:

https://www.statista.com/statistics/263710/unemployment-rate-in-the-united-states/

Stigler, G. J. (1971). The Theory of Economic Regulation. *The Bell Journal of Economics and Management Science*, 2(1), 3-21.

Stiglitz, J. E. (2010). Lessons from the Global Financial Crisis of 2008. *Seoul Journal of Economics*, 23(3), 321-339.

Stonham, P. (1996). Whatever happened at Barings? Part one: The lure of derivatives and collapse. *European Management Journal*, 14(2), 167-175.

Sullivan, G. M., and Artino, A. R. (2013). *Journal of Graduate Medical Education*, 5(4), 541-542.

Szolnoki, G., and Hoffman, D. (2013). Online, face-to-face and telephone surveys—Comparing different sampling methods in wine consumer research. *Wine Economics and Policy*, 2(2), 57-66.

Tabachnick, B. G., and Fidell, L. S. (2014). *Using Multivariate Statistics: Pearson New International Edition*. Essex: Pearson Education Limited.

Taylor, A. (2017). High Court reaffirms FOS discretion to depart from the law when "fair and reasonable". *Squire Patton Boggs UK Financial Services & Regulatory Investigations Group*. Retrieved from: <u>https://www.finance-</u> disputes.co.uk/2017/03/high-court-reaffirms-fos-discretion-to-depart-from-the-law-

when-fair-and-reasonable/

Taylor, M. (1995). "Twin Peaks": A regulatory structure for the new century. *Centre for the Study of Financial Innovation*. Retrieved from:

https://static1.squarespace.com/static/54d620fce4b049bf4cd5be9b/t/55241159e4b 0c8f3afe1d11e/1428427097907/Twin+Peaks+A+regulatory+structure+for+the+new +century.pdf

Taylor, M. W. (2009). 'Twin Peaks' Revisited...a second chance for regulatory reform. *Centre for the Study of Financial Innovation*. Retrieved from: <u>http://static1.squarespace.com/static/54d620fce4b049bf4cd5be9b/t/55241044e4b0</u> 3769e017208a/1428426820095/Twin+Peaks+Revisited.pdf

Tejada, J. J., & Punzalan, J. R. B. (2012). On the Misuse of Slovin 's Formula. *The Philippine Statistician*, *61*(1), 129-136.

Thaler, R. and Sunstein, C. (2009). *Nudge. Improving decisions about health, wealth and happiness* London: Penguin Books.

The United States Department of Justice. (2012). *HSBC Holdings Plc. And HSBC* \$1.256 Billion in Deferred Prosecution Agreement. Retrieved from: <u>https://www.justice.gov/opa/pr/hsbc-holdings-plc-and-hsbc-bank-usa-na-admit-anti-money-laundering-and-sanctions-violations</u>

The United States Department of Justice. (2016). *HSBC Holdings Plc Agrees to Pay More Than \$100 Million to Resolve Fraud Charges*. Retrieved from: <u>https://www.justice.gov/opa/pr/hsbc-holdings-plc-agrees-pay-more-100-million-resolve-fraud-charges</u>

Thompson, B. (1995). Stepwise Regression and Stepwise Discriminant Analysis Need Not Apply here: A Guidelines Editorial. *Educational and Psychological Measurement*, 55(4), 525-534.

Treanor, J. (2016, April 11). Misconduct 'has cost UK's banks £53bn over 15 years'. *The Guardian*. Retrieved from:

https://www.theguardian.com/money/2016/apr/11/misconduct-has-cost-uks-banks-53bn-over-15-years-ppi

Trochim, W. (2000). *The Research Methods Knowledge Base*. (2nd ed.) Cincinnati: Atomic Dog Publishing.

Vassalos, Y. (2018). *Financial Regulators and the private sector: permanent revolving door at DG FISMA.* Corporate Europe Observatory. Retrieved from: <u>https://corporateeurope.org/en/power-lobbies-revolving-doors/2018/04/financial-regulators-and-private-sector-permanent-revolving</u>

van Teijlingen, E., and Hundley, V. (2001). The importance of pilot studies. *Social Research Update 35.* University of Surrey. Retrieved from: <u>http://sru.soc.surrey.ac.uk/SRU35.html</u>

Velardo, V. (2016). Recursive Ontology: A Systemic Theory of Reality. *Axiomathes*, 26(1), 89-114.

Vigeo Eiris. (2017). Responsibility and ethical culture in banking and finance. *Sustainability Focus*. Retrieved from: <u>http://www.vigeo-eiris.com/wp-</u> <u>content/uploads/2017/09/VigeoEiris_SustainabilitFocus_Responsible-Business-</u> <u>Conduct_Banking-1.pdf?x62552</u>

Virgin Money. (2012). Virgin Money agrees to buy £465m high quality mortgage book from UKAR. Retrieved from: <u>https://uk.virginmoney.com/virgin/news-centre/press-</u> releases/2012/virgin-money-to-buy-mortgage-book-from-ukar.jsp

Wainwright, S.P., (1997). A new paradigm for nursing. *Journal of Advanced Nursing*, 26(6), 1262-1271.

PhD April 2020

Walker, G.A. (2013). Reference Rate Regulation and LIBOR. *Banking and Finance Law Review*, 28(3), 351-388.

Wallace, T. (2014, October 21). Banking Standards Review Council: Ethics panel for banks on the brink. *City A.M.* Retrieved from:

http://www.cityam.com/1413857530/ethics-panel-banks-brink

Walliman, N. (2015). Social Research Methods: The Essentials. (2nd ed.). London: SAGE Publications Ltd.

Watt, G. (2013). *How behavioural economics can keep on the right side of the FCA*. Retrieved from: <u>https://www.moneymarketing.co.uk/how-behavioural-</u>economics-can-keep-on-the-right-side-of-the-fca/

Webley, S. and Werner, A. (2008). Corporate codes of ethics: necessary but not sufficient. *Business Ethics: A European Review*, 17(4), 405-415.

Wells, J. (2011). Another fine mess. *Financial Adviser*. Retrieved from <u>https://www.proquest.com/</u>

Werhane, P.H. (1985). *Persons, Rights, and Corporations*. New Jersey: Prentice-Hall.

Which? (2019). *Best and worst banks*. Retrieved from: <u>https://www.which.co.uk/money/banking/bank-accounts/best-and-worst-banks-a3q5d8c6di7y</u>

Withers, I. (2017, December 13). South Africa whistleblower aids HSBC money laundering investigation. *The Telegraph*. Retrieved from: https://www.telegraph.co.uk/business/2017/12/13/south-africa-whistleblower-aids-

hsbc-money-laundering-investigation/

Wood, G. (2001). Too much regulation? *Journal of Financial Regulation and Compliance*, 9(4), 350-358.

Wu, J., and Lee, V. (2001). Protection for Banking Consumers in the United Kingdom and the United States of America: Fees and Charges. *Research and Library Services Division Legislative Council Secretariat.* Retrieved from: <u>https://www.legco.gov.hk/yr01-02/english/sec/library/0102rp01e.pdf</u>

YouGov. (2018). *Ten years after the financial crisis – two thirds of British People don't trust banks.* Retrieved from: <u>https://yougov.co.uk/topics/finance/articles-reports/2018/08/29/ten-years-after-financial-crisis-two-thirds-britis</u>

APPENDIX A: EXAMPLE CONSENT FORM AND EMAIL COMMUNICATION

(i) Example Consent Form

Edinburgh Napier University requires that all individuals who participate in research studies give their formal consent to do so. Please read the following and sign below to confirm you understand and agree to participate in the study.

1. I freely and voluntarily consent to be a participant in the pilot study of the research project on the topic of 'Consumer confidence in Retail banking regulation' conducted by Claire McCafferty, a PhD student at Edinburgh Napier University.

The main aim of this research is to measure consumer confidence in the different areas that regulation seeks to address.

3. I understand that no data will be collected at this stage. The purpose of this group session is to test the validity of the consumer survey. Unless agreed*, my organisation will not be linked with the research materials, and I will not be identified or identifiable in any report subsequently produced by the researcher.

4. I also understand that if, at any time, during the pilot group session I feel unable or unwilling to continue, I am free to leave. That is, my participation in the study is completely voluntary, and I may withdraw from it without negative consequences.

Additionally, should I not wish to answer any particular question or questions, I am free to decline.

In addition to the researcher, two more individuals will be present; the transcriber and recorder. Neither of these individuals are taking part in the study.

7. I have read and understood the above and consent to participate in this pilot study. My approval is not a waiver of my legal rights. Furthermore, I understand that I will be able to keep a copy of the informed consent form for my records upon request.

Signed_____ Date_____

*I agree for the name of my organisation to be included so gratitude for participation can be expressed in the thesis.

YES/NO

Signed _____ (Answer must be selected and signed if YES).

(ii) Example email to participate in the research design

Dear John

Thank you for agreeing to take part in my pilot study.

- To ensure clarity and ease of completion I have provided the following instructions:

 - 1. Read attached consent form
 2. Read background to study (see below)
 3. Read consumer survey questions on page 3 of survey (only page 3 is being 'tested')
 4. Answer pilot questions (see below)
 5. Return feedback and completed consent form

Background to the study

Is the retail banking regulator failing the consumer?

Despite being one of the most heavily regulated industries in the world there have been many recent, and indeed historic, cases of consumer detriment at the hands of retail banks in the UK. Given the nature of financial services there are many economic and social rationales for regulation (Llewellyn, 1999, Cartwright, 2004, Goodhart et al, 2013). Conversely, there are economists who believe market solutions are preferable to government intervention (Dowd, 1993, Benston, 1996). Though most academics and industry experts agree that some degree of regulation is required, just what level and type of regulation is appropriate is widely debated. An interesting finding from the literature review was the consensus, from both sides of the debate, that the need for consumer confidence is grounds for regulating the banking industry. Indeed, a statutory objective of one of the retail banking regulators, the Financial Conduct Authority (FCA), is to 'maintain confidence in the UK financial system' under the terms of the Financial Services and Markets Act 2000.

A quantitative approach to data collection is employed in the research. Pilot studies will be undertaken with industry experts, academics and consumers to test the survey questions. A survey of 400 consumers will then be conducted.

A stratified sample of the Scottish adult (18+ years) population will be used.

This study will measure whether UK retail banking regulation is 'fit for purpose' by measuring consumer confidence levels in the achievement of each of the rationales for regulation. Seven key rationales for regulation were identified in the literature: externalities (secure), asymmetric information (transparent), moral hazard (just), paternalism (professional), distributive justice (fair), community values (equitable) and corruption (honest). The researcher could find no other studies which have captured this data.

Key references

Benston, G. and Kaufman, G. (1996) 'The Appropriate Role of Bank Regulation', The Economic Journal, 106(436), pp.688-697 Cartwright, P. (2004) Banks, Consumers and Regulation Oxford; Hart Publishing Dowd, K. (1993) 'Re-Examining the Case for Government Deposit Insurance' Southern Economic Journal, 59(3), pp.363-370 Goodhart, C. Hartmann, P. Llewellyn, D. Rojas-Suarez, L. Weisbrod, S. (2013) Financial Regulation. Why, how and where now? London: Routledge

Pilot guestions

1. Do you believe other people may have difficulty in answering this questionnaire? If so, which question may pose a problem?

Llewellyn, D. (1999) 'The Economic Rationale for Financial Regulation' Occasional Paper No 1, Financial Services Authority.

- 2. Given the purpose of my study outlined in the introduction section, would you suggest a question, may perhaps be added or removed to enhance the study?
- 3. How long did it take you to answer the questionnaire?
- 4. Did you read the instructions?
- 5. Were the instructions clear?
- 6. Did you object to answering any of the questions?
- 7. Was the layout of the questionnaire clear and logical?
- 8. Is there any question that the wording could be enhanced to clarify?

I would be grateful if you could complete and return the feedback and consent form by Sunday 25 June.

Again, thank you for agreeing to take part and I would reiterate that no part(s) of the study should be shared with others.

Kind regards
APPENDIX B: EXAMPLE SURVEY

Claire McCafferty



Consent

Edinburgh Napier University requires that all individuals who participate in research studies give their formal consent to do so. Please read the following and tick the box to confirm you understand and agree to participate in the study.

1. I freely consent to be a participant in the research project on the topic of 'Consumer confidence in Retail banking regulation' conducted by Claire McCafferty, a PhD student at Edinburgh Napier University.

2. The main aim of this research is to measure consumer confidence in the different areas that regulation seeks to address and should take no longer than 20 minutes to complete.

3. I understand that my answers are anonymous and confidential. My name will not be linked with the research materials, and I will not be identified or identifiable in anything published by the researcher.

4. I also understand that if, at any time, during the survey I feel unable or unwilling to continue, I am free to leave. That is, my participation in the study is completely voluntary, and I may withdraw from it without negative consequences. However, after data has been collected it will not be possible for my data to be removed.

5. Additionally, should I not wish to answer any particular question or questions, I am free to decline. A partial completion will indicate a wish to withdraw. Partial completions will not be included in the research and will be void.

6. I have read and understood the above and consent to participate in this study. My approval is not a waiver of my legal rights. Furthermore, I understand that I will be able to acquire a copy of the informed consent form for my records upon request.

Please tick this box to confirm that you have read the consent information above $_1$, and agree to participate in this study.

□ I agree

Introduction

My name is Claire McCafferty and I am a PhD student at Edinburgh Napier University.

As part of my research I am collecting data on consumer confidence in retail banking regulation.

For the purposes of this study, **retail banking** is defined as personal or individual accounts held at a high street or online bank. The type of accounts this includes are: current accounts (with or without overdrafts), savings accounts, personal loans, credit cards and mortgages.

It does **not** include investments, pensions or other specialised products.

In total there are **three** pages of questions in this survey. The first page has a total of **eight** demographic questions. The second and third page have a combined total of **ten** questions which ask you to rate to what extent you agree with a particular statement. The scale runs from strongly agree to strongly disagree.

Please select the option that most reflects your individual view. Whether you agree or disagree with a statement, and how strongly you feel about this, will depend on your personal experience and therefore there are no right or wrong answers.

If you would like to be informed of the findings of this research, please contact me on the details on the last page of this survey.

Please note this survey is only open to participants living in Scotland.

Thank you for participating!

Demographic questions

| 1 | 1 Gender | | | | | | | | | | | | | | | |
|---|-------------|---|----------------|-----------------|----------|----------|----------|----------|----------|----------|----------|----------|----------|-------|----------|----|
| | 0 | Male | 0 | Fema | le C | Pre | efer n | ot to | say | ° 0 | ther | | | | | |
| | | | | | | | | | | | | | | | | |
| 2 | Age | range | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ | _ |
| | 0 18- | 0 20- | O 25- | 0 30- | O 35- | O 40- | O 45- | O 50- | O 55- | O 60- | O 65- | O 70- | O 75- | 0 | 0 85- | 0 |
| | 10 19 | 24 | 29 | 34 | 39 | 44 | 49 | 50 54 | 59 | 64 | 69 | 74 | 79 | 84 | 89 | + |
| 3 | Do γ | | rront | ly ha | 70 2 L | IK hai | | count | 2 | | | | | | | |
| | | | | ly llav | /e a u | IN Dal | IK au | count | | | | | | | | |
| | | Yes | ~ N | 0 | | | | | | | | | | | | |
| 4 | Do y | ou cu | rrent | ly live | e in S | cotlar | nd? | | | | | | | | | |
| | **Th | This survey is only open to participants living in Scotland** | | | | | | | | | | | | | | |
| | | Yes | | | | | | | | | | | | | | |
| 5 | High | est ai | Jalifio | cation | ı held | | | | | | | | | | | |
| | 0 | Secon | darv | scho | | പ | اممو | ο 1 | Inde | oradi | ateu | 0 | Posta | radu | oto | |
| | · | Jecon | uury | Senot | 51 | 001 | lege | , | Jiidei | grau | uate | | 1 0315 | lauu | itt | |
| 6 | Emp | loyme | ent st | atus | | | | | | | | | | | | |
| • | 0 | Work | ing fu | ıll- | 0 | Wor | king p | oart- | C |) | | 0 | Not c | urrer | itly | |
| | time | | | | tim | e | | | St | tuden | t | wor | king | | | |
| 7 | Inco | me ra | nge | | | | | | | | | | | | | |
| · | 0 | | 0 | 5 _{£1} | 3,501 | [- | 0 | £27,6 | 01- | 0 | £43 | 3,001 | - | 0 | | |
| | <£13 | 3,500 | £ | 27,60 | 00 | | £43, | ,000 | | £1 | 150,0 | 00 | | £15 | 0,001 | l+ |
| | Do v | Do you currently work in the financial services sector? | | | | | | | | | | | | | | |
| 8 | 0, | Vac | о _м | | | | | | | | | | | | | |
| · | | 162 | IN | U | | | | | | | | | | | | |

Core questions

I have confidence the UK retail banking system is stable and secure

| 1. C Strongly | 0 | Slightly | Slightly | 0 | Strongly |
|---------------|-------|----------|----------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

I have confidence that the money in my retail bank account(s) is protected should 2. my bank fail

| Strongly | 0 | Slightly | C Slightly | 0 | Strongly |
|----------|-------|----------|------------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

I have confidence that my retail bank will provide me with sufficient information to 3. make informed decisions

| Strongly | 0 | Slightly | Slightly | 0 | • Strongly |
|----------|-------|----------|----------|----------|------------|
| agree | Agree | agree | disagree | Disagree | disagree |

4. I have confidence that my retail bank will do the right thing

| Strongly | 0 | Slightly | Slightly | 0 | Strongly |
|----------|-------|----------|----------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

5. I have confidence that I am protected from retail bank misconduct

| Strongly | 0 | C Slightly | Slightly | 0 | Strongly |
|----------|-------|------------|----------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

6. I have confidence that I can trust the expertise of my retail bank

| Strongly | 0 | Slightly | Slightly | 0 | C Strongly |
|----------|-------|----------|----------|----------|------------|
| agree | Agree | agree | disagree | Disagree | disagree |

7. I have confidence that the price of retail banking products is fair

| Strongly | 0 | Slightly | Slightly | 0 | • Strongly |
|----------|-------|----------|----------|----------|------------|
| agree | Agree | agree | disagree | Disagree | disagree |

I have confidence that I will not be unlawfully discriminated against by my retail 8. bank

| • Strongly | 0 | C Slightly | Slightly | 0 | Strongly |
|------------|-------|------------|----------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

In general, I have confidence that as a retail bank customer I am adequately 9. protected

| Strongly | 0 | Slightly | Slightly | 0 | Strongly |
|----------|-------|----------|----------|----------|----------|
| agree | Agree | agree | disagree | Disagree | disagree |

I have confidence that adequate protection is provided by:

□ The retail banking regulator

10. \Box Consumer Protection laws

- \square Both of the above
- □ Neither of the above
- Don't know

Thank you

Thank you for taking the time to complete this survey.

If you wish to contact me you can do so at the following addresses:

email: c.mccafferty@napier.ac.uk

post: Room 3/45, Edinburgh Napier University Business School, Craiglockhart Campus, 219 Colinton Road, Edinburgh, EH14 1DJ

APPENDIX C: SHAPIRO-WILK TEST OF NORMALITY

(i) Shapiro-Wilk: Age

| Age Group | | Shapiro-Wilk | | lk |
|--|-------|--------------|-----|------|
| | | Statistic | df | Sig. |
| H1: I have confidence the UK retail banking system is stable and secure | 18-29 | .884 | 71 | .000 |
| | 30-44 | .893 | 126 | .000 |
| | 45-59 | .874 | 149 | .000 |
| | 60+ | .896 | 86 | .000 |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail | 18-29 | .863 | 71 | .000 |
| | 30-44 | .872 | 126 | .000 |
| | 45-59 | .866 | 149 | .000 |
| | 60+ | .814 | 86 | .000 |
| H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions | 18-29 | .811 | 71 | .000 |
| | 30-44 | .895 | 126 | .000 |
| | 45-59 | .897 | 149 | .000 |
| | 60+ | .809 | 86 | .000 |
| H4: I have confidence that my retail bank will do the right thing | 18-29 | .831 | 71 | .000 |
| | 30-44 | .925 | 126 | .000 |
| | 45-59 | .883 | 149 | .000 |
| | 60+ | .848 | 86 | .000 |
| H5: I have confidence that I am protected from retail bank misconduct | 18-29 | .866 | 71 | .000 |

| | 30-44 | .893 | 126 | .000 |
|---|-------|------|-----|------|
| | 45-59 | .896 | 149 | .000 |
| | 60+ | .860 | 86 | .000 |
| H6: I have confidence that I can trust the expertise of my retail bank | 18-29 | .780 | 71 | .000 |
| | 30-44 | .916 | 126 | .000 |
| | 45-59 | .894 | 149 | .000 |
| | 60+ | .905 | 86 | .000 |
| H7: I have confidence that the price of retail banking products is fair | 18-29 | .876 | 71 | .000 |
| | 30-44 | .929 | 126 | .000 |
| | 45-59 | .915 | 149 | .000 |
| | 60+ | .908 | 86 | .000 |
| H8: I have confidence that I will not be unlawfully discriminated against by my retail bank | 18-29 | .853 | 71 | .000 |
| | 30-44 | .871 | 126 | .000 |
| | 45-59 | .865 | 149 | .000 |
| | 60+ | .784 | 86 | .000 |
| H9: In general, I have confidence that as a retail bank customer I am adequately protected | 18-29 | .859 | 71 | .000 |
| | 30-44 | .824 | 126 | .000 |
| | 45-59 | .865 | 149 | .000 |
| | 60+ | .839 | 86 | .000 |

(ii) Shapiro-Wilk: Education

| Highest Qualification H | Sł | apiro-Wilk | | | |
|--|------------------|------------|-----|------|--|
| | • | Statistic | df | Sig. | |
| | Secondary School | .825 | 98 | .000 | |
| H1: I have confidence the UK retail banking system is stable and secure | College | .902 | 114 | .000 | |
| | Undergraduate | .907 | 108 | .000 | |
| | Postgraduate | .906 | 112 | .000 | |
| | Secondary School | .761 | 98 | .000 | |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail | College | .879 | 114 | .000 | |
| | Undergraduate | .893 | 108 | .000 | |
| | Postgraduate | .852 | 112 | .000 | |
| H3: I have confidence that my retail bank | Secondary School | .773 | 98 | .000 | |
| will provide me with sufficient information to make informed decisions | College | .882 | 114 | .000 | |
| | Undergraduate | .877 | 108 | .000 | |
| | Postgraduate | .904 | 112 | .000 | |
| | Secondary School | .784 | 98 | .000 | |
| H4: I have confidence that my retail bank will do the right thing | College | .910 | 114 | .000 | |
| | Undergraduate | .898 | 108 | .000 | |
| | Postgraduate | .913 | 112 | .000 | |

| | Secondary School | .775 | 98 | .000 |
|---|------------------|------|-----|------|
| H5: I have confidence that I am protected from retail bank misconduct | College | .899 | 114 | .000 |
| | Undergraduate | .911 | 108 | .000 |
| | Postgraduate | .873 | 112 | .000 |
| | Secondary School | .823 | 98 | .000 |
| Ho: I have confidence that I can trust the expertise of my retail bank | College | .890 | 114 | .000 |
| | Undergraduate | .902 | 108 | .000 |
| | Postgraduate | .928 | 112 | .000 |
| | Secondary School | .870 | 98 | .000 |
| H7: I have confidence that the price of retail banking products is fair | College | .920 | 114 | .000 |
| | Undergraduate | .934 | 108 | .000 |
| | Postgraduate | .930 | 112 | .000 |
| | Secondary School | .794 | 98 | .000 |
| H8: I have confidence that I will not be unlawfully discriminated against by my retail bank | College | .835 | 114 | .000 |
| | Undergraduate | .875 | 108 | .000 |
| | Postgraduate | .876 | 112 | .000 |
| | Secondary School | .730 | 98 | .000 |
| H9: In general, I have confidence that as a retail bank customer I am adequately protected | College | .878 | 114 | .000 |
| | Undergraduate | .866 | 108 | .000 |
| | Postgraduate | .883 | 112 | .000 |

(iii) Shapiro-Wilk: Income

| Income range | | Sh | apiro-Wi | lk |
|--|------------------|-----------|----------|------|
| | | Statistic | df | Sig. |
| | <£13,500 | .820 | 128 | .000 |
| H1: I have confidence the UK retail banking system is stable and secure | £13,501-£27,600 | .900 | 122 | .000 |
| | £27,601-£43,000 | .917 | 114 | .000 |
| | £43,001-£150,000 | .925 | 62 | .001 |
| | £150,001+ | .640 | 6 | .001 |
| | <£13,500 | .822 | 128 | .000 |
| H2:I have confidence that the money in my retail bank account(s) is protected | £13,501-£27,600 | .906 | 122 | .000 |
| should my bank fail | £27,601-£43,000 | .838 | 114 | .000 |
| | £43,001-£150,000 | .852 | 62 | .000 |
| | £150,001+ | .496 | 6 | .000 |
| | <£13,500 | .809 | 128 | .000 |
| H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions | £13,501-£27,600 | .862 | 122 | .000 |
| | £27,601-£43,000 | .897 | 114 | .000 |
| | £43,001-£150,000 | .917 | 62 | .000 |
| | £150,001+ | .907 | 6 | .415 |
| | <£13,500 | .848 | 128 | .000 |

| H4: I have confidence that my retail bank | £13,501-£27,600 | .854 | 122 | .000 |
|--|------------------|------|-----|------|
| will do the right thing | £27,601-£43,000 | .907 | 114 | .000 |
| | £43,001-£150,000 | .937 | 62 | .003 |
| | £150,001+ | .908 | 6 | .421 |
| | <£13,500 | .863 | 128 | .000 |
| H5: I have confidence that I am protected from retail bank misconduct | £13,501-£27,600 | .859 | 122 | .000 |
| | £27,601-£43,000 | .879 | 114 | .000 |
| | £43,001-£150,000 | .927 | 62 | .001 |
| | £150,001+ | .775 | 6 | .035 |
| | <£13,500 | .843 | 128 | .000 |
| H6: I have confidence that I can trust the expertise of my retail bank | £13,501-£27,600 | .874 | 122 | .000 |
| | £27,601-£43,000 | .901 | 114 | .000 |
| | £43,001-£150,000 | .921 | 62 | .001 |
| | £150,001+ | .805 | 6 | .065 |
| | <£13,500 | .888 | 128 | .000 |
| H7: I have confidence that the price of retail banking products is fair | £13,501-£27,600 | .921 | 122 | .000 |
| | £27,601-£43,000 | .925 | 114 | .000 |
| | £43,001-£150,000 | .926 | 62 | .001 |
| | £150,001+ | .920 | 6 | .505 |

| | <£13,500 | .783 | 128 | .000 |
|--|------------------|------|-----|------|
| H8: I have confidence that I will not be unlawfully discriminated against by my | £13,501-£27,600 | .863 | 122 | .000 |
| | £27,601-£43,000 | .851 | 114 | .000 |
| | £43,001-£150,000 | .916 | 62 | .000 |
| | £150,001+ | .822 | 6 | .091 |
| | <£13,500 | .817 | 128 | .000 |
| H9: In general, I have confidence that as a retail bank customer I am adequately | £13,501-£27,600 | .866 | 122 | .000 |
| protected | £27,601-£43,000 | .832 | 114 | .000 |
| | £43,001-£150,000 | .907 | 62 | .000 |
| | £150,001+ | .822 | 6 | .091 |

APPENDIX D: DESCRIPTIVE STATISTICS

(i) T-tests: Employed in the financial services sector

| T-Test | Work in FS | Ν | Mean | Std. Deviation | t | df | Sig (2- tailed) |
|---|------------|-----|------|-------------------|-----------|------|--------------------|
| H1: I have confidence the UK retail banking system is stable | Yes | 18 | 4.83 | .857 | -3.446 | 430 | .001 |
| | No | 414 | 3.86 | 1.190 | | | |
| H2 I have confidence that the money in my retail bank | Yes | 18 | 5.17 | .707 | | | |
| account(s) is protected should my bank fail | No | 414 | 4.45 | 1.112 | -4.104 | 430 | .001 |
| H3 I have confidence that my retail bank will provide me with | Yes | 18 | 4.06 | 1.162 | | | |
| sufficient information to make informed decisions | No | 414 | 4.18 | 1.145 | 0.455 | 430 | .649 |
| H4: I have confidence that my | Yes | 18 | 3.83 | 1.339 | 0.457 | 100 | 070 |
| retail bank will do the right thing | No | 414 | 3.88 | 1.211 | 0.157 | 430 | .876 |
| H5: I have confidence that I am | Yes | 18 | 4.06 | 1.162 | - | 10.0 | |
| protected from retail bank misconduct | No | 414 | 4.03 | 1,172 | .0.77 | 430 | .939 |
| H6: I have confidence that I can | Yes | 18 | 3.83 | 1.505 | 054 | 100 | 700 |
| trust the expertise of my retail bank | No | 414 | 3.93 | 1.174 | .354 | 430 | .723 |
| H7: I have confidence that the | Yes | 18 | 3.61 | 1.461 | | | |
| price of retail banking products is fair | No | 414 | 3.57 | 1.244 | 128 | 430 | .898 |
| H8: I have confidence that I will | Yes | 18 | 4.44 | 1.464 | | | |
| not be unlawfully discriminated against by my retail bank | No | 414 | 4.43 | 1.050 | 047 | 430 | .963 |
| H9 In general, I have confidence | Yes | 18 | 4.44 | 1.097 | | | |
| that as a retail bank customer I am adequately protected | No | 414 | 4.28 | 1.045 | 661 | 430 | .509 |

| Mann- Whitney U | Work in FS | Ν | Median | U | Z | Ρ |
|-------------------------------------|------------|-----|--------|--------|-------|------|
| H1: I have confidence the UK | Yes | 18 | 5 | | | |
| retail banking system is stable | | | | 5410.5 | 3.374 | .001 |
| and secure | No | 414 | 4 | | | |
| H2 I have confidence that the | Yes | 18 | 5 | _ / | | |
| money in my retail bank | No | 414 | 5 | 5186.0 | 2.991 | .003 |
| account(s) is protected should my | | | | | | |
| H2 Lbave confidence that my | Voc | 10 | 1 | | | |
| retail bank will provide me with | 165 | 10 | 4 | 3362.5 | - 738 | 461 |
| sufficient information to make | No | 414 | 4 | 0002.0 | .700 | .401 |
| informed decisions | | | | | | |
| H4: I have confidence that my | Yes | 18 | 4 | | | |
| retail bank will do the right thing | No | 414 | 4 | 3581.0 | 290 | .772 |
| H5: I have confidence that I am | Yes | 18 | 4 | | | |
| protected from retail bank | No | 414 | 4 | 3741.5 | .031 | .975 |
| misconduct | | | • | | | |
| H6: I have confidence that I can | Yes | 18 | 4 | | | |
| trust the expertise of my retail | No | 414 | 4 | 3649.5 | 153 | .878 |
| bank | | | | | | |
| H7: I have confidence that the | Yes | 18 | 4 | | | |
| price of retail banking products is | No | 414 | 4 | 3751.0 | .050 | .960 |
| fair | | | | | | |
| H8: I have confidence that I will | Yes | 18 | 5 | | | 400 |
| not be unlawfully discriminated | No | 414 | 5 | 4114.0 | .802 | .422 |
| against by my retail bank | | 10 | | | | |
| Hy in general, I have confidence | Yes | 18 | 5 | 1150 F | 070 | 204 |
| inal as a retail bank customer I | No | 414 | 4.5 | 4152.5 | .876 | .381 |
| an auequalely prolected | | | | | | |

(ii) Mann-Whitney U tests: Employed in the financial services sector

(iii) T-tests: Gender

| T-Test | Gende r | Ν | Mean | Std. Deviation | t | df | Sig (2- tailed) |
|---|------------|-----|------|-------------------|--------|-----|--------------------|
| H1: I have confidence the | Male | 147 | 4.01 | 1.22 | 1 34 | 427 | 180 |
| stable and secure | Female | 282 | 3.85 | 1.17 | 1.04 | 721 | .100 |
| H2 I have confidence that the money in my retail bank | Male | 147 | 4.55 | 1.16 | | | |
| account(s) is protected should my bank fail | Female | 282 | 4.45 | 1.07 | .928 | 427 | .354 |
| H3 I have confidence that my retail bank will provide | Male | 147 | 4.21 | 1.18 | 290 | 407 | 704 |
| information to make informed decisions | Female | 282 | 4.17 | 1.12 | .300 | 421 | .704 |
| H4: I have confidence that my retail bank will do the | Male | 147 | 3.97 | 1.27 | 1.103 | 427 | .271 |
| right thing | Female | 282 | 3.84 | 1.18 | | | |
| H5: I have confidence that I am protected from retail | Male | 147 | 4.10 | 1.25 | .743 | 427 | .458 |
| bank misconduct | Female | 282 | 4.01 | 1.12 | | | |
| H6: I have confidence that I can trust the expertise of my | Male | 147 | 3.93 | 1.29 | -0.035 | 427 | .972 |
| retail bank | Female | 282 | 3.94 | 1.13 | | | |
| H7: I have confidence that the price of retail banking | Male | 147 | 3.66 | 1.36 | 1.002 | 427 | .317 |
| products is fair | Female | 282 | 3.53 | 1.20 | | | |
| H8: I have confidence that I will not be unlawfully | Male | 147 | 4.47 | 1.14 | | | |
| discriminated against by my retail bank | Female | 282 | 4.42 | 1.03 | .436 | 427 | .663 |
| H9 In general, I have | Male | 147 | 4.41 | 1.05 | | | |
| confidence that as a retail bank customer I am adequately protected | Female | 282 | 4.22 | 1.04 | 1.809 | 427 | .071 |

| Mann- Whitney U | Gender | Ν | Median | U | Z | р |
|--|--------|-------|--------|---------|--------|-------|
| H1: I have confidence the UK | Male | 147 | 4 | | | |
| retail banking system is stable and secure | Female | 282 | 4 | 18981.5 | -1.488 | 0.137 |
| H2: I have confidence that | Male | 147 | 5 | | | |
| the money in my retail bank | | | | | | |
| account(s) is protected | Female | 282 | 5 | 19215.5 | -1.318 | 0.185 |
| should my bank fall | Mala | 4 4 7 | | | | |
| H3: I have confidence that | wale | 147 | 5 | | | |
| me with sufficient information | Female | 282 | 4 | 20002 5 | - 626 | 532 |
| to make informed decisions | | | | 20002.0 | .020 | .002 |
| H4: I have confidence that | Male | 147 | 4 | | | |
| my retail bank will do the | | | | 19104.5 | -1.382 | .167 |
| right thing | Female | 282 | 4 | | | |
| H5: I have confidence that I | Male | 147 | 4 | | | |
| am protected from retail bank | | | | 19362.5 | -1.169 | .242 |
| misconduct | Female | 282 | 4 | | | |
| H6: I have confidence that I | Male | 147 | 4 | | | |
| can trust the expertise of my | Famala | 202 | 4 | 20337.5 | 332 | .740 |
| retail bank | Female | 282 | 4 | | | |
| H/: I have confidence that | Male | 147 | 4 | 10100 | | 2005 |
| products is fair | Female | 282 | 4 | 19406 | -1.114 | .205 |
| H8: I have confidence that I | Male | 147 | 5 | | | |
| will not be unlawfully | Wale | 1-17 | Ŭ | 19554 | -1.032 | .302 |
| discriminated against by my | Female | 282 | 5 | | | .002 |
| retail bank | | | | | | |
| H9: In general, I have | Male | 147 | 5 | | | |
| confidence that as a retail | | | | 18467 | -1.976 | .048 |
| bank customer I am | Female | 282 | 5 | | | |
| adequately protected | | | | | | |

(iv) Mann-Whitney U tests: Gender

(v) ANOVA and Tukey HSD test: Age

| ANOVA | Age | df1 | df2 | F | Sig. | Age groups | Mean |
|--|-----------------|-----|-------|-------|-------|------------|---|
| | groups | | | | | | Difference |
| | | | | | | 30-44 | .337 |
| | 18-29 | | | | | 45-59 | .329 |
| | | | | | | 60+ | .236 |
| | | | | | | 18-29 | 337 |
| H1 I have | 30-44 | | | | | 45-59 | 008 |
| confidence the | | | 400 | 4 400 | 0.040 | 60+ | 101 |
| UK retail banking | | 3 | 428 | 1.483 | 0.218 | 18-29 | 329 |
| system is stable | 45-59 | | | | | 30-44 | .008 |
| and secure | | _ | | | | 60+ | 093 |
| | | | | | | 18-29 | 236 |
| | 60+ | | | | | 30-44 | 101 |
| | | | | | | 45-59 | .093 |
| | | | | | | 30-44 | 303 |
| | 18-29 | | | | | 45-59 | 367 |
| | | _ | | | | 60+ | 326 |
| H2: I have | | | | | | 18-29 | .303 |
| confidence that | e money in my | | | | 0.400 | 45-59 | 064 |
| the money in my | | | 400 | 1 000 | | 60+ | 023 |
| retall bank | 3 | 428 | 1.699 | 0.129 | 18-29 | .367 | |
| account(s) is | votected should | | | | | 30-44 | .064 |
| my bank fail | | _ | | | | 60+ | .041 |
| | | | | | | 18-29 | .326 |
| | 60+ | | | | | 30-44 | Mean Difference .337 .329 .236 .337 .008 .101 .329 .008 .101 .329 .008 .101 .329 .008 .0033 .236 .033 .303 .303 .303 .303 .303 .303 .303 .303 .303 .303 .303 .303 .303 .303 .3041 .023 .041 .326 .023 .041 .326 .023 .041 .326 .023 .041 .724* .582* .443 .280 .139 |
| | | | | | | 45-59 | 041 |
| | 10.00 | | | | | 30-44 | .724* |
| | 18-29 | | | | | 45-59 | .582* |
| H3: I have | | | _ | | | | 60+ |
| confidence that | 00.44 | | | 6.702 | .000 | 18-29 | /24* |
| nrovido mo with | 30-44 | | | | | 45-59 | 142 |
| sufficient | | 3 | 128 | | | 60+ | 280 |
| information to | 45 50 | 5 | 420 | | | 18-29 | 582* |
| make informed | 45-59 | | | | | 30-44 | .142 |
| decisions | | _ | | | | 60+ | 139 |
| | <u> </u> | | | | | 18-29 | 443 |
| | 60+ | | | | | 30-44 | .280 |
| | 40.00 | | | | | 45-59 | .139 |
| | 18-29 | | | | | 30-44 | .941* |
| | | | | | | 45-59 | .000 |
| | 00.44 | | | | | 60+ | .418 |
| | 30-44 | | | | | 18-29 | 941^ |
| confidence that | | | | | | 45-59 | 275 |
| confidence that my retail bank will do the right thing | 45.50 | 3 | 428 | 10 5 | 000 | 60+ | 523* |
| | 45-59 | Ŭ | 720 | 10.0 | .000 | 18-29 | 000 |
| | | | | | | 30-44 | .275 |
| | 601 | - | | | | 60+ | 248 |
| | 00+ | | | | | 18-29 | 418 |
| | | | | | | 30-44 | .523 |
| | 19.00 | | | | | 45-59 | .248 |
| | 18-29 | 1 | | | | 30-44 | .504* |

| | | | | | | 45-59 | .496* |
|------------------------|-------|---|-----|--------|------|--------------|----------------------------|
| | | | | | | 60+ | .281 |
| | 30-44 | | | | | 18-29 | -504* |
| H5: I have | | | | | | 45-59 | 008 |
| confidence that I | | | | | | 60+ | 223 |
| am protected | 45-59 | 3 | 428 | 3.721 | 0.12 | 18-29 | 496* |
| from retail bank | | | | | | 30-44 | .008 |
| misconduct | | | | | | 60+ | 215 |
| | 60+ | | | | | 18-29 | 281 |
| | | | | | | 30-44 | .223 |
| | | | | | | 45-59 | .215 |
| | 18-29 | | | | | 30-44 | .915* |
| | | | | | | 45-59 | .827* |
| H6: I have | | | | | | 60+ | .617* |
| confidence that I | 30-44 | | | | | 18-29 | .915* |
| can trust the | | | | | | 45-59 | 088 |
| expertise of my | | | | | | 60+ | 298 |
| retail bank | 45-59 | 3 | 428 | 10.955 | .000 | 18-29 | 827* |
| | | | | | | 30-44 | .088 |
| | | | | | | 60+ | - 210 |
| | 60+ | 1 | | | | 18-29 | - 617* |
| | 001 | | | | | 30-44 | 298 |
| | | | | | | 45-59 | 210 |
| | 18-29 | | | | | 30-44 | 931* |
| | 10 20 | | | | | 45-59 | 913* |
| | | | | | | | 679* |
| H7: I have | 30-44 | | | | | 18-29 | - 031* |
| confidence that | 00 44 | | | | | 45-59 | - 017 |
| the price of retail | | | | | | | - 234* |
| banking products | 45-59 | 3 | 428 | 10.936 | .000 | 18-29 | - 013* |
| is fair | 40 00 | | | | | 30-44 | 017 |
| | | | | | | 60+ | - 234 |
| | 60+ | - | | | | 18-29 | .20 4 - 679* |
| | 001 | | | | | 30-44 | 257 |
| | | | | | | 45-59 | 234 |
| | 18-29 | | | | | 30-44 | 360 |
| | 10 20 | | | | | 45-59 | 300 |
| | | | | | | | .300 |
| H8 [.] I have | 30-44 | - | | | | 18-20 | - 360 |
| confidence that I | 50 44 | | | | | 10 20 | - 061 |
| will not be | | | | | | 40 00 60± | - 257 |
| unlawfully | 15-59 | 3 | 428 | 2.354 | 0.71 | 18-20 | 207 |
| discriminated | 40-09 | _ | _ | | - | 30-44 | 300 |
| against by my | | | | | | | - 106 |
| retail bank | 60+ | - | | | | 18-20 | 130 |
| | 00+ | | | | | 30-44 | 104 |
| | | | | | | 45-50 | .207 |
| | | | | | | 40-08 | .190 |
| | 18-29 | | | | | 30-44 | 542* |
| | 1 | | | | | 45-59 | .357 |
| | | | | | | 60+ | .199 |
| H9: In general, I | 30-44 | | | | | 18-29 | 542* |
| have confidence | | | | | | 45-59 | 185 |

| that as a retail | | | | | | 60+ | 343 |
|------------------|-------|---|-----|------|-------|-------|------|
| bank customer I | 45-59 | 3 | 428 | 4.67 | 0.003 | 18-29 | 357 |
| am adequately | | | | | | 30-44 | .185 |
| protected | | | | | | 60+ | 159 |
| | 60+ | | | | | 18-29 | 199 |
| | | | | | | 30-44 | .343 |
| | | | | | | 45-59 | .159 |

(vi) Kruskal-Wallis and pairwise comparison test: Age

| Kruskal-Wallis | Age groups | Md | df | Ν | Sig (2- sided) | Age groups | Adj. Sig |
|--|---------------|----|----|-----|-------------------|------------|----------|
| H1: I have | 18-29 | 4 | | | , | | |
| retail banking | 30-44 | 4 | | | | | |
| system is stable and secure | 45-59 | 4 | 3 | 432 | .241 | N/A | N/A |
| | 60+ | 4 | | | | | |
| H2: I have | 18-29 | 5 | | | | | |
| money in my retail | 30-44 | 5 | 0 | 400 | 000 | N1/A | |
| protected should | 45-59 | 5 | 3 | 432 | .206 | N/A | N/A |
| my bank fail | 60+ | 5 | | | | | |
| H3: I have confidence that my retail bank will | 18-29 | 5 | | | | N/A | N/A |
| provide me with | 30-44 | 4 | | | | 18-29 | .000 |
| sufficient information to | | | 3 | 432 | .000 | N/A | N/A |
| make informed decisions | 45-59 | 4 | | | | 18-29 | .001 |
| | 60+ | 5 | | | | N/A | N/A |
| | 001 | Ű | | | | | |
| H4: I have | 18-29 | 5 | | | | N/A | N/A |
| confidence that my | 30-44 | 4 | | | | 18-29 | .000 |
| retail bank will do | | | 3 | 432 | .000 | N/A | N/A |
| the light thing | 45 50 | 1 | | | | 60+ | .008 |
| | 40-09 | 4 | | | | 10-29 | .000 |
| | 60+ | 4 | | | | N/A | N/A |
| | 18-29 | 5 | | | | N1/A | NI/A |
| H5: I have | | | | | | N/A | N/A |
| confidence that I | 30-44 | 4 | | | | 18-29 | .038 |
| am protected from retail bank | | | 3 | 432 | .009 | N/A | N/A |
| misconduct | 45-59 | 4 | | | | 18-29 | .012 |
| | 60+ | 4 | | | | N/A | N/A |
| | | | | | | | |

| | 18-29 | 5 | | | | N/A | N/A |
|-------------------------------------|-------|---|---|-----|------|-------|-------------|
| H6: I have | | | | | | | |
| confidence that I | 30-44 | 4 | | | | 18-29 | .000 |
| can trust the | •••• | - | | | | N/A | N/A |
| expertise of my | 45 50 | | 3 | 432 | 000 | 40.00 | |
| | 45-59 | 4 | 5 | 452 | .000 | 18-29 | .000 N/A |
| | | | | | | IN/A | N/A |
| | 60+ | 4 | | | | 18-29 | .002 |
| | | | | | | N/A | N/A |
| | 18-29 | 5 | | | | | |
| | | | | | | N/A | N/A |
| H7: I have | 30-44 | 3 | | | | 18-29 | .000 |
| confidence that the | | _ | | | | N/A | N/A |
| banking products is | 15-59 | 3 | 3 | 432 | .000 | 18-20 | 000 |
| fair | -5-55 | 5 | | | | N/A | N/A |
| | 60+ | 1 | | | | 18-20 | 005 |
| | 007 | - | | | | N/A | 005 N/A |
| | | | | | | | |
| H8: I have | 18-29 | 5 | | | | | |
| will not be | 30-44 | 5 | | | | | |
| unlawfully | 45-59 | 5 | 3 | 432 | .096 | N/A | N/A |
| against by my retail | 601 | F | | | | | |
| bank | 00+ | 5 | | | | | |
| | 18-29 | 5 | | | | | |
| H9: In general, I | | | | | | N/A | N/A |
| have confidence that as a retail | 30-44 | 4 | | | | 18-29 | _011 |
| bank customer I | | | 3 | 432 | .008 | | |
| am adequately | 45-59 | 4 | | | | N/A | N1/A |
| | 60+ | 5 | | | | | IN/A |
| | | | | | | | |

| Secondary School Secondary School C Difference 0 H1: I have confidence the UK retail satule and secure College | ANOVA | Education | df1 | df2 | F | Sia. | Education | Mean |
|---|-------------------|---------------|-----|-----|-------|------|-----------|------------|
| H1: I have confidence the UK retail banking system is stable and secure Secondary College 3 428 2.689 0.46 C 387 H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail Secondary School 3 428 2.689 0.46 SS 387 H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College 3 428 1.744 1.774 9 0.046 SS 315 H3: I have confidence that my retail bank will provide me with sufficient Secondary School 3 428 1.744 .157 SS 187 PG .020 .020 .020 .020 .020 .020 VIG .217 SS .157 SS .187 .016 .218 H3: I have confidence that my retail bank will provide me will do the right thing Undergraduate 3 428 3.446 .017 SS .279 College .173 .026 .017 .016 .210 PG .210 .210 | | | | | | 5 | | Difference |
| School School UG 407 H1: I have confidence the ubkring system is stable and secure College 3 428 2.689 0.046 SS 307 Undergraduate 0 225 023 SS 304 Undergraduate 0 225 023 SS 304 H2: I have confidence that the money in my retail bank fail College 225 UG .187 Visconfidence that the money in my retail bank fail College 225 UG .187 Postgraduate 225 225 UG .187 H3: I have confidence that my retail bank will powiden decisions College 225 225 H3: I have confidence that my retail bank will powiden decisions College 225 225 H4: I have confidence that my retail bank will do the right will do the right will do the right thing College 225 017 College 225 017 SS 279 College 225 021 022 Undergraduate 23 | | Secondary | | | | | С | .387 |
| H1: I have confidence the UK retail banking system is stable and secure College 3 428 2.689 .046 PG .364 Postgraduate 3 428 2.689 .046 C .020 Postgraduate Postgraduate 3 428 2.689 .046 C .020 Postgraduate Postgraduate 3 428 2.689 .046 C .020 Postgraduate Postgraduate 3 428 1.744 .046 .046 .046 H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College 3 428 1.744 1.577 SS .187 H3: I have confidence that my retail bank will provide me will provide me decisions Postgraduate 3 428 3.446 .017 SS .127 H4: I have confidence that my retail bank will provide me will ot the my will ot the my will ot the my thin sufficient Secondary School 428 3.446 .017 SS .279 College - .017 SS .210 | | School | | | | | UG | .407 |
| H1: I have confidence the banking system is stable and secure College 3 428 2.689 .046 SS 364 Undergraduate 0 .020 .026 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>PG</td> <td>.364</td> | | | | | | | PG | .364 |
| confidence the UK retail banking system is stable and secure Undergraduate 3 428 2.689 .046 G 020 Postgraduate 3 428 2.689 .046 G .020 Secure Postgraduate 3 428 2.689 .046 G .020 Postgraduate Postgraduate 3 428 2.689 .046 G .020 Postgraduate Secondary School Secondary School Retrieve the confidence that account(s) is protected College .046 .046 .023 H3: I have confidence that my retail bank fail Undergraduate 3 428 1.744 .157 SS .315 H3: I have confidence that my retail bank fail College .1744 .157 SS .167 H4: I have confidence that my retail bank will provide me with sufficient information to make informed decisions College 3 428 3.446 .017 .017 .017 Secondary School Secondary School Retrieve the secondary School | H1: I have | College | | | | | SS | 387 |
| UK retail banking system is stable and secure Undergraduate 3 428 2.689 .046 PG .023 Secure Postgraduate 3 428 2.689 .046 SS 407 Secure Postgraduate 3 428 2.689 .046 SS 407 Postgraduate Postgraduate | confidence the | 0 | | | | | UG | .020 |
| banking system is stable and secure Undergraduate 3 428 2.689 .046 SS 407 Secure Postgraduate Postgraduate Postgraduate PG 020 Postgraduate Secondary School Secondary SS 364 H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College 428 1.744 .157 SS 315 H3: I have confidence that minformation to make informed decisions Postgraduate 3 428 1.744 .157 SS 187 H3: I have confidence that minformation to make informed decisions Secondary School 428 3.446 .017 SS 173 H4: I have confidence that minformation to make informed decisions Secondary School 428 3.446 .017 SS 279 H4: I have confidence that minformation to make informed decisions Secondary School 3 428 7.177 000 SS 400 School | UK retail | | | | | | PG | 023 |
| Is stable and secure Postgraduate Postgraduate Postgraduate Postgraduate PC 020 PG 023 PG 023 PG 023 UC 0.023 UC 0.031 0.023 UC 0.021< | banking system | Undergraduate | 3 | 428 | 2.689 | .046 | SS | 407 |
| Secure Postgraduate Pestgraduate Pestgraduate Pestgraduate Secondary | is stable and | _ | | | | | С | 020 |
| Postgraduate Postgraduate SS 364 C .023 .0060 .003 .0060 | secure | | | | | | PG | 023 |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College 3 428 1.744 .157 C .000 .187 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Postgraduate 3 428 1.744 .157 C .000 .187 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School A 428 3.446 .017 C .128 H4: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School 3 428 3.446 .017 SS .017 H4: I have confidence that my retail bank will do the right thing Secondary School 3 428 3.446 .017 SS .279 C C .017 SS .279 .017 .016 .021 H4: I have confidence that my retail bank will do the right thing Secondary School 3 428 7.177 .000 SS .401 Undergraduate 3 428 <td></td> <td>Postgraduate</td> <td></td> <td></td> <td></td> <td></td> <td>SS</td> <td>364</td> | | Postgraduate | | | | | SS | 364 |
| Secondary School Secondary School UG .043 .315 H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College .3 428 1.744 .157 G .315 PG .060 SS 315 UG .187 PG .060 SS 315 UG .218 PG .265 UG .218 PG .218 Postgraduate Postgraduate .157 SS .127 Secondary School Secondary School .127 SS .060 C .255 .016 .255 .016 .017 SS .127 H3: I have confidence that my retail bank will provide me with sufficient information to decisions Undergraduate 3 428 3.446 .017 SS .279 H4: I have confidence that my retail bank will do the right thing College 3 428 7.177 .000 SS .401 Undergraduate 3 428 7.177 .000 SS <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td><td>С</td><td>.023</td></t<> | | | | | | | С | .023 |
| Secondary School Secondary School Secondary School Secondary College Secondary Confidence that My retail bank will provide me with sufficient information to make informed decisions Secondary School Secondary College Secondary School Secondary College Secondary College Secondary College Secondary College Secondary College Secondary College Secondary College Secondary College Secondary College Secondary Confidence that My retail bank will do the right thing Secondary College College | | | | | | | UG | .043 |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College Undergraduate 3 428 1.744 .157 UG .187 Machine Undergraduate 3 428 1.744 .157 PG .0000 Secondary should my bank fail Postgraduate 3 428 1.744 .157 C .128 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School A 428 3.446 .017 FG .017 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 3.446 .017 FG .017 FG .017 FG .017 FG .017 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS 401 LG 217 FG 210 210 210 210 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 < | | Secondary | | | | | С | .315 |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail College 93 428 1.744 .157 PG .060 Undergraduate 3 428 1.744 .157 SS 187 protected should my bank fail Postgraduate 3 428 1.744 .157 SS 187 H3: I have confidence that my retail bank wilt provide me decisions Secondary School Secondary School S .017 G .173 Undergraduate 3 428 3.446 .017 SS .173 H3: I have confidence that my retail bank wilt provide me decisions College 3 428 3.446 .017 FG .017 PG .017 SS .279 .173 UG .173 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .463* PG .210 .210 .210 .210 .210 .210 H4: I have confidence that my retail bank will | | School | | | | | UG | .187 |
| H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail Undergraduate 3 428 1.744 .157 SS 315 Postgraduate 0 428 1.744 .157 SS 187 Should my bank fail Postgraduate 0 428 1.744 .157 SS 060 R Secondary School Secondary School S 060 .255 UG .173 S 060 .265 .060 UG .173 S 060 .0173 .016 .0173 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Undergraduate 3 428 3.446 .017 .017 .017 PG .317 .017 SS .279 .017 .017 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .4401 GC .017 .016 .016 .061 .06 | | | | | | | PG | .060 |
| confidence that the money in my retail bank account(s) is protected should my bank fail Undergraduate 3 428 1.744 .157 G 218 PG 225 .128 Bank fail Postgraduate 3 428 1.744 .157 SS .121 Bank fail Postgraduate Secondary School | H2: I have | College | | | | | SS | 315 |
| the money in my retail bank account(s) is protected should my bank fail Undergraduate 3 428 1.744 .157 PG 255 Postgraduate 3 428 1.744 .157 SS 187 should my bank fail Postgraduate 3 428 1.744 .157 SS 167 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School A 428 3.446 .017 FG .279 VIndergraduate 3 428 3.446 .017 FG .210 H3: I have confidence that information to make informed decisions Undergraduate 3 428 3.446 .017 FG .210 H4: I have confidence that my retail bank will do the right thing Secondary School Secondary School S 401 SS .420 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .401 UG .210 | confidence that | | | | | | UG | 218 |
| my retail bank account(s) is protected should my bank fail Undergraduate 3 428 1.744 .157 SS 187 Protected should my bank fail Postgraduate Postgraduate 1.744 .157 SS 187 Protected should my bank fail Postgraduate Postgraduate 1.744 .157 SS 187 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School A28 3.446 .017 SS 127 H4: I have confidence that my retail bank will provide me with sufficient information to make informed decisions Secondary School 3 428 3.446 .017 SS 279 H4: I have confidence that my retail bank will do the right thing Secondary School S 490' C 317 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS 463' PG .210 .016 .026 .359 .016 .061 PG .238 .177 .000 <t< td=""><td>the money in</td><td></td><td></td><td>100</td><td></td><td></td><td>PG</td><td>255</td></t<> | the money in | | | 100 | | | PG | 255 |
| account(s) is protected should my bank fail Postgraduate Postgraduate PC .128 PG 127 SS .060 C .255 UG .127 UG .127 H3: I have confidence that my retail bank will provide me unt sufficient information to make informed decisions College 428 3.446 .017 C .173 Undergraduate 3 428 3.446 .017 PG .317 UG .107 PG .210 .210 .210 .210 H4: I have confidence that my retail bank will do the right thing Postgraduate 3 428 7.177 .000 SS .420* H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .420* PG .210 .210 .210 .210 .210 .210 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .4283 PG | my retail bank | Undergraduate | 3 | 428 | 1./44 | .157 | SS | 187 |
| Protected should my bank fail Postgraduate PG 127 Should my bank fail Postgraduate SS 060 C .255 UG .127 H3: I have confidence that information to make informed decisions Secondary School A28 3.446 .017 C .173 Undergraduate 3 428 3.446 .017 SS .279 PG 490* .017 SS .279 .010 .017 Make informed decisions Postgraduate 3 428 3.446 .017 SS .279 H4: I have confidence that my retail bank will do the right thing Secondary School .428 7.177 .000 SS .401* UG .210 .210 .210 .210 .210 .210 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .401* UG .210 .210 .210 .210 .210 .210 | account(s) is | s) is | | | | | С | .128 |
| Should fly bank fail Postgraduate SS 060 C .255 UG .127 UG .127 UG .173 School C .173 H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions College 3 428 3.446 .017 SS .173 Undergraduate 3 428 3.446 .017 SS .279 PG .317 UG .107 SS .279 PG .210 .017 SS .279 PG .210 .017 SS .210 PG .210 .017 SS .210 PG .210 .017 SS .210 School .017 SS .490* .017 H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS .463* PG .359 .051 .051 .051 | protected | | | | | | PG | 127 |
| ConstrainedC | bank fail | Postgraduate | | | | | SS | 060 |
| Secondary School Secondary School Alternation of the secondary School | | | | | | | C | .255 |
| Secondary School Secondary School Secondary School Secondary School Secondary School Secondary School Secondary SChool Secondary SS Secondary SChool Secondary SS | | | | | | | UG | .127 |
| H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions College 3 428 3.446 .017 Message SS 173 UG .017 PG 017 SS 279 .017 SS 173 Undergraduate 3 428 3.446 .017 SS 279 PG 107 SS 279 017 SS 279 C 017 SS 279 017 SS 279 PG 107 SS 219 017 SS 219 H4: I have confidence that my retail bank will do the right thing Secondary School S 401 UG 463* VID dergraduate 3 428 7.177 000 SS 463* PG 359 463* 463* 463* 463* UG 661 632 632 632 632 PG 298 632 179 6 | | Secondary | | | | | С | .173 |
| H3: I have confidence that my retail bank will provide meCollege 3 428 3.446 0.017 PG $.490^*$ With sufficient information to make informed decisionsUndergraduate 3 428 3.446 $.017$ PG $.317$ PG 0.017 PG 0.017 PG 0.017 PG 0.017 PG 0.017 PG 0.017 PG 0.017 H4: I have confidence that my retail bank will do the right thing PG 0.011 PG 0.061 PG 0.011 0.011 0.011 0.011 0.011 0.011 PG 0.011 0.011 0.011 0.011 0.011 <td< td=""><td></td><td>School</td><td rowspan="5">429</td><td></td><td></td><td>UG</td><td>.279</td></td<> | | School | | 429 | | | UG | .279 |
| Confidence that my retail bank will provide me with sufficient information to make informed decisions Undergraduate 3 428 3.446 .017 SS 173 PG .317 PG .317 C 107 make informed decisions Postgraduate 3 428 3.446 .017 SS 279 PG .210 PG .210 SS 490* C 317 UG 210 UG 210 Secondary confidence that my retail bank will do the right thing Secondary A A A A A A A A A PG .000 SS 401 UG .463* A H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS 463* PG .298 SS 298 SS 298 SS 298 PG .298 SS 289 SS 289 SS 28 | H3: I have | | | | | | PG | .490* |
| Implement of the provide me with sufficient information to make informed decisions Undergraduate 3 428 3.446 .017 PG .317 Multiprovide me with sufficient information to make informed decisions Undergraduate 3 428 3.446 .017 SS 279 Postgraduate Postgraduate - - - - - - - 0 - | confidence that | College | | | 2.446 | | SS | 173 |
| With provide the with sufficient information to make informed decisionsUndergraduate3428 3.446 $.017$ PGG $.317$ Make informed decisionsPostgraduate 3 428 3.446 $.017$ SS 279 PostgraduatePostgraduate -107 PG $.210$ Postgraduate -107 -107 PG $.210$ Postgraduate -107 -107 -107 SecondarySecondary -107 UG -210 School -107 -107 UG -210 H4: I have confidence that my retail bank will do the right thing -107 -107 -107 Postgraduate -107 -107 -107 -107 Po | my retail bank | | | | | 017 | UG | .107 |
| will sufficient information to make informed decisionsUndergraduate3428 3.440 $.017$ SS 279 Make informed decisionsPostgraduate 0.017 SS 279 PostgraduatePostgraduate PG $.210$ PostgraduateSecondary School- 0.017 SS 490^* H4: I have confidence that my retail bank will do the right thingSecondary College- 428 7.177 0.000 SS 401 Undergraduate3 428 7.177 0.000 SS 463^* 0.61 PG 0.061 0.61 0.61 0.61 0.61 0.61 Postgraduate3 428 7.177 0.000 SS 463^* Undergraduate 0.017 0.000 SS 463^* 0.061 PG 0.298 0.011 0.001 0.021 0.021 Postgraduate 0.017 0.000 0.011 0.021 PG 0.021 0.011 0.021 0.021 PG 0.021 0.021 0.021 0.021 PG 0.021 $0.$ | will provide me | | 2 | | | | PG | .317 |
| Initiation to make informed decisions Postgraduate Image: Constraint of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system of the system | information to | Undergraduate | 3 | 420 | 3.440 | .017 | SS | 279 |
| Induct monitor PG .210 decisions Postgraduate SS 490* Postgraduate C 317 UG 210 UG Secondary School .000 College VIII UG .401 VIII College .000 .061 Viiii Undergraduate 3 428 7.177 .000 SS 401 VIII Undergraduate 3 428 7.177 .000 SS 403* PG .001 SS 401 .061 PG .359 PG .000 SS 403* .000 SS 463* VIII O SS 463* .001 SS 298 PG .0298 SS .532 | make informed | | | | | | C | 107 |
| Bestgraduate SS 490* C 317 UG 210 UG 210 UG 210 UG .001 UG .210 Kecondary School UG .401 College College .000 .663* College .000 SS .401 UG .061 .061 .061 PG .359 .403* Undergraduate 3 428 7.177 .000 SS .463* PG .359 .463* .000 SS .463* PG .298 .001 SS .463* PG .298 .001 SS .532 C .179 UG .289 SS .532 .179 UG .289 | decisions | | - | | | | PG | .210 |
| H4: I have confidence that my retail bank will do the right thing Secondary School A 428 7.177 .000 C 317 UG 210 UG 210 .401 UG .401 UG .401 UG .401 UG .401 UG .401 UG .401 UG .401 UG .401 UG .061 .061 .061 PG .359 .403* PG .298 .428 PG .298 .532 C .179 UG .289 Postgraduate Image: Content of the content o | | Postgraduate | | | | | 55 | 490^ |
| Secondary School Secondary School Addition H4: I have confidence that my retail bank will do the right thing College | | | | | | | U U | 317 |
| Secondary School Secondary School Advin H4: I have confidence that my retail bank will do the right thing College 428 7.177 .000 SS 401 Undergraduate 3 428 7.177 .000 SS 463* PG .359 .000 SS 463* C 061 PG .298 SS 532 C 179 UG .289 | | 0 | | | | | UG | 210 |
| H4: I have confidence that my retail bank will do the right thing College 3 428 7.177 .000 G .463* Undergraduate 3 428 7.177 .000 SS 401 DG .061 .061 .061 .061 .061 .061 PG .359 .000 SS 463* .000 SS 463* PG .298 .001 SS 298 | | Secondary | | | | | | .401 |
| H4: I have confidence that my retail bank will do the right thing College 3 428 7.177 .000 SS 401 Undergraduate 3 428 7.177 .000 SS 463* PG .359 .000 SS 463* PG .298 .001 SS 532 POstgraduate - - - 0 VIDG 532 532 179 0 | | School | | | | | UG | .463** |
| H4: I have confidence that my retail bank will do the right thing Undergraduate 3 428 7.177 .000 SS 401 PG .359 PG .359 PG .359 PG .298 PG .298 C 179 UG .298 Postgraduate .000 SS C 179 UG .289 | | Callaga | - | | | | PG | .760** |
| Indice confidence that my retail bank will do the right thingUndergraduate34287.177.000SS.00190.0619359.0610.0610.0610.06190.0610.0610.0610.0610.0610.06110000.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.0610.0610.0610.0610.06190.061 <t< td=""><td>H4. I have</td><td>College</td><td></td><td></td><td></td><td></td><td>55</td><td>401</td></t<> | H4. I have | College | | | | | 55 | 401 |
| Will do the right thing Undergraduate 3 428 7.177 .000 SS 463* Postgraduate | confidence that | | | | | | UG | .061 |
| will do the right thingOndergraduateo1201100333463PostgraduateC061C061PostgraduateSS532C179UG289UG289 | my retail bank | Undergreduete | 3 | 428 | 7 177 | 000 | PG | .309 |
| thing Postgraduate Postgraduate Postgraduate C C C C C C C C C C C C C C C C C C C | will do the right | Undergraduate | Ŭ | 0 | | | | 403 |
| Postgraduate FG .290 0 SS 532 0 C 179 UG 289 0 C 353 | thing | | | | | | | 001 |
| C179 | | Postaraduata | | | | | | .290 |
| UG179 UG289 | | r usigraduate | | | | | 33 | 032 |
| | | | | | | | | 179 |
| | | | | | | | 00 C | 209 |

(vii) ANOVA and Tukey HSD test: Education

| | Secondary | | | | | UG | .243 |
|------------------|-----------------|---|-----|-------|------|----------|--------|
| | School | | | | | PG | .532* |
| H5: I have | College | | | | | SS | 353 |
| confidence that | | | | | | UG | 110 |
| I am protected | | | 400 | 0.044 | 040 | PG | .179 |
| from retail bank | Undergraduate | 3 | 428 | 3.841 | .010 | SS | 243 |
| misconduct | | | | | | C | .110 |
| | | - | | | | PG | .289 |
| | Postgraduate | | | | | SS | .532* |
| | | | | | | C | 179 |
| | O a a a a da ma | | | | | UG | 289 |
| | Secondary | | | | | | .096 |
| H6. I have | 301001 | | | | | | .332 |
| confidence that | Collogo | | | | | | .009 |
| I can trust the | College | | | | | 33 | 090 |
| expertise of my | | | | | | DG PG | .230 |
| retail bank | Undergraduate | 3 | 428 | 4.909 | .002 | | .403 |
| | Undergraduate | - | | | | 33 | 332 |
| | | | | | | PG | 230 |
| | Postaraduate | | | | | 22 | - 559* |
| | 1 USIGIAUUUUU | | | | | 0 | - 463* |
| | | | | | | UG | 227 |
| | Secondary | | | | | 00 C | 284 |
| | School | | | | | UG | .398 |
| | | | | | | PG | .577* |
| H7: I have | College | | | | | SS | 284 |
| confidence that | | | | | | UG | .114 |
| the price of | | | | | | PG | .293 |
| retail banking | Undergraduate | 3 | 428 | 3.950 | .008 | SS | 398 |
| products is fair | - | | | | | С | 114 |
| | | | | | | PG | .179 |
| | Postgraduate | | | | | SS | 577* |
| | | | | | | С | 293 |
| | | | | | | UG | 179 |
| | Secondary | | | | | С | .052 |
| | School | | | | | UG | .182 |
| | | - | | | | PG | .267 |
| H8: I have | College | | | | | SS | 052 |
| confidence that | | | | | | UG | .182 |
| | | 3 | 128 | 1 281 | 248 | PG | .267 |
| discriminated | Undergraduate | 5 | 420 | 1.501 | .240 | SS | 182 |
| against by my | | | | | | C | 129 |
| retail bank | Destanduate | | | | | PG | .085 |
| | Postgraduate | | | | | 55 | 267 |
| | | | | | | | 214 |
| | | | | | | 00 | 005 |
| | Secondary | | | | | C | .308 |
| | School | | | | | UG | .266 |
| | | | | | | PG | .536* |
| H9: In general, | College | | | | | SS | 308 |
| I have | | | | | | UG | 042 |

| confidence that | | | | | | PG | .227 |
|------------------|---------------|---|-----|-------|------|----|------|
| as a retail bank | Undergraduate | 3 | 428 | 4.715 | .003 | SS | 266 |
| customer I am | | | | | | С | .042 |
| adequately | | | | | | PG | .270 |
| protected | Postgraduate | | | | | SS | 536* |
| | | | | | | С | 227 |
| | | | | | | UG | 270 |

| Kruskal-Wallis | Education | Md | df | Ν | Sig (2- sided) | Education | Adj. Sig |
|---|---------------------|-----|----|-----|-------------------|---------------------|----------|
| H1: I have confidence the | Secondary School | 4 | | | olucuj | | |
| UK retail | College | 4 | 3 | 132 | 040 | N/A | NI/A |
| is stable and | Undergraduate | 4 | 5 | 402 | | N/A | IN/71 |
| secure | Postgraduate | 4 | | | | | |
| H2: I have confidence that | Secondary School | 5 | | | | | |
| the money in | College | 4.5 | 2 | 400 | 105 | N1/A | N1/A |
| account(s) is | Undergraduate | 5 | 3 | 432 | .155 | N/A | N/A |
| should my bank | Postgraduate | 5 | | | | | |
| H3: I have confidence that | Secondary School | 5 | | | | | |
| my retail bank | College | 4.5 | 3 | 432 | 005 | N/A | N/A |
| with sufficient | Undergraduate | 4 | 0 | 102 | | | |
| make informed decisions | Postgraduate | 4 | | | | Secondary School | .003 |
| | | | | | | N/A | N/A |
| | Secondary School | 5 | | | | N/A | N/A |
| H4: I have | College | 4 | | | | Secondary School | .033 |
| confidence that my retail bank | | | 3 | 432 | .000 | N/A | N/A |
| will do the right thing | Undergraduate | 4 | | | | Secondary School | .027 |
| | | | | | | N/A | N/A |
| | Postgraduate | 4 | | | | Secondary School | .000 |
| | | | | | | N/A | N/A |
| | Secondary School | 5 | | | | N/A | N/A |
| H5: I have confidence that I am protected from retail bank | College | 4 | | | | Secondary School | .049 |
| | Undergraduate | 4 | 3 | 432 | .002 | N/A | N/A |
| misconduct | Postgraduate | 4 | | | | Secondary School | .001 |

(viii) Kruskal-Wallis and pairwise comparison test: Education

| | | | | | | N/A | N/A |
|---|---------------------|---|---|-----|------|---------------------|------|
| | Secondary School | 5 | 3 | 432 | .000 | | |
| H6: I have | College | 4 | | | | N/A | N/A |
| I can trust the | Undergraduate | 4 | | | | | |
| retail bank | Postgraduate | 4 | | | | Secondary School | .000 |
| | | | | | | College | .012 |
| | | | | | | N/A | N/A |
| | Secondary School | 4 | | | | | |
| H7: I have confidence that | College | 4 | | | | N/A | N/A |
| the price of retail banking | Undergraduate | 4 | 3 | 432 | .006 | | |
| products is fair | Postgraduate | 3 | | | | Secondary School | .003 |
| | | | | | | N/A | N/A |
| H8: I have confidence that | Secondary School | 5 | | | | | |
| l will not be unlawfully | College | 5 | 3 | 432 | .257 | N/A | N/A |
| discriminated | Undergraduate | 5 | | | | | |
| retail bank | Postgraduate | 5 | | | | | |
| H9: In general, | Secondary School | 5 | | | | | |
| I have confidence that as a retail bank customer I am adequately protected | College | 4 | 3 | 432 | .002 | N/A | N/A |
| | Undergraduate | 4 | | | | | |
| | Postgraduate | 4 | | | | Secondary School | .001 |
| protected | | | | | | N/A | N/A |

| ANOVA | Income | df1 | df2 | F | Sig. | Income | Mean |
|---------------------|-----------|-----|-----|-------|------|-----------------|------------|
| | | | | | | | Difference |
| | | | | | | £13,501-£27,600 | .283 |
| | | | | | | £27,601-£43,000 | .323 |
| | <£13,500 | | | | | £43,001- | 400 |
| | | | | | | £150,000 | .199 |
| | | | | | | £150,001+ | 247 |
| | | | | | | <£13,500 | 283 |
| | £13,501- | | | | | £27,601-£43,000 | .040 |
| H1: I have | £27,600 | | | | | £43,001- | 004 |
| confidence the UK | | | | | | £150,000 | 084 |
| retail banking | | | | | | £150,001+ | 530 |
| system is stable | | 4 | 427 | 1.557 | .185 | <£13,500 | 323 |
| and secure | £27,601- | | | | | £13,501-£27,600 | 040 |
| | £43,000 | | | | | £43,001- | 40.4 |
| | | | | | | £150,000 | 124 |
| | | | | | | £150,001+ | 570 |
| | | | | | | <£13,500 | 199 |
| | £43,001- | | | | | £13,501-£27,600 | .084 |
| | £150,000 | | | | | £27,601-£43,000 | .124 |
| | | | | | | £150,001+ | 446 |
| | | | | | | <£13,500 | .247 |
| | | | | | | £13,501-£27,600 | .530 |
| | £150,001+ | | | | | £27,601-£43,000 | .570 |
| | | | | | | £43,001- | 440 |
| | | | | | | £150,000 | .446 |
| | | | | | | £13,501-£27,600 | .201 |
| | | | | | | £27,601-£43,000 | 234 |
| | <£13,500 | | | | | £43,001- | 004 |
| | | | | | | £150,000 | 384 |
| | | | | | | £150,001+ | 760 |
| H2: I have | | | | | | <£13,500 | 201 |
| confidence that the | £13,501- | | | | | £27,601-£43,000 | 435* |
| money in my retail | £27,600 | | | 4.567 | .001 | £43,001- | * |
| bank account(s) is | | 4 | 427 | | | £150,000 | 585 |

(ix) ANOVA and Tukey HSD test: Income

Claire Lynne McCafferty

| protected should | | | | | | £150,001+ | 962 |
|---------------------|-----------|---|-----|-------|------|-----------------|-------------------|
| my bank fail | | | | | | <£13,500 | .234 |
| | £27,601- | | | | | £13,501-£27,600 | .435* |
| | £43,000 | | | | | £43,001- | |
| | | | | | | £150,000 | 150 |
| | | | | | | £150,001+ | 526 |
| | £43,001- | | | | | <£13,500 | .384 |
| | £150,000 | | | | | £13,501-£27,600 | .585* |
| | | | | | | £27,601-£43,000 | .150 |
| | | | | | | £150,001+ | 376 |
| | £150,001+ | | | | | <£13,500 | .760 |
| | | | | | | £13,501-£27,600 | .962 |
| | | | | | | £27,601-£43,000 | .526 |
| | | | | | | £43,001- | 070 |
| | | | | | | £150,000 | .376 |
| | <£13,500 | | | | | £13,501-£27,600 | .312 |
| | | | | | | £27,601-£43,000 | .563 [*] |
| | | | | | | £43,001- | 200 |
| | | | | | | £150,000 | .388 |
| | | | | | | £150,001+ | 1.151 |
| | | | | | | <£13,500 | 312 |
| | £13,501- | | | | | £27,601-£43,000 | .251 |
| H3: I have | £27,600 | | | | | £43,001- | 075 |
| confidence that my | | 4 | 427 | 4.787 | .001 | £150,000 | .075 |
| retail bank will | | | | | | £150,001+ | .839 |
| provide me with | | | | | | <£13,500 | 563 [*] |
| sufficient | £27,601- | | | | | £13,501-£27,600 | 251 |
| information to make | £43,000 | | | | | £43,001- | 470 |
| informed decisions | | | | | | £150,000 | 176 |
| | | | | | | £150,001+ | .588 |
| | | | | | | <£13,500 | 388 |
| | £43,001- | | | | | £13,501-£27,600 | 075 |
| | £150,000 | | | | | £27,601-£43,000 | .176 |
| | | | | | | £150,001+ | .763 |
| | | | | | | <£13,500 | -1.151 |
| | | | | | | £13,501-£27,600 | 839 |

| | £150,001+ | | | | | £27,601-£43,000 | 588 |
|---------------------|-----------|---|-----|-------|------|-----------------|---------------------|
| | | | | | | £43,001- | 700 |
| | | | | | | £150,000 | 763 |
| | | | | | | £13,501-£27,600 | .342 |
| | | | | | | £27,601-£43,000 | .457* |
| | <£13,500 | | | | | £43,001- | 070* |
| | | | | | | £150,000 | .679 |
| | | | | | | £150,001+ | 1.378* |
| | | | | | | <£13,500 | 342 |
| | £13,501- | | | | | £27,601-£43,000 | .114 |
| | £27,600 | | | | | £43,001- | 207 |
| | | | | | | £150,000 | .337 |
| H4: I have | | | | | | £150,001+ | 1.036 |
| confidence that my | | | | | | <£13,500 | 457 [*] |
| retail bank will do | £27,601- | | | | | £13,501-£27,600 | 114 |
| the right thing | £43,000 | 4 | 427 | 5.262 | .000 | £43,001- | 000 |
| | | | | | | £150,000 | .222 |
| | | | | | | £150,001+ | .921 |
| | | | | | | <£13,500 | 679 [*] |
| | £43,001- | | | | | £13,501-£27,600 | 337 |
| | £150,000 | | | | | £27,601-£43,000 | 222 |
| | | | | | | £150,001+ | .699 |
| | £150,001+ | | | | | <£13,500 | -1.378 [*] |
| | | | | | | £13,501-£27,600 | -1.036 |
| | | | | | | £27,601-£43,000 | 921 |
| | | | | | | £43,001- | 000 |
| | | | | | | £150,000 | 699 |
| | | | | | | £13,501-£27,600 | .187 |
| | | | | | | £27,601-£43,000 | .187 |
| | <£13,500 | | | | | £43,001- | 000 |
| | | | | | | £150,000 | .308 |
| | | | | | | £150,001+ | 1.029 |
| | | 1 | | | | <£13,500 | 187 |
| | £13,501- | | | | | £27,601-£43,000 | 001 |
| | £27,600 | | | | | £43,001- | 404 |
| | | 4 | 427 | 1.715 | .146 | £150,000 | .121 |

| H5: I have | | | | | | £150,001+ | .842 |
|-------------------|-----------|---|-----|-------|------|-----------------|---------------------|
| confidence that I | | | | | | <£13,500 | 187 |
| am protected from | £27,601- | | | | | £13,501-£27,600 | .001 |
| retail bank | £43,000 | | | | | £43,001- | |
| misconduct | | | | | | £150,000 | .122 |
| | | | | | | £150,001+ | .842 |
| | | | | | | <£13,500 | 308 |
| | £43,001- | | | | | £13,501-£27,600 | 121 |
| | £150,000 | | | | | £27,601-£43,000 | 122 |
| | | | | | | £150,001+ | .720 |
| | | | | | | <£13,500 | -1.029 |
| | £150,001+ | | | | | £13,501-£27,600 | 842 |
| | | | | | | £27,601-£43,000 | 842 |
| | | | | | | £43,001- | |
| | | | | | | £150,000 | 720 |
| | | | | | | £13,501-£27,600 | .203 |
| | | | | | | £27,601-£43,000 | .448* |
| | <£13,500 | | | | | £43,001- | * |
| | | | | | | £150,000 | .598 |
| | | | | | | £150,001+ | 1.378 [*] |
| | | | | | | <£13,500 | 203 |
| | £13,501- | | | | | £27,601-£43,000 | .245 |
| | £27,600 | | | | | £43,001- | 005 |
| | | 4 | 427 | 5.052 | .001 | £150,000 | .395 |
| H6: I have | | | | | | £150,001+ | 1.175 |
| confidence that I | | | | | | <£13,500 | 448 [*] |
| can trust the | £27,601- | | | | | £13,501-£27,600 | 245 |
| expertise of my | £43,000 | | | | | £43,001- | |
| retail bank | | | | | | £150,000 | .150 |
| | | | | | | £150,001+ | .930 |
| | | | | | | <£13,500 | 598 [*] |
| | £43,001- | | | | | £13,501-£27,600 | 395 |
| | £150,000 | | | | | £27,601-£43,000 | 150 |
| | | | | | | £150,001+ | .780 |
| | | | | | | <£13,500 | -1.378 [*] |
| | | | | | | £13,501-£27,600 | -1.175 |

| | £150,001+ | | | | | £27,601-£43,000 | 930 |
|---------------------|-----------|---|-----|-------|------|-----------------|-------------------|
| | | | | | | £43,001- | |
| | | | | | | £150,000 | 780 |
| | | | | | | £13,501-£27,600 | .542 [*] |
| | | | | | | £27,601-£43,000 | .537 [*] |
| | <£13,500 | | | | | £43,001- | 740* |
| | | | | | | £150,000 | .742 |
| | | | | | | £150,001+ | .651 |
| | | | | | | <£13,500 | 542 [*] |
| H7: I have | £13,501- | | | | | £27,601-£43,000 | 005 |
| confidence that the | £27,600 | | | | | £43,001- | 201 |
| price of retail | | 4 | 427 | 5.425 | .000 | £150,000 | .201 |
| banking products is | | | | | | £150,001+ | .109 |
| fair | | | | | | <£13,500 | 537 [*] |
| | £27,601- | | | | | £13,501-£27,600 | .005 |
| | £43,000 | | | | | £43,001- | 005 |
| | | | | | | £150,000 | .205 |
| | | | | | | £150,001+ | .114 |
| | | | | | | <£13,500 | 742 [*] |
| | £43,001- | | | | | £13,501-£27,600 | 201 |
| | £150,000 | | | | | £27,601-£43,000 | 205 |
| | | | | | | £150,001+ | 091 |
| | | | | | | <£13,500 | 651 |
| | | | | | | £13,501-£27,600 | 109 |
| | £150,001+ | | | | | £27,601-£43,000 | 114 |
| | | | | | | £43,001- | |
| | | | | | | £150,000 | .091 |
| | | | | | | £13,501-£27,600 | .368* |
| | | | | | | £27,601-£43,000 | .249 |
| | <£13,500 | | | | | £43,001- | * |
| | | | | | | £150,000 | .558 |
| | | | | | | £150,001+ | .354 |
| | | | | | | <£13,500 | 368* |
| | £13,501- | | | | | £27,601-£43,000 | 119 |
| H8: I have | £27,600 | | | | | £43,001- | |
| confidence that I | | 4 | 427 | 3.507 | .008 | £150,000 | .191 |

| will not be | | | | | | £150,001+ | 014 |
|---|---------------------|---|-----|-------|------|-----------------|------------------|
| unlawfully | | | | | | <£13,500 | 249 |
| discriminated | £27,601- | | | | | £13,501-£27,600 | .119 |
| against by my retail | £43,000 | | | | | £43,001- | |
| bank | | | | | | £150,000 | .310 |
| | | | | | | £150,001+ | .105 |
| | | | | | | <£13,500 | 558 [*] |
| | £43,001- | | | | | £13,501-£27,600 | 191 |
| | £150,000 | | | | | £27,601-£43,000 | 310 |
| | | | | | | £150,001+ | 204 |
| | | | | | | <£13,500 | 354 |
| | | | | | | £13,501-£27,600 | .014 |
| | £150,001+ | | | | | £27,601-£43,000 | 105 |
| | | | | | | £43,001- | |
| | | | | | | £150,000 | .204 |
| | | | | | | £13,501-£27,600 | .160 |
| | <£13,500 | | | 1.261 | | £27,601-£43,000 | .150 |
| | | | | | | £43,001- | .357 |
| H9: In general, I have confidence that as a retail bank | | | | | | £150,000 | |
| | | | | | | £150,001+ | .089 |
| | | 4 | 427 | | | <£13,500 | 160 |
| | £13,501- £27,600 | | | | .285 | £27,601-£43,000 | 010 |
| | | | | | | £43,001- | 100 |
| | | | | | | £150,000 | .198 |
| | | | | | | £150,001+ | 071 |
| customer I am | | | | | | <£13,500 | 150 |
| adequately | £27,601- | | | | | £13,501-£27,600 | .010 |
| protected | £43,000 | | | | | £43,001- | |
| | | | | | | £150,000 | .207 |
| | | | | | | £150,001+ | 061 |
| | | | | | | <£13,500 | 357 |
| | £43,001- | | | | | £13,501-£27,600 | 198 |
| | £150,000 | | | | | £27,601-£43,000 | 207 |
| | | | | | | £150,001+ | 269 |
| | | 1 | | | | <£13,500 | 089 |
| | | | | | | £13,501-£27,600 | .071 |

| £150,001+ | | £27,601-£43,000 | .061 |
|-----------|--|-----------------|------|
| | | £43,001- | |
| | | £150,000 | .269 |

| (x) Kruskal-Wallis test and | pairwise | comparison | test: Income |
|-----------------------------|----------|------------|--------------|
|-----------------------------|----------|------------|--------------|

| Kruskal-Wallis | Income | Md | df | Ν | Sig (2- | Income | Adj. Sig |
|-----------------|------------------|----|----|-----|---------|----------|----------|
| | | | | | sided) | | |
| | <£13,500 | 4 | | | | | |
| H1: I have | 640 504 607 600 | A | | | | | |
| confidence the | £13,501-£27,600 | 4 | | | | | |
| UK retail | 607 604 642 000 | | | | | | |
| banking system | £27,601-£43,000 | 4 | 4 | 432 | .125 | N/A | N/A |
| is stable and | | | | | | | |
| secure | £43,001-£150,000 | 4 | | | | | |
| | | | | | | | |
| | £150,001+ | 4 | | | | | |
| | | 5 | | | | N/A | N/A |
| | <£13,500 | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | | 4 | | | | | |
| H2: I have | £13,501-£27,600 | | | | | £27,601- | .029 |
| confidence that | | | | | | £43,000 | |
| the money in | | | 4 | 432 | .001 | | |
| my retail bank | | | | | | £43,001- | .009 |
| account(s) is | | | | | | £150,000 | |
| protected | | | | | | | |
| should my bank | £27,601-£43,000 | 5 | | | | | |
| fail | | | | | | | |
| | | | | | | | |
| | £43,001-£150,000 | 5 | | | | N/A | N/A |
| | | | | | | | |
| | £150,001+ | 5 | | | | | |
| | | | | | | | |
| | <£13,500 | 5 | | | | | |
| | | | | | | N/A | N/A |
| | £13,501-£27,600 | 4 | | | | | |
| | £27,601-£43,000 | 4 | | | | <£13,500 | .000 |

| | | | 4 | 432 | .000 | N/A | N/A |
|---|--|-------------------------|---|-----|------|----------|------|
| H3: I have | | | | | | | |
| confidence that | | | | | | | |
| my retail bank | £43,001-£150,000 | 4 | | | | | |
| will provide me | | | | | | | |
| with sufficient | | | | | | | |
| information to | | | | | | | |
| make informed | £150,001+ | 3.5 | | | | | |
| decisions | | | | | | | |
| | | | | | | | |
| | | | | | | | |
| | <£13.500 | 5 | | | | | |
| | | C C | | | | N/A | N/A |
| | £13,501-£27,600 | 4 | | | | | |
| | 607 604 642 000 | 4 | | | | .042 500 | 000 |
| | £27,601-£43,000 | 4 | | | | <£13,500 | .008 |
| H4: I have | | | | | | | |
| confidence that | | | 4 | 432 | 000 | N/A | N/A |
| my retail bank | | | | 102 | .000 | | |
| will do the right | £43,001-£150,000 | 3.5 | | | | <£13,500 | .001 |
| thing | | | | | | | |
| unig | | | | | | | |
| | | | | | | N/A | N/A |
| | £150,001+ | 3 | | | | | |
| | | | | | | | |
| | <f13 500<="" td=""><td>5</td><td></td><td></td><td></td><td></td><td></td></f13> | 5 | | | | | |
| | <£13,500 | 5 | | | | | |
| H5: I have | <£13,500 | 5 | | | | | |
| H5: I have | <£13,500 £13,501-£27,600 | 5 | | | | | |
| H5: I have confidence that | <£13,500 £13,501-£27,600 | 5 | 4 | 432 | 044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank | <£13,500 £13,501-£27,600 £27,601-£43,000 | 5 | 4 | 432 | .044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank | <£13,500 £13,501-£27,600 £27,601-£43,000 | 5 | 4 | 432 | .044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank misconduct | <£13,500 £13,501-£27,600 £27,601-£43,000 £43,001-£150,000 | 5 4 4 4 | 4 | 432 | .044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank misconduct | <£13,500 £13,501-£27,600 £27,601-£43,000 £43,001-£150,000 | 5 4 4 4 | 4 | 432 | .044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank misconduct | <£13,500 £13,501-£27,600 £27,601-£43,000 £43,001-£150,000 £150,001+ | 5 4 4 4 3.5 | 4 | 432 | .044 | N/A | N/A |
| H5: I have confidence that I am protected from retail bank misconduct | <£13,500 £13,501-£27,600 £27,601-£43,000 £43,001-£150,000 £150,001+ | 5 4 4 4 3.5 5 | 4 | 432 | .044 | N/A | N/A |
| | £13,501-£27,600 | 4 | | | | N/A | N/A |
|-------------------|------------------|-----|---|-----|------|------------------|------|
| H6: I have | £27,601-£43,000 | 4 | | | | <£13,500 | 0.15 |
| L can trust the | | | 4 | 432 | 000 | | |
| expertise of my | | | 7 | 452 | .000 | N/A | N/A |
| retail bank | | | | | | | |
| | £43,001-£150,000 | 3 | | | | <£13,500 | .002 |
| | | | | | | | |
| | £150,001+ | 3 | | | | N/A | N/A |
| | <£13,500 | 4 | | | | N/A | N/A |
| | £13,501-£27,600 | 3 | | | | <£13,500 | .001 |
| H7: I have | | | | | | N/A | N/A |
| confidence that | £27,601-£43,000 | 3.5 | | | | <£13,500 | .004 |
| the price of | | | 4 | 432 | .000 | N/A | N/A |
| retail banking | £43,001-£150,000 | 3 | | | | <£13,500 | .004 |
| products is fair | | | | | | | |
| | £150,001+ | 3.5 | | | | N/A | N/A |
| | <£13,500 | 5 | | | | | |
| | | | | | | N1/A | N1/A |
| H8: I have | | | | | | IN/A | IN/A |
| confidence that | £13 501-£27 600 | 5 | | | | ∠ £13 500 | 037 |
| I will not be | 210,001 227,000 | Ū | | | | ~~10,000 | |
| unlawfully | £27 601-£43 000 | 5 | | | | | |
| discriminated | 227,001 240,000 | 5 | 4 | 432 | .007 | N/A | N/A |
| against by my | | | | | | | |
| retail bank | £43,001-£150,000 | 4 | | | | <£13,500 | .009 |
| | | | | | | | |
| | £150,001+ | 4.5 | | | | N/A | N/A |
| | | | | | | | |
| H9: In general, I | <£13,500 | 5 | | | | | |
| have | £13,501-£27,600 | 4 | | | | | |
| confidence that | | | 4 | 400 | | K1/A | N1/A |
| as a retail bank | £27,601-£43,000 | 4 | 4 | 432 | .111 | N/A | N/A |

| customer I am | £43,001-£150,000 | 4 |
|---------------|------------------|-----|
| adequately | | |
| protected | £150,001+ | 4.5 |
| | | |

APPENDIX E: CORRELATION ANALYSIS

(i) Pearson product-moment correlation coefficient: Age

| H6: I have confidence that I can trust the expertise of my | Pearson Correlation Sig. (2-tailed) | 120 0.12 |
|--|--|-------------|
| retail bank | Ν | 432 |
| H7: I have confidence that the | Pearson Correlation | .136 |
| price of retail banking products | Sig. (2-tailed) | .005 |
| is fair | Ν | 432 |

(ii) Pearson product-moment correlation coefficient: Education

| H1: I have confidence the UK | Pearson Correlation | 098 |
|---|---------------------|------|
| and secure | Sig. (2-tailed) | .042 |
| | Ν | 432 |
| H3: I have confidence that my | Pearson Correlation | 152 |
| sufficient information to make | Sig. (2-tailed) | .001 |
| informed decisions | Ν | 432 |
| H4: I have confidence that my | Pearson Correlation | 211 |
| thing | Sig. (2-tailed) | .000 |
| | N | 432 |
| H5: I have confidence that I | Pearson Correlation | 138 |
| misconduct | Sig. (2-tailed) | .004 |
| | N | 432 |
| H6: I have confidence that I | Pearson Correlation | 180 |
| retail bank | Sig. (2-tailed) | .000 |
| | N | 432 |
| H7: I have confidence that the | Pearson Correlation | 161 |
| is fair | Sig. (2-tailed) | .001 |
| | N | 432 |
| H8: I have confidence that I | Pearson Correlation | 097 |
| discriminated against by my retail bank | Sig. (2-tailed) | .044 |
| | N | 432 |
| H9: In general, I have | Pearson Correlation | 164 |
| bank customer I am | Sig. (2-tailed) | .001 |
| adequately protected | Ν | 432 |

(ii) Pearson product-moment correlation coefficient: Income

| H2: I have confidence that the money | Pearson Correlation | .153 |
|--|---------------------|------|
| in my retail bank account(s) is | Sig. (2-tailed) | .001 |
| protected should my bank fail | Ν | 432 |
| H3: I have confidence that my retail | Pearson Correlation | 175 |
| bank will provide me with sufficient | Sig. (2-tailed) | .000 |
| information to make informed decisions | Ν | 432 |
| H4: I have confidence that my retail | Pearson Correlation | 208 |
| bank will do the right thing | Sig. (2-tailed) | .000 |
| | Ν | 432 |
| H5: I have confidence that I am | Pearson Correlation | 105 |
| protected from retail bank misconduct | Sig. (2-tailed) | .029 |
| | Ν | 432 |
| H6: I have confidence that I can trust | Pearson Correlation | 206 |
| the expertise of my retail bank | Sig. (2-tailed) | .000 |
| | Ν | 432 |
| H7: I have confidence that the price of | Pearson Correlation | 192 |
| retail banking products is fair | Sig. (2-tailed) | .000 |
| | Ν | 432 |
| H8: I have confidence that I will not be | Pearson Correlation | 143 |
| unlawfully discriminated against by my | Sig. (2-tailed) | .003 |
| retail bank | Ν | 432 |

APPENDIX F: STEPWISE REGRESSION ANALYSIS

(i) H1: I have confidence the UK retail banking system is stable and secure

| H1 | В | Sig. | R square |
|------------------------------|--------|------|----------|
| Work In FS (No) | -1.045 | .000 | .059 |
| Education (Secondary School) | .397 | .003 | |
| Age (18-29) | .376 | .013 | |

(ii) H2: I have confidence that the money in my retail bank account(s) is protected should my bank fail

| H1 | В | Sig. | R square |
|--------------------------|------|------|----------|
| Income (£13,501-£27,600) | 369 | .002 | .049 |
| Age (18-29) | 324 | .022 | |
| Work In FS (Yes) | .598 | .023 | |

(iii) H3: I have confidence that my retail bank will provide me with sufficient information to make informed decisions

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .557 | .000 | .061 |
| Education (Secondary School) | .313 | .015 | |
| Income (£27,601-£43,000) | 243 | .050 | |

(iv) I have confidence that my retail bank will do the right thing

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .723 | .000 | .084 |
| Education (Secondary School) | .562 | .000 | |

(v) I have confidence that I am protected from retail bank misconduct

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .461 | .002 | .040 |
| Education (Secondary School) | .390 | .003 | |

(vi) I have confidence that I can trust the expertise of my retail bank

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .801 | .000 | .103 |
| Education (Secondary School) | .464 | .001 | |
| Education (College) | .349 | .008 | |
| Income (£150,001+) | 975 | .037 | |

(vii) I have confidence that the price of retail banking products is fair

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .744 | .000 | .097 |
| Education (Secondary School) | .378 | .008 | |
| Income (<£13,500) | .294 | .035 | |

(viii) I have confidence that I will not be unlawfully discriminated against by my retail bank

| H1 | В | Sig. | R square |
|-------------------|------|------|----------|
| Income (<£13,500) | .362 | .001 | .024 |

(ix) In general, I have confidence that as a retail bank customer I am adequately protected

| H1 | В | Sig. | R square |
|------------------------------|------|------|----------|
| Age (18-29) | .382 | .001 | .042 |
| Education (Secondary School) | .398 | .003 | |